UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Amendment No. 1

to

FORM S-1

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

Precision BioSciences, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

2836 (Primary Standard Industrial Classification Code Number)

302 East Pettigrew St., Suite A-100 Durham, North Carolina 27701 (919) 314-5512

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

Matthew Kane
President and Chief Executive Officer
Precision BioSciences, Inc.
302 East Pettigrew St., Suite A-100
Durham, North Carolina 27701
(919) 314-5512

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Peter N. Handrinos Nathan Ajiashvili Latham & Watkins LLP 200 Clarendon Street Boston, Massachusetts 02116 (617) 948-6000 20-4206017 (I.R.S. Employer Identification Number)

Divakar Gupta
Darren DeStefano
Yvan-Claude Pierre
Kristin VanderPas
Cooley LLP
1114 Avenue of the Americas
New York, New York 10036
(212) 479-6000

Amy M. Batten
Heyward D. Armstrong
Smith, Anderson, Blount, Dorsett,
Mitchell & Jernigan, LLP
150 Fayetteville Street, Suite 2300
Raleigh, North Carolina 27601
(919) 821-1220

Michael P. Saber

Approximate date of commencement of proposed sale to the public: As soon as practicable after this Registration Statement is declared effective.		
f any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securitions:	es Act of 1933 check the following	
f this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following by	ox and list the Securities Act registra	tion
f this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities earlier effective registration statement for the same offering.	Act registration statement number o	f the
f this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities earlier effective registration statement for the same offering.	Act registration statement number o	f the
ndicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting com accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.	pany. See the definitions of "large	
.arge accelerated filer □	Accelerated filer	
Non-accelerated filer ⊠	Smaller reporting company	Σ
	Emerging growth company	Σ
f an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with tandards provided pursuant to Section 7(a)(2)(B) of the Securities Act □	any new or revised financial accour	nting
The registrant hereby amends this registration statement on such date or dates as may be necessary to delay its effective date until the registration of the registrat		

The registrant hereby amends this registration statement on such date or dates as may be necessary to delay its effective date until the registrant shall file a further amendment that specifically states that this registration statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933, as amended, or until the registration statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

Explanatory note

This Amendment No. 1 ("Amendment No. 1") to the Registration Statement on Form S-1 ("Registration Statement") is being filed solely for the purpose of filing Exhibit 10.1 and Exhibit 10.4 as indicated in Part II. This Amendment No. 1 does not modify any provision of the prospectus that forms a part of the Registration Statement, and accordingly, such prospectus has been omitted.

Part II

Information not required in prospectus

Item 13. Other expenses of issuance and distribution.

The following table indicates the expenses to be incurred in connection with the offering described in this registration statement, other than underwriting discounts and commissions, all of which will be paid by us. All amounts are estimated except the Securities and Exchange Commission registration fee, the Financial Industry Regulatory Authority, Inc., or FINRA, filing fee and the Nasdaq listing fee.

	Amount
Securities and Exchange Commission registration fee	\$ 12,210
FINRA filing fee	14,850
Nasdaq initial listing fee	*
Legal fees and expenses	*
Accounting fees and expenses	*
Blue sky fees and expenses	*
Printing and engraving expenses	*
Transfer agent fees and expenses	*
Miscellaneous	*
Total expenses	\$ *

^{*} To be filed by amendment.

Item 14. Indemnification of directors and officers.

Section 102 of the General Corporation Law of the State of Delaware permits a corporation to eliminate the personal liability of directors of a corporation to the corporation or its stockholders for monetary damages for a breach of fiduciary duty as a director, except where the director breached his or her duty of loyalty, failed to act in good faith, engaged in intentional misconduct or knowingly violated a law, authorized the payment of a dividend or approved a stock repurchase in violation of the General Corporation Law of the State of Delaware or obtained an improper personal benefit. Our amended and restated certificate of incorporation will provide that none of our directors shall be personally liable to us or to our stockholders for monetary damages for any breach of fiduciary duty as a director, notwithstanding any provision of law imposing such liability, except to the extent that the General Corporation Law of the State of Delaware prohibits the elimination or limitation of liability of directors for breaches of fiduciary duty.

Section 145 of the General Corporation Law of the State of Delaware provides that a corporation has the power to indemnify a director, officer, employee or agent of the corporation, or a person serving at the request of the corporation for another corporation, partnership, joint venture, trust or other enterprise in related capacities, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by the person in connection with an action, suit or proceeding to which he or she was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding by reason of such position, if such person acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the corporation and, in any criminal action or proceeding, had no reasonable cause to believe his or her conduct was unlawful, except that, in the case of actions brought by or in the right of the corporation, no indemnification shall be made with respect to any claim, issue or matter as to which such person shall have been adjudged to be liable to the corporation unless and only to the extent that the Court of Chancery or other adjudicating court determines that, despite the adjudication of liability but in

view of all of the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

Our amended and restated bylaws will provide that we will indemnify each person who was or is a party or threatened to be made a party to any threatened, pending or completed action, suit or proceeding (other than an action by or in the right of our company) by reason of the fact that he or she is or was, or has agreed to become, a director or officer, or is or was serving, or has agreed to serve, at our request as a director, officer, partner, employee or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (all such persons being referred to as an "Indemnitee"), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred in connection with such action, suit or proceeding and any appeal therefrom, if such Indemnitee acted in good faith and in a manner he or she reasonably believed to be in, or not opposed to, our best interests, and, with respect to any criminal action or proceeding, he or she had no reasonable cause to believe his or her conduct was unlawful. Our amended and restated certificate of incorporation and amended and restated bylaws will provide that we will indemnify any Indemnitee who was or is a party to an action or suit by or in the right of our company to procure a judgment in our favor by reason of the fact that the Indemnitee is or was, or has agreed to become, a director or officer, or is or was serving, or has agreed to serve, at our request as a director, officer, partner, employee or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise, or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees) and, to the extent permitted by law, amounts paid in settlement actually and reasonably incurred in connection with such action, suit or proceeding, and any appeal therefrom, if the Indemnitee acted in good faith and in a manner he or she reasonably believed to be in, or not opposed to, our best interests, except that no indemnification shall be made with respect to any claim, issue or matter as to which such person shall have been adjudged to be liable to us, unless a court determines that, despite such adjudication but in view of all of the circumstances, he or she is entitled to indemnification of such expenses. Notwithstanding the foregoing, to the extent that any Indemnitee has been successful, on the merits or otherwise, he or she will be indemnified by us against all expenses (including attorneys' fees) actually and reasonably incurred in connection therewith. Expenses must be advanced to an Indemnitee under certain circumstances.

We have entered into indemnification agreements with each of our directors and executive officers. These indemnification agreements may require us, among other things, to indemnify our directors and executive officers for expenses, including attorneys' fees, judgments, fines and settlement amounts, incurred by a director or executive officer in any action or proceeding arising out of his or her service as one of our directors or executive officers, or any of our subsidiaries or any other company or enterprise to which the person provides services at our request.

We maintain a general liability insurance policy that covers certain liabilities of directors and officers of our corporation arising out of claims based on acts or omissions in their capacities as directors or officers.

In any underwriting agreement we enter into in connection with the sale of common stock being registered hereby, the underwriters will agree to indemnify, under certain conditions, us, our directors, our officers and persons who control us within the meaning of the Securities Act of 1933, as amended, or the Securities Act, against certain liabilities.

Item 15. Recent sales of unregistered securities.

Set forth below is information regarding shares of capital stock issued by us within the past three years. Also included is the consideration received by us for such shares and information relating to the section of the

Securities Act, or rule of the Securities and Exchange Commission, under which exemption from registration was claimed.

(a) Issuances of Capital Stock and Securities Convertible into Capital Stock.

From May 2018 to July 2018, we issued 21,956,095 shares of Series B Preferred Stock to certain accredited investors at a price of \$5.01 per share for aggregate proceeds of approximately \$110.0 million.

In March 2019, we entered into a note purchase agreement with certain individual and institutional accredited investors, pursuant to which we sold and issued, or agreed to sell and issue, approximately \$39.6 million aggregate principal amount of convertible promissory notes in exchange for aggregate cash proceeds of approximately \$39.6 million.

(b) Equity Grants and Issuances under Stock Incentive Plans.

Since January 1, 2016, we have granted stock option and restricted stock awards to purchase an aggregate of 13,042,000 shares of our common stock to employees, consultants and directors under our 2015 Stock Incentive Plan with exercise or purchase prices ranging between \$0.19 and \$6.46 per share, and we have issued 850,563 shares of restricted common stock to employees, consultants and directors under our 2015 Stock Incentive Plan. In addition, since January 1, 2016, we have also issued 682,911 shares of restricted common stock to employees, consultants and directors in connection with the exercise of stock options granted under our 2006 Stock Incentive Plan.

The issuances of the above securities were deemed to be exempt from registration under the Securities Act in reliance upon Section 4(a)(2) of the Securities Act or Regulation D promulgated thereunder, or Rule 701 promulgated under Section 3(b) of the Securities Act, as transactions by an issuer not involving any public offering or pursuant to benefit plans and contracts relating to compensation as provided under Rule 701. Individuals who purchased securities as described above represented their intention to acquire the securities for investment only and not with a view to or for sale in connection with any distribution thereof, and appropriate legends were affixed to the share certificates issued in such transactions.

None of the foregoing transactions involved any underwriters, underwriting discounts or commissions or any public offering.

Item 16. Exhibits and financial statement schedules.

(a) Exhibits.

Exhibit number	Description
1.1*	Form of Underwriting Agreement
3.1**	Amended and Restated Certificate of Incorporation of the Registrant, as currently in effect
3.2*	Form of Amended and Restated Certificate of Incorporation of the Registrant, to be effective upon the closing of this offering
3.3**	Amended and Restated By-laws of the Registrant, as currently in effect
3.4*	Form of Amended and Restated Bylaws of the Registrant, to be effective upon the closing of this offering
4.1*	Specimen Common Stock Certificate
4.2*	Amended and Restated Investors' Rights Agreement among the Registrant and certain of its stockholders and convertible noteholders, dated May 25, 2018, as amended

Exhibit number	Description
5.1*	Opinion of Smith, Anderson, Blount, Dorsett, Mitchell & Jernigan, LLP
10.1†	Development and Commercial License Agreement by and between Les Laboratoires Servier and the Registrant, dated February 24, 2016, as amended
10.2†**	License Agreement by and between Duke University and the Registrant, dated April 17, 2006, as amended
10.3†**	Patent Cross-License Agreement by and between Cellectis SA and the Registrant, dated January 23, 2014
10.4†	Collaboration and License Agreement by and between Gilead Sciences, Inc. and the Registrant, dated September 10, 2018
10.5*	Lease Agreement between the Registrant and VC Owner, dated April 5, 2010, as amended
10.6**	Lease Agreement between Elo Life Systems, Inc. and ARE-NC Region No. 17, LLC, dated March 29, 2018, as amended
10.7**	Lease Agreement between Registrant and Durham TW Alexander, LLC, dated October 2, 2018
10.8**	2006 Stock Incentive Plan, as amended, and form of award agreements thereunder
10.9**	2015 Stock Incentive Plan, as amended, and form of award agreements thereunder
10.10*	2019 Incentive Award Plan, and form of award agreements thereunder
10.11*	2019 Employee Stock Purchase Plan
10.12*	Employment Agreement between the Registrant and Matthew Kane, dated
10.13*	Employment Agreement between the Registrant and Derek Jantz, dated
10.14*	Employment Agreement between the Registrant and Abid Ansari, dated
10.15*	Employment Agreement between the Registrant and David Thomson, dated
10.16*	Employment Agreement between the Registrant and Fayaz Khazi, dated
10.17*	Form of Indemnification Agreement between the Registrant and its directors and officers
10.18*	Non-Employee Director Compensation Plan
21.1**	Subsidiaries of the Registrant
23.1	Consent of Deloitte & Touche LLP
23.2*	Consent of Smith, Anderson, Blount, Dorsett, Mitchell & Jernigan, LLP (included as part of Exhibit 5.1)
24.1	Power of Attorney (included on signature page)

^{*} To be filed by amendment.

^{**} Previously filed.

[†] Portions of this exhibit (indicated by asterisks) have been omitted pursuant to a request for confidential treatment pursuant to Rule 406 under the Securities Act of 1933, as amended, and have been filed separately with the Securities and Exchange Commission.

⁽b) Financial Statement Schedules. Schedules not listed above have been omitted because the information required to be set forth therein is not applicable or is shown in the consolidated financial statements or notes thereto.

Item 17. Undertakings.

The undersigned registrant hereby undertakes to provide to the underwriter, at the closing specified in the underwriting agreement, certificates in such denominations and registered in such names as required by the underwriter to permit prompt delivery to each purchaser.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

The undersigned hereby undertakes that:

- (1) For purposes of determining any liability under the Securities Act, the information omitted from the form of prospectus filed as part of this registration statement in reliance upon Rule 430A and contained in a form of prospectus filed by the registrant pursuant to Rule 424(b) (1) or (4) or 497(h) under the Securities Act shall be deemed to be part of this registration statement as of the time it was declared effective.
- (2) For the purpose of determining any liability under the Securities Act, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

Signatures

Pursuant to the requirements of the Securities Act of 1933, the registrant has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Durham, State of North Carolina, on this 13th day of March, 2019.

PRECISION BIOSCIENCES, INC. (Registrant)

By: /s/ Matthew Kane

Matthew Kane

President and Chief Executive Officer

Power of attorney and signatures

We, the undersigned officers and directors of Precision BioSciences, Inc., hereby severally constitute and appoint Matthew Kane and Abid Ansari, and each of them singly (with full power to each of them to act alone), our true and lawful attorneys-in-fact and agents, with full power of substitution and resubstitution in each of them for him and in his name, place and stead, and in any and all capacities, to sign any and all amendments (including post-effective amendments) to this registration statement (or any other registration statement for the same offering that is to be effective upon filing pursuant to Rule 462(b) under the Securities Act of 1933), and to file the same, with all exhibits thereto and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite or necessary to be done in and about the premises, as full to all intents and purposes as he might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents or any of them, or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following persons in the capacities indicated.

Signature	Title	Date
/s/ Matthew Kane Matthew Kane	President, Chief Executive Officer and Director (Principal Executive Officer)	March 13, 2019
/s/ Abid Ansari Abid Ansari	Chief Financial Officer (Principal Financial and Accounting Officer)	March 13, 2019
* Robert Adelman, M.D.	Director	March 13, 2019
* Derek Jantz, Ph.D.	Director	March 13, 2019
/s/ Raymond Schinazi Raymond Schinazi, Ph.D.	Director	March 13, 2019

Signature	Title	Date
* Shalini Sharp	Director	March 13, 2019
* Tony Yao, M.D., Ph.D.	Director	March 13, 2019

*By: /s/ Abid Ansari Attorney-in-fact

DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT

By and Among

BAXALTA INCORPORATED,

BAXALTA US INC.,

BAXALTA GMBH,

and

PRECISION BIOSCIENCES, INC.

[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

$\frac{\textbf{FINAL EXECUTION VERSION}}{\textbf{CONFIDENTIAL}}$

TABLE OF CONTENTS

ARTICLE I.	DEFINITIONS	Page 1
ARTICLE II.	DISCOVERY AND DEVELOPMENT	20
2.1 2.2 2.3 2.4 2.5 2.6 2.7	Selection of Included Targets Development Plans; Development Obligations Development of Unselected Candidates Inactive Targets; Precision Targets Development Licenses Isolex Co-Development Know-How Database	20 24 26 27 29 30 30
ARTICLE III.	GOVERNANCE	31
3.1 3.2	Joint Steering Committee Alliance Managers	31 35
ARTICLE IV.	COMMERCIAL OPTIONS; LICENSES	35
4.1 4.2 4.3 4.4 4.5	Commercial Option Development Transfer Licenses to Baxalta Sublicenses No Other Licenses	35 37 37 38 38
ARTICLE V.	REGULATORY MATTERS	39
5.1 5.2 5.3 5.4 5.5 5.6	Regulatory Roles Cooperation; Effort Right of Reference Adverse Event Reporting Notification of Threatened Action Remedial Actions	39 40 41 41 41 41
ARTICLE VI.	MANUFACTURING AND SUPPLY	42
6.1 6.2 6.3	Manufacturing and Supply Roles Supply Agreement Quality Agreement	42 43 44
ARTICLE VII.	COMMERCIALIZATION OF LICENSED PRODUCTS	44
7.1 7.2 7.3	Commercialization Responsibility Diligence Co-Development and Co-Promote Option	44 44 44

TABLE OF CONTENTS

(continued)

ARTICLE VIII.	PAYMENTS	Page 46
8.1	Upfront Fee	46
8.2	Commercial Option Exercise Fee	46
8.3	Milestones	46
8.4	Royalty Payment; Audits	49
8.5	Taxes	51
8.6	Late Payments	51
8.7	Reporting	51
8.8	Currency; Exchange Rate	51
ARTICLE IX.	INVENTIONS; ACCESS TO IMPROVEMENTS; PATENTS; TRADEMARKS	52
9.1	Ownership	52
9.2	Prosecution of Patents	53
9.3	Licenses to Enabling Technologies	55
9.4	Duke IP	56
9.5	Infringement of Patents by Third Parties	56
9.6	Infringement of Third Party Rights in the Territory	58
9.7	Patent Oppositions and Other Proceedings	59
9.8	Patent Term Extensions in the Territory	60
9.9	Registration of License	61
9.10	[***]	61
9.11	Trademarks	61
9.12	Patent Marking	61
ARTICLE X.	REPRESENTATIONS AND WARRANTIES	62
10.1	The Parties' Representations and Warranties	62
10.2	Precision's Representations and Warranties	62
ARTICLE XI.	CERTAIN COVENANTS	64
11.1	Covenants	64
11.2	Dartistics Comments	(5

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

TABLE OF CONTENTS

(continued)

ARTICLE XII.	CONFIDENTIALITY	Page 67
12.1 12.2 12.3 12.4	Confidentiality Obligations of Baxalta Confidentiality Obligations of Precision Scientific Publication Press Releases; Publicity	67 68 70 70
ARTICLE XIII.	INDEMNIFICATION	72
13.1 13.2 13.3 13.4	Precision Indemnity Baxalta Indemnity Indemnification Procedure Limitation of Liability; Exclusion of Damages; Disclaimer	72 73 73 75
ARTICLE XIV.	TERM; TERMINATION	76
14.1 14.2 14.3	Term Early Termination Effects of Termination	76 76 77
ARTICLE XV.	DISPUTE RESOLUTION	83
15.1 15.2	Resolution by Executive Officers; Mediation Arbitration	83 83
ARTICLE XVI.	CERTAIN ADDITIONAL MATTERS	86
16.1 16.2 16.3 16.4	Change of Control of Precision Effect of Final Determination of [***] Effect of Final Determination of [***] Other Remedies	86 87 87 88
ARTICLE XVII.	GENERAL PROVISIONS	88
17.1 17.2 17.3 17.4 17.5	Assignment Allocation of Costs Headings; Rules of Construction No Implied Waiver Notices	88 89 89 89
17.6 17.7	Severability Entire Agreement	90 91

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

TABLE OF CONTENTS

(continued)

		Page
17.8	Amendment; Waiver	91
17.9	Counterparts	91
17.10	Agency	91
17.11	Further Actions	91
17.12	Compliance with Laws	91
17.13	Governing Law; Jurisdiction	91
17.14	Rights in Bankruptcy	92
17.15	Force Majeure	93
17.16	Consideration	94
17.17	Antitrust	94

EXHIBITS

EXHIBIT A: Duke IP EXHIBIT B: Supply Agreement Term Sheet EXHIBIT C: Quality Agreement Term Sheet Form CDCP Agreement Term Sheet EXHIBIT D: EXHIBIT E: Precision Patents and Precision Platform Patents EXHIBIT F: Press Release EXHIBIT G: Initial Summary CD19 Development Plan EXHIBIT H: Precision Wire Instructions

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT

This DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT (together with the exhibits and schedules hereto, as may be amended pursuant hereto, collectively, this "<u>Agreement</u>") is entered into on this 24th day of February, 2016 (the "<u>Effective Date</u>"), by and among BAXALTA INCORPORATED, a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("<u>BUSI</u>"), BAXALTA US INC., a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("<u>BUSI</u>"), BAXALTA GMBH, a company organized under the laws of Switzerland with its principal place of business at Postfach 8010, Zurich, Switzerland ("<u>BGMBH</u>" and, together jointly and severally with BI and BUSI, collectively, "<u>Baxalta</u>"), and PRECISION BIOSCIENCES, INC., a Delaware corporation with its principal place of business at 302 E Pettigrew St A-100, Durham, NC 27701 ("<u>Precision</u>"). Baxalta and Precision may each be referred to herein individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>."

RECITALS

WHEREAS, Precision has expertise in and owns or controls proprietary nuclease-based genome engineering technology (known as ARCUSTM) and other technologies relating to genetically engineering novel human T cells with chimeric antigen receptors for allogeneic use (as further defined below, "CAR-T") to enable such cells to target and destroy cancer cells; and

WHEREAS, Baxalta is a global biopharmaceutical company that clinically develops and commercializes products in oncology and other therapeutic areas; and

WHEREAS, Precision and Baxalta desire to establish a collaboration whereby Precision conducts research and development activities on specified CAR-T product candidates directed to antigen targets selected by Baxalta and, if successful, Baxalta has the Commercial Option (as defined below) to conduct further clinical development and commercialization of such product candidates for the treatment of cancer and other indications in humans, all under the terms and conditions set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the mutual promises and covenants set forth below and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

ARTICLE I. DEFINITIONS

The following terms as used in this Agreement shall have the meanings set forth in this Article I:

1.1 "Acquirer" means, collectively, (a) any Third Party that acquires Precision after the Effective Date (whether by transfer or sale of all or any portion of Precision's assets, equity or business, or by a merger, consolidation or similar business combination transaction or otherwise) and (b) the Affiliates of such Third Party, but excluding Precision and Precision's Affiliates existing immediately prior to the closing of such acquisition of Precision.

- 1.2 "Affiliate" means, with respect to a particular Person, any other Person that directly or indirectly is controlled by, controls or is under common control with such Person. For the purposes of this definition only, the term "control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to a particular Person that is an entity, means (a) if such Person is a corporate entity, direct or indirect ownership of voting securities entitled to cast at least fifty percent (50%) of the votes in the election of directors of such Person, (b) if such Person is a non-corporate entity, direct or indirect ownership of at least fifty percent (50%) of the equity interests of such Person or (c) the power to direct the management and policies of such Person.
 - 1.3 "Agents" has the meaning set forth in Section 13.1.
 - 1.4 "Agreement" has the meaning set forth in the preamble.
 - 1.5 "Alliance Manager" has the meaning set forth in Section 3.2.
- 1.6 "Allogeneic" means the treatment of a patient with human cells derived from a donor other than the patient, where such donor is genetically non-identical. "Allogeneic" does not include the treatment of a patient with human cells derived from such patient's own cells (autologous treatment).
 - 1.7 "Antitrust Laws" has the meaning set forth in Section 17.17.2.
- 1.8 "Antitrust Clearance" means, with respect to any Commercial Option, (a) that any applicable waiting periods (and any extensions thereof) for exercise of such Commercial Option under the HSR Act and any other applicable Antitrust Laws will have expired or otherwise been terminated and (b) that there exists (i) no requirement for a waiver, consent or approval of the FTC, the DOJ or any other applicable Governmental Authority with respect to the exercise of such Commercial Option, (ii) no judgment, injunction, decree or any other order in any suit or proceeding preventing the exercise of such Commercial Option, and (iii) no other matter relating to actions under any Antitrust Laws that would preclude, impede or delay materially the grant of the rights and licenses that would be granted upon the exercise of such Commercial Option.
 - 1.9 "Background IP" has the meaning set forth in Section 9.1.1.
 - 1.10 "Bankruptcy Laws" has the meaning set forth in Section 17.14.1.
 - 1.11 "Batch Success Achievement Plan" has the meaning set forth in Section 4.1.3.
 - 1.12 "Baxalta" has the meaning set forth in the preamble.
 - 1.13 "Baxalta Confidential Information" has the meaning set forth in Section 12.2.1(a).

- 1.14 "Baxalta-Developed Included Target" has the meaning set forth in Section 16.3.
- 1.15 "Baxalta-Developed Licensed Product" has the meaning set forth in Section 16.2.
- 1.16 "Baxalta-Developed Licensed Product Candidate" has the meaning set forth in Section 16.2.
- 1.17 "Baxalta Indemnitees" has the meaning set forth in Section 13.1.
- 1.18 "Baxalta Invention" has the meaning set forth in Section 9.1.3(b).
- 1.19 "Baxalta Know-How" means the Know-How Controlled by Baxalta or any of its Affiliates as of the Effective Date, or that comes to be Controlled by Baxalta or any of its Affiliates at any time during the Term that is reasonably necessary or useful to Develop, manufacture, use and Commercialize Licensed Product Candidates or Licensed Product(s) in the Field, but excluding Know-How Controlled by Baxalta specifically related to the Isolex Platform Technology. Baxalta Know-How shall include Baxalta's interest in any Joint Inventions. For clarity, Baxalta Know-How does not include Patent Rights claiming Baxalta Know-How.
 - 1.20 "Baxalta Parent Group" has the meaning set forth in Section 9.10.
- 1.21 "Baxalta Patents" means the Patent Rights Controlled by Baxalta or any of its Affiliates as of the Effective Date, or that come to be Controlled by Baxalta or any of its Affiliates at any time during the Term, that claim Baxalta Know-How. Baxalta Patents shall include Baxalta's interest in any Joint Patents. Baxalta Patents do not include Isolex Platform Patents.
 - 1.22 "Baxalta Restrictive Covenants" has the meaning set forth in Section 11.2.2(a).
 - 1.23 "BI" has the meaning set forth in the preamble.
 - 1.24 "BGMBH" has the meaning set forth in the preamble.
 - 1.25 [***]
 - 1.26 "BUSI" has the meaning set forth in the preamble.
- 1.27 "Business Day" means a day other than Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required by applicable Law to close.
- 1.28 "Calendar Quarter" means the successive periods of three (3) consecutive calendar months ending on March 31, June 30, September 30 or December 31, for so long as this Agreement is in effect.
- [***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.29 "Calendar Year" means any year beginning on January 1 and ending on December 31 of such year.
- 1.30 "Candidate Proposal Date" has the meaning set forth in Section 2.1.5(a).
- 1.31 "CAR-T" means genetically engineered human T cells with Chimeric Antigen Receptor(s) for Allogeneic use.
- 1.32 "CD19" has the meaning set forth in Section 2.1.2.
- 1.33 "CDCP Agreement" has the meaning set forth in Section 7.3.2(a).
- 1.34 "CDCP Option" has the meaning set forth in Section 7.3.1.
- 1.35 "CDCP Option Fee" has the meaning set forth in Section 7.3.2(b).
- 1.36 "Cellectis Agreement" has the meaning set forth in Section 4.5.
- 1.37 "Cellectis Patents" has the meaning set forth in Section 4.5.
- 1.38 "Change" means any change with respect to manufacturing, quality system or validation/qualification status, including the list of changes set forth in the Quality Agreement.
- 1.39 "<u>Change of Control of Precision</u>" means the occurrence of any of the following: (i) a sale of all or substantially all of the assets of Precision to a Third Party, (ii) a Third Party acquires beneficial ownership of more than fifty percent (50%) of the stock of Precision or stock possessing fifty percent (50%) or more of the total voting power of the stock of Precision, or (iii) the merger, consolidation or similar business combination transaction of Precision into a Third Party.
- 1.40 "<u>Chimeric Antigen Receptor</u>" means a genetically engineered molecule that (a) when present on the surface of human T cells, enables the T cells to recognize and bind to specific antigens that are present on the surface of cells, and (b) comprises a single-chain antibody fragment (scFv), a transmembrane domain, and at least one intracellular signaling domain.
- 1.41 "Clinical Trial" means a clinical study conducted on certain numbers of human subjects (depending on the phase of the trial) that is designed to (a) establish that a product for the treatment of human diseases and conditions is reasonably safe for continued testing, (b) investigate the safety and efficacy of the product for its intended use, and to define warnings, precautions and adverse reactions that may be associated with the product in the dosage range to be prescribed, or (c) support Marketing Approval or Reimbursement Approval of such product or label expansion of such product.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.42 "CMO" means contract manufacturing organization.
- 1.43 "Code" has the meaning set forth in Section 8.5.
- 1.44 "Combination Product" has the meaning set forth in Section 1.132.3.
- 1.45 "Combination Target" has the meaning set forth in Section 2.1.1.
- 1.46 "Commercial License" has the meaning set forth in Section 4.3.1.
- 1.47 "Commercial License Fee" has the meaning set forth in Section 8.2.
- 1.48 "Commercial Option" has the meaning set forth in Section 4.1.1.
- 1.49 [***]
- 1.50 "Commercial Option Exercise Notice" has the meaning set forth in Section 4.1.2.
- 1.51 "Commercial Option Exercise Period" has the meaning set forth in Section 4.1.1.
- 1.52 "Commercialization" means all activities related to the commercial exploitation of products for the treatment of human diseases and conditions, including manufacturing, importation, exportation, marketing, Promotion, distribution, pre-launch, launch, sale, and offering for sale of such products. When used as a verb, "Commercialize" or "Commercializing" means to engage in Commercialization.
 - 1.53 "Commercialization Enabling Technology" has the meaning set forth in Section 9.3.2.
 - 1.54 "Commercially Reasonable Efforts" means:
- 1.54.1 with respect to the obligations of a Party under this Agreement relating to Development activities, the level of efforts and expenditure of resources required to carry out such obligation in a sustained manner consistent with the efforts and resources such Party typically devotes to a product of similar market potential, resulting from its own research efforts or development and commercialization collaborations for which it is responsible, at a similar stage in its development or product life, taking into account Relevant Factors; or
 - 1.54.2 with respect to the level of obligations of a Party under this Agreement relating to Commercialization activities,
- (a) as it relates to Baxalta, the level of efforts and expenditure of resources required to carry out such obligation in a sustained manner consistent with the efforts and resources Baxalta typically devotes to a product of similar market potential, resulting from its own research efforts or development and commercialization collaborations for which it is responsible, at a similar stage in its development or product life, taking into account Relevant Factors;

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (b) as it relates to Precision, the level of efforts and expenditure of resources required to carry out such obligation in a sustained manner consistent with the efforts and resources of a typical Third Party biopharmaceutical company of similar size and with similar resources as Precision typically devotes to a product of similar market potential, at a similar stage in its development or product life, taking into account Relevant Factors; or
- 1.54.3 with respect to the obligations of a Party under this Agreement relating to any other objective, reasonable, good-faith efforts, taking into account industry practices.

Provided that, [***].

- 1.55 "Competitive Program" means [***].
- 1.56 "Competitor" means any Person, other than the Parties and their Affiliates, that is conducting any Competitive Program, for so long as such conduct continues.
 - 1.57 "Confidential Information" means Baxalta Confidential Information and Precision Confidential Information.
- 1.58 "Confidentiality Agreement" means that certain Mutual Confidential Disclosure Agreement between Baxter Healthcare Corporation and Precision dated October 6, 2014, which was assigned in its entirety by Baxter Healthcare Corporation to Baxalta effective as of July 1, 2015.
- 1.59 "Control" or "Controlled" means, with respect to any Know-How, Patent Rights or other intellectual property rights, that a Party has the legal authority or right (whether by ownership, license or otherwise) to grant a license, sublicense, access or right to use (as applicable) under such Know-How, Patent Rights, or other intellectual property rights, including to the other Party on the terms and conditions set forth herein, as applicable, in each case without breaching the terms of any agreement with a Third Party.
- 1.60 "Cover" or "Covering" means, with respect to Patent Rights and a particular Licensed Product, that such Patent Rights include one or more Valid Claims that would, but for the licenses granted under this Agreement be infringed by Development, manufacture, use or Commercialization of the applicable Licensed Product or the Licensed Product Candidate comprising such Licensed Product in the applicable country in which any such activity occurred.
 - 1.61 "CPR" has the meaning set forth in Section 15.1.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.62 "CPR Panels" has the meaning set forth in Section 15.1.
- 1.63 "Damages" has the meaning set forth in Section 13.1.
- 1.64 "Debtor Party" has the meaning set forth in Section 14.2.3.
- 1.65 "Defense Election Notice" has the meaning set forth in Section 13.3.1.
- 1.66 "Development" means all activities related to the development of products for the treatment of human diseases and conditions and obtaining Marketing Approval for such products, including all activities related to research, development, preclinical testing, preclinical toxicology, stability testing, toxicology, formulation, Clinical Trials, regulatory affairs, statistical analysis, report writing, manufacturing process scale up (including without limitation, registration batches/process validation, engineering studies qualification and validation, process validation, characterization and stability, scale and technology transfer to CMOs), qualification and validation activities, quality assurance/quality control development, Regulatory Filing creation and submission related to obtaining Marketing Approval for a product, and all other activities directed to obtaining Marketing Approval and/or Reimbursement Approval. When used as a verb, "Develop" means to engage in Development.
 - 1.67 "Development Achievement Notice" has the meaning set forth in Section 2.2.5.
 - 1.68 "Development Milestone" has the meaning set forth in Section 8.3.1.
- 1.69 "<u>Development Plan</u>" means each plan setting forth the specific activities to be undertaken by each of the Parties, as applicable, in connection with the Development of each Licensed Product Candidate and Licensed Product, as may be amended as set forth in this Agreement.
- 1.70 "<u>Directed to</u>" means, when used to describe the relationship between an engineered human T cell and a Target, that the T cell (a) is designed or being developed to bind to the Target (or a portion thereof), (b) is designed or being developed to exert its biological effect in whole or in part through binding to such Target (or such portion thereof), and (c) is not designed or developed to bind to or exert its effect on any other Target (or a portion thereof). Notwithstanding the foregoing, when used to describe the relationship between an engineered human T cell and a Combination Target, "Directed to" means that the T cell (i) is designed or being developed to bind to both of the Targets (or portions thereof) that comprise the Combination Target, (ii) is designed or being developed to exert its biological effect in whole or in part through binding to both such Targets (or such portions thereof), and (iii) is not designed or developed to bind to or exert its effect on any other Target (or a portion thereof).
- 1.71 "<u>Divest</u>" means, as it relates to a Competitive Program: (i) the sale of all right, title and interest in such Competitive Program, including all technology, intellectual property and other assets relating solely thereto, to a Third Party, without the retention or reservation of any rights, license or interest (other than an economic interest such as a right to receive payments) by the selling entity or its Affiliates; or (ii) the complete termination and/or shut-down of such Competitive Program such that no technology, intellectual property or other asset solely relating thereto is used by the terminating entity or its Affiliates for the conduct of such Competitive Program.

- 1.72 "DOJ" has the meaning set forth in Section 17.17.1.
- 1.73 "<u>Duke Agreement</u>" means the License Agreement entered into by Precision and Duke University ("<u>Duke</u>") on April 17, 2006, as amended from time to time.
- 1.74 "<u>Duke IP</u>" means all Patent Rights and Know-How licensed to Precision under the Duke Agreement that constitute Precision Platform Patents or Precision Platform Technology. The patent numbers and patent application numbers of the Patent Rights that are included within the Duke IP as of the Effective Date are set forth in <u>Exhibit A</u>.
 - 1.75 "Effective Date" has the meaning set forth in the preamble.
 - 1.76 "Election Time Period" has the meaning set forth in Section 13.3.1.
 - 1.77 "EMA" means the European Medicines Agency or any successor agency or agencies thereto.
 - 1.78 "Enforcing Party" has the meaning set forth in Section 9.5.3.
 - 1.79 "Escrow Agent" has the meaning set forth in Section 2.1.3(b).
 - 1.80 "European Union" or "EU" means all countries of the European Union, as may be included from time to time.
 - 1.81 "Executive Officers" has the meaning set forth in Section 3.1.6(b).
 - 1.82 "FCPA" means the Foreign Corrupt Practices Act, as amended (15 U.S.C. §§ 78dd-1, et. seq.).
 - 1.83 "FDA" means the U.S. Food and Drug Administration or any successor agency or agencies thereto.
 - 1.84 "FDCA" means the United States Food, Drug and Cosmetic Act, as amended (21 U.S.C. §§ 301, et. seq.).
 - 1.85 "Field" means all uses in humans.
 - 1.86 "Final Candidate Proposal Date" has the meaning set forth in Section 2.1.5(b).
- 1.87 "Final Determination" means, in connection with any disputed event or occurrence, the resolution of the dispute (i) pursuant to the dispute resolution provisions set forth in <u>Article XV</u> or (ii) by mutual agreement of the Parties.

1.88 [***]	
1.89 [***]	
1.90 [***]	

1.91 "First Commercial Sale" means, with respect to a particular Licensed Product in a particular country or other jurisdiction, the first arms'-length sale of such Licensed Product by Baxalta or any of its Affiliates or Sublicensees, in each case to a Third Party for use in the Field in such country or other jurisdiction after such Licensed Product has been granted (a) Marketing Approval and Reimbursement Approval if such sale is in any country in the EU or (b) Marketing Approval only if such sale is not in a country in the EU. For avoidance of doubt, no sale or other disposition of a Licensed Product shall be deemed the "First Commercial Sale" if such sale or other disposition would not be included in the calculation of Net Sales.

1.92 [***]
1.93 [***]
1.94 [***]
1.95 "Force Majeure" has the meaning set forth in Section 17.15.1.
1.96 "FTC" has the meaning set forth in Section 17.17.1.

1.97 "Global Dossier" means, with respect to a particular Licensed Product, a set of documents that follows the principles of the International Conference on Harmonization and that contains all of the technical data (including quality, non-clinical and clinical data) and administrative information of a product intended for the treatment of human disease and conditions necessary for such product to be approved or registered for Commercialization in the countries and regions in the Territory. The Global Dossier must include data proving that the Licensed Products have quality, efficacy and safety properties suitable for the applicable intended uses, as well as additional administrative documents that follow applicable local and regional guidance.

1.98 "Governmental Authority" means any nation or government, any state, local or other political subdivision thereof, and any entity, department, commission, bureau, agency, authority, board, court, official or officer, domestic or foreign, exercising executive, judicial, regulatory or administrative governmental functions.

1.99 "Hourly Rate" has the meaning set forth in Section 6.1.3.

1.100 "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.101 "Inactive Target" has the meaning set forth in Section 2.4.1.
- 1.102 "Inactive Target Evaluation Data" has the meaning set forth in Section 2.4.3(a).
- 1.103 "Inactive Target Freeze Period" has the meaning set forth in Section 2.4.3(d).
- 1.104 "Inactive Target Non-Disclosure Period" has the meaning set forth in Section 2.4.3(b).
- 1.105 "Included Target" means a Target that has been Nominated by Baxalta and has become an Included Target pursuant to Section 2.1.2 or Section 2.1.4 of this Agreement.
 - 1.106 "Included Target-Specific Information" means with respect to an Included Target [***].
- 1.107 "IND" means any investigational new drug application, clinical trial application, clinical trial exemption or similar or equivalent application or submission (including investigator-initiated applications) for approval to conduct human Clinical Trials filed with or submitted to a Regulatory Authority in conformance with the requirements of such Regulatory Authority.
 - 1.108 [***]
 - 1.109 "Indemnification Claim Notice" has the meaning set forth in Section 13.3.1.
 - 1.110 "Indemnified Party" has the meaning set forth in Section 13.3.1.
 - 1.111 "Indemnifying Party" has the meaning set forth in Section 13.3.1.
 - 1.112 "Indemnitee" means, as the context requires, the Precision Indemnitees and/or the Baxalta Indemnitees.
- 1.113 "<u>Independently Active Therapeutic Ingredient</u>" means, with respect to a Combination Product, an active therapeutic ingredient having a different Target or mode of action, or which is otherwise treated or designated by the applicable Regulatory Authority as a separate active ingredient, than the applicable Licensed Product.
 - 1.114 "Initial CD19 Licensed Product Candidates" means [***] Directed to CD19 [***].
- 1.115 "Initiate" or "Initiation" means, with respect to a Clinical Trial of a Licensed Product Candidate or Licensed Product, the first dosing of the first patient for such Clinical Trial.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.116 "<u>Invention</u>" means any and all discoveries, developments, improvements, modifications, formulations, compositions of matter, processes and other inventions (whether patentable or not patentable) that are invented (as determined in accordance with U.S. patent laws) in the course of activities performed under this Agreement by or on behalf of either Party or both Parties, including all rights, title and interest in and to the intellectual property rights therein.
- 1.117 "Isolex Platform Patents" means the Patent Rights Controlled by Baxalta as of the Effective Date, or that come to be Controlled by Baxalta at any time during the Term, that claim the Isolex Platform Technology.
- 1.118 "Isolex Platform Technology" means the technology Controlled by Baxalta as of the Effective Date, or that come to be Controlled by Baxalta at any time during the Term, relating to the Isolex cell separator machine, disposables, reagents and methods of use thereof.
 - 1.119 "Joint Inventions" has the meaning set forth in Section 9.1.2.
 - 1.120 "Joint Patents" has the meaning set forth in Section 9.1.2.
 - 1.121 "JSC" has the meaning set forth in Section 3.1.1.
- 1.122 "Know-How" means techniques, data, inventions, practices, methods, trade secrets, knowledge, sources of supply, patent positioning, know-how, skill, experience, test data (including manufacturing, pharmacological, toxicological, preclinical and clinical test data) and analytical and quality control data or descriptions, including all proprietary information submitted to relevant Regulatory Authorities to support an application for Marketing Approval or an application for Reimbursement Approval, and in each case in written, oral, electronic or other form.
 - 1.123 "Launch Milestone" has the meaning set forth in Section 8.3.2.
- 1.124 "Law" means all laws, statutes, rules, regulations, ordinances, orders, judgments and other pronouncements having the effect of law of any federal, national, multinational, state, provincial, county, city or other political subdivision, domestic or foreign, including all such laws, statutes, rules, regulations, ordinances, orders, judgments and other pronouncements pertaining to the pharmaceutical industry or the healthcare industry and all anti-bribery or anti-corruption laws, including the FDCA and the FCPA and their implementing regulations and all foreign equivalents thereof.
- 1.125 "<u>Licensed Product</u>" means a Licensed Product Candidate for which the Commercial Option Exercise Date has occurred, including in any preparation, formulation, dosage, packaging or method of administration.
 - 1.126 "Licensed Product Candidate" means each Variant Directed to an Included Target that is the subject of a Development Plan.
 - 1.127 "Licensed Product Marks" has the meaning set forth in Section 9.11.
 - 1.128 "Litigation Conditions" has the meaning set forth in Section 13.3.1.

- 1.129 "Major EU Countries" means [***].
- 1.130 "Manufacturing and Supply Role" has the meaning set forth in Section 6.1.
- 1.131 "Marketing Approval" means, with respect to a particular product for the treatment of human disease and conditions in a particular country or regulatory jurisdiction, the registrations, authorizations and approvals of the applicable Regulatory Authority or other Governmental Authority in such country or regulatory jurisdiction (including, but not limited to, the FDA and EMA) that are necessary to market, sell or otherwise Commercialize such product in such country or regulatory jurisdiction.

1.132 Net Sales Definition.

- 1.132.1 "Net Sales" means, with respect to a particular Licensed Product in a country in the Territory, the gross revenues invoiced for sales of such Licensed Product by Baxalta or any of its Affiliates or Sublicensees in connection with the sale, lease or other transfer for value to a Third Party in such country in a *bona fide* arm's length transaction, less the following deductions from such gross revenues, in each case to the extent attributable to such Licensed Product and to the extent actually incurred or reasonably accrued and to the extent not already deducted in the amount invoiced:
 - (a) trade and quantity and/or cash discounts actually allowed or taken;
- (b) governmental customs, duties, sales, withholding and similar taxes (including, for the avoidance of doubt value added or import/export taxes, sales taxes and excise taxes but excluding taxes based on income), if any, imposed on the Licensed Product, to the extent directly related to such sale; as well as any amount of branded prescription pharmaceutical fee allocable to the sale of Licensed Products and paid by Baxalta, its Affiliates or Sublicensees under section 9008 of the Affordable Care Act (P.L. 111-148), as amended;
- (c) amounts actually allowed or credited by reason of rejections, return of goods (including as a result of recalls), any retroactive price reductions or allowances specifically identifiable as relating to the Licensed Product (including those resulting from inventory management or similar agreements with wholesalers);
 - (d) amounts incurred resulting from government-mandated rebate programs, including programs mandated by any agency thereof;
 - (e) rebates actually given to a Third Party specifically for the Licensed Product;

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (f) freight, postage, shipping and applicable insurance charges, to the extent same are separately itemized in the invoice price and charged to the buyer;
 - (g) patient discount programs and chargebacks or similar price concessions related to the sale of the Licensed Product; and
- (h) to the extent agreed by the Parties in writing, such agreement not to be unreasonably withheld, any other specifically identifiable appropriate allowances or deductions that were actually credited as may be similar to those deductions listed above;

provided, however, that any rebate to Baxalta, gift, excess payment on other compounds or similar compensation received by Baxalta from a Third Party whether in the applicable country or any other and whether intended to be applicable to the Licensed Product or not shall be added to Net Sales.

- 1.132.2 The Parties agree that the following shall not be considered Net Sales for purposes of Section 1.132.1: (a) the transfer of Licensed Product (i) between Baxalta or its Affiliates, on one hand, and Baxalta's Sublicensees, on the other hand, for resale (which resale will give rise to Net Sales), or (ii) between or among Baxalta and its Affiliates or between Baxalta's Affiliates, in each case unless such Affiliate is the end user of such Licensed Product; (b) use of Licensed Product in a preclinical trial or a Clinical Trial; (c) use of Licensed Product as free marketing samples; or (d) the transfer of Licensed Product by Baxalta or its Affiliates or Sublicensees to a Third Party as a sale or donation for charitable or compassionate use purposes.
- 1.132.3 In the event a Licensed Product is co-packaged, co-formulated or otherwise sold in a manner that includes one or more Independently Active Therapeutic Ingredients in addition to a Licensed Product (such Licensed Product, a "Combination Product"), then Net Sales, for purposes of determining royalty payments under Section 8.4 on such Combination Product, will be calculated by [***].
- 1.132.4 All calculations set forth in this <u>Section 1.132</u> shall be determined in accordance with U.S. GAAP on a basis consistent with Baxalta's annual audited financial statements.
- 1.133 "Nominate" or "Nomination" means delivery by Baxalta of a written notice to Precision indicating that Baxalta desires to designate a Target as an Included Target.
- 1.134 "Nomination Period" means the period commencing on the Effective Date and ending on the fourth (4th) anniversary of the Effective Date or such later date to which such period has been extended pursuant to Section 2.1.6, if applicable; provided, however, that the Nomination Period shall terminate immediately on the first date upon which there are six (6) Included Targets.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.135 "Non-Debtor Party" has the meaning set forth in Section 14.2.3.
- 1.136 "Other Enabling Technology" has the meaning set forth in Section 9.3.3.
- 1.137 "Party" and "Parties" has the meaning set forth in the preamble.
- 1.138 "Patent" means (a) unexpired and currently in force letters patent (or other equivalent legal instrument), including utility and design patents, and including any extension, substitution, registration, confirmation, reissue, re-examination or renewal thereof, (b) applications for letters patent, a reissue application, a continuation application, a continuation-in-part application, a divisional application or any equivalent of the foregoing applications, that are pending before a government patent authority and (c) all foreign or international equivalents of any of the foregoing in any country.
 - 1.139 "Patent Challenge" has the meaning set forth in Section 14.2.2.
 - 1.140 "Patent Rights" means all rights in, to and under Patents.
 - 1.141 "Patent Term Extensions" has the meaning set forth in Section 9.8.
 - 1.142 "Paying Party" has the meaning set forth in Section 8.5.
- 1.143 "Person" means any individual, firm, corporation, partnership, limited liability company, trust, business trust, joint venture, Governmental Authority, association or other entity.
- 1.144 "Phase I Clinical Trial" means a Clinical Trial in which a product is administered to human subjects with the primary purpose of determining (as appropriate) safety and pharmacokinetic and pharmacodynamic properties of the product, and which is consistent with 21 C.F.R. § 312.21(a) or equivalent regulation in countries other than the US.
 - 1.145 "Phase I Notice" has the meaning set forth in Section 2.2.4.
 - 1.146 [***]
- 1.147 "Phase II Clinical Trial" means a Clinical Trial that would satisfy the requirements of 21 C.F.R. § 312.21(b) or equivalent regulation in countries other than the US.
 - 1.148 "Phase II Ready Batch" means the first batch of Phase II Clinical Trial material for the applicable Licensed Product Candidate [***].
- 1.149 "Phase II Ready Batch Success" means, with respect to any Licensed Product Candidate, that Precision has delivered to Baxalta a Phase II Ready Batch.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.150 "Phase II Ready Status" means, with respect to a particular Licensed Product Candidate, [***].
- 1.151 [***].
- 1.152 "Phase III Clinical Trial" means a controlled or uncontrolled human Clinical Trial of a Licensed Product that would satisfy the requirements of 21 C.F.R. § 312.21(c) or equivalent regulation in countries other than the US.
 - 1.153 [***].
 - 1.154 "Platform Enabling Technology" has the meaning set forth in Section 9.3.1.
 - 1.155 "Precision" has the meaning set forth in the preamble.
 - 1.156 "Precision Confidential Information" has the meaning set forth in Section 12.1.1(a).
 - 1.157 "Precision Indemnitees" has the meaning set forth in Section 13.2.
 - 1.158 "Precision Invention" has the meaning set forth in Section 9.1.3(a).
- 1.159 "Precision Know-How" means the Know-How (a) Controlled by Precision as of the Effective Date, or that comes to be Controlled by Precision at any time during the Research Phase, or (b) that comes to be Controlled by Precision at any time thereafter during the Term and that Precision elects to provide to Baxalta, in each case that is reasonably necessary or useful to clinically Develop, manufacture and Commercialize Licensed Product Candidate(s) or Licensed Product(s) in the Field; in each case of (a) and (b), excluding Know-How Controlled by Precision specifically related to the Precision Platform Technology. Precision Know-How shall include Precision's interest in any Joint Inventions. For clarity, Precision Know-How does not include Patent Rights claiming Precision Know-How.
- 1.160 "Precision Patents" means the Patent Rights (a) Controlled by Precision as of the Effective Date, (b) that come to be Controlled by Precision at any time during the Term, or (c) that are jointly owned by Baxalta and Precision pursuant to this Agreement; in each case of (a), (b) and (c), that claim (x) any Licensed Product Candidate or Licensed Product or a method of manufacture or use thereof, or (y) any constructs that are used to create any Licensed Product Candidate or Licensed Product, any gene products (*i.e.*, RNA and protein resulting from expression, which may include ARCUSTM nucleases) of such constructs, the replication of such constructs, or the use of such constructs or such nucleases to Develop, manufacture, use or Commercialize any Licensed Product Candidate or Licensed Product. Precision Patents shall include Precision's interest in any Joint Patents. Precision Patents do not include Precision Platform Patents.
- 1.161 "Precision Platform Patents" means the Patent Rights Controlled by Precision as of the Effective Date, or that come to be Controlled by Precision at any time during the Term, that claim Precision Platform Technology.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.162 "<u>Precision Platform Technology</u>" means the technology Controlled by Precision as of the Effective Date, or that comes to be Controlled by Precision at any time during the Research Phase, relating to (a) the Development, manufacture or use of any genome engineering tools (excluding use of ARCUSTM nucleases that are used to create any Licensed Product Candidate or Licensed Product) or (b) the Development or manufacture of any precursor, intermediate or construct (including any such CAR-T construct), which tools, precursors, intermediates or constructs, as applicable, are used in the creation of any Licensed Product Candidate or Licensed Product.
 - 1.163 "Precision Restrictive Covenants" has the meaning set forth in Section 11.2.1(b).
 - 1.164 "Precision Target" has the meaning set forth in Section 2.4.4.
 - 1.165 "Product Infringement" has the meaning set forth in Section 9.5.1.
 - 1.166 "Product-Related Patents" has the meaning set forth in Section 9.2.2.
- 1.167 "Promote" means those activities normally undertaken by a pharmaceutical company's sales force in accordance with applicable Laws to implement marketing plans and strategies aimed at encouraging the appropriate use of a particular product for the treatment of human diseases and conditions. When used as a verb, "Promote" shall mean to engage in such activities. The word "Promotion" has correlative meaning.
 - 1.168 "Prosecuting Party" has the meaning set forth in Section 9.2.3.
 - 1.169 "Quality" has the meaning set forth in Section 6.3.
 - 1.170 "Quality Agreement" has the meaning set forth in Section 6.3.
 - 1.171 "Quarantined Information" means [***] in each case that are in Precision's possession [***].
 - 1.172 "Recipient Party" has the meaning set forth in Section 8.5.
 - 1.173 "Redacted Form of Agreement" has the meaning set forth in Section 12.4.2.
- 1.174 "Regulatory Authority" means any national, supra national, regional, state or local regulatory authority, department, bureau, commission, council or other Governmental Authority (including the FDA and EMA) that is responsible for overseeing the Development, use, manufacture, transport, storage or Commercialization of a Licensed Product Candidate or a Licensed Product.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.175 "Regulatory Exclusivity" means the right or protection, granted by a Regulatory Authority or other Governmental Authority, as applicable, in a jurisdiction, providing with respect to a Licensed Product in such jurisdiction: (a) marketing exclusivity that prevents the Regulatory Authority or other Governmental Authority from accepting an application for Marketing Approval from, or from granting Marketing Approval to, a Third Party (other than any Sublicensee or authorized designee of Baxalta or any of its Affiliates or Sublicensees) for a pharmaceutical or biologic product (including a generic, biosimilar, similar medicinal product or generic or competing version of a pharmaceutical product) that is the same or a bioequivalent of the Licensed Product, such as through new molecular entity or biological product or orphan drug or pediatric exclusivity designation by the applicable Regulatory Authority or Governmental Authority, or an exclusive right to sell pursuant to data exclusivity provisions such as those under EC Directives 2004/27/EC and 2001/83/EC and Regulation 726/2004/EC or, in each of the foregoing cases, any foreign equivalent; or (b) data protection for regulatory data relating to the Licensed Product against unfair commercial use or public release, such as that provided for in Article 39.3 of Annex 1C. Part II. Section 7 of the Agreement on Trade-Related Aspects of Intellectual Property Rights (TRIPS Agreement), or any foreign equivalent.
- 1.176 "Regulatory Filings" means any application for Marketing Approval, any application for Reimbursement Approval, and any notification or other submission made to or with a Regulatory Authority that is necessary or reasonably desirable to Develop (including to conduct Clinical Trials), use, manufacture, transport, store or Commercialize a particular product for the treatment of human diseases and conditions in a particular country or regulatory jurisdiction, whether made before or after receipt of Marketing Approval in the country or regulatory jurisdiction. The term "Regulatory Filings" shall include all amendments and supplements to any of the foregoing and all proposed labels, labeling, package inserts, monographs and packaging for a Licensed Product in a particular country.
 - 1.177 "Regulatory Role" has the meaning set forth in Section 5.1.
 - 1.178 "Reimbursement Approval" means with respect to a particular Licensed Product and a particular country or regulatory jurisdiction, [***].
- 1.179 "Relevant Factors" means all factors that are relevant to the Development, manufacture or Commercialization of a product, including its safety and efficacy, product profile, cost to develop, cost and availability of supply, the time required to complete Development, the competitiveness of the marketplace (including the proprietary position and anticipated market share of the product), the patent position with respect to such product (including the ability to obtain or enforce, or have obtained or enforced, such patent rights), the third-party patent landscape relevant to the product, the regulatory structure involved, the likelihood of regulatory approval, the anticipated or actual profitability of the applicable product and other technical, commercial, legal, scientific, regulatory and medical considerations.
 - 1.180 "Remedial Action" has the meaning set forth in Section 5.6.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.181 "Research Phase" means for each Licensed Product Candidate the period starting with the establishment of a Development Plan and ending upon Phase II Ready Status for such Licensed Product Candidate.
 - 1.182 "Restrictive Covenants" has the meaning set forth in Section 11.2.2(a).
- 1.183 "Reversion IP" means any Baxalta Patents, Baxalta Know-How and Joint Patents that (a) come to be Controlled by Baxalta or any of its Affiliates at any time during the Term as a result of activities in connection with this Agreement, (b) are in existence as of the date of termination of this Agreement (in whole or in part) by Precision pursuant to Section 14.2.1, Section 14.2.2 or Section 14.2.3, by Baxalta in accordance with Section 14.2.1 (for any reason other than for the events described in Section 14.3.2(a)), or by Baxalta pursuant to Section 14.2.5 and (c) are reasonably necessary or useful in connection with the Development, manufacture, use or Commercialization of Reversion Products in the Territory.
 - 1.184 "Reversion Patents" has the meaning set forth in Section 14.3.1(a)(i).
- 1.185 "Reversion Products" means, in the event of termination of this Agreement by Precision in accordance with Section 14.2.1, Section 14.2.2 or Section 14.2.3, by Baxalta in accordance with Section 14.2.1 (for any reason other than for the events described in Section 14.3.2(a)), or by Baxalta in accordance with Section 14.2.5, in each case in its entirety or with respect to one or more Included Targets, (a) any product that (i) was a Licensed Product Candidate or Licensed Product immediately prior to such termination and (ii) is no longer a Licensed Product Candidate or Licensed Product as a result of such termination, and (b) any other engineered human T cells with Chimeric Antigen Receptors Directed to any Target that was an Included Target immediately prior to such termination and is no longer an Included Target as a result of such termination.
 - 1.186 "Royalty Term" has the meaning set forth in Section 8.4.3.
 - 1.187 "Safety Agreement" has the meaning set forth in Section 5.4.
 - 1.188 "Sales Milestone" has the meaning set forth in Section 8.3.3.
- 1.189 "<u>Sales Report</u>" means, with respect to each Calendar Quarter, a report detailing for such Calendar Quarter, on a on a Licensed Product-by-Licensed Product and country-by-country basis: (a) the amount of Net Sales of the Licensed Products in the Territory, (b) a calculation of the royalty payment due on such Net Sales, and (c) the exchange rates and dates used to convert any amounts to USD, as applicable.
 - 1.190 [***].

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.191 "Sublicensee" means, with respect to Baxalta, any Third Party to which Baxalta sublicenses all or any portion of the rights granted to it under Section 4.3.
- 1.192 "Supply Agreement" means a Manufacturing and Supply Agreement between Precision and Baxalta pursuant to which Precision will, by itself or through one or more Third Party CMOs, supply to Baxalta its requirements of clinical trial materials for Phase II Clinical Trials for each of the Licensed Products.
- 1.193 "Target" means a single unique molecular species that (a) is chemically distinct from other molecules, (b) is an antigenic protein or antigenic glycolipid protein complex that is expressed on or in a human cell (including all epitopes of such antigenic protein or antigenic glycolipid protein complex), and (c) wherein a binding entity derives recognized therapeutic value from binding such molecular species.
 - 1.194 "Target Information Package" has the meaning set forth in Section 2.1.4.
 - 1.195 "Target Proposal Date" has the meaning set forth in Section 2.1.4.
- 1.196 "<u>Technology Transfer Plan</u>" means, with respect to a given Licensed Product, the mutually-agreed technology transfer plan, timeline, budget and assignment of project managers from both Baxalta and Precision, which shall be developed and agreed upon in good faith by the Parties.
 - 1.197 "Term" has the meaning set forth in Section 14.1.
 - 1.198 "Territory" means worldwide.
 - 1.199 "Third Party" means any Person other than Precision and Baxalta and their respective Affiliates.
- 1.200 "Third Party License Agreement" means any agreement (including any settlement agreement) entered into after the Effective Date with a Third Party, whereby royalties are to be paid to such Third Party based on the grant of rights under valid Patent Rights Controlled by such Third Party in a country or countries, which Patent Rights are Necessary to Commercialize the Licensed Product free from infringement of such Patent Rights. For purposes of this definition, "Necessary to Commercialize" means, with respect to a particular Licensed Product and Third Party Patent Rights in a particular country or countries, [***]. For clarity, an agreement under which rights are obtained with respect to any Independently Active Therapeutic Ingredient is not a Third Party License Agreement.
 - 1.201 "Threshold Decrease" has the meaning set forth in Section 8.4.4.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.202 "<u>Unavailable Target</u>" means a Target that becomes the subject of an agreement entered into by Precision or any of its Affiliates and a Third Party after the Effective Date relating to any Development and/or Commercialization of any products with respect to such Target, for so long as such Target remains the subject of such an agreement.
 - 1.203 "Unavailable Target Information" has the meaning set forth in Section 2.1.3(b).
 - 1.204 "Upfront Fee" has the meaning set forth in Section 8.1.
- 1.205 "<u>Unselected Candidate</u>" means any Variant that is proposed by Precision to Baxalta in accordance with <u>Section 2.1.4</u> or <u>Section 2.1.5</u> and that does not become the subject of a Development Plan upon the completion of the Variant selection processes that are set forth in such Sections.
 - 1.206 "U.S." or "US" means the United States of America, its territories and possessions.
 - 1.207 "U.S. GAAP" means U.S. Generally Accepted Accounting Principles.
 - 1.208 "USD" means U.S. Dollars.
- 1.209 "Valid Claim" means (a) a claim of an issued and unexpired patent (as may be extended through supplementary protection certificate or patent term extension or the like) that has not been revoked, held invalid or unenforceable by a patent office, court or other governmental agency of competent jurisdiction in a final and non-appealable judgment (or judgment from which no appeal was taken within the allowable time period) and which claim has not been disclaimed, denied or admitted to be invalid or unenforceable through reissue, re-examination or disclaimer or otherwise; or (b) a pending claim of an unissued patent application that is supported by the specification and that has been pending for no longer than [***] from the earliest claimable priority date.
 - 1.210 "Variant" means a CAR-T Directed to a particular Target, which contains [***].
 - 1.211 "Withholding Taxes" has the meaning set forth in Section 8.5.

ARTICLE II.

DISCOVERY AND DEVELOPMENT

- 2.1 Selection of Included Targets.
- 2.1.1 <u>Maximum Number of Included Targets</u>. Baxalta shall have the right to designate up to six (6) Included Targets during the Nomination Period in accordance with this <u>Section 2.1</u>, including the Target (CD19) that is designated as of the Effective Date pursuant to <u>Section 2.1.2</u>. [***]. Any Target is eligible to become an Included Target unless it is an Unavailable Target or a Precision Target. Such designation right includes the right, during the Nomination
- [***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Period, for Baxalta to designate combinations of [***] Included Targets to be a single Included Target (each such combination, a "Combination Target"); provided, however, that (a) a combination of Targets may not be designated as a Combination Target unless (i) [***] the Targets within such combination are also Included Targets on an individual basis, and (ii) each of such Included Targets is not itself a Combination Target, and (b) a Combination Target shall constitute a separate Included Target for all purposes under this Agreement, except that a Combination Target shall not constitute a separate Included Target for purposes of Baxalta's limit of designating up to six (6) Included Targets.

2.1.2 Initial Included Target. The Parties agree that B-lymphocyte antigen CD19 ("CD19") is hereby designated, as of the Effective Date, as the initial Included Target. A summary draft Development Plan for CD19 is attached hereto as Exhibit G. No later than [***] days after the Effective Date, Precision shall deliver to Baxalta a full Development Plan for CD19, [***]. As promptly as practicable thereafter (but in no event later than [***] after delivery of such proposed Development Plan), the JSC shall meet to review and approve such Development Plan. The Parties will work together in good faith to ensure that the Development Plan for CD19 has been approved by the JSC by no later than [***] after Precision's delivery of the full Development Plan for CD19.

2.1.3 Unavailable Targets.

- (a) Precision represents and warrants that as of the Effective Date, there are no Unavailable Targets. Except as expressly set forth in this Section 2.1.3, Baxalta shall have no rights with respect to any Unavailable Targets, and Precision and its Affiliates may freely undertake and engage in, including in collaboration with Third Parties, any research, Development, Commercialization or any other activities of any kind in any field with respect to all Unavailable Targets without any obligation to Baxalta. For clarity, without limiting the obligations set forth in Section 2.4, nothing in this Agreement shall be construed to preclude, limit, restrict or otherwise affect any right or ability of Precision or any of its Affiliates to enter into any agreement that would cause any Target that is not an Included Target to become an Unavailable Target at any time.
- (b) If any Target becomes an Unavailable Target after the Effective Date, Precision shall submit a complete and accurate list of Unavailable Targets along with a copy of the applicable agreement related to each of such Unavailable Targets, which copies may be submitted in redacted form except to the extent necessary to confirm that such agreement relates to any Development and/or Commercialization of products with respect to such Unavailable Target (the "Unavailable Target Information"), to an independent escrow agent mutually agreed to by the Parties (the "Escrow Agent"), and shall provide the Escrow Agent with updated Unavailable Target Information promptly upon any Target becoming an Unavailable Target or losing its status as an Unavailable Target. The Unavailable Target Information shall be held by the Escrow Agent in confidence. If Precision notifies Baxalta that any Target is an Unavailable Target, the Escrow Agent shall provide confirmation to Baxalta that such Target is an Unavailable Target. The Escrow Agent shall not provide to Baxalta the identity of any other Target that appears on the list of Unavailable Targets or any details regarding any agreement related to any Unavailable Target. Precision represents and warrants that the Unavailable Target Information maintained with the Escrow Agent shall be promptly updated by Precision in order to remain accurate, complete and up-to-date at all times during the Nomination Period.
- 2.1.4 Nomination of Additional Targets. At any time during the Nomination Period, Baxalta may designate up to (i) [***] additional Targets that are not Unavailable Targets or Precision Targets, or (ii) subject to Section 2.1.1, any number of Combination Targets, as Included Targets in accordance with this Section 2.1.4. Baxalta may Nominate a Target for potential designation as an Included Target by providing Precision with written notice identifying the Target and providing such other information (if any) that it determines may be reasonably

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

necessary or useful to Precision for preparing the associated Development Plans (the "Target Information Package"). If Baxalta Nominates pursuant to any provision of this Agreement any Target that is an Unavailable Target as of the date of Nomination by Baxalta, Precision shall promptly, but in no event more than [***] following Nomination, provide Baxalta with written notice that such Target is an Unavailable Target, and shall request confirmation by the Escrow Agent in accordance with Section 2.1.3(b), and such Nomination by Baxalta shall have no further effect. If Baxalta Nominates pursuant to any provision of this Agreement any Target that is a Precision Target as of the date of Nomination by Baxalta, such Nomination by Baxalta shall have no effect. If the Nominated Target is not an Unavailable Target or a Precision Target as of the date of Nomination by Baxalta, Baxalta and Precision will meet to discuss the Target Information Package through a meeting of the JSC, including any technical or other concerns Precision may have with respect to any such Target. Provided that Baxalta desires to move forward with Nomination after such meeting, Precision shall prepare and deliver to the JSC within [***] after the date of such meeting (i) a proposed Development Plan for such Target for review and approval by the JSC, and (ii) a list of Variants that Precision believes in good faith are capable of being generated by Precision via Precision Platform Technology for such Target (the date such proposal and Variant list are delivered, the "Target Proposal Date"). A Target will be deemed an Included Target if (and only if) (x) Baxalta has selected at least one (1) Variant to be the subject of a Development Plan and (y) the Development Plan for such Variant(s) and Target has been approved in writing by the JSC; in each case within [***] after the Target Proposal Date (or any extended period as mutually agreed by the Parties). If Baxalta does not move forward with Nomination or select one or more Variants, or if the JSC does not approve the proposed Development Plan for at least one of the Variants proposed by Precision and selected by Baxalta, the Parties shall discuss same at the following JSC meeting. For the avoidance of doubt, if any Target Nominated by Baxalta, other than any Unavailable Target or Precision Target, fails to become an Included Target through the foregoing process, Baxalta may later re-Nominate such Target, but such Target will not be an Included Target unless the process described in this Section 2.1.4 has been repeated resulting in such Target becoming an Included Target.

2.1.5 Additional Variants for Included Targets.

(a) <u>Proposal of Additional Variants</u>. From time to time during the Term, Precision may notify Baxalta in writing of one or more additional Variants (which may include Unselected Candidates) that Precision believes in good faith are capable of being generated by Precision via Precision Platform Technology for an Included Target (the date such notice is provided, the "<u>Candidate Proposal Date</u>"). If, within [***] after such notification, Baxalta indicates in writing that it is interested in reviewing a proposed Development Plan for such Variant(s), then Precision will prepare a proposed Development Plan that includes such Variant(s) for review and approval by the JSC within [***] after receipt of such indication of interest from Baxalta. Each proposed Variant will be deemed a Licensed Product Candidate for such Included Target if (and only if) a Development Plan that includes such Variant has been approved in writing by the JSC,

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

which approval shall be provided within [***] after the Candidate Proposal Date (or any extended period as mutually agreed by the Parties). If
(a) Baxalta does not indicate that it is interested in reviewing a proposed Development Plan for any one or more of such Variants within the applicable time period above or (b) the JSC does not approve a Development Plan for any one or more of such Variants within the applicable time period above, such Variants shall thereafter be (or continue to be, as applicable) deemed Unselected Candidates.

(b) Final Proposal of Additional Variants. During any period in which (i) Baxalta does not hold a Commercial License for any one or more Licensed Products Directed to a particular Included Target, and (ii) there is no Licensed Product Candidate Directed to such Included Target that is currently the subject of Development activities under a Development Plan pursuant to Section 2.2, this Section 2.1.5(b) shall apply with respect to proposals of additional Variants for such Included Target in lieu of Section 2.1.5(a). During any such period with respect to a particular Included Target, Precision may notify Baxalta of one or more additional Variants (which may include Unselected Candidates) that Precision believes in good faith are capable of being generated by Precision via Precision Platform Technology for such Included Target and if Baxalta indicates in writing that it is interested in reviewing a proposed Development Plan for such Variant(s), which it must do within [***] after the date of Precision's notice, Precision shall promptly deliver a proposed Development Plan that includes such Variant(s) for review and approval by the JSC (the later of the expiration of such [***] period or the date such Development Plan(s) are provided, the "Final Candidate Proposal Date"). Each proposed Variant will be deemed a Licensed Product Candidate for such Included Target if (and only if) a Development Plan that includes such Variant has been approved in writing by the JSC, which approval shall be provided within [***] after the Final Candidate Proposal Date (or any extended period as mutually agreed by the Parties). If the JSC does not approve a Development Plan for any one or more of such Variants within such time period above, or if Baxalta does not express interest in any such Variants within the [***] period set forth above, then notwithstanding anything to the contrary in this Agreement, Baxalta shall be deemed to have terminated this Agreement with respect to such Included Target (including all Licensed Product Candid

2.1.6 Extension of the Nomination Period. Baxalta may (in its sole discretion) request to extend the Nomination Period for a single additional term of [***] by providing written notice of such request to Precision; provided, however, that such extension shall only become effective upon mutual written agreement of the Parties (in the Parties' sole discretion) regarding funding commitments and other terms related to such extension. For clarity, while the Parties will be obligated to discuss the proposed extension in good faith, neither Party shall be obligated to agree to extend the Nomination Period.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

2.2 Development Plans; Development Obligations.

- 2.2.1 Precision shall, subject to the oversight of the JSC, use Commercially Reasonable Efforts to direct, coordinate, perform and manage the Development of Licensed Product Candidates for each of the Included Targets, in each case in accordance with this Agreement and the applicable Development Plan.
- 2.2.2 Through the JSC, each Party shall have the right to propose changes to the Development Plans on an ongoing basis as necessary. The JSC shall have the authority to review and approve such changes, provided that each Development Plan shall at all times contain terms that are consistent with this <u>Article II</u>. If the terms of any Development Plan contradict, or create inconsistencies or ambiguities with, the terms of this Agreement, then the terms of this Agreement shall govern.
- 2.2.3 Precision shall be solely responsible for, shall use Commercially Reasonable Efforts to carry out, and shall pay all costs and expenses incurred by Precision in carrying out, Development activities for each Licensed Product Candidate in accordance with the applicable Development Plan (including any fees and associated costs and expenses for Regulatory Filings to be made by Precision pursuant to the Development Plans). Precision may not subcontract any of Precision's Development obligations (other than to a CMO approved pursuant to Section 6.2) without the JSC's consent to the subcontracting of such obligations (but which shall not require the JSC's consent to any particular subcontractors that are not contract research organizations that will conduct Clinical Trials under this Agreement or CMOs), and, if the JSC provides such consent, Precision agrees that any subcontractor (including any contract research organization or CMO) shall be bound by written obligations consistent with those required under Section 11.1.1(a) and written obligations of confidentiality and non-use consistent with this Agreement, and such other provisions to permit Precision to comply with all the terms and conditions of this Agreement. Precision shall remain responsible for compliance with this Agreement and shall be fully responsible for any breach of this Agreement by any of its subcontractors.
- 2.2.4 Promptly, but in no event more than [***], following receipt of trial phase data for the first Phase I Clinical Trial dose escalation study with respect to any Licensed Product Candidate, Precision shall deliver written notice to the JSC, together with a copy of the relevant data regarding the results of such study (the "Phase I Notice"). The JSC shall promptly (but in any event within [***] after the Phase I Notice) convene a meeting to review and discuss such data with respect to such Licensed Product Candidate and determine (a) whether activities related to the manufacture of Phase II Clinical Trial material for such Licensed Product Candidate should be initiated under the applicable Development Plan, and (b) the amount of such material that will be sufficient to conduct the initial Phase II Clinical Trial with respect to such Licensed Product Candidate. If the JSC does not determine to initiate such activities, then the JSC shall meet to consider whether it is commercially reasonable to continue to seek to achieve Phase II

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Ready Status for such Licensed Product Candidate and, upon a positive determination, develop a plan for such achievement. If the JSC determines that it is not commercially reasonable to continue to seek to achieve Phase II Ready Status for such Licensed Product Candidate, or fails to approve a plan for such achievement within [***] after such meeting of the JSC, then neither Party shall be required nor permitted to conduct any further Development or Commercialization activities under this Agreement with respect to such Licensed Product Candidate.

2.2.5 Promptly, but in no event more than [***] following the achievement of Phase II Ready Status (which may or may not include, for purposes of Precision's notice as set forth in this sentence, the JSC having identified a pivotal Phase II Clinical Trial dose for the applicable Licensed Product Candidate) with respect to any Licensed Product Candidate, Precision shall deliver written notice to the JSC (the "Development Achievement Notice"), together with a copy of the data package regarding the results of Precision's research and development program for such Licensed Product Candidate. The JSC shall promptly, but in no event more than [***] following receipt of such notice, convene a meeting to review and discuss such notice and data package with respect to such Licensed Product Candidate and, if not yet identified by the JSC, to identify a pivotal Phase II Clinical Trial dose for such Licensed Product Candidate. Additionally, for [***] following the end of such meeting, Baxalta may request, and Precision shall thereafter provide as promptly as practicable, any additional information that could reasonably be expected to assist Baxalta in determining whether it desires to exercise the Commercial Option for the applicable Licensed Product Candidate in accordance with Section 4.1. If the JSC determines that the Development Achievement Notice was not properly provided, or that it cannot identify a pivotal Phase II Clinical Trial dose for the applicable Licensed Product Candidate based on the available Phase I Clinical Trial data, then the Development Achievement Notice shall be deemed not to have been delivered by Precision. In the event that the JSC determines that it cannot identify a pivotal Phase II Clinical Trial dose for the applicable Licensed Product Candidate based on the available Phase I Clinical Trial data, (a) the JSC shall meet to consider whether it is commercially reasonable to continue to seek to achieve Phase II Ready Status for such Licensed Product Candidate and, upon a positive determination, develop a plan for such achievement and (b) if the JSC determines that it is not commercially reasonable to continue to seek to achieve Phase II Ready Status for such Licensed Product Candidate, or fails to approve a plan for such achievement within [***] after such meeting of the JSC, then neither Party shall be required nor permitted to conduct any further Development or Commercialization activities under this Agreement with respect to such Licensed Product Candidate.

2.2.6 Following the Commercial Option Exercise Date for each Licensed Product Candidate, Baxalta shall be solely responsible for and shall have sole control of, and shall use Commercially Reasonable Efforts to carry out, all further Development activities for such Licensed Product in the Territory, at Baxalta's sole cost and expense (including any fees and associated costs and expenses for Regulatory Filings made following such exercise). With respect to determinations by Baxalta to discontinue Development activities for a particular Licensed Product in all countries throughout the Territory, Baxalta shall give Precision written notice of such determination within a reasonable period of time, but in any event within [***] after any such determination is made, and upon provision of such notice Baxalta shall be deemed to have terminated this Agreement pursuant to Section 14.2.5 with respect to such Licensed Product.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

2.2.7 Neither Party makes any representation, warranty or guarantee that the Development activities conducted under the Development Plan will be successful or that any particular results will be achieved.

2.3 <u>Development of Unselected Candidates</u>.

- 2.3.1 Precision and its Affiliates shall be free to Develop Unselected Candidates (but not any other Variant for an Included Target that is not a Licensed Product Candidate) at their own expense, provided that (a) Precision complies with its obligations under the Development Plans, (b) no Unselected Candidate may be the subject of any Clinical Trial unless approved in advance by the JSC, and (c) no Unselected Candidate may be the subject of Development efforts conducted in collaboration with or on behalf of any Third Party without prior approval of the JSC. Precision will deliver written updates to Baxalta no less frequently than quarterly regarding the status of Precision's Development of any Unselected Candidates, including all reasonably necessary data to allow Baxalta to assess Precision's progress towards achieving a functional CAR-T Directed to the applicable Included Target with any Unselected Candidate, including details regarding the first in-vivo proof of concept and pre-IND information relating to each such Unselected Candidate. If any Unselected Candidate is approved by the JSC for Clinical Trials, Precision shall provide no less than [***] advance written notice prior to any planned IND filing with respect to such Unselected Candidate. If safety concerns related to the Unselected Candidates arise from such Development activities, Precision must inform Baxalta immediately. Any publication or presentation with respect to Unselected Candidates shall be subject to Section 12.3.
- 2.3.2 Baxalta may elect to designate any Unselected Candidate as a Licensed Product Candidate for a particular Included Target at any time until the [***] anniversary of the date upon which such Included Target became an Included Target, by notifying Precision in writing that it is interested in reviewing a proposed Development Plan for such Unselected Candidate. In such event, Precision will prepare a proposed Development Plan for such Unselected Candidate for review and approval by the JSC within [***] after receipt of such indication of interest from Baxalta. The Unselected Candidate will be deemed a Licensed Product Candidate for such Included Target if (and only if) a Development Plan for such Unselected Candidate has been approved in writing by the JSC, which approval shall be provided within [***] after such written indication of interest from Baxalta (or any extended period as mutually agreed by the Parties). If such Unselected Candidate becomes a Licensed Product Candidate, and if such Unselected Candidate has previously achieved any Development Milestone(s) [***] prior to

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

becoming a Licensed Product Candidate, Precision may immediately invoice Baxalta for such achieved Development Milestone(s), and Baxalta shall make the applicable payment within [***] following receipt of such invoice. If the JSC does not approve a Development Plan for such Unselected Candidate within the applicable time period above, such Unselected Candidate shall remain an Unselected Candidate.

2.3.3 For the avoidance of doubt, the rights set forth in this <u>Section 2.3</u> shall be subject to Precision's obligations under the Precision Restrictive Covenants.

2.4 Inactive Targets; Precision Targets.

- 2.4.1 <u>Inactive Targets</u>. From the Effective Date until the end of the Nomination Period, Precision shall notify Baxalta in writing prior to Precision initiating Development of any CAR-T Directed to any Target that is clinically relevant to oncology and not an Included Target, Unavailable Target or Precision Target (each, an "<u>Inactive Target</u>"); provided that, the JSC must determine that a Target (other than an Unavailable Target or Precision Target) is not clinically relevant to oncology prior to Precision being relieved of any obligations with respect to such Target pursuant to this <u>Section 2.4</u>. For clarity, Precision and its Affiliates may freely undertake and engage in, including in collaboration with Third Parties, any research, Development, Commercialization or any other activities of any kind in any field with respect to Inactive Targets at their own expense without any obligation to Baxalta except as expressly set forth in this Section 2.4.
- 2.4.2 Nomination of Inactive Targets. At any time during the Nomination Period, Baxalta may Nominate any Inactive Target. In such event, the process set forth in Section 2.1.4 shall commence with respect to the Nomination of such Inactive Target. If an Inactive Target becomes an Included Target as a result of such process, such Inactive Target shall thereafter cease to be an Inactive Target and shall be deemed to be an Included Target, and if any Variant for such Inactive Target has previously achieved any Development Milestone(s) [***] prior to becoming a Licensed Product Candidate, Precision may immediately invoice Baxalta for such achieved Development Milestone(s), and Baxalta shall make the applicable payment within [***] following receipt of such invoice. If such Inactive Target fails to become an Included Target as a result of such process, then such Inactive Target shall continue to be an Inactive Target, subject to the terms of this Agreement.

2.4.3 Updates; Inactive Target Non-Disclosure Period.

(a) During the Nomination Period, Precision will deliver written updates to Baxalta no less frequently than Calendar Quarterly regarding the status of Precision's Development of any Inactive Targets that are clinically relevant to oncology, including all reasonably necessary data to allow Baxalta to assess Precision's progress towards achieving a functional CAR-T Directed to the applicable Inactive Target, including details regarding the first in-vivo proof of concept and pre-IND information relating to Variants Directed to each Inactive Target (the "Inactive Target Evaluation Data").

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (b) Precision will not provide any Inactive Target Evaluation Data to any Third Party in connection with any discussions with or inquiries of a Third Party that, at the time such data would be provided, relate to a transaction that would result in such Inactive Target becoming an Unavailable Target during the Nomination Period, unless Precision has first provided notice to Baxalta of its desire to share such information with such a Third Party, together with a proposed Development Plan and list of Variants that Precision believes in good faith are capable of being generated by Precision via Precision Platform Technology for such Inactive Target, and Baxalta has had at least [***] after such notice and delivery of a proposed Development Plan and list of Variants (the "Inactive Target Non-Disclosure Period") to determine whether Baxalta desires to Nominate such Inactive Target as an Included Target in accordance with this Agreement.
- (c) If Baxalta provides notice Nominating an Inactive Target within the applicable Inactive Target Non-Disclosure Period (which it may do only if it has a bona fide interest in such Inactive Target becoming an Included Target), then the following shall apply with respect to such Nomination in lieu of Section 2.4.2. Such Inactive Target shall be deemed an Included Target if (and only if) (i) Baxalta has selected at least one such Variant to be the subject of the Development Plan for such Inactive Target and (ii) an initial Development Plan including such Variant as a Licensed Product Candidate for such Inactive Target has been approved in writing by the JSC; in each case within the Inactive Target Non-Disclosure Period (or any extended period as mutually agreed by the Parties). If such Inactive Target becomes an Included Target as a result of such process, such Inactive Target shall thereafter cease to be an Inactive Target and shall be deemed to be an Included Target, and if any Variant for such Inactive Target has previously achieved any Development Milestone(s) [***] prior to becoming a Licensed Product Candidate, Precision may immediately invoice Baxalta for such achieved Development Milestone(s), and Baxalta shall make the applicable payment within [***] following receipt of such invoice.
- (d) If, prior to the expiration of the applicable Inactive Target Non-Disclosure Period for an Inactive Target, Baxalta does not Nominate such Inactive Target as set forth in Section 2.4.3(g), or if Baxalta Nominates such Inactive Target during the applicable Inactive Target Non-Disclosure Period and such Inactive Target fails to become an Included Target as the result of the process described above, then Precision and its Affiliates shall thereafter be free to share the applicable Inactive Target Evaluation Data for such Inactive Target with any Third Party for a period of [***] directly following the conclusion of the Inactive Target Non-Disclosure Period for such Inactive Target (such [***] period, the "Inactive Target Freeze Period"). During the Inactive Target Freeze Period, (i) Precision's obligation to share the applicable Inactive Target Evaluation Data with Baxalta under Section 2.4.3(a) with respect to such Inactive Target shall not apply, and (ii) Baxalta shall have no right to Nominate such Inactive Target. Following the conclusion of the Inactive Target Freeze Period, unless such Inactive Target has become an Unavailable Target during such period, Precision's obligations with respect to such Inactive Target under Section 2.4.3(a) shall resume (including with respect to Inactive Target Evaluation Data generated during the Inactive Target Freeze Period), and Baxalta's ability to Nominate such Inactive Target shall resume, but Precision's obligations with respect to such

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Inactive Target under Section 2.4.3(b) shall cease to apply with respect to any Third Party to which the Inactive Target Evaluation Data was disclosed during the applicable Inactive Target Freeze Period or any of such Third Party's Affiliates.

- (e) Precision may, but shall not be required to, commence additional Inactive Target Freeze Periods (including during any current Inactive Target Freeze Period) by providing the notice and opportunity to Nominate the Inactive Target as set forth in Section 2.4.3(b); provided, however, if Precision provides any such notice during the pendency of any Inactive Target Freeze Period for any Inactive Target, Baxalta shall, notwithstanding Section 2.4.3(d), be permitted to Nominate the Target within the [***] Inactive Target Non-Disclosure Period in accordance with Section 2.4.3(c).
- (f) Notwithstanding anything to the contrary in this Agreement, if Baxalta Nominates an Inactive Target during the applicable Inactive Target Non-Disclosure Period for such Inactive Target, then regardless of whether or not such Inactive Target becomes an Included Target as a result of the process described above, the number of available Included Targets that may be designated by Baxalta in accordance with this Agreement, as set forth in Section 2.1, shall be reduced by one.
- 2.4.4 <u>Precision Targets</u>. Precision shall provide no less than [***] advance written notice prior to any planned IND filing by Precision with respect to any Variant Directed to an Inactive Target. If Baxalta does not Nominate such Inactive Target in accordance with <u>Section 2.4.2</u> within [***] after the date of such notice, or thereafter such Inactive Target fails to become an Included Target as a result of the process initiated by such Nomination, such Inactive Target shall thereafter cease to be an Inactive Target and shall be deemed to be a "<u>Precision Target</u>". Precision and its Affiliates may freely undertake and engage in, including in collaboration with Third Parties, any research, Development, Commercialization or any other activities of any kind in any field with respect to all Precision Targets with no further obligation to Baxalta.

2.5 <u>Development Licenses</u>.

2.5.1 Subject to the terms and conditions of this Agreement, Precision hereby grants to Baxalta a non-exclusive, royalty-free, non-transferable (except in accordance with <u>Section 17.1</u>), non-sublicensable license under the Precision Patents and Precision Know-How, solely to conduct Baxalta's activities set forth in the applicable Development Plans or as contemplated pursuant to <u>Section 16.2</u> or <u>16.3</u> (if applicable) with respect to the Licensed Product Candidates and to otherwise conduct internal research and Development relating to Licensed Product Candidates.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 2.5.2 Subject to the terms and conditions of this Agreement, Baxalta hereby grants to Precision a non-exclusive, royalty-free, non-transferable (except in accordance with Section 17.1), non-sublicensable license under the Baxalta Patents and Baxalta Know-How, solely to conduct Precision's activities set forth in the applicable Development Plans or the Supply Agreement with respect to the Licensed Product Candidates and Licensed Products, to otherwise conduct internal research and Development relating to Licensed Product Candidates as expressly permitted pursuant to this Agreement, and to comply with all other obligations of Precision under this Agreement.
- 2.5.3 Each Party is permitted to exercise its rights and perform its obligations under the licenses set forth in this <u>Section 2.5</u>, by itself or through the engagement of any of its Affiliates without the prior written consent of the other Party.

2.6 Isolex Co-Development.

- 2.6.1 If the JSC determines that it is desirable to use the Isolex Platform Technology for clinical and/or commercial-scale manufacturing of one or more Licensed Product Candidates or Licensed Products, then:
- (a) Baxalta agrees to grant and hereby grants to Precision a non-exclusive, royalty-free, non-transferable (except in accordance with Section 17.1), non-sublicensable license under the Isolex Platform Patents and related Know-How Controlled by Baxalta that is necessary or useful to practice and further develop the Isolex Platform Technology for CAR-T manufacturing, solely to conduct Precision's activities set forth in the applicable Development Plans or the Supply Agreement with respect to the Licensed Product Candidates and Licensed Products, to otherwise conduct internal research and Development relating to Licensed Product Candidates as expressly permitted pursuant to this Agreement, and to comply with all other obligations of Precision under this Agreement; and
- (b) The Parties shall use Commercially Reasonable Efforts to enter into an agreement pursuant to which the Parties will use Commercially Reasonable Efforts to co-develop the Isolex Platform Technology for commercial-scale manufacturing, which agreement shall include the right for Precision to use the Isolex Platform Technology for its own products and services on commercially reasonable terms and shall otherwise be mutually agreed between the Parties within [***] after such determination by the JSC.

(c) [***]

2.7 <u>Know-How Database</u>. The Parties will establish and maintain a Know-How database to identify the Precision Know-How provided to Baxalta in connection with this Agreement as well as any Baxalta Know-How provided to Precision by Baxalta in connection with this Agreement.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE III. GOVERNANCE

3.1 Joint Steering Committee.

- 3.1.1 <u>Establishment; Responsibility and Authority</u>. Within [***] after the Effective Date, the Parties will establish a joint steering committee to oversee the planning and execution of the activities under the Development Plans (the "<u>JSC</u>"). The JSC's responsibilities shall include the following:
- (a) reviewing and approving the Development Plan, and overseeing and evaluating implementation of the Development Plan, with respect to each Included Target, including monitoring progress of preclinical and clinical studies of the Licensed Product Candidates and Licensed Products and otherwise monitoring compliance with the Development Plan;
- (b) proposing, reviewing and approving amendments to, including implementing all agreed upon extensions or delays, the Development Plan;
- (c) reviewing, commenting on and approving all Regulatory Filings and other regulatory submissions and all material correspondence with Regulatory Authorities occurring prior to the Commercial Option Exercise Date;
- (d) overseeing development of the Isolex Platform Technology in relation to Licensed Product Candidates and Licensed Products (as applicable);
- (e) determining in accordance with <u>Section 2.2.4</u> whether any activities related to the manufacture of Phase II Clinical Trial material for any Licensed Product Candidate should be initiated under the applicable Development Plan and establishing the amount of such material that will be sufficient to conduct the initial Phase II Clinical Trial with respect to any such Licensed Product Candidate;
- (f) reviewing and discussing the Development Achievement Notices provided by Precision pursuant to <u>Section 2.2.5</u> and identifying a pivotal Phase II Clinical Trial dose for each Licensed Product Candidate;
- (g) overseeing manufacture and supply of pre-clinical and clinical trial materials necessary for Development of Licensed Product Candidates through Phase II Ready Batch Success by Precision or its CMO(s) and reviewing and approving all Changes to the manufacturing process for clinical trial materials manufactured by Precision or its CMO(s);
- (h) reviewing and approving the protocols of all preclinical and Clinical Trials for the Licensed Product Candidates planned prior to the Commercial Option Exercise Date;
- (i) reviewing, commenting on and approving all publication strategy (in accordance with <u>Section 12.3)</u> with respect to each Licensed Product and Licensed Product Candidate prior to the Commercial Option Exercise Date;
- [***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (j) evaluating and establishing the patent prosecution and maintenance strategy together with appropriate legal counsel for each Party in accordance with <u>Article IX</u>;
- (k) overseeing any technology transfers required pursuant to this Agreement, and addressing any intellectual property or technology related issues, including issues under Article IX;
- (l) attempting to resolve disputes arising under this Agreement among the Parties, the Alliance Managers or any project teams of the Parties (provided, that for clarity, the JSC shall not have authority to amend this Agreement or to resolve any disputes between the Parties regarding whether a Party has fulfilled or breached this Agreement, and such disputes shall instead be resolved pursuant to <u>Article XV</u>); and
- (m) performing such other tasks and undertaking such other responsibilities as designated to it under this Agreement or the Development Plans.
- 3.1.2 <u>Composition; Voting</u>. Within [***] after the Effective Date, each Party shall appoint three (3) employees or consultants to serve on the JSC, each of which shall have such expertise as is appropriate to the activities of the JSC. Each Party may replace its JSC representatives by written notice to the other Party. Each Party shall have one (1) vote on all matters and decisions that are within the responsibility of the JSC, regardless of the number of such Party's representatives on the JSC, and any decision or other action by the JSC may only be made by unanimous consensus of the Parties. The members of the JSC will use good faith efforts to reach unanimous consensus on all decisions and other actions that are within the responsibility of the JSC.
- 3.1.3 <u>Co-Chairpersons</u>. Each Party shall designate one of its JSC representatives to serve as co-chairperson. The co-chairpersons shall be jointly responsible for calling meetings and shall be jointly responsible for setting the agenda (which shall include a list of all participants expected at a meeting). The co-chairpersons shall alternate responsibility for circulating such agenda at least [***] prior to each meeting and distributing minutes of the meetings pursuant to <u>Section 3.1.5</u> within [***] following such meeting, but will not otherwise have any greater power (including voting power) or authority than any other member of the JSC.
- 3.1.4 Meetings. The JSC shall, after appointment of its initial members, meet at least once every Calendar Quarter at times mutually agreed upon by the Parties, and at least two (2) of such meetings each year shall be held in person. The location of the meetings of the JSC to be held in person shall be agreed upon by the Parties (with the intent that it should alternate between the Parties' respective headquarters locations or be held at the time and sites of major medical conferences attended by both Parties). Additionally, either Party may call a special meeting of the JSC upon written notice to the other (and which meeting shall be scheduled promptly at mutually agreeable times) (a) to make any determination under this Agreement that cannot reasonably be postponed until the next scheduled JSC meeting, (b) for the purpose of

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

resolving disputes in connection with, or for the purpose of reviewing or making a decision pertaining to, any material matter within the purview of the JSC, the examination or resolution of which cannot reasonably be postponed until the next scheduled JSC meeting or (c) as reasonably necessary to review other matters occurring between JSC meetings. Each such special meeting of the JSC shall be convened at such time as may be mutually agreed upon by the Parties, but in any event shall be held within [***] after delivery of the written notice described in the immediately preceding sentence. Each Party shall use reasonable efforts to cause its representatives to attend the meetings of the JSC. If a representative of a Party is unable to attend a meeting, such Party may designate an alternate to attend such meeting in place of the absent representative. Each Party shall bear all the expenses of its representatives on the JSC. Either Party may invite personnel or consultants of the Parties (other than the members of the JSC) having applicable expertise to participate in discussions of the JSC from time to time as appropriate to assist in the activities of the JSC; provided, that neither Party's legal counsel may attend a JSC meeting without prior notice to the other Party reasonably sufficient to allow the other Party's legal counsel to attend or to expressly waive attendance, and any such non-member shall be (x) bound by confidentiality and non-use obligations equivalent in scope to those set forth in ARTICLE XII of this Agreement and (y) under a written obligation to assign to the Party inviting such non-member any inventions of such non-member in the course of or as a result of attending any such meeting.

3.1.5 Minutes. The minutes of each JSC meeting shall be distributed to the members within [***] after the completion of the relevant meeting and shall provide a description in reasonable detail of the discussions held at the meeting and a list of any actions, decisions or determinations approved by the JSC. Minutes of each JSC meeting shall be approved or disapproved, and revised as necessary, within [***] after the applicable JSC meeting and shall be considered Confidential Information of both Parties.

3.1.6 Matter Resolution; Escalation.

- (a) The Alliance Managers shall use good faith efforts to mediate potential disagreements between the Parties regarding the Development Plans with the goal of resolving such disagreements without requiring escalation. If the Alliance Managers are unable to reach resolution on an issue for which the JSC has responsibility hereunder, the issue will be escalated to the JSC for review.
- (b) If the JSC is unable to reach unanimous consensus on a particular issue for which the JSC has responsibility hereunder within [***] after such issue is first presented to the JSC, such issue shall be referred to an executive officer of each Party or their designees selected for such purpose and authorized to resolve the issue and notified to other Party (the "Executive Officers") for resolution. Each Party shall use good faith efforts to resolve issues as promptly as practicable and at the lowest level possible within the governance structure established by this Article III and, notwithstanding the escalation procedures set forth herein, to limit any such escalations.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (c) If the Executive Officers are unable to resolve a particular issue referred to them pursuant to Section 3.1.6(b) within [***] after such issue is referred to them, then such issue shall be finally resolved pursuant to the dispute resolution provisions set forth in Article XV.
- (d) For the avoidance of doubt, other than amendments of the Development Plan in accordance with <u>Section 2.2.2</u>, the JSC shall not have the power to amend or waive compliance with this Agreement, determine any such issue in a manner that would conflict with the express terms and conditions of this Agreement, require any Party to perform any act that is inconsistent with applicable Law or, without the consent of the affected Party, to materially increase or reduce the obligations of the Parties under this Agreement.
 - 3.1.7 Limitations on JSC Responsibility after the Commercial Option Exercise Date; Annual Reports.
- (a) With respect to each Licensed Product for which Precision has not exercised the CDCP Option, from and after the Commercial Option Exercise Date with respect to such Licensed Product Candidate, the JSC's responsibility with respect to the corresponding Licensed Product shall be limited as follows: (i) the JSC will not have any decision making or oversight rights with respect to such Licensed Product; and (ii) the JSC's role shall be limited to reviewing and discussing reports provided by Baxalta as set forth in the following sentence. On an annual basis, Baxalta will be obligated to (x) deliver to Precision a report describing the status of the Development and Commercialization efforts with respect to such Licensed Product, for informational purposes only (but sufficient to establish Baxalta's compliance with its Development and Commercialization obligations under this Agreement and for Precision to comply with its disclosure obligations (if any) under any Law applicable to the public sale of securities or status as a public or listed company, provided that Precision has provided Baxalta a written request describing the type of information it needs in order to fulfill such disclosure obligations), and (y) upon Precision's request, meet with Precision through the JSC to discuss such report. In addition, within [***] after Baxalta's completion of any Phase II Clinical Trial with respect to a particular Licensed Product, Baxalta shall notify Precision of Baxalta's determination of whether such Licensed Product has an acceptable safety profile to proceed to Phase III Clinical Trials.
- (b) With respect to each Licensed Product for which Precision has exercised the CDCP Option, a separate joint co-development and co-promotion committee will be established for each such Licensed Product to oversee clinical Development, manufacturing, registration, Commercialization and marketing in the U.S. pursuant to the CDCP Agreement, and the JSC under this Agreement will have no role with respect to any such Licensed Product.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

3.2 <u>Alliance Managers</u>. Promptly after the Effective Date, each Party shall appoint an individual to act as alliance manager for such Party, which may be one of the representatives of such Party on the JSC (each, an "<u>Alliance Manager</u>"). The Alliance Managers shall be the primary point of contact for the Parties regarding the activities contemplated by this Agreement and shall facilitate all such activities hereunder and shall be responsible for progressing the alliance activities, otherwise facilitating communication and being the first line of dispute resolution. The Alliance Managers shall attend all meetings of the JSC and shall be responsible for assisting the JSC in performing its oversight responsibilities. The name and contact information for each Party's Alliance Manager, as well as any replacement(s) chosen by such Party, in its sole discretion, from time to time, shall be promptly provided to the other Party in accordance with Section 17.5. Each Party shall provide its Alliance Manager with sufficient resources for the Alliance Manager to perform his or her role under this Agreement.

ARTICLE IV. COMMERCIAL OPTIONS; LICENSES

4.1 Commercial Option.

- 4.1.1 <u>Commercial Option Exercise Period</u>. With respect to each Licensed Product Candidate Directed to a particular Included Target, subject to the terms and conditions of this Agreement, Precision hereby grants to Baxalta an exclusive option to obtain a Commercial License for such Licensed Product Candidate (the "<u>Commercial Option</u>"). Baxalta will be allowed to (but will not be obligated to) provide notice that it desires to exercise its Commercial Option, on a Licensed Product Candidate-by-Licensed Product Candidate basis, for each Licensed Product Candidate Directed to a particular Included Target at any time during the period beginning on the date on which Precision provides the Development Achievement Notice for such Licensed Product Candidate and ending [***] (such period, the "<u>Commercial Option Exercise Period</u>").
- 4.1.2 Commercial Option Exercise. Baxalta may begin the process for exercising the Commercial Option by providing written notice to Precision during the Commercial Option Exercise Period indicating its intent to exercise its Commercial Option with respect to the applicable Licensed Product Candidate (the "Commercial Option Exercise Notice"). Upon provision of the Commercial Option Exercise Notice with respect to a particular Licensed Product Candidate, (a) Precision will promptly initiate or continue, as applicable (at Precision's sole cost and expense) manufacturing and quality testing of a Phase II Ready Batch of such Licensed Product Candidate and (b) if Antitrust Clearance is required, the Parties will comply with their obligations under Section 17.17. The Commercial Option with respect to a Licensed Product Candidate shall be deemed exercised on the Commercial Option Exercise Date, and such Licensed Product Candidate shall thereafter be a Licensed Product. Notwithstanding anything to the contrary set forth in this Agreement, if Antitrust Clearance is required for the exercise of any Commercial Option and is not obtained within [***] following delivery of the Commercial Option Exercise Notice, the Commercial Option Exercise Period for the applicable Licensed Product Candidate will be deemed to have occurred without issuance of a Commercial Option Exercise Notice, and shall be treated in accordance with Section 4.1.4.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 4.1.3 <u>Batch Failure</u>. If Precision is unable to achieve Phase II Ready Batch Success within [***] after the date of the Commercial Option Exercise Notice with respect to a Licensed Product Candidate, then the JSC shall meet to consider (a) whether Phase II Ready Batch Success can be achieved, (b) whether it is commercially reasonable to continue to seek to achieve Phase II Ready Batch Success and, (c) upon positive determination of (a) and (b), to develop a plan for such achievement (the "<u>Batch Success Achievement Plan</u>"). Following receipt of the Batch Success Achievement Plan from the JSC, Precision will use Commercially Reasonable Efforts to perform the activities set forth in such plan as promptly as practicable. If, subsequent to the performance of the activities set forth in the Batch Success Achievement Plan, Phase II Ready Batch Success has not been achieved, the JSC shall meet to discuss the considerations set forth in clauses (a) through (c) above, and any further obligations of Precision in relation to manufacturing and quality testing of the Phase II Ready Batch shall be subject to the JSC's approval of an additional Batch Success Achievement Plan.
- 4.1.4 Effect of Failure to Deliver Commercial Option Exercise Notice. If Precision provides a Development Achievement Notice for a Licensed Product Candidate Directed to a particular Included Target and Baxalta does not provide a Commercial Option Exercise Notice with respect to such Licensed Product Candidate during the applicable Commercial Option Exercise Period, then:
- (a) during any period in which (i) Baxalta holds a Commercial License for any one or more Licensed Products Directed to such Included Target, or (ii) a different Licensed Product Candidate Directed to such Included Target is currently the subject of Development activities under a Development Plan pursuant to Section 2.2, then, subject to Section 4.1.4(b), upon the first date following the expiration of such Commercial Option Exercise Period, neither Party shall be required nor permitted to conduct any further Development or Commercialization activities under this Agreement with respect to the Licensed Product Candidate for which Precision provided the Development Achievement Notice; and
- (b) during any period in which (i) Baxalta does not hold a Commercial License for any one or more Licensed Products Directed to such Included Target, and (ii) there is no Licensed Product Candidate Directed to such Included Target that is currently the subject of Development activities under a Development Plan pursuant to Section 2.2, then Precision shall notify Baxalta of remaining Variants (which may include Unselected Candidates), if any, that Precision believes in good faith are capable of being generated by Precision via Precision Platform Technology for such Included Target in accordance with Section 2.1.5(b). If (x) (i) the JSC determines that no such Variants exist, or (ii) the JSC does not approve a Development Plan for any one or more of such Variants for which Baxalta has expressed interest in accordance with the process set forth in Section 2.1.5(b) within [***] after the Final Candidate Proposal Date, and (y) Baxalta does not provide a Commercial Option Exercise Notice with respect to any other Licensed Product Candidate Directed to such Included Target (which may include a Licensed Product Candidate for which Baxalta has previously declined to timely provide a Commercial Option

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Exercise Notice) within [***] after the Final Candidate Proposal Date, then notwithstanding anything to the contrary in this Agreement, Baxalta shall be deemed to have terminated this Agreement with respect to such Included Target (including all Licensed Product Candidates, Unselected Candidates, and Variants Directed to such Included Target) pursuant to <u>Section 14.2.5</u>.

- 4.2 <u>Development Transfer</u>. With respect to a particular Licensed Product, as promptly as practicable following the Commercial Option Exercise Date for such Licensed Product, Precision shall deliver to Baxalta a copy of all Precision Know-How then existing that relates to such Licensed Product, and a description of the status of the Development efforts to date for such Licensed Product.
- 4.3 <u>Licenses to Baxalta</u>. Effective immediately upon the Commercial Option Exercise Date for a particular Licensed Product Candidate in accordance with <u>Section 4.1</u> (or if applicable, <u>Section 14.3.2(b)</u>, <u>Section 14.3.2(c)</u>, <u>Section 16.2.4</u>, or <u>Section 16.3.4</u>) subject to the terms and conditions of this Agreement, Precision hereby grants to Baxalta the following licenses:
- 4.3.1 Precision hereby grants to Baxalta an exclusive (even as to Precision except as set forth in Section 4.3.3), royalty-bearing (in accordance with Section 8.4), transferable (in accordance with Section 17.1), sub-licensable (in accordance with Section 4.4) right and license under the Precision Patents and Precision Know-How, in each case to make, have made, use, sell, offer for sale, import and otherwise Develop and Commercialize such Licensed Product with respect to the applicable Included Target in the Field in the Territory, in each case alone or in combination with other products or therapies (with respect to each such Licensed Product and Included Target, for so long as such license remains in effect during the Term, a "Commercial License"). For clarity, the foregoing license does not include any rights under the Precision Platform Patents or Precision Platform Technology or any rights with respect to other products or therapies with which a Licensed Product may be combined.
- 4.3.2 Precision hereby grants to Baxalta a non-exclusive, royalty-bearing (in accordance with <u>Section 8.4</u>), transferable (in accordance with <u>Section 17.1</u>), sub-licensable (in accordance with <u>Section 4.4</u>) right and license under the Precision Platform Patents solely to the extent necessary to permit Baxalta's conduct of the activities licensed to Baxalta under the Precision Patents and Precision Know-How under <u>Section 4.3.1</u> with respect to such Licensed Product.
- 4.3.3 Notwithstanding the exclusive licenses set forth in <u>Section 4.3.1</u>, and subject to the terms and conditions of this Agreement (including the Precision Restrictive Covenants), Precision and its Affiliates shall retain the right under the Precision Patents and Precision Know-How to:
 (a) practice the Precision Patents and Precision Know-How to exercise its rights and perform its obligations under this Agreement; (b) conduct research related to the Precision Platform Technology; and (c) practice and license Precision Patents and Precision Know-How outside the scope of the licenses granted to Baxalta under <u>Section 4.3.1</u>.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

4.4 Sublicenses.

- 4.4.1 Baxalta may exercise its rights and perform its obligations under this Agreement by itself or through the engagement of any of its Affiliates without Precision's prior written consent.
- 4.4.2 Baxalta may grant sublicenses of the rights granted to it under Section 4.3 to one or more Third Parties without Precision's prior written consent [***]. Notwithstanding the foregoing, Baxalta may not grant a sublicense to any potential Sublicensee if such potential Sublicensee, its Affiliates or its or their respective Agents (a) has ever been debarred or is subject to debarrent or, to Baxalta's knowledge after reasonable inquiry, convicted of a crime for which a Person could be debarred before a Regulatory Authority under applicable Laws, or (b) to such Baxalta's knowledge after reasonable inquiry, has ever been under indictment for a crime for which a Person could be debarred under such Laws.
- 4.4.3 Baxalta shall remain responsible for all of its obligations under this Agreement that have been delegated, subcontracted or sublicensed to any of its Affiliates, Sublicensees or subcontractors. All such delegation, subcontracting and sublicensing shall be established by written agreements consistent with the terms and conditions of this Agreement in all material respects (including without limitation written obligations consistent with those required under Section 11.1.1(a) and written obligations of confidentiality and non-use consistent with this Agreement), and Baxalta shall be fully responsible for any breach of this Agreement by any of its Affiliates, Sublicensees or subcontractors. In addition, each sublicense granted under any one or more of the Precision Patents and/or Precision Platform Patents must grant the same scope of rights under each such Patent that is included in such sublicense grant.
- 4.5 No Other Licenses. Neither Party grants to the other Party any rights, licenses or covenants in or to any intellectual property, whether by implication, estoppel, or otherwise, other than the license rights that are expressly granted under this Agreement. Without limiting the foregoing, and except for the limited rights set forth in Section 16.2.2 or Section 16.3.2, if applicable, nothing in this Agreement shall be deemed to grant Baxalta any right to access or receive any Precision Platform Technology or to design, Develop or manufacture any genome engineering tools, or any precursor, intermediate or construct, within the Precision Platform Technology. Baxalta acknowledges and agrees that rights under certain Precision Patents and Precision Platform Patents are licensed to Precision by Cellectis S.A. (the "Cellectis Patents") under that certain Patent Cross-License Agreement between Cellectis S.A. and Precision dated January 23, 2014 (the "Cellectis Agreement"), and, notwithstanding any exclusive license granted to Baxalta under this Agreement, (a) Cellectis S.A. retains rights under the Cellectis Patents and is not restricted from granting rights to Third Parties under the Cellectis Patents, (b) any licenses and rights granted by Precision to Baxalta under the Cellectis Patents are granted only within the permissible scope of sublicenses granted under the Cellectis Agreement, and (c) pursuant to the Cellectis Agreement, Cellectis S.A. retains non-exclusive rights under certain Precision Patents and Precision Platform Patents identified in the Cellectis Agreement, which may be further sublicensed by Cellectis S.A. without Precision control or consent. Baxalta acknowledges and

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

agrees that any exercise of any right by Cellectis S.A, or by any Third Party through Cellectis S.A, under the Cellectis Agreement shall not constitute a breach of this Agreement by Precision. Each Party shall not, and shall not permit any of its Affiliates or sublicensees to, practice any Patent Rights or Know-How licensed to it by the other Party outside the scope of the license granted to it under this Agreement.

ARTICLE V. REGULATORY MATTERS

- 5.1 <u>Regulatory Roles</u>. The Parties shall have responsibility for the following activities with respect to Regulatory Filings for each Licensed Product Candidate or Licensed Product, as applicable (each Party's activities, its "<u>Regulatory Role</u>"):
- 5.1.1 With respect to each Licensed Product Candidate, Precision shall (subject to the oversight of the JSC, as provided in Section 3.1.1) prepare and file all Regulatory Filings (including an IND) for such Licensed Product Candidate that are to be filed at or prior to Phase II Ready Status for such Licensed Product Candidate pursuant to the applicable Development Plan, in each case in the name of Precision, and shall prepare and file all amendments to such Regulatory Filings that are required by applicable Regulatory Authorities or that are otherwise necessary to comply with applicable Laws. Without limiting the foregoing, Precision shall be responsible for attendance at all meetings (whether occurring in person or by telephone or other remote means) with applicable Regulatory Authorities with respect to each such Licensed Product Candidate; provided that Precision, where permitted by applicable Law, shall permit or may require Baxalta to designate up to two (2) Baxalta employees to attend such meetings. As promptly as practicable following and, in any event, unless prohibited by applicable regulations, within [***] following, the Commercial Option Exercise Date for a particular Licensed Product Candidate, Precision shall transfer and, if applicable, assign to Baxalta all clinical trial exemptions, all Regulatory Filings (including the IND file and supporting master files) and worldwide sponsorship for such Licensed Product Candidate, and will provide a copy of all related regulatory documents and regulatory information (including clinical, safety/pharmacovigilance and other data) in Precision's Control, including all communications with Regulatory Authorities, source documentation, analysis files, source program files, validation reports, study reports, Clinical Trial raw data sets (including all collected raw data, including, for example CRF data and lab data) in electronic form (or, if not available electronically, in paper form), study submission data sets in submission-ready electronic form, study documentation and all other documents and data reasonably necessary for Regulatory Filings, in each case regarding such Licensed Product Candidate; provided, that all such data, documents and other information, including the design history file (DHF) established to support such Licensed Product Candidate as required for any Regulatory Filings, shall be transferred to Baxalta using a mutually acceptable secure and validated transfer method; provided, further, that Precision shall be entitled to retain copies of any or all of the foregoing.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 5.1.2 Following the Commercial Option Exercise Date for each Licensed Product Candidate, Baxalta shall, at Baxalta's sole cost and expense, be solely responsible for and, without limiting Baxalta's obligations under Section 2.2.6, have sole control of all clinical, nonclinical and quality studies, creation of the Global Dossier and all regulatory matters (including ownership of all Regulatory Filings and all correspondence with Regulatory Authorities) for such Licensed Product, and shall use Commercially Reasonable Efforts to prepare and file all Regulatory Filings necessary to obtain Marketing Approvals and Reimbursement Approvals, and thereafter maintain such Marketing Approvals and Reimbursement Approvals, throughout the Territory in the name of Baxalta with respect to such Licensed Product. Baxalta shall own and be the license holder for all Marketing Approvals and Reimbursement Approvals for the Licensed Products and shall be responsible for complying with all requirements of Regulatory Authorities in the Territory after the Commercial Option Exercise Date.
 - 5.1.3 Each Party shall bear its own costs and expenses for effecting its respective Regulatory Role.
- 5.2 Cooperation; Effort. Each Party will, at its sole cost and expense, cooperate with the other Party in providing technical regulatory expertise for assistance in developing the submission strategy for Regulatory Filings and defining technical content and will provide reasonable support to the other Party to ensure timely Regulatory Filings and other regulatory submissions reasonably necessary to obtain Marketing Approvals and Reimbursement Approvals, and any post-Marketing Approval or post-Reimbursement Approval Regulatory Filings or other regulatory submissions, in each case for each Licensed Product Candidate or Licensed Product, as applicable. Additionally, Precision shall provide to Baxalta, as promptly as practicable following Baxalta's request and at Baxalta's sole cost and expense, such assistance, cooperation and input (including documents and data) reasonably necessary for Baxalta to obtain Marketing Approvals and Reimbursement Approvals, together with any post-Marketing Approval or post-Reimbursement Approval Regulatory Filings or other regulatory submissions, in each case with respect to each Licensed Product, including information, data and documents reasonably necessary for Baxalta's creation of the Global Dossier. Baxalta shall designate a global regulatory affairs representative and Precision shall invite such representative to attend any substantive in-person or other meetings (including telephonic meetings) with Regulatory Authorities associated with Precision's Regulatory Role. Through the JSC, Baxalta shall have the right to review the content of all Regulatory Filings for each Licensed Product Candidate that are prepared or filed by Precision, and Precision agrees to consider in good faith Baxalta's comments with respect to each such Regulatory Filing; provided, that Precision shall provide such proposed Regulatory Filings to the members of the JSC at least [****] prior to filing such Regulatory Filings, together with the proposed filing date.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 5.3 <u>Right of Reference</u>. Each Party hereby grants to the other Party a "Right of Reference", as that term is defined in 21 C.F.R. § 314.3(b), to any data Controlled by such Party or its Affiliates (a) that relates to the Licensed Product Candidates or Licensed Products and (b) that each Party reasonably believes may be necessary or useful to the Development, manufacture or Commercialization of any Licensed Product Candidate or Licensed Product pursuant to this Agreement, and each Party will provide a signed statement to the foregoing effect, if so requested by the other Party in accordance with 21 C.F.R. § 314.50(g)(3).
- 5.4 Adverse Event Reporting. Promptly after the transfer by Precision to Baxalta of the Regulatory Filings with respect to a Licensed Product, Baxalta and Precision shall discuss in good faith whether their respective activities would require them to enter into a pharmacovigilance and adverse event reporting agreement setting forth the worldwide pharmacovigilance procedures for Baxalta and Precision with respect to the Licensed Product, such as safety data sharing, adverse event reporting and prescription events monitoring (the "Safety Agreement"). If Baxalta and Precision agree that a Safety Agreement is necessary or otherwise advisable, the Parties will work together in good faith to execute a Safety Agreement for such Licensed Product no later than [***] after transfer by Precision to Baxalta of the Regulatory Filings with respect to the applicable Licensed Product, provided that the procedures set forth in such agreement shall be in accordance with, and enable the Parties to fulfill, local and national regulatory reporting obligations under applicable Laws. In any event, Baxalta shall maintain the global adverse event database for each Licensed Product in the Territory at its cost and shall be responsible for adverse event reporting in accordance with applicable Laws related to the Licensed Product to the applicable Regulatory Authorities in the Territory, as well as responding to safety issues and to all requests of Regulatory Authorities related to the Licensed Products in the Territory. Each Party hereby agrees to comply with its respective obligations under each Safety Agreement and to cause its Affiliates and sublicensees to comply with such obligations.
- 5.5 Notification of Threatened Action. Each Party shall promptly notify the other Party of any information it receives regarding any threatened or pending action, inspection or communication by any Regulatory Authority, which such Party believes is reasonably likely to have a material adverse effect on any rights or licenses granted in this Agreement, including as may affect the claims of any Licensed Product Candidate or Licensed Product or the continued marketing of any Licensed Product. Upon receipt of such information, Baxalta and Precision shall consult with each other in an effort to arrive at a mutually acceptable procedure for taking appropriate action; however, Baxalta shall have final decision making authority with respect to any such action related to Licensed Product Candidates or Licensed Products. Baxalta shall have the right to determine whether or not to continue the marketing of any Licensed Product in the Field in any jurisdiction based on communications by Regulatory Authorities.
- 5.6 <u>Remedial Actions</u>. Each Party shall notify the other Party promptly, and promptly confirm such notice in writing, if it obtains information indicating that any Licensed Product Candidate or Licensed Product may be subject to any recall, corrective action or other regulatory action with respect to such Licensed Product Candidate or Licensed Product in the Field taken by virtue of applicable Laws (a "<u>Remedial Action</u>"). The Parties shall assist each other in gathering and evaluating such information as is necessary to determine the necessity of conducting a Remedial Action. Each Party shall, and shall ensure that its Affiliates and sublicensees will,

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

maintain adequate records to permit the Parties to trace the manufacture of each Licensed Product Candidate and Licensed Product and the distribution and use of each Licensed Product Candidate and Licensed Product. Precision shall have sole discretion with respect to any matters relating to any Remedial Action directed towards any Licensed Product Candidate, including the decision to commence such Remedial Action and the control over such Remedial Action, at its sole cost and expense. Baxalta shall have sole discretion with respect to any matters relating to any Remedial Action directed towards any Licensed Product, including the decision to commence such Remedial Action and the control over such Remedial Action, at its sole cost and expense.

ARTICLE VI. MANUFACTURING AND SUPPLY

- 6.1 <u>Manufacturing and Supply Roles</u>. The Parties shall have responsibility for the following activities with respect to manufacture and supply of each Licensed Product Candidate or Licensed Product, as applicable (each Party's activities, its "<u>Manufacturing and Supply Role</u>"):
- 6.1.1 With respect to each Licensed Product Candidate, Precision shall be solely responsible (subject to the oversight of the JSC), by itself or through one or more CMOs approved as set forth in <u>Section 6.2</u>, to use Commercially Reasonable Efforts to manufacture and supply pre-clinical and clinical trial materials necessary for Development of such Licensed Product Candidate through Phase II Ready Batch Success, including, as applicable, the Phase II Ready Batch.
- 6.1.2 With respect to each Licensed Product, Baxalta shall be solely responsible, by itself or through one or more CMOs, to use Commercially Reasonable Efforts to (a) manufacture and supply such Licensed Product for Development use after the Phase II Ready Batch and (b) manufacture and supply such Licensed Product for Commercialization, in each case for use in the Field in the Territory.
- 6.1.3 Following the Commercial Option Exercise Date for a Licensed Product Candidate, at such time as Baxalta notifies Precision in writing that Baxalta will initiate the manufacture of clinical or commercial supply of any Licensed Product itself or will engage a CMO to conduct such manufacture and supply pursuant to Baxalta's obligations under Section 6.1.2, Precision shall, at Baxalta's cost and expense, use Commercially Reasonable Efforts to perform a technology transfer of Precision's (or its CMOs') manufacturing process for the corresponding Licensed Product to Baxalta (or its designated Affiliate or CMO) pursuant to the applicable Technology Transfer Plan for such Licensed Product. Promptly following such written notice from Baxalta, the Parties shall establish the Technology Transfer Plan for the applicable Licensed Product, for the purpose of enabling technology transfer of Precision Know-How sufficient to allow Baxalta (or its Affiliates or CMOs) to manufacture and supply the Licensed Product for further Development and Commercialization of such Licensed Product. Baxalta shall pay Precision (or its Affiliates or CMOs) the Hourly Rate with respect to technical support provided for each hour of activities undertaken pursuant to the applicable Technology Transfer Plan. For

purposes of this Section 6.1.3, the "Hourly Rate" means [***]. Such technology transfer shall be subject to reasonable and appropriate restrictions on use and disclosure of Precision Know-How, including appropriate confidentiality agreements between Precision and any Third Party that is involved. Any representatives of Baxalta or its Affiliate or CMO, while present at Precision's facilities, shall comply with all terms and conditions of this Agreement and such rules, regulation, policies and procedures as are from time to time applicable to Precision's employees and such other reasonable requests as Precision may make from time to time. Without limiting the foregoing, Precision may isolate or require such representatives to abide by firewalls or other protective procedures designed to protect the confidential and proprietary information of Precision and Third Parties to which Baxalta does not have access rights under this Agreement. Baxalta shall be responsible for all activities of such representatives under this Agreement. In no event shall Precision be required to disclose or make available any Precision Platform Technology. The Parties acknowledge and agree that Baxalta or its CMO(s) shall hold all manufacturing licenses with applicable Regulatory Authorities for the Licensed Products and shall be solely responsible through itself or one or more CMOs for such manufacture.

- 6.1.4 Except to the extent otherwise expressly provided in <u>Section 6.1.3</u>, each Party shall bear its own costs and expenses for effecting its respective Manufacturing and Supply Role. For clarity, Precision has no obligation to perform any manufacture or supply of Licensed Products for Commercialization.
- 6.2 <u>Supply Agreement</u>. On or before a date to be established by the JSC but in no event later than [***] prior to initiation of manufacture of any Licensed Product Candidate that will be used by Baxalta in Clinical Trials, the Parties shall enter into a Supply Agreement pursuant to which Precision or its CMO, approved by the JSC as set forth below, will manufacture and supply clinical trial materials for Phase II Clinical Trials, including Phase II Ready Batches, to Baxalta. The terms of such Supply Agreement shall be negotiated in good faith by the Parties and will contain the terms attached hereto as <u>Exhibit B</u> and other customary terms and conditions that are consistent with this Agreement. All CMOs manufacturing and supplying clinical trial materials of Licensed Products for Clinical Trials must be approved by the JSC, except that Precision and [***] are deemed to be approved suppliers as of the Effective Date. The Supply Agreement will provide that Precision, and any contract service provider used by Precision, must maintain supplier compliance/cGMP status and ensure the production of any clinical product meets cGMP requirements.
- 6.2.1 Based on risk, the JSC will determine if other contract service providers in the Licensed Product supply chain will be required to be approved by the JSC.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

6.3 Quality Agreement. In connection with the negotiation and execution of a Supply Agreement, the Parties shall also enter into a separate agreement governing the quality control, quality assurance and validation (the "Quality") of any clinical trial materials for Phase II Clinical Trials delivered by Precision (or its CMO) to Baxalta under the Supply Agreement, including the requirement that Baxalta and Precision will implement a joint quality team to facilitate communication and consensus on issues related to quality of such clinical trial materials (the "Quality Agreement"). The Quality Agreement shall be negotiated in good faith by the Parties and will contain the terms attached hereto as Exhibit C and other customary terms and conditions that are consistent with this Agreement. The Quality Agreement will provide that Precision will have final functional Quality decision-making authority with respect to the clinical trial materials for Phase II Clinical Trials delivered to Baxalta and that Precision will have responsibility for Quality oversight of all Third Party manufacturers of such clinical trial materials that are engaged by Precision; provided, that no clinical trial materials may be used in humans without approval by the JSC.

ARTICLE VII. COMMERCIALIZATION OF LICENSED PRODUCTS

- 7.1 <u>Commercialization Responsibility</u>. Subject to <u>Section 7.2</u> and <u>Section 7.3</u>, Baxalta shall be solely responsible for and have sole control of all Commercialization activities for Licensed Products in the Field in the Territory, at Baxalta's sole cost and expense.
- 7.2 <u>Diligence</u>. After receiving Marketing Approval and, in countries of the EU, Reimbursement Approval, of a particular Licensed Product, for a country in the Territory, Baxalta shall use Commercially Reasonable Efforts to Commercialize such Licensed Product in such country in accordance with the terms of this Agreement and in compliance with all applicable Laws. With respect to determinations by Baxalta to discontinue Commercialization activities for a particular Licensed Product in all countries throughout the Territory, Baxalta shall give Precision written notice of such determination within a reasonable period of time, but in any event within [***] after any such determination is made, and upon provision of such notice Baxalta shall be deemed to have terminated this Agreement pursuant to <u>Section 14.2.5</u> with respect to such Licensed Product.

7.3 Co-Development and Co-Promote Option.

- 7.3.1 On a Licensed Product-by-Licensed Product basis, Baxalta hereby grants to Precision an exclusive option to co-Develop and co-Promote such Licensed Product with Baxalta in the United States, and to share profits from the sales of any such co-Developed and co-Promoted Licensed Product in the United States with Baxalta, pursuant to the terms and conditions of <u>Section 7.3.2</u> and a CDCP Agreement (each, a "<u>CDCP Option</u>").
- 7.3.2 Upon Precision's written request delivered to Baxalta within [***] after the Commercial Option Exercise Date for a particular Licensed Product, Baxalta shall prepare in good faith a summary Development plan for such Licensed Product for the United States and an estimated budget for the Development activities set forth in such plan. Baxalta shall provide a

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

copy of such plan and budget to Precision within [***] of such request by Precision. Precision may exercise the CDCP Option for such Licensed Product by providing Baxalta with written notice within [***] after Precision's receipt of such plan and budget for such Licensed Product. The CDCP Option shall be exercisable by the delivery by Precision of written notice to Baxalta. Upon exercise of the CDCP Option for a Licensed Product, the following shall apply:

- (a) Subject to Section 7.3.4, Baxalta and Precision shall promptly enter into good faith negotiations of and execute a co-Development, co-Promotion and profit share agreement consistent with the key terms set forth in Exhibit D as described in Section 7.3.3 (and in any event, no later than [***] after the date of exercise of the CDCP Option), pursuant to which the Parties will, notwithstanding anything to the contrary in this Agreement, equally share all remaining Development, manufacturing and Commercialization costs, and all profits and losses, with respect to the applicable Licensed Product in the United States (each, a "CDCP Agreement"), and
- (b) On or prior to [***] after the execution of a CDCP Agreement with respect to the first Licensed Product Directed to a particular Included Target for which the CDCP Option is exercised, following Precision's receipt of Baxalta's invoice therefor, Precision shall pay [***] (the "CDCP Option Fee") to Baxalta.

For the avoidance of doubt, after payment of the CDCP Option Fee with respect to a Licensed Product Directed to a particular Included Target, Precision may elect to exercise the CDCP Option for any subsequent Licensed Product Directed to the same Included Target without payment of an additional CDCP Option Fee.

- 7.3.3 The key terms of the CDCP Agreement are set forth in <u>Exhibit D</u>. The Parties will negotiate in good faith a form of CDCP Agreement consistent with such terms and other terms and conditions that are typical for such agreements with respect to similarly situated products between similarly situated parties in the pharmaceutical industry promptly after the Effective Date (and in any event, no later than [***] after the Effective Date), and such form of CDCP Agreement shall be attached to this Agreement as <u>Exhibit D-1</u> upon mutual agreement between the Parties.
- 7.3.4 In the event the Parties are unable to agree upon the terms of any CDCP Agreement, including the form of CDCP Agreement described in <u>Section 7.3.3</u>, the Parties will resolve the disagreement in accordance with the dispute resolution process set forth in <u>ARTICLE XV</u>.
- 7.3.5 During any period in which a CDCP Agreement is in effect with respect to a particular Licensed Product, notwithstanding anything to the contrary in <u>ARTICLE VIII</u>: (a) the obligation for Baxalta to pay [***] shall cease to apply; (b) the Calendar Year global Net Sales of such Licensed Product for purposes of calculating the Sales Milestones shall exclude [***]; and (c) the cumulative Calendar Year global Net Sales of such Licensed Product, for purposes of determining royalty rates payable under Section 8.4.1, shall [***].

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE VIII. PAYMENTS

- 8.1 <u>Upfront Fee</u>. In consideration of the Commercial Options, licenses, technology access and other rights granted to Baxalta under this Agreement, in addition to the payments specified in <u>Section 8.2</u>, <u>Section 8.3</u> and <u>Section 8.4</u>, Baxalta shall pay to Precision, on or prior to the date that is fifteen (15) days after the Effective Date or, if such date is not a Business Day, on the next Business Day, a one time, non-refundable, non-creditable fee equal to one hundred five million USD (\$105,000,000) (the "<u>Upfront Fee</u>"). Precision may invoice Baxalta for the Upfront Fee on the Effective Date.
- 8.2 <u>Commercial Option Exercise Fee</u>. In consideration for exercising its Commercial Option with respect to a particular Included Target, Baxalta shall pay to Precision a one-time, non-creditable, non-refundable exercise fee with respect to such Included Target, [***] (the "<u>Commercial License Fee</u>"). Precision may invoice Baxalta for the Commercial License Fee upon the Commercial Option Exercise Date for the first Licensed Product Candidate Directed to each Included Target. Baxalta shall pay such invoice within [***] after receipt of such invoice.

	Commercial
Achievement of Phase II Ready Batch Success	License Fee
On or prior to the date that is [***]	[***]
After the date that is [***] but on or prior to the date that is [***].	[***]
After the date that is [***]	[***]

[***]

8.3 <u>Milestones</u>. In addition to the payments specified in <u>Section 8.1</u>, <u>Section 8.2</u> and <u>Section 8.4</u>, Baxalta shall, conditioned upon achievement of the applicable milestone, for each Licensed Product Candidate and Licensed Product, make the following payments to Precision in consideration of the licenses, technology access and other rights granted to Baxalta under this Agreement:

8.3.1 <u>Development Milestones</u>.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

(a) Baxalta shall pay to Precision the following non-creditable, non-refundable amounts for the first achievement of the following milestone events for each Licensed Product Candidate (as applicable) (each, a "<u>Development Milestone</u>"):

	Milestone
Development Milestone	Payment (USD)
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

(b) In addition, with respect to each Licensed Product, Baxalta shall pay to Precision the following non-creditable, non-refundable amounts for the first achievement of the following milestone events for such Licensed Product by Baxalta or any of its Affiliates or Sublicensees (each, also a "Development Milestone"):

	Milestone
Development Milestone	Payment (USD)
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

(c) As to the Licensed Product Candidate or Licensed Product, as applicable, each applicable milestone payment set forth in this Section 8.3.1 shall be payable by Baxalta upon the first achievement of the applicable Development Milestone [***]. In each case, Precision shall provide Baxalta with a corresponding invoice and Baxalta shall pay to Precision the applicable milestone payment described above within [***] after receipt of such invoice.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

8.3.2 <u>Launch Milestones</u>. As to each Licensed Product, Baxalta shall pay to Precision the following non-creditable, non-refundable amounts for the first achievement of the following milestone events for such Licensed Product by Baxalta of any of its Affiliates or Sublicensees (each, a "<u>Launch Milestone</u>"):

Launch Milestone	Milestone Payment (USD)
First Commercial Sale of the Licensed Product in [***]	[***]
First Commercial Sale of the Licensed Product in [***]	[***]
First Commercial Sale of the Licensed Product in [***]	[***]
First Commercial Sale of the Licensed Product in [***]	[***]
First Commercial Sale of the Licensed Product in [***]	[***]
First Commercial Sale of the Licensed Product in [***]	[***]

[***]

Each milestone payment set forth in this Section 8.3.2 shall be payable by Baxalta upon the first achievement of the applicable Launch Milestone, and Baxalta shall provide notice to Precision of such achievement within [***] following such achievement. Following Precision's receipt of a notice described in the immediately preceding sentence, Precision shall provide Baxalta with a corresponding invoice and Baxalta shall pay to Precision the applicable milestone payment described above within [***] after receipt of such invoice.

8.3.3 <u>Sales Milestones</u>. As to each Licensed Product, Baxalta shall pay to Precision the following non-creditable, non-refundable amounts for the first achievement of the following milestone events for such Licensed Product (each, a "<u>Sales Milestone</u>"):

Sales Milestone	Milestone Payment (USD)
First achievement of [***]	[***]
First achievement of [***]	[***]
First achievement of [***]	[***]

For the avoidance of doubt, (a) each of the milestone payments set forth above in this Section 8.3.3 shall be payable no more than one time for any particular Licensed Product, (b) Net Sales of the applicable Licensed Product shall be aggregated globally without regard to the identity of the selling entity in any particular country of the Territory (e.g., Baxalta or any of its Affiliates or Sublicensees) for purposes of determining whether the above Net Sales thresholds have been achieved, and (c) Net Sales of different Licensed Products shall not be aggregated for purposes of determining whether the above Net Sales thresholds have been achieved.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Each milestone payment set forth in this <u>Section 8.3.3</u> shall be payable by Baxalta upon the first achievement of the applicable Sales Milestone, and Baxalta shall provide notice to Precision of such achievement within [***] following such achievement. Following Precision's receipt of a notice described in the immediately preceding sentence, Precision shall provide Baxalta with a corresponding invoice and Baxalta shall pay to Precision the applicable milestone payment described above within [***] after receipt of such invoice.

8.4 Royalty Payment; Audits.

8.4.1 <u>Royalty Payments</u>. In addition to the payments specified in <u>Section 8.1</u>, <u>Section 8.2</u> and <u>Section 8.3</u>, in consideration of the licenses, technology access and other rights granted to Baxalta under this Agreement, on a Licensed Product-by-Licensed Product basis, Baxalta shall pay to Precision non-creditable, non-refundable royalty payments at the following royalty rates on the applicable portion of cumulative Calendar Year global Net Sales of the applicable Licensed Product:

Cumulative Calendar Year Global Net Sales of the Applicable Licensed Product	Royalty Rate
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

[***]

[***]

8.4.2 <u>Royalty Stacking</u>. Subject to the terms herein, if Baxalta, its Affiliate or Sublicensee enters into one or more Third Party License Agreement(s) with respect to a particular Licensed Product in a particular country or countries, then Baxalta's obligation to pay royalties to Precision with respect to sales of such Licensed Product in such country or countries in a particular Calendar Quarter shall be reduced by [***] provided that no royalty payment to Precision for a Licensed Product hereunder shall be reduced, pursuant to this <u>Section 8.4.2</u>, to less than [***] of the royalty payment that would otherwise be due to Precision in the absence of a reduction pursuant to this <u>Section 8.4.2</u>.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

8.4.3 <u>Royalty Term</u>. Baxalta's obligation to pay royalties pursuant to this Section 8.4 shall expire, on a country-by-country and Licensed Product-by-Licensed Product basis, upon the last to occur of: (a) the expiration of the last to expire Valid Claim of all Precision Patents Covering such Licensed Product, (b) expiration of all Regulatory Exclusivity with respect to such Licensed Product in the applicable country of sale, and (c) ten (10) years after the First Commercial Sale of such Licensed Product in such country (with respect to a particular Licensed Product in a particular country, the "<u>Royalty Term</u>"). [***]

8.4.4 <u>Royalty Reductions</u>. Notwithstanding <u>Section 8.4.3</u>, if, during any portion of the Royalty Term, there is no [***], the royalty payments due pursuant to <u>Section 8.4.1</u> during such portion of the Royalty Term for sales of such Licensed Product in such country shall be reduced to [***].

8.4.5 [***].

8.4.6 <u>Royalty Payment Timing</u>; <u>Royalty Reports</u>. Within [***] following the end of each Calendar Quarter during which royalties accrue, Baxalta shall provide Precision with a Sales Report and any other information reasonably required by Precision for the purpose of calculating royalties and Sales Milestone payments due under this Agreement. Any royalty payments due to Precision will be paid on the date of delivery of such Sales Report. In the event that either Party determines that the calculation of Net Sales for a Calendar Quarter deviates from the amounts previously reported to Precision for any reason (such as, on account of additional amounts collected or Licensed Product returns), Baxalta and Precision shall reasonably cooperate to reconcile any such deviations to the extent necessary under applicable legal or financial reporting requirements.

8.4.7 <u>Audit</u>. Until the expiration of all royalty payment obligations hereunder and for a period of [***] thereafter, Baxalta shall keep complete and accurate records pertaining to the sale or other disposition of Licensed Products by Baxalta, its Affiliates and Sublicensees in sufficient detail to permit Precision to confirm the accuracy of the royalties and Sales Milestone payments due hereunder. Precision shall have the right to cause an independent internationally recognized accounting firm reasonably acceptable to Baxalta to audit such records for the sole purpose of confirming Net Sales and royalties for a period covering not more than the preceding [***]. Baxalta may require such accounting firm to execute a reasonable confidentiality agreement with Baxalta prior to commencing the audit. Such audits may be conducted during normal business hours upon reasonable prior written notice to Baxalta, but no more frequently than [***]. No accounting period of Baxalta shall be subject to audit more than one time by Precision, unless after an accounting period has been audited by Precision, Baxalta restates its financial results for such accounting period, in which event Precision may conduct a second audit of such accounting period

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

in accordance with this <u>Section 8.4.7</u>. Adjustments (including remittances of underpayments or overpayments disclosed by such audit) shall be made by the Parties to reflect the results of such audit, which adjustments shall be paid (plus interest as set forth in <u>Section 8.6</u>) promptly following receipt of an invoice therefor. Precision shall bear the full cost and expense of such audit unless such audit discloses an underpayment by Baxalta of [***] or more of the amount of royalties due under this Agreement for the audited period, in which case Baxalta shall bear and reimburse Precision for the full cost and expense of such audit.

- 8.5 <u>Taxes</u>. All payments under this Agreement shall be made without any deduction or withholding for or on account of any tax, except as set forth in this <u>Section 8.5</u>. The Parties agree to cooperate with one another and use reasonable efforts to minimize under applicable Law obligations for any and all income or other taxes required by applicable Law to be withheld or deducted from any of the royalty and other payments made by or on behalf of a Party hereunder ("<u>Withholding Taxes</u>"). The applicable paying Party under this Agreement (the "<u>Paying Party</u>") shall, if required by applicable Law, deduct from any amounts that it is required to pay to the recipient Party hereunder (the "<u>Recipient Party</u>") an amount equal to such Withholding Taxes; provided that the Paying Party shall give the Recipient Party reasonable notice prior to paying any such Withholding Taxes. Such Withholding Taxes shall be paid to the proper taxing authority for the Recipient Party's account and, if available, evidence of such payment shall be secured and sent to Recipient Party within [***] after such payment. The Paying Party shall, at the Recipient Party's sole cost and expense, as mutually agreed by the Parties, do all such lawful acts and things and sign all such lawful deeds and documents as the Recipient Party may reasonably request to enable the Parties to avail themselves of any applicable legal provision or any double taxation treaties with the goal of paying the sums due to the Recipient Party hereunder without deducting any Withholding Taxes. For US federal income tax purposes, Baxalta will report the payments made under this Agreement in the manner required by the US Internal Revenue Code (the "<u>Code</u>"). The Parties agree that this Agreement does not constitute a financial option for US federal income tax purposes as described in section 1234 of the Code.
- 8.6 <u>Late Payments</u>. If Precision does not receive payment of any sum due to it under this Agreement on or before the due date, interest shall thereafter accrue on the sum due to Precision from the due date until the date of payment, such interest to be calculated at a rate equal to [***].
 - 8.7 Reporting. All financial reporting hereunder shall be, if applicable, on the basis of U.S. GAAP, consistently applied.
- 8.8 <u>Currency; Exchange Rate</u>. All payments to be made under this Agreement shall be made in USD by bank wire transfer in immediately available funds to a bank account designated by written notice from Precision. With respect to sales not denominated in USD, Baxalta shall convert each applicable quarterly sales in foreign currency into USD by using the then current and reasonable standard exchange rate methodology applied by Baxalta in its worldwide accounting practices, consistent with U.S. GAAP, consistently applied. Based on the resulting sales in USD, the then applicable royalties shall be calculated. The initial wire transfer instructions for Precision are set forth on <u>Exhibit H</u>.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE IX. INVENTIONS; ACCESS TO IMPROVEMENTS; PATENTS; TRADEMARKS

9.1 Ownership.

- 9.1.1 With respect to all Patents, Know-How and other intellectual property Controlled by a Party prior to the Effective Date or first invented (as determined in accordance with U.S. patent laws) outside of the course of activities performed under this Agreement ("Background IP"), as between the Parties, such Background IP shall be deemed owned by the Controlling Party. Without limiting the foregoing, Precision Platform Technology and Precision Platform Patents shall be deemed Precision's Background IP, and Isolex Platform Technology and Isolex Platform Patents shall be deemed Baxalta's Background IP.
- 9.1.2 Subject to Section 9.1.3, (a) any Invention invented (as determined in accordance with U.S. patent laws) solely by Agents of a Party or its Affiliates in the course of performing activities under this Agreement, together with all intellectual property rights therein, shall be owned by such Party and (b) any Invention invented (as determined in accordance with U.S. patent laws) jointly by at least one (1) Agent of each Party or such Party's Affiliate, together with all intellectual property rights therein ("Joint Inventions", and all Patents claiming such Joint Inventions, hereinafter, "Joint Patents"), shall be owned jointly by the Parties in accordance with joint ownership interests of co-inventors under U.S. patent laws, with each joint Party having, unless otherwise set forth in this Agreement, an equal, undivided interest therein, with the unrestricted right to practice, exploit, license and grant its rights to sublicense any such Joint Invention without a duty of accounting or an obligation to seek consent from the other Party, subject to the exclusive licenses granted to the other Party, the payment obligations under this Agreement and the Restrictive Covenants set forth in Section 11.2. Each Party shall promptly disclose to the other Party in writing any Inventions and any written Invention disclosures, or other similar documents, submitted to it by its Agents describing each and every Invention that constitutes an Invention owned by the other Party or a Joint Invention, and all Know-How relating to such Invention that is in the disclosing Party's possession.

9.1.3 Notwithstanding Section 9.1.2:

(a) Subject to Section 9.1.3(c), to the extent any Invention comprises an improvement or modification of Precision Platform Technology (a "Precision Invention"), such Precision Invention will be solely owned by Precision (as Precision Platform Technology or Precision Platform Patents, as applicable) irrespective of inventorship, and Baxalta agrees to assign and hereby assigns all of its right, title and interest in and to the Precision Inventions (and any intellectual property rights thereto) to Precision and agrees to execute such documents and perform such other acts as Precision may reasonably request to obtain, perfect and enforce its rights to such Precision Inventions and the assignment thereof.

- (b) Subject to Section 9.1.3(c), to the extent any Invention comprises an improvement or modification of Isolex Platform Technology (a "Baxalta Invention"), such Baxalta Invention will be solely owned by Baxalta (as Isolex Platform Technology or Isolex Platform Patents, as applicable) irrespective of inventorship, and Precision agrees to assign and hereby assigns all of its right, title and interest in and to the Baxalta Inventions (and any intellectual property rights thereto) to Baxalta and agrees to execute such documents and perform such other acts as Baxalta may reasonably request to obtain, perfect and enforce its rights to such Baxalta Inventions and the assignment thereof.
- (c) If any Invention comprises only an improvement or modification of both the Precision Platform Technology and the Isolex Platform Technology, such Invention shall be jointly owned by the Parties (as Joint Inventions or Joint Patents, as applicable) irrespective of inventorship, and to the extent ownership is not already vested jointly by inventorship, each Party agrees to assign and hereby assigns such joint right, title and interest in and to such Joint Inventions (and any intellectual property rights thereto) to the other Party as is necessary for each of the Parties to own an equal, undivided interest therein, and agrees to execute such documents and perform such other acts as the other Party may reasonably request to obtain, perfect and enforce its rights to such Joint Inventions and the assignment thereof. Such Joint Inventions and Joint Patents shall be subject to all terms of this Agreement applicable to Joint Inventions and Joint Patents.

9.2 Prosecution of Patents.

- 9.2.1 Non-Product-Related Precision Patents. Precision shall have sole discretion and authority, at its sole cost and expense, with respect to filing, prosecuting and maintaining (a) all Precision Patents that do not include claims directed to CAR-Ts, human T cell engineering, one or more Licensed Product Candidates or Licensed Products, or manufacture or components of one or more Licensed Product Candidates or Licensed Products, and (b) all Precision Platform Patents. Baxalta acknowledges and agrees that Precision has no rights or responsibility for filing, prosecuting or maintaining the Cellectis Patents.
- 9.2.2 <u>Product-Related Precision Patents</u>. Precision shall have the first right and authority to file, prosecute, and maintain the Precision Patents, other than those described in <u>Section 9.2.1</u> and other than Joint Patents (collectively, "<u>Product-Related Patents</u>"), on a worldwide basis in its sole discretion, subject to this <u>Section 9.2.2</u>. Precision shall be solely responsible for all costs and expenses incurred in connection with such filing, prosecution and maintenance in the Territory. Precision shall provide Baxalta with the opportunity to review and comment on any and all prosecution efforts, but in no case less than [***] prior to any filing deadlines, regarding the Product-Related Patents within the Territory; provided, that Precision shall have final control over such prosecution efforts after reasonably considering Baxalta's comments, if any (and if Baxalta does not provide comments within [***] after such opportunity, Baxalta shall be deemed to have no comment on such prosecution efforts). Precision shall provide Baxalta with a copy of material communications from Patent authorities in the Territory regarding

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

the Product-Related Patents, and shall provide drafts of any material filings or responses to be made to such Patent authorities with respect to such Product-Related Patents in a timely manner. Notwithstanding the foregoing, if Precision determines in its sole discretion to abandon or not maintain in the Territory any Product-Related Patents (other than allowing the lapse of any provisional patent application, or abandonment of any patent application in favor of a continuation), Precision shall provide Baxalta with at least [***] prior written notice of such determination and shall assign to Baxalta an equal, undivided interest in such Product-Related Patent and permit Baxalta, at its discretion and expense, to continue filing, prosecution and maintenance of such Product-Related Patent. Such Product-Related Patent shall thereafter be deemed to be a Joint Patent. Baxalta's filing, prosecution or maintenance of such Product-Related Patent shall not change the Parties' respective rights and obligations under this Agreement with respect to such Product-Related Patent other than those expressly set forth in this Section 9.2.2. Baxalta will have the right to deduct any costs or expenses it incurs in the performance of filing, prosecution or maintenance of such Product-Related Patents from any amounts payable by Baxalta to Precision under this Agreement.

9.2.3 Joint Patents. With respect to any potentially patentable Joint Inventions, the Parties shall meet and agree in good faith upon which Party shall file, prosecute and maintain Patent applications claiming such Joint Invention in particular countries and jurisdictions throughout the Territory. The Parties will designate one or the other Party to file, prosecute and maintain each such Patent application and coordinate their efforts as appropriate to make such prosecution activities as efficient, convenient, and harmonious as possible. All costs and expenses of filing, prosecuting and maintaining a Joint Patent shall be shared equally by the Parties. The Party that prosecutes a particular Joint Patent (the "Prosecuting Party") shall provide the other Party the opportunity to review and comment on any and all such prosecution efforts, but in no case less than [***] prior to any filing deadlines, regarding the applicable Joint Patent in the particular countries and jurisdictions in the Territory, and such other Party shall provide the Prosecuting Party reasonable assistance in such efforts; provided, that the Prosecuting Party shall have final control over such prosecution efforts after reasonably considering the other Party's comments, if any (and if the other Party does not provide comments within [***] after such opportunity, such Party shall be deemed to have no comment on such prosecution efforts). The Prosecuting Party shall provide the other Party with a copy of all material communications from Patent Authorities in the Territory regarding the Joint Patent being prosecuted by such Party, and shall provide drafts of any material filings or responses to be made to such Patent authorities with respect to such Joint Patent in a timely manner. In particular, each Party agrees to provide the other Party with all information necessary or desirable to enable the other Party to comply with any duty of candor or duty of disclosure requirements of any Patent authority. Notwithstanding the foregoing, if the Prosecuting Party determines in its sole discretion to abandon or not maintain in any country or jurisdiction of the Territory any Joint Patent (other than allowing the lapse of any provisional patent application, or abandonment of any patent application in favor of a continuation), the Prosecuting Party shall provide the other Party with at least [***] prior written notice of such determination and the other Party shall thereafter have the right, but not the obligation, at its sole discretion and expense, to continue filing, prosecution and maintenance of such Joint Patent.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 9.2.4 <u>Baxalta Patents</u>. Baxalta shall have sole discretion and authority, at its sole cost and expense, with respect to filing, prosecuting and maintaining (a) all Baxalta Patents, and (b) all Isolex Platform Patents.
- 9.2.5 <u>Cooperation in Prosecution</u>. Each Party shall, at its sole cost and expense, provide the other Party all reasonable assistance and cooperation in the Patent prosecution efforts described above in this <u>Article IX</u>, including providing any necessary powers of attorney and executing any other required documents or instruments for such prosecution. Such cooperation may further include coordinating filing or prosecution of applications to avoid potential issues during prosecution (including novelty, enablement, estoppel, double-patenting and execution of amendments), and the assistance of each Party's relevant personnel. Each Party will use reasonable efforts via consultation with the JSC to avoid creating potential issues in prosecution of the applications for Precision Patents, Precision Platform Patents, Isolex Platform Patents and Joint Patents.

9.3 Licenses to Enabling Technologies.

- 9.3.1 If Precision, in its good faith discretion, identifies any enabling technologies that are reasonably necessary for the use of Precision Platform Technology to engineer CAR-Ts to create Licensed Product Candidates in accordance with the Development Plans under this Agreement (such technology, "<u>Platform Enabling Technology</u>"), (a) Precision shall promptly inform the JSC of such Platform Enabling Technology, and (b) if approved by the JSC for use under the Development Plans in accordance with this Agreement [***].
- 9.3.2 If Baxalta, in its good faith discretion, identifies any enabling technologies that are reasonably necessary for the commercial-scale manufacturing, Development after Phase II Ready Status or Commercialization of Licensed Products in accordance with this Agreement (such technology, "Commercialization Enabling Technology,"), (a) Baxalta shall promptly inform the JSC of such Commercialization Enabling Technology, and (b) [***].
- 9.3.3 If either Party, in its good faith discretion, identifies any other enabling technologies that are not clearly within the scope of Platform Enabling Technology or Commercialization Enabling Technology (such technology, "Other Enabling Technology."), (a) such Party shall promptly inform the JSC of such Other Enabling Technology, and (b) if approved by the JSC for use in the Development or Commercialization of the Licensed Product in accordance with this Agreement, [***].

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 9.3.4 With respect to any Platform Enabling Technology, Commercialization Enabling Technology, Other Enabling Technology or any other technology Controlled by either Party and used in connection with this Agreement, each Party acknowledges that such technology may be subject to certain restrictions or other terms and conditions set forth in the agreement or agreements under which such Party obtained access to such technology. Prior to either Party delivering any such technology to the other Party, the use of which would be subject to any such restrictions or other terms or conditions not imposed by any other provision of this Agreement, such Party shall notify the JSC in writing of such applicable restrictions or other terms or conditions, and as a condition of using such technology, such other Party shall comply with such restrictions, terms and conditions.
- 9.4 <u>Duke IP</u>. Baxalta acknowledges and agrees that any licenses and rights granted by Precision to Baxalta under the Duke IP are granted subject to the terms and conditions of the Duke Agreement, including Duke's right to practice under the Duke IP for its own internal, non-commercial, educational, research and clinical purposes, and subject to the rights of the United States Government and applicable limitations under 37 C.F.R. § 401, Public Law 96-517 and Public Law 98-620 resulting from the United States Government's funding of research leading to creation of the Duke IP. Without limiting the foregoing, Baxalta agrees to comply with any obligations resulting from such government rights with respect to its practice of the Duke IP (if any) under this Agreement.

9.5 Infringement of Patents by Third Parties.

9.5.1 Notification. Each Party shall promptly notify the other Party in writing of any existing or threatened infringement by a Third Party of the Precision Patents, Precision Platform Patents, Baxalta Patents, Isolex Platform Patents or the Joint Patents of which it becomes aware in the Territory, and shall provide to the other Party any and all evidence and information available to such Party regarding such alleged infringement. Any such infringement that results from making, using, importing, offering for sale, or selling any Licensed Product Directed to an Included Target in the Field during any period in which Baxalta holds a Commercial License to such Licensed Product with respect to such Included Target, including any "patent certification" filed in the United States under 21 U.S.C. § 355(b)(2) or 21 U.S.C. § 355(j)(2) or similar provisions in other jurisdictions is referred to as "Product Infringement" for purposes of this Section 9.5.

9.5.2 Product Infringement in the Territory.

(a) Baxalta shall have the first right, but not the obligation, to bring an appropriate suit or other action against any Third Party engaged in Product Infringement in the Territory based on Precision Patents or Joint Patents, subject to Section 9.5.2(d). Baxalta shall have a period of [***] after such notification to or by Precision, to elect to so enforce such Precision Patent or Joint Patent, as applicable, in the Territory. If Baxalta does not so elect, Baxalta shall so notify Precision in writing during such [***] period, or [***] prior to any deadline relating to loss of any rights with respect to the Product Infringement, whichever is earlier, and Precision shall have the right, but not the obligation, to commence a suit or take action to enforce any such Precision Patent or Joint Patent against such Third Party allegedly perpetrating such Product

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Infringement. Each Party shall provide to the Party enforcing any such rights under this <u>Section 9.5.2(a)</u> (the "<u>Enforcing Party</u>") reasonable assistance in such enforcement, including joining an action as a party plaintiff if so required by applicable Laws to pursue such action, at the Enforcing Party's sole expense. The Enforcing Party shall keep the other Party regularly informed of the status and progress of such enforcement efforts, and shall reasonably consider the other Party's comments on any such efforts. The Enforcing Party shall bear and be responsible for all costs incurred in connection with each Party's activities under this <u>Section 9.5.2(a)</u>.

- (b) Baxalta shall have the sole and exclusive right, but not the obligation, in its sole discretion to bring an appropriate suit or other action against any Third Party engaged in Product Infringement in the Territory based on the Baxalta Patents or Isolex Platform Patents.
- (c) Precision shall have the sole and exclusive right, but not the obligation, in its sole discretion to bring an appropriate suit or other action against any Third Party engaged in Product Infringement in the Territory based on the Precision Platform Patents.
- (d) The Party not bringing an action with respect to Product Infringement under this <u>Section 9.5.2</u> shall be entitled to separate representation in such matter by counsel of its own choice and at its own expense, but such Party shall at all times cooperate fully with the Party bringing such action. Additionally, the Party not bringing an action under this <u>Section 9.5.2</u> may have an opportunity to participate in such action, at its sole cost and expense, to the extent that the Parties may mutually agree at the time the Enforcing Party elects to bring such action hereunder.

9.5.3 Non-Product Infringement in the Territory.

(a) For infringement of the Joint Patents in the Territory that is not Product Infringement, the Parties shall confer to determine which Party shall have the first right to bring an appropriate suit or other action against the Third Party engaged in such infringement, and the manner in which the Parties shall bear costs of and share related damage recoveries from such suit or action. The Party that brings such suit or actions (also the "Enforcing Party") shall keep the other Party regularly informed of the status and progress of such enforcement efforts, and shall reasonably consider the other Party's comments on any such efforts. The other Party shall cooperate with the Enforcing Party in enforcing Joint Patents against such infringement, including joining an action as a party plaintiff if so required by applicable Laws to pursue such action. If the Parties are unable to reach agreement upon which a Party shall bring an appropriate suit or other action against any Third Party engaged in such infringement of such Joint Patent within [***] prior to any deadline relating to loss of any rights with respect to such infringement, whichever is earlier, then Baxalta shall have the first right, but not the obligation, to bring such suit or other actions against such infringement in the Territory at its sole expense. The Party that does not bring such suit or action shall have the right to participate in such actions at its expense upon written notice to the other Party. The Enforcing Party shall bear and be responsible for all costs incurred in connection with each Party's activities under this Section 9.5.3(a).

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (b) Baxalta shall have the sole and exclusive right, but not the obligation, in its sole discretion to bring an appropriate suit or other action against any Third Party engaged in infringement that does not constitute Product Infringement in the Territory, based on the Baxalta Patents or Isolex Platform Patents.
- (c) Precision shall have the sole and exclusive right, but not the obligation, in its sole discretion to bring an appropriate suit or other action against any Third Party engaged in infringement that does not constitute Product Infringement in the Territory, based on the Precision Patents or Precision Platform Patents.
- 9.5.4 <u>Settlement</u>. No Party shall settle any claim, suit or action that it brings under this <u>Section 9.5</u> involving Precision Patents or Joint Patents in any manner that would, in the other Party's reasonable judgment, materially and adversely impact the other Party or that would have the effect of diminishing any rights or licenses granted hereunder, including settlements involving the ownership, validity or enforceability of any of the Precision Patents or Joint Patents, or that do not include a full and unconditional release from all liability of the other Party, without the prior written consent of the other Party, which shall not be unreasonably withheld, conditioned or delayed.
- 9.5.5 <u>Allocation of Proceeds</u>. Except as otherwise provided in this <u>Section 9.5</u>, if either Party recovers monetary damages from any Third Party in a suit or action described in this <u>Section 9.5</u>, such recovery shall be allocated first to the repayment of costs and expenses of the Party(ies) with respect to the action (on a pro rata basis), and any remaining damages [***].
- 9.5.6 <u>Certain Limitations</u>. Baxalta acknowledges and agrees that (a) Precision has no rights or responsibility for enforcing the Cellectis Patents, and therefore all references to Precision Patents or Precision Platform Patents in this <u>Section 9.5</u> shall be deemed to exclude the Cellectis Patents for all purposes, (b) prior to initiating enforcement actions against a Third Party with respect to certain Precision Patents or Precision Platform Patents which were subject to the non-exclusive license granted by Precision to Cellectis S.A. pursuant to the Cellectis Agreement, Precision is required by the Cellectis Agreement to confirm that Cellectis S.A has not granted a license to such Third Party under such Precision Patents or Precision Platform Patents, and Baxalta will cooperate with Precision in taking such actions as required by the Cellectis Agreement, and (c) Duke retains discretion as to whether to become a party plaintiff and has certain rights with respect to enforcement of Patent Rights contained within the Duke IP in the event Precision does not enforce such Patent Rights.
 - 9.6 Infringement of Third Party Rights in the Territory.
- 9.6.1 <u>Notice</u>. If the Development, use, manufacture or Commercialization of any Licensed Product in the Field and in the Territory results in a claim for Patent infringement by a Third Party, the Party first having notice of such claim shall promptly notify the other Party in writing of such a claim. Following such notice, the Parties agree to enter into either a joint defense or common interest agreement, under which agreement the Parties can share the known facts of such infringement in reasonable detail, if they are advised to do so by counsel.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

9.6.2 Third Party Claims. Baxalta shall assume control of the defense of any claims brought by Third Parties alleging infringement of Third Party intellectual property rights in connection with Baxalta's Development, manufacture, use or Commercialization of any Licensed Product in the Field in the Territory, represented by its own counsel. If requested by Baxalta, Precision agrees to join in any such litigation, and in any event shall reasonably cooperate with Baxalta, at Baxalta's sole cost and expense. Baxalta shall have the exclusive right to settle any such claim without the consent of Precision, unless such settlement shall materially and adversely affect Precision's rights under the Precision Patents, Precision Platform Patents, or Joint Patents, including on the ownership, validity or enforceability thereof. Any costs and expenses incurred in defending any such claims shall be solely the responsibility of Baxalta (without limiting Baxalta's rights under Section 8.4.2), provided that if Precision elects to retain separate representation in the defense of any such claims brought by Third Parties using counsel of its own choice, it shall do so solely at its cost and expense.

9.6.3 <u>Potential Third Party Claims</u>. In the event that either Party becomes aware of a Third Party intellectual property right that might reasonably be expected to give rise to a claim of infringement under this <u>Section 9.6</u>, that Party shall promptly notify the other Party of such intellectual property right and all relevant facts and circumstances known to the discovering Party. Following such notice, the Parties agree to enter into either a joint defense or common interest agreement, under which agreement the Parties can share the known facts of such infringement in reasonable detail, if they are advised to do so by counsel, and to consult thereafter regarding any corrective or preventive action to be taken to address such potential claim.

9.7 Patent Oppositions and Other Proceedings.

9.7.1 By the Parties. If either Party desires to bring an opposition, action for declaratory judgment, nullity action, interference, declaration for non-infringement, reexamination, inter partes review or other attack upon the validity, title, or enforceability of a Patent owned or controlled by a Third Party that claims, in the Territory, any Licensed Product Candidate or Licensed Product, or the manufacture, use or Commercialization of any Licensed Product Candidate or Licensed Product (except insofar as such action is a counterclaim to or defense of, or accompanies a defense of, a Third Party's claim or assertion of infringement under Section 9.6, in which case the provisions of Section 9.6 shall govern), such Party shall so notify the other Party, and the Parties shall promptly confer to determine whether to bring such action or the manner in which to settle such action. Baxalta shall have the first right, but not the obligation, to bring in its sole control and at its sole cost and expense such action in the Territory. If Baxalta does not bring such action within [***] after notification thereof pursuant to this Section 9.7 (or earlier, if required by the nature of the proceeding), then Precision shall have the right, but not the obligation, to bring, in Precision's sole control and at its sole cost and expense, such action. The

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Party not bringing an action under this <u>Section 9.7</u> shall join the action as a joint party plaintiff if required to enable the other Party to bring such action, at the other Party's sole cost and expense. Additionally, if appropriate, the Party not bringing the applicable action under this <u>Section 9.7</u> shall be entitled to separate representation, at its sole cost and expense, in such proceeding by counsel of its own choice, and shall reasonably cooperate with the Party bringing such action. Any awards or amounts received in bringing any such action shall be first allocated to reimburse the Parties' expenses in such action, and any remaining amounts shall be retained by the Party bringing such action.

9.7.2 By Third Parties.

- (a) <u>Precision Patents</u>. If a Precision Patent becomes the subject of any proceeding commenced by a Third Party in the Territory in connection with an opposition, reexamination request, action for declaratory judgment, nullity action, interference, inter partes review or other attack upon the validity, title or enforceability thereof, then Precision shall have the first right, but not the obligation, to control such defense at its sole cost and expense. Precision shall permit Baxalta to participate in the proceeding to the extent permissible under applicable Laws, and to be represented by its own counsel in such proceeding, at Baxalta's sole cost and expense. If Precision decides that it does not wish to defend against such action, then Baxalta shall have a right to assume defense of such Third Party action.
- (b) <u>Joint Patents</u>. If a Joint Patent becomes the subject of any proceeding commenced by a Third Party in the Territory in connection with an opposition, reexamination request, action for declaratory judgment, nullity action, interference, inter partes review or other attack upon the validity, title or enforceability thereof (except insofar as such action is a counterclaim to or defense of, or accompanies a defense of, an action for infringement against a Third Party under <u>Section 9.5</u>, in which case the provisions of <u>Section 9.5</u> shall govern), then Baxalta shall have the first right, but not the obligation, to control such defense at its sole cost and expense. Baxalta shall permit Precision to participate in the proceeding to the extent permissible under applicable Laws, and to be represented by its own counsel in such proceeding, at Precision's sole cost and expense. If Baxalta decides that it does not wish to defend against such action, then Precision shall have a right to assume defense of such Third Party action.
- (c) <u>Allocation of Costs and Proceeds</u>. Except as set forth above, all expenses incurred by the Parties in an action under this <u>Section 9.7</u> shall be borne by the Party controlling the defense of the Third Party action. Any awards or amounts received in defending any such Third Party action, if any, shall be allocated between the Parties as provided in <u>Section 9.5.5</u> as if the Party controlling the defense of the Third Party action were the Party that brought an action against an alleged infringer.
- 9.8 <u>Patent Term Extensions in the Territory.</u> The patent counsel of each Party shall discuss and recommend for which, if any, of the Precision Patents and Joint Patents in the Territory the Parties should seek any term extensions, supplementary protection certificates, and equivalents thereof offering Patent protection beyond the initial term with respect to any issued Patents ("<u>Patent Term Extensions</u>") licensed to Baxalta hereunder in the Territory. If Precision consents to applying for any such Patent Term Extension (such consent not to be unreasonably withheld,

conditioned or delayed), Baxalta shall thereafter have (i) the final decision making authority with respect to applying for any such Patent Term Extensions in the Territory; provided that Baxalta shall not unreasonably fail or refuse to do so, and (ii) the sole right to apply for any such Patent Term Extensions Baxalta decides to seek, at its expense; provided, however, that to the extent any such application for Patent Term Extension must be filed in the name of Precision, Precision hereby grants (effective as of the date of its consent) Baxalta the power to file such application on behalf of or as agent for Precision. Precision shall cooperate fully with Baxalta, at Baxalta's expense, in making such filings or taking any related actions, for example and without limitation, making available all required regulatory data and information and executing any required authorizations to apply for such Patent Term Extension.

9.9 <u>Registration of License</u>. Precision agrees that Baxalta may, if applicable, register its license under the Precision Patents or Joint Patents with the Patent authorities in the Territory. Baxalta shall, at its expense, prepare and deliver to Precision such instruments and other documents reasonably necessary and in proper form for such registration. The Parties shall mutually agree on the form of documents to be used for such purpose, and shall cooperate to preserve confidentiality of this Agreement to the extent permitted under applicable Laws in the relevant country. Precision shall execute and return to Baxalta such instruments and documents within [***] from the receipt thereof.

```
9.10 [***]

9.10.1 [***]

9.10.2 [***]

9.10.3 [***]
```

- 9.11 <u>Trademarks</u>. Baxalta shall have the sole right to brand the Licensed Products using Baxalta related trademarks, service marks, names, logos, design or trade dress it determines appropriate for the Licensed Products, which may vary by country or within a country ("<u>Licensed Product Marks</u>"). Baxalta shall own all rights in the Licensed Product Marks in the Territory and shall register and maintain the Licensed Product Marks in the countries and regions in the Territory that it determines reasonably necessary, at Baxalta's cost and expense. Under a separate trademark agreement, Baxalta and Precision may mutually agree to the use of certain Precision trademarks for the benefit of branding, including co-branding.
- 9.12 <u>Patent Marking</u>. Baxalta shall use Commercially Reasonable Efforts to mark all Licensed Products in accordance with the applicable patent marking Law, and shall require all of its Affiliates and Sublicensees to do the same to the extent required by Law.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE X. REPRESENTATIONS AND WARRANTIES

- 10.1 <u>The Parties' Representations and Warranties</u>. Each Party hereby represents and warrants to the other Party, as of the Effective Date, as set forth below.
- 10.1.1 Such Party (a) is a corporation or other entity duly organized and subsisting under the applicable Laws of its jurisdiction of incorporation or organization, and (b) has full power and authority and the legal right to own and operate its property and assets and to carry on its business as it is now being conducted and as it is contemplated to be conducted by this Agreement.
- 10.1.2 Such Party has the power, authority and legal right, and is free to, enter into and perform its obligations under this Agreement and, in so doing, will not violate or conflict with (a) any other agreement to which such Party is a party as of the Effective Date or (b) any instrument or binding understanding, oral or written, to which such Party is a party or by which it is otherwise bound.
- 10.1.3 This Agreement has been duly executed and delivered on behalf of such Party and constitutes a legal, valid, and binding obligation of such Party and is enforceable against it in accordance with its terms.
 - 10.1.4 Such Party has taken all corporate action necessary to authorize the execution and delivery of this Agreement.
- 10.1.5 Except with respect to required approvals for the exercise of the Commercial Option pursuant to applicable Antitrust Laws and Marketing Approvals and Reimbursement Approvals for Licensed Products or as otherwise described in this Agreement, such Party has obtained all necessary consents, approvals, and authorizations of all Regulatory Authorities and other Third Parties required to be obtained by such Party in connection with the execution and delivery of this Agreement and the performance of its obligations hereunder.
- 10.1.6 The execution and delivery of this Agreement and the performance of such Party's obligations hereunder (a) do not conflict with or violate any requirement of applicable Laws or any provision of the articles of incorporation, bylaws, limited partnership agreement, or any similar instrument of such Party, as applicable, in any material way, and (b) do not conflict with, violate, or breach or constitute a default or require any consent under, any applicable Laws or any contractual obligation or court or administrative order by which such Party is bound.
- 10.1.7 Such Party is neither a party to nor bound by any corporate integrity agreement or similar compliance agreement to which any Governmental Authority or Third Party payor is a counterparty.
 - 10.2 Precision's Representations and Warranties. Precision hereby represents and warrants to Baxalta, as of the Effective Date, as set forth below.

- 10.2.1 Exhibit E sets forth a complete and correct list of patent numbers and patent application numbers of all Precision Patents and Precision Platform Patents as of the Effective Date, other than the Cellectis Patents. Precision is the owner of, or otherwise has Control of, each Precision Patent and Precision Platform Patent set forth in Exhibit E and, subject to the terms of the Cellectis Agreement, the Cellectis Patents. Except as would not have a material adverse effect on the rights granted to Baxalta herein, each Precision Patent and Precision Platform Patent owned by Precision has been filed in good faith, has been prosecuted in accordance with any applicable duty of candor and has been maintained in a manner consistent with Precision's standard practice, in each case in each applicable jurisdiction in which such Precision Patent or and Precision Platform Patent rights have been filed, and applicable fees (to the extent such fees have come due) have been paid on or before the due date for payment.
- 10.2.2 Precision and its Affiliates have sufficient legal or beneficial rights or title under their respective intellectual property rights necessary to grant all of the licenses contained in this Agreement with respect to such intellectual property rights.
- 10.2.3 Neither Precision nor any of its Affiliates has granted any right or license, or agreed to grant any right or license, to any Third Party relating to any of the intellectual property rights that are licensed by Precision or any of its Affiliates to Baxalta pursuant to this Agreement that conflict with, or limit the scope of, any of the rights or licenses granted to Baxalta pursuant to this Agreement.
- 10.2.4 All of Precision's and its Affiliates' Agents that are involved in research and Development work have executed agreements requiring assignment or licensing to Precision or its Affiliate, as applicable, of all Inventions made during the course of and as a result of their association with Precision or its Affiliate, except to the extent that any such assignment would not be permitted pursuant to applicable Law, as applicable, and obligating the individual to maintain as confidential the Confidential Information of Precision or its Affiliate, as applicable (except where the failure to have entered into such an agreement would not have a material adverse effect on the rights granted to Baxalta herein); provided, however, that for employees of Precision or its Affiliates based in Germany, Austria, or any other jurisdiction where a prior obligation to assign is not permitted, the obligation under this paragraph will be deemed satisfied if (a) each such employee is obligated to notify his employer of such Inventions and (b) the employer has an established program for receiving such notifications and timely claiming ownership of or exclusive rights to such Inventions after notification.
- 10.2.5 None of Precision, its Affiliates or their respective officers or employees (a) has ever been debarred or is subject to debarment or, to Precision's knowledge, convicted of a crime for which a Person could be debarred before a Regulatory Authority under applicable Laws, or (b) to Precision's knowledge, has ever been under indictment for a crime for which a Person could be debarred under such Laws. To Precision's knowledge, none of the independent contractors, consultants or agents of Precision, or its Affiliates (a) has ever been debarred or is subject to debarment or convicted of a crime for which a Person could be debarred before a Regulatory Authority under applicable Laws, or (b) has ever been under indictment for a crime for which a Person could be debarred under such Laws.

10.2.6 [***].

- 10.2.7 Other than as disclosed to Baxalta prior to the Effective Date, neither Precision nor any of its Affiliates has received any written communications alleging that it has infringed, misappropriated or otherwise violated, or that it would infringe, misappropriate or otherwise violate, through the use of the Precision Platform Technology, Precision Patents and/or Precision Know-How, to manufacture, use, import, export, sale, or offer for sale of any of the products in the Field and in the Territory, any intellectual property rights Controlled by any Third Party.
- 10.2.8 The Duke Agreement is a legal and valid obligation binding upon Precision and, to Precision's knowledge, Duke, and authorizes Precision to grant the rights under the Duke IP granted to Baxalta under this Agreement. As of the Effective Date, Precision is not in breach of the Duke Agreement. Precision shall comply in all material respects with the Duke Agreement and shall not modify or amend the Duke Agreement in a manner that would reduce Baxalta's access or rights to the Duke IP without the prior written consent of Baxalta.
- 10.2.9 To Precision's knowledge, no invention claimed by any Precision Patent owned by Precision was made or reduced to practice using any funding of the U.S. Government.
- 10.2.10 To Precision's knowledge, Precision is not in possession of any in-licensed Third Party Know-How or Patent Rights that are necessary to Develop, use, manufacture or Commercialize any Licensed Product Candidate or Licensed Product but are not Controlled by Precision. As of the Effective Date, the only Third Party Patents that are included within the Precision Patents and the Precision Platform Patents are the Cellectis Patents and Patents contained within the Duke IP.

ARTICLE XI. CERTAIN COVENANTS

11.1 Covenants.

- 11.1.1 Mutual Covenants. Each Party hereby covenants throughout the Term as set forth below:
- (a) All of such Party's and its Affiliates' Agents working under this Agreement will be under the obligation to assign to such Party or such Party's Affiliate, as applicable, in each case as the sole owner, all right, title and interest in and to their inventions and discoveries arising in the performance of such work, whether or not patentable, either immediately upon invention or, if applicable Law so provides, upon disclosure to and demand made by such Party or such Party's Affiliates; provided, however, that for employees based in Germany, Austria, or any other jurisdiction where a prior obligation to assign is not permitted, the obligation under this paragraph will be deemed satisfied if (i) each such employee is obligated to notify his employer of such inventions and (ii) the employer has an established program for receiving such notifications and timely claiming ownership of or exclusive rights to such inventions after notification.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (b) Such Party will not, and will cause its Affiliates not to, employ or use any Agent that employs any individual or entity (i) that has been debarred by a Regulatory Authority under applicable Laws or convicted of a crime for which such Person could be so debarred, or (ii) that is the subject of a debarment investigation or proceeding of a Regulatory Authority under applicable Laws, in each case of clauses (i) and (ii), in the conduct of such Party's or its Affiliates' activities under this Agreement. If during the Term, a Party has reason to believe that actions or omissions have occurred that will cause such Party to breach the covenant in the immediately preceding sentence, then such Party promptly shall notify the other Party of same in writing.
- (c) Such Party shall not, and shall cause its Affiliates not to, enter into any agreement or other arrangement with a Third Party that conflicts with the rights granted to the other Party under this Agreement.
- 11.1.2 <u>Precision Covenants</u>. Precision covenants that it will update <u>Exhibit E</u> on a regular basis during the Term of this Agreement, at least once per year upon each anniversary of the Effective Date (provided updates are required at such time). In addition, Precision covenants that it will not amend the Cellectis Agreement in any manner that would have the effect of diminishing the rights granted by Precision to Baxalta hereunder without Baxalta's prior written consent, not to be unreasonably withheld.

11.2 Restrictive Covenants.

11.2.1 Precision Restrictive Covenants

- (a) In consideration for the payments and rights granted to it under this Agreement, Precision agrees that for so long as [***], Precision will not [***].
 - (b) If [***] (this Section 11.2.1(b), together with Section 11.2.1(a), the "Precision Restrictive Covenants").
- (c) Notwithstanding anything to the contrary in this Agreement, but without limiting the obligations of Precision as it relates to its Affiliates (including an Acquirer) as set forth in Section 11.2.1(a), Section 11.2.1(b), Section 16.1.2(a) or Section 16.1.2(b), [***].
- (d) Notwithstanding Section [***] shall not be a breach of the Precision Restrictive Covenants, provided that (i) Precision notifies Baxalta of such [***] period following such [***], (ii) within such [***] and (iii) during any such [***].

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

11.2.2 Baxalta Restrictive Covenants.

- (a) In consideration for the payments, rights and licenses granted to it under this Agreement, Baxalta agrees that for so long as [***], Baxalta will not, and will cause its Affiliates not to, [***] (the "Baxalta Restrictive Covenants" and, together with the Precision Restrictive Covenants, the "Restrictive Covenants").
- (b) Notwithstanding Section 11.2.2(a) [***] shall not be a breach of the Baxalta Restrictive Covenants, provided that Baxalta notifies Precision of such [***] and either: (i) Baxalta provides written notice, within [***], that Baxalta is terminating this Agreement with respect to [***] in accordance with Section 14.2.5 (Termination for Convenience), with an effective date of such termination not later than [***] days after the lapse of such [***] period, or (ii) Baxalta notifies Precision of [***] within the [***] period following such [***], and within such [***] period, Baxalta or its Affiliate, as applicable, [***].
- 11.2.3 <u>Injunctive Relief</u>. Each Party acknowledges and agrees that the other Party may be irreparably damaged if the Restrictive Covenants are not performed by a Party in accordance with their specific terms, and that any such breach by a Party may not be adequately compensated by monetary damages alone. Each Party shall be entitled to enforce any Restrictive Covenant by a decree of specific performance and to temporary, preliminary and permanent injunctive relief to prevent breaches or threatened breaches of the Restrictive Covenants, without posting any bond or other undertaking.
- 11.2.4 <u>Attorneys' Fees</u>. If any arbitration or other legal proceeding is brought by a Party for the enforcement of the Restrictive Covenants, or to recover damages or other applicable remedy based on an alleged dispute, breach or default in connection with Restrictive Covenants, the prevailing Party shall be entitled to recover reasonable attorneys' fees and other reasonable costs and expenses incurred in enforcing the Restrictive Covenants, and such Party shall be entitled to (i) cause the other Party to reimburse such reasonable fees, costs and expenses, or (ii) offset such reasonable fees, costs and expenses against any amounts payable to the other Party under this Agreement.
- 11.2.5 <u>Offset of Damages</u>. Each Party shall be entitled to cause the other Party to pay or to offset, against amounts payable to the other Party under this Agreement, any amounts of damages determined in an arbitration or other legal proceeding, to be owed to such Party by the other Party based on the other Party's breach of the Restrictive Covenants.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE XII. CONFIDENTIALITY

12.1 Confidentiality Obligations of Baxalta.

- 12.1.1 Subject to Section 12.3, during the Term and for a period of [***] Baxalta:
- (a) shall hold in strict confidence any and all information disclosed to it by Precision, including Precision Know-How and including all information disclosed to it by Precision prior to the Effective Date under the Confidentiality Agreement (together "Precision Confidential Information"), and shall not use, nor disclose or supply to any Third Party, nor permit any Third Party, to have access to the Precision Confidential Information, without first obtaining the written consent of Precision, except in connection with the performance of its obligations and exercise of its rights under this Agreement;
- (b) shall take all reasonable precautions necessary or prudent to prevent material in its possession or control that contains or refers to Precision Confidential Information from being destroyed or lost, or discovered, received, used, intercepted or copied by any Third Party; and
- (c) may disclose the Precision Confidential Information only to its Agents, Affiliates, actual and potential Sublicensees, actual and potential collaborators, and actual and potential investors or acquirers, in each case solely to the extent reasonably necessary for the purposes of, and for those matters undertaken pursuant to, the performance of Baxalta's obligations and exercise of Baxalta's rights under this Agreement, and may disclose Precision Confidential Information contained in reports provided by Precision pursuant to this Agreement to Baxalta's actual and potential investors or acquirers; provided in each case that such Agents, Affiliates, actual and potential Sublicensees, actual and potential collaborators and actual and potential investors or acquirers are bound by terms and conditions of confidentiality no less protective than the terms and conditions that bind Baxalta hereunder; provided, however, that the duration of such terms and conditions of confidentiality for such recipients shall be no less than five (5) years.

For the avoidance of doubt, it is understood that Baxalta shall be liable for any breach of the confidentiality obligation under this <u>Section 12.1</u> by any Person to whom the Precision Confidential Information is disclosed by Baxalta.

- 12.1.2 Baxalta's obligations of confidentiality and non-use under this <u>Section 12.1</u> shall not apply to, and Baxalta shall have no further obligations under this <u>Section 12.1</u> with respect to, any of the Precision Confidential Information, to the extent that such Precision Confidential Information:
 - (a) is or becomes part of the public domain without breach by Baxalta of this Agreement;
- (b) was rightfully in Baxalta's possession before disclosure by Precision to Baxalta and was not acquired directly or indirectly from Precision, as documented by Baxalta's written records;
- (c) is obtained from a Third Party with no applicable obligation of confidentiality to Precision, who has a right to disclose it to Baxalta;
- (d) is developed independently by Baxalta without use of or reference to the Precision Confidential Information, as evidenced by Baxalta's written records; or
- [***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

(e) is required to be revealed in response to a court decision or administrative order, or to otherwise comply with applicable Law, applicable rules of any recognized stock exchange or quotation system or applicable rules or requirements of the Securities and Exchange Commission or other Governmental Authority or Regulatory Authority, provided, that in each such case Baxalta shall inform Precision immediately by written notice and cooperate with Precision using its Commercially Reasonable Efforts either to seek protective measures for such Precision Confidential Information, or to seek confidential treatment of such Precision Confidential Information, and in any case Baxalta shall disclose only such portion of the Precision Confidential Information which is so required to be disclosed.

Any combination of features or disclosures shall not be deemed to fall within the foregoing exclusions merely because individual features are published or available to the general public or in the rightful possession of Baxalta unless the combination itself and principle of operation are published or available to the general public or in the rightful possession of Baxalta.

12.1.3 Nothing herein shall prevent Baxalta from disclosing any Precision Confidential Information to the extent that such Precision Confidential Information is required to be used or disclosed for the purposes of seeking or obtaining approvals of Licensed Product Candidates or Licensed Products from Regulatory Authorities, including Marketing Approvals and Reimbursement Approvals, or seeking or maintaining patent protection for Inventions it owns or has responsibility for prosecuting under Article IX.

12.2 Confidentiality Obligations of Precision.

- 12.2.1 Subject to Section 12.3, during the Term and for a period of [***] Precision:
- (a) shall hold in strict confidence any and all information disclosed to it by Baxalta, including Baxalta Know-How and including all information disclosed to it by Baxalta prior to the Effective Date under the Confidentiality Agreement, as well as the fact that any Target is an Included Target under this Agreement (together "Baxalta Confidential Information"), and shall not use, nor disclose or supply to any Third Party nor permit any Third Party to have access to the Baxalta Confidential Information, without first obtaining the written consent of Baxalta, except in connection with the performance of its obligations and exercise of its rights under this Agreement;
- (b) shall take all reasonable precautions necessary or prudent to prevent material in its possession or control that contains or refers to Baxalta Confidential Information from being destroyed or lost, or discovered, received, used, intercepted or copied by any Third Party;

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (c) may disclose the Baxalta Confidential Information to its Agents, Affiliates, actual and potential collaborators and actual and potential investors or acquirers, in each case solely to the extent reasonably necessary for the purposes of, and for those matters undertaken pursuant to, the performance of Precision's obligations and exercise of Precision's rights under this Agreement, and may disclose Baxalta Confidential Information contained in reports provided by Baxalta pursuant to this Agreement to Precision's actual and potential investors or acquirers; provided in each case that such Agents, Affiliates, actual and potential collaborators and actual and potential investors or acquirers are bound by terms and conditions of confidentiality no less protective than the terms and conditions that bind Precision hereunder; provided, however, that the duration of such terms and conditions of confidentiality for such recipients shall be no less than five (5) years; and
- (d) [***] For the avoidance of doubt, it is understood Precision shall be liable for any breach of the confidentiality obligation under this <u>Section 12.2</u> by any Person to whom the Baxalta Confidential Information is disclosed by Precision.
- 12.2.2 Precision's obligations of confidentiality and non-use under this <u>Section 12.2</u> shall not apply to, and Precision shall have no further obligations under this <u>Section 12.2</u> with respect to, any of the Baxalta Confidential Information to the extent that such Baxalta Confidential Information:
 - (a) is or becomes part of the public domain without breach by Precision of this Agreement;
- (b) was rightfully in Precision's possession before disclosure by Baxalta to Precision and was not acquired directly or indirectly from Baxalta, as documented by Precision's written records;
- (c) is obtained from a Third Party with no applicable obligation of confidentiality to Baxalta, who has a right to disclose it to Precision;
- (d) is developed by Precision without use of or reference to the Baxalta Confidential Information, as evidenced by Precision's written records; or
- (e) is required to be revealed in response to a court decision or administrative order, or to otherwise comply with applicable Law, applicable rules of any recognized stock exchange or quotation system or applicable rules or requirements of the Securities and Exchange Commission or other Governmental Authority or Regulatory Authority, provided, that in each such case Precision shall inform Baxalta immediately by written notice and cooperate with Baxalta using its Commercially Reasonable Efforts either to seek protective measures for such Baxalta Confidential Information, or to seek confidential treatment of such Baxalta Confidential Information, and in any case Precision shall disclose only such portion of the Baxalta Confidential Information which is so required to be disclosed.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Any combination of features or disclosures shall not be deemed to fall within the foregoing exclusions merely because individual features are published or available to the general public or in the rightful possession of Precision unless the combination itself and principle of operation are published or available to the general public or in the rightful possession of Precision.

- 12.2.3 Nothing herein shall prevent Precision from disclosing any Baxalta Confidential Information to the extent that such Baxalta Confidential Information is required to be used or disclosed for the purposes of seeking or obtaining approvals of Licensed Product Candidates [***] from Regulatory Authorities, including Marketing Approvals and Reimbursement Approvals, or seeking or maintaining patent protection for Inventions it owns or has responsibility for prosecuting under Article IX.
- 12.2.4 In addition, Precision shall be permitted to disclose Baxalta Confidential Information and the terms and conditions of this Agreement to Duke solely as and to the extent necessary to fulfill Precision's reporting obligations under the Duke Agreement, provided that such information is disclosed subject to the confidentiality provisions of the Duke Agreement.
- 12.3 Scientific Publication. Each Party recognizes that the publication of papers regarding results of and other information regarding Licensed Products, Licensed Product Candidates, and Unselected Candidates, including oral presentations and abstracts, may be beneficial to both Parties, provided that publications are subject to reasonable controls to protect Confidential Information and the Parties' mutual interest in obtaining Patent Rights and protecting trade secret information. Accordingly, the Party proposing publication shall deliver to the JSC for review and approval a copy of any proposed publication or presentation that pertains to any Licensed Product, Licensed Product Candidate or Unselected Candidate at least [***] prior to submitting the material to a publisher or initiating any other publication or presentation. The other Party shall have the right (a) to require modifications of the publication or presentation (i) to protect each Party's respective Confidential Information, or (ii) for trade secret reasons or business reasons; (b) to delay such submission for an additional [***] as may be reasonably necessary to seek patent protection for the information disclosed in such proposed submission; and/or (c) to prevent such presentation or publication based on a good faith determination that such publication or presentation will have an adverse effect on the non-publishing Party's ability to procure a patent or, with regard to Baxalta, to Commercialize Licensed Products. Notwithstanding the foregoing, after the Commercial Option Exercise Date, Baxalta shall not be required to share proposed presentations and publications with Precision that relate to any Licensed Product provided that any such presentations and publications (x) do not include any of Precision's Confidential Information and (y) do not contain any information that has a reasonable likelihood of adversely affecting Precision's ability to obtain a Patent on any invention owned by Precision relating to Precision Platform Technology or the

12.4 Press Releases; Publicity.

[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 12.4.1 <u>Press Release</u>. Except with respect to the press release attached hereto as <u>Exhibit F</u> which will be jointly issued by the Parties upon execution of this Agreement, and except as otherwise permitted under this Agreement, no disclosure shall be made by either Party concerning the execution of this Agreement or the terms and conditions hereof without the prior written consent of the other Party, which shall not be unreasonably withheld, conditioned or delayed.
- 12.4.2 <u>Disclosure of Agreement to Third Parties.</u> Notwithstanding the foregoing in <u>Section 12.4.1</u>, each Party may disclose a copy of this Agreement without the prior written consent of the other Party pursuant to <u>Section 12.1.2(e)</u> or <u>Section 12.2.2(e)</u>, as applicable. In addition, the Parties will use good faith efforts to agree in writing upon a redacted form of this Agreement within [***] after the Effective Date that redacts information that each Party considers (in good faith) to be of a competitively sensitive nature (the "<u>Redacted Form of Agreement</u>"), and each Party may disclose a copy of the Redacted Form of Agreement without the prior written consent of the other Party in connection with a due diligence process associated with the negotiation or exploration of a possible financing transaction by such Party or the negotiation or exploration of a possible acquisition transaction involving such Party (e.g., transfer or sale of all or any portion of such Party's assets, equity or business, or a merger or consolidation or similar business combination transaction or otherwise); provided that (a) such disclosure is made in the course of such diligence, negotiation or exploration pursuant to confidentiality and nonuse obligations consistent with those set forth in this Agreement and (b) that the duration of such terms and conditions of confidentiality for such recipients shall be no less than five (5) years. Notwithstanding the foregoing, either Party may provide an unredacted copy to the Third Party after such Party and the Third Party have executed a non-binding term sheet or letter of intent or have otherwise entered into a period of exclusive negotiation with respect to a possible financing transaction or acquisition transaction involving such Party.
- 12.4.3 <u>General</u>. Except as otherwise permitted under this Agreement, (a) Precision may not issue a press release or public announcement concerning the Development of any particular Licensed Product Candidates or Licensed Products without Baxalta's prior written consent, and (b) prior to the Commercial Option Exercise Date for any Licensed Product Candidate, Baxalta may not issue a press release or public announcement concerning the Development of any particular Licensed Product Candidates without Precision's prior written consent.
- 12.4.4 <u>Disclosures Required by Law</u>. Each Party agrees that it shall cooperate fully and in a timely manner with the other Party with respect to all disclosures required by the Securities and Exchange Commission and any other Governmental Authority or Regulatory Authority, including requests for confidential treatment of Confidential Information of either Party included in any such disclosure. Notwithstanding any other provision of this Agreement, either Party may issue any public announcement or other disclosure that it is advised by legal counsel is required under applicable Laws or the applicable rules of any recognized stock exchange or

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

quotation system, provided that the Parties shall coordinate with each other with respect to the timing, form and content of such required disclosure to the extent practicable under the circumstances, and the Party seeking such disclosure shall use Commercially Reasonable Efforts to provide the other Party with reasonable advance notice thereof (including a copy of the proposed disclosure) and, in any event, at least [***] advance notice. Any request for revision to the content of said disclosure by the non-disclosing Party shall be furnished to the disclosing Party as promptly as necessary for the disclosing Party to comply with such requirements in a timely manner. Without limiting the foregoing, each Party shall consult in good faith with the other Party on the provisions of this Agreement (for the avoidance of doubt, including the schedules and exhibits hereto) to be redacted in any filings made by either Party with the Securities and Exchange Commission or as otherwise required by applicable Law.

12.4.5 <u>Right to Further Disclose</u>. Once a public disclosure that is required pursuant to applicable Law, pursuant to applicable rules of any recognized stock exchange or quotation system or by the Securities and Exchange Commission or other Governmental Authority or Regulatory Authority, is made, in each case in accordance with this <u>Section 12.4</u>, the content of such disclosure (or any portion thereof) may be repeated in one or more subsequent disclosures without any obligation of the disclosing Party to give any notices or obtain any consents for such disclosure that would otherwise be required under this <u>Section 12.4</u>.

ARTICLE XIII. INDEMNIFICATION

13.1 Precision Indemnity. Precision shall defend, indemnify and hold harmless Baxalta and its Affiliates, successors and permitted assignees and each of its and their respective Agents (collectively, the "Baxalta Indemnitees") from and against any and all liabilities, losses, costs, damages and expenses, including reasonable attorneys' fees (collectively, "Damages"), to the extent arising out of or resulting from any claim, suit, action, demand or other proceeding made or brought against one or more Baxalta Indemnitees by a Third Party in connection with (a) the gross negligence, recklessness, or intentional wrongful acts or omissions of Precision or its Affiliates or their respective employees, directors, representatives, consultants, independent contractors or agents (collectively, "Agents"), in connection with the performance by or on behalf of Precision of Precision's obligations or exercise of Precision's rights under this Agreement, (b) any material breach by Precision or its Affiliates or their respective Agents of any representation, warranty, covenant or obligation of Precision set forth in this Agreement, and (c) [***]; except, in any such case, to the extent such Damages are reasonably primarily attributable to any gross negligence, recklessness, willful misconduct or breach of this Agreement by Baxalta or a Baxalta Indemnitee (other than any breach by Baxalta or a Baxalta Indemnitee that primarily resulted from Precision's or its Affiliates' or their respective Agents' breach of this Agreement).

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

13.2 <u>Baxalta Indemnity</u>. Baxalta shall defend, indemnify and hold harmless Precision and its Affiliates, successors and permitted assignees and each of its and their respective Agents (collectively, the "<u>Precision Indemnitees</u>") from and against any and all Damages, to the extent arising out of or resulting from any claim, suit, action, demand or other proceeding made or brought against one or more Precision Indemnitees by a Third Party in connection with (a) the gross negligence, recklessness, or intentional wrongful acts or omissions of Baxalta or its Affiliates or their respective Agents, in connection with the performance by or on behalf of Baxalta of Baxalta's obligations or exercise of Baxalta's rights under this Agreement, (b) any material breach by Baxalta or its Affiliates or their respective Agents of any representation, warranty, covenant or obligation of Baxalta set forth in this Agreement, and (c) [***]; except, in any such case, to the extent such Damages are reasonably primarily attributable to any gross negligence, recklessness, willful misconduct or breach of this Agreement by Precision or a Precision Indemnitee (other than any breach by Precision or a Precision Indemnitee that primarily resulted from Baxalta's or its Affiliates' or their respective Agents' breach of this Agreement).

13.3 <u>Indemnification Procedure</u>.

13.3.1 Each Party shall notify the other in the event it becomes aware of a claim for which indemnification may be sought pursuant to this Article XIII. In case any claim, suit, action, demand or other proceeding (including any governmental investigation) shall be instituted involving any Party or its Indemnitees in respect of which indemnity may be sought pursuant to this Article XIII, such Party (the "Indemnified Party") shall promptly notify the other Party (the "Indemnifying Party") in writing (an "Indemnification Claim Notice"); provided, that the failure to promptly provide an Indemnification Claim Notice shall not relieve the Indemnifying Party of its indemnification obligations except, and only to the extent, that the Indemnifying Party is actually incrementally damaged as a result of such failure. The Indemnifying Party and Indemnified Party shall promptly meet to discuss how to respond to any claims that are the subject matter of such proceeding. At its option, the Indemnifying Party may assume the defense of a Third Party claim subject to indemnification as provided for in this Section 13.3 with competent counsel free of any conflict of interest with the Indemnified Party by giving written notice (a "<u>Defense Election Notice</u>") to the Indemnified Party within [***] after its receipt of the applicable Indemnification Claim Notice (the "Election Time Period"), solely for claims (a) that solely seek monetary damages and (b) as to which the Indemnifying Party expressly agrees in writing that, as between the Indemnifying Party and the Indemnified Party, the Indemnifying Party shall be solely obligated to satisfy and discharge the claim in full (the matters described in (a) and (b), the "Litigation Conditions"). If the Indemnifying Party does not deliver a Defense Election Notice to the Indemnified Party during the applicable Election Time Period, or if any Litigation Condition is not satisfied, the Indemnified Party will assume responsibility for and control such defense and, without limiting the Indemnifying Party's indemnification obligations, the Indemnifying Party will reimburse the Indemnified Party for all costs and expenses, including reasonable attorneys' fees, incurred by the Indemnified Party in defending itself.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 13.3.2 Upon assuming the defense of a Third Party claim in accordance with this Section 13.3, the Indemnifying Party shall be entitled to appoint competent counsel free of any conflict of interest with the Indemnified Party in the defense of the Third Party claim. Should the Indemnifying Party assume and continue the defense of a Third Party claim, except as otherwise set forth in this Section 13.3, the Indemnifying Party will not be liable to the Indemnified Party for any legal expenses subsequently incurred by such Indemnified Party after the date of assumption of defense in connection with the analysis, defense, countersuit or settlement of the Third Party claim. Without limiting this Section 13.3, any Indemnified Party will be entitled to participate in, but not control, the defense of a Third Party claim for which it has sought indemnification hereunder and to engage counsel of its choice for such purpose; provided, however, that such engagement will be at the Indemnified Party's own cost and expense unless (a) the engagement thereof has been specifically requested by the Indemnifying Party in writing, or (b) the Indemnifying Party has failed to assume and actively further the defense and engage counsel in accordance with this Section 13.3 (in which case the Indemnified Party will control the defense), or (c) the Indemnifying Party no longer satisfies the Litigation Conditions.
- 13.3.3 Subject to the Litigation Conditions continuing to be satisfied, the Indemnifying Party will have the sole right to consent to the entry of any judgment, enter into any settlement or otherwise dispose of such Damages, on such terms as the Indemnifying Party, in its reasonable discretion, will deem appropriate (provided, however, that such terms (a) shall include a complete and unconditional release of the Indemnified Party from all liability with respect thereto and (b) shall not include any admission of fault by, or impose any liability or obligation on, the Indemnified Party), and will transfer to the Indemnified Party all amounts which said Indemnified Party will be liable to pay pursuant to such settlement or disposal of such claim prior to the time such payments become due by the Indemnified Party. With respect to all other entries of judgment, entries into settlements or other dispositions of Damages in connection with a Third Party claim for which the Indemnifying Party has assumed the defense in accordance with this Section 13.3, the Indemnifying Party will only have authority to consent to the entry of such judgment, entry into such settlement or such other disposition of Damages if it has obtained the Indemnified Party's prior written consent, not to be unreasonably withheld, conditioned or delayed.
- 13.3.4 The Indemnifying Party that has assumed the defense of the Third Party claim in accordance with this Section 13.3 (and continues to maintain control of such defense pursuant to this Section 13.3) will not be liable for any settlement or other disposition of any Damages by an Indemnified Party that is reached without the prior written consent of such Indemnifying Party. The Indemnified Party will not admit any liability with respect to, or settle, compromise or discharge, any Third Party claim without first offering to the Indemnifying Party the opportunity to assume the defense of the Third Party claim in accordance with this Section 13.3. If the Indemnifying Party chooses to defend or prosecute any Third Party claim, the Indemnified Party will cooperate in the defense or prosecution thereof and will furnish such records, information and testimony, provide such witnesses including to the extent possible, former employees and attend such conferences, discovery proceedings, hearings, trials and appeals as may be reasonably requested in connection with such Third Party claim. Such cooperation will include access during normal business hours afforded to the Indemnifying Party to, and reasonable retention by the Indemnified Party of, records and information that are reasonably relevant to such

Third Party claim, and making Agents available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. The Indemnifying Party will reimburse the Indemnified Party for all its reasonable out of pocket expenses incurred in connection with such cooperation.

13.3.5 Each Party shall maintain, at its cost, a program of insurance or self-insurance against liability and other risks associated with its activities and obligations under this Agreement, including its Clinical Trials, its Development, use, manufacture and Commercialization of any Licensed Product Candidates or Licensed Products and its indemnification obligations hereunder, in such amounts, subject to such deductibles and on such terms as are customary for the activities to be conducted by it under this Agreement. All insurance required by this Section 13.3.5 shall be maintained during the Term and each Party shall, from time to time, provide copies of certificates of such insurance to the other Party upon request. Further, each Party shall list the other Party as an additional insured on all insurance policies. All insurance required by this Section 13.3.5 shall be maintained for at least [***] following expiration or termination of this Agreement.

13.4 <u>Limitation of Liability</u>; <u>Exclusion of Damages</u>; <u>Disclaimer</u>.

13.4.1 EXCEPT TO THE EXTENT A PARTY IS REQUIRED TO PROVIDE INDEMNIFICATION UNDER SECTION 13.1 OR SECTION 13.2, AND EXCEPT IN THE CASE OF A BREACH OF ARTICLE XII, AND WITHOUT LIMITING THE LIABILITY OF A PARTY FOR INFRINGEMENT OR MISAPPROPRIATION OF THE INTELLECTUAL PROPERTY RIGHTS OF THE OTHER PARTY OR ANY OF ITS AFFILIATES OR FOR FRAUD OR WILLFUL MISCONDUCT, NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY FOR SPECIAL, INDIRECT, INCIDENTAL, PUNITIVE, OR CONSEQUENTIAL DAMAGES (INCLUDING DAMAGES RESULTING FROM LOSS OF USE, LOSS OF PROFITS, INTERRUPTION OR LOSS OF BUSINESS, DIMINUTION OF VALUE, OR OTHER ECONOMIC LOSS) ARISING OUT OF THIS AGREEMENT OR WITH RESPECT TO A PARTY'S PERFORMANCE OR NON PERFORMANCE HEREUNDER.

13.4.2 EXCEPT AS EXPRESSLY PROVIDED IN THIS AGREEMENT, NEITHER PARTY PROVIDES ANY REPRESENTATIONS OR WARRANTIES, WHETHER WRITTEN OR ORAL, EXPRESS OR IMPLIED, REGARDING ANY SUBJECT MATTER OF THIS AGREEMENT AND EACH PARTY HEREBY DISCLAIMS ALL OTHER REPRESENTATIONS AND WARRANTIES, WHETHER WRITTEN OR ORAL, EXPRESS AND IMPLIED, INCLUDING REGARDING TITLE, VALIDITY, PATENTABILITY, ENFORCEABILITY OF PATENT RIGHTS, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND FREEDOM FROM INFRINGEMENT OF THIRD PARTY RIGHTS, AND ANY WARRANTIES ARISING FROM A COURSE OF DEALING, USAGE OR TRADE PRACTICES.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE XIV. TERM; TERMINATION

14.1 Term. This Agreement shall become effective as of the Effective Date and shall remain in effect until the first to occur of (a) the instance that, at any time from and after the expiration of the Nomination Period, there are no Included Targets, (b) expiration of all Commercial Option Exercise Periods for all Licensed Product Candidates if no Commercial Option Exercise Dates have occurred, or (c) the expiration of the last to expire Royalty Term in the Territory and satisfaction of all of Baxalta's payment obligations under this Agreement; in each case of (a), (b) and (c), unless this Agreement is earlier terminated in its entirety pursuant to Section 14.2 (the "Term"). In the event of expiration of this Agreement pursuant to clause (c) of this Section 14.1, (x) the Commercial Licenses previously granted to Baxalta upon exercise of Commercial Options shall continue in full force and effect, shall not be revocable and shall be considered to be fully paid up and (y) subject to the confidentiality obligations contained in Article XII, Baxalta shall have the right to freely use all Know-How disclosed to it by Precision hereunder solely in connection with the Development, manufacture and Commercialization of Licensed Products and Licensed Product Candidates Directed to Included Targets in the Field.

14.2 Early Termination.

14.2.1 By Either Party for Material Breach. Without prejudice and in addition to any other contractual remedy the non-breaching Party may have with respect to this Agreement, either Party may, upon a material breach of this Agreement (as it relates to this Agreement in its entirety or to any particular Licensed Product Candidate or Licensed Product) by the other Party, terminate this Agreement (at the non-breaching Party's option, either (a) in its entirety or (b) for any material breach of the Agreement relating only to certain of the Licensed Product Candidate(s) or Licensed Product(s), (i) with respect to the affected Licensed Product Candidate(s) or Licensed Product(s), or (ii) with respect to the affected Included Target, including all Licensed Product Candidates and Licensed Products Directed to such Included Target) by providing [***] prior written notice [***] to the breaching Party, specifying in such notice the breaching Party's material breach and demanding its cure, with such termination being effective upon the end of such [***] cure period or, if applicable, the end of the extended cure period set forth in the immediately following sentence, in each case if the applicable material breach has not then been cured. [***]. Neither Party may terminate this Agreement pursuant to this Section 14.2.1 if the other Party's material breach of this Agreement.

14.2.2 By Precision for Patent Challenge. Precision shall have the right to terminate this Agreement in its entirety immediately upon written notice to Baxalta if Baxalta, itself or through an Affiliate (other than any Person within the Baxalta Parent Group), or any of its Sublicensees, directly or through assistance granted to a Third Party commences or participates in any administrative, judicial or similar proceeding challenging the validity, enforceability and/or

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

patentability of any Precision Patent or Precision Platform Patent (except where such action is ordered by a court, patent office or other tribunal or is required by Law), (such action, a "<u>Patent Challenge</u>"), however, Patent Challenge shall exclude (a) any *bona fide* patent infringement defense against a claim of infringement of such Patent Rights brought by Precision, a Precision Affiliate or a Third Party and (b) in response to a claim by Precision that a royalty payment is due, assertion by Baxalta that the Royalty Term has expired because there is no longer a Valid Claim of a Precision Patent Covering a Licensed Product in the applicable country.

- 14.2.3 By Either Party for Insolvency. Either Party (the "Non-Debtor Party") may terminate this Agreement in its entirety effective immediately upon delivery of written notice to the other Party (the "Debtor Party") if the Debtor Party is dissolved or liquidated, files or has filed against it a petition as a debtor under Title 11 of the U.S. Bankruptcy Code (excluding a reorganization proceeding under Chapter 11 of the U.S. Bankruptcy Code if the Debtor Party is continuing to perform all of its obligations under this Agreement) that is not dismissed within [***], makes an assignment of all or substantially all of its property for the benefit of its creditors or has a receiver or trustee appointed for all or substantially all of its property.
 - 14.2.4 For Force Majeure. This Agreement may be terminated in its entirety as set forth in Section 17.15.2.
- 14.2.5 By Baxalta for Convenience. Baxalta may terminate this Agreement (a) in its entirety or (b) with respect to one or more Included Targets (including all Licensed Product Candidates and Licensed Products Directed to such Included Targets), or with respect to one or more Licensed Product Candidates or Licensed Products, in each case effective at any time after the second anniversary of the Effective Date, by providing [***] prior written notice to Precision, specifying in such notice which Included Targets, Licensed Product(s) or Licensed Product Candidate(s) the termination applies to and that the termination is for convenience, with such termination being effective upon the later of the end of such [***] notice period or the second anniversary of the Effective Date. The Parties shall continue to perform their respective obligations hereunder with respect to the Included Targets, Licensed Product(s) and Licensed Product Candidate(s) to which the termination applies until the effective date of the termination as described in the immediately preceding sentence. For clarity, if Baxalta has terminated this Agreement with respect to all Licensed Product Candidates and Licensed Products Directed to a particular Included Target, then Baxalta shall be deemed to have terminated this Agreement with respect to such Included Target pursuant to this Section 14.2.5.

14.3 Effects of Termination.

- 14.3.1 Effect of Termination by Precision for [***] or Termination by Baxalta for [***].
- (a) In the event of termination by Precision [***], in each case without prejudice and in addition to any contractual remedy either Party may have with respect to this Agreement, the following shall apply:
- [***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

(i) Baxalta agrees to grant and hereby grants to Precision and its Affiliates, effective upon such termination, a royalty-bearing, worldwide, transferrable (solely in connection with a transfer of the rights related to the applicable Reversion Product), irrevocable license (with the right to grant sublicenses through multiple tiers) under the Reversion IP to make, have made, use, sell, offer for sale, import and otherwise Develop and Commercialize Reversion Products in the Field in the Territory. Such license will be exclusive (even as to Baxalta and its Affiliates) or non-exclusive, on a Patent-by-Patent basis, based on Precision's election made in a written notice to Baxalta with respect to each of the Patents included in the Reversion IP ("Reversion Patents"). Precision may also decline to accept the foregoing license on a Patent-by-Patent basis by providing written notice thereof to Baxalta. With respect to any Reversion Patents licensed to Precision on an exclusive basis, Precision shall have all of the rights granted to Baxalta in (A) Sections 9.5, 9.7, 9.8 and 9.9 determined mutatis mutandis as if such Reversion Patents were Precision Patents under such Sections, and (B) Section 9.2 determined mutatis mutandis as if such Reversion Patents were Product-Related Patents under such Section. During the period in which a Valid Claim of any Reversion Patent licensed to Precision would, but for the license granted in this Section 14.3.1(a)(i) be infringed by Development, manufacture, use or Commercialization of a particular Reversion Product in the country of sale, on a Reversion Product-by-Reversion Product, country-by-country basis, Precision shall pay to Baxalta annual (once per Calendar Year) non-refundable, non-creditable royalty payments equal to [***]. Such royalty payment shall be subject to Sections 8.4.2, 8.4.6, 8.4.7, 8.5, 8.6, 8.7 and 8.8, each determined mutatis mutandis with respect to sales by Precision, its Affiliates and sublicensees and Reversion Patents (and without Section 9.3 limiting Precis

(ii) Upon Precision's request and, except in connection with termination by Precision in accordance with <u>Section 14.2.1</u> or <u>Section 14.2.2</u>, at Precision's expense (including payment at the then-prevailing FTE rate for work provided by Baxalta employees), as to any or all of the following, Baxalta shall:

(1) to the extent permitted by Law or the terms of any applicable Third Party agreements, assign to Precision its and its Affiliates' entire right, title and interest in and to all materials, licenses, Third Party agreements, preclinical and clinical data, safety data and all other supporting data that is in Baxalta's or its Affiliates' Control, and deliver to Precision a copy of all relevant Know-How, in each case that relates to, and to the extent reasonably necessary for, Precision to continue the Development, use, manufacture and Commercialization of the Reversion Products; provided, that Baxalta shall not be obligated to translate or reformat any data or to convert or adapt any database or software (it being understood that such data and databases shall be transferred on an as-is basis);

(2) furnish Precision with reasonable cooperation to transition to Precision the management and continued performance of any clinical or other studies in progress then being conducted by Baxalta or its Affiliates related to the Reversion Products which Precision determines to continue in compliance with applicable Laws and ethical guidelines applicable to the transfer or termination of any such studies; <u>provided</u>, that in the event that Precision informs Baxalta that it does not intend to continue specific Development activities then in progress, Baxalta shall bear its own costs and expenses incurred in closing out such activities;

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

(3) transfer to Precision any and all Regulatory Filings, Global Dossiers and related regulatory data and Know-How, Marketing Approvals and Reimbursement Approvals and all other filings and submissions with and to Regulatory Authorities with respect to the Reversion Products; and, to this end, Baxalta shall use Commercially Reasonable Efforts to file for transfer with the relevant Regulatory Authorities and to give all other notifications and approvals necessary under applicable Law for the transfer of Regulatory Filings, Global Dossiers and related regulatory data and Know-How, Marketing Approvals and Reimbursement Approvals and such other filings and submissions;

(4) use Commercially Reasonable Efforts to transfer to Precision the manufacturing processes, documents, materials and other Know-How reasonably necessary for the manufacture, testing and supply of the Reversion Products, in each case to the extent (A) Controlled by Baxalta, (B) Baxalta is legally or contractually, as applicable, permitted to so transfer, and (C) such processes, documents, materials and other Know-How are actually used (at the time of the applicable termination) by or on behalf of Baxalta or its Affiliates in the manufacture of the Reversion Products (in all cases, provided that Baxalta shall not be obligated to incur any costs or expenses from the use of such transferred processes, documents, materials or other Know-How by or on behalf of Precision);

(5) provide reasonable technical assistance relating to the manufacture, testing and supply of the Reversion Products as necessary for Precision to be qualified or to qualify a Third Party for the manufacturing of such Reversion Products, such assistance being limited to assistance that a manufacturer familiar with, and having experience with equipment for, manufacturing of CAR-T, and in any case not to exceed a total of [***];

(6) after fulfillment of Baxalta's existing commitments to its customers (including its distributors), sell to Precision Baxalta's then-existing inventory of Reversion Products, at [***]; <u>provided</u> that Precision shall not be obligated to purchase such inventory and such sale shall only occur if Precision shall notify Baxalta within thirty (30) days after the effective date of termination that Precision elects to exercise such right;

(7) to the extent permitted by Law, transfer to Precision ongoing Clinical Trials of Reversion Products being conducted by or under authority of Baxalta as of the date of the applicable termination notice, continue to conduct such Clinical Trials up to such transfer or, if requested by Precision, terminate such Clinical Trials in a manner conforming to applicable Laws; <u>provided</u>, that Baxalta shall not be obligated to translate or reformat documents or databases or to convert or adapt any database or software (it being understood that such documents and databases shall be transferred on an as-is basis);

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (8) Baxalta shall not withdraw or cancel any Reversion Product Marketing Approval or Reimbursement Approval or application for either, unless expressly instructed so by Precision in writing; <u>provided</u> that Precision shall be responsible for all costs and expenses for the maintenance of all Marketing Approvals and Reimbursement Approvals following receipt of notice of termination;
- (9) Baxalta shall negotiate in good faith with Precision a royalty-free, fully paid up license for use of the Licensed Product Marks that have been used in commerce solely with Reversion Products (excluding any corporate name or logo of Baxalta or any of its Affiliates and any trademarks that are used by Baxalta or any of its Affiliates on products that are not Reversion Products), together with all goodwill relevant thereto, throughout the Territory;
- (10) Baxalta shall thereafter refrain from making any statement, public or otherwise, regarding any Reversion Products unless Baxalta is required to make such statement pursuant to applicable Law, such statement is limited to the fact that Baxalta is no longer Developing or Commercializing the Reversion Products or Precision shall have approved any such statement in writing; and
- (11) following written request by Precision, Baxalta shall take such other actions and execute such other instruments, assignments and documents that are reasonably necessary to effect the transfers and grants of rights under this <u>Section 14.3.1</u> to Precision.

Following the foregoing assignments and transfer, all information and Know-How so assigned or transferred that was previously Confidential Information of Baxalta shall thereafter be deemed the Confidential Information of both Parties under <u>Article XII</u>. In addition, following the foregoing assignments and transfer and without limiting Precision's other rights and remedies under this Agreement for such breach, Baxalta shall be deemed to no longer be in continuing breach of this Agreement.

- (iii) For the avoidance of doubt, in the event that this Agreement is terminated in its entirety or with respect to one or more Included Targets, the terminated Included Target(s) shall not be subject to any further Nomination right of Baxalta under this Agreement, and the Reversion Products shall not be subject to the Precision Restrictive Covenants.
 - (b) With respect only to termination by Baxalta [***] upon Precision's request, Baxalta will [***].
- 14.3.2 Effect of Termination by Baxalta for Precision's Uncured Material Breach of the Precision Restrictive Covenants or Enablement of an Acquirer to Conduct a Competitive Program; [***].
- (a) The provisions set forth in subsections (b) and (c) of this Section 14.3.2 shall apply if Baxalta terminates this Agreement with respect to an Included Target pursuant to Section 14.2.1 based on a Final Determination that one of the following has occurred with respect to such Included Target:
- (i) Precision has materially breached the Precision Restrictive Covenants and fails to cure such breach (if capable of cure) within ninety (90) days after receipt of Baxalta's written notice of such breach (or, in the case of a good faith dispute as to whether a breach occurred, within ninety (90) days after a Final Determination that such breach occurred), provided, that, without limiting the foregoing in this clause (i), any such breach of the Precision Restrictive Covenants shall be deemed material and incapable of cure if such breach resulted in Commercialization of any product comprising engineered human T cells with Chimeric Antigen Receptors in the Field (whether Allogeneic, autologous or otherwise) that is Directed to the applicable Included Target; or
- (ii) Precision does any of the following (each of which shall be deemed a material breach of this Agreement) during the term the Firewall is required to be in effect and ending on the Commercial Option Exercise Date for the Included Target that is the subject of the Firewall: (x) affirmatively performs a technology transfer to the Acquirer of the Precision Know-How or Precision Platform Technology in a manner that enables the Acquirer to conduct a Competitive Program Directed to such Included Target based on the Precision Know-How or Precision Platform Technology, (y) grants the Acquirer a license or a covenant not to assert under the Precision Know-How, Precision Platform Technology, Precision Platform Patents or Precision Patents in a manner that enables the Acquirer to conduct such a Competitive Program, or (z) transfers ownership of the Precision Know-How, Precision Platform Technology, Precision Platform Patents or Precision Patents to the Acquirer in the absence of restrictions requiring that such Precision Know-How, Precision Platform Technology, Precision Platform Patents or Precision Patents not be used to conduct such a Competitive Program; and fails to cure such breach (if capable of cure) within ninety (90) days after receipt of Baxalta's written notice of such breach (or, in the case of a good faith dispute as to whether a breach occurred, within ninety (90) days after a Final Determination that such breach occurred).
- (b) The licenses granted to Baxalta under Section 2.5.1 and Section 4.3 of this Agreement with respect to the Licensed Product Candidates and Licensed Products Directed to the Included Target that was the subject of such breach will become royalty-free, fully paid up, perpetual and irrevocable, with Baxalta being deemed to have previously exercised the Commercial Option for any then-existing Licensed Product Candidates Directed to such Included Target (but shall have no obligation to pay the Commercial Option Exercise Fee).

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

(c) [***]

- 14.3.3 <u>Effect of Termination for Any Reason</u>. In the event of termination of this Agreement (in whole or in part) for any reason (including the reasons set forth in <u>Section 14.3.1</u> and <u>Section 14.3.2</u>), in each case without prejudice and in addition to any contractual remedy either Party may have with respect to this Agreement, in addition to the rights and obligations set forth in <u>Section 14.3.1</u> and <u>Section 14.3.2</u> (as and if applicable), the following shall apply:
- (a) If such termination is with respect to one or more, but not all, Licensed Product Candidates or Licensed Products, or one or more, but not all, Included Targets, the effects of termination set forth in Section 14.3.1 or Section 14.3.2, as applicable, and Section 14.3.3(b) shall apply solely as to such Licensed Product Candidates or Licensed Products, or in the case of such termination with respect to an Included Target apply solely as to such Included Target and all Licensed Product Candidates and Licensed Products Directed to such Included Target. Except as otherwise expressly set forth in this Section 14.3.3 or in Section 14.3.1, Section 14.3.2 or Section 14.3.4, all rights and obligations of the Parties under this Agreement with respect to any terminated Licensed Product Candidates, terminated Licensed Products and terminated Included Targets shall cease.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (b) Except as expressly set forth in Section 14.3.1(a)(i), Section 14.3.2(b) or Section 14.3.2(d) (as and if applicable), all licenses granted to either Party under this Agreement, including all sublicenses thereunder, shall immediately terminate. Notwithstanding the foregoing, if this Agreement is terminated by Precision pursuant to Section 14.2.3 (Termination for Insolvency), any sublicenses granted by Baxalta prior to the effective date of such termination shall continue provided that (i) the Sublicensee consents to the assignment of its sublicense agreement to Precision and agrees in writing with Precision to render all performance required under its sublicense agreement directly to Precision, (ii) the Sublicensee cures any uncured material breach of Baxalta existing at such time within [***] of such termination, and (iii) the Sublicensee agrees in writing with Precision that Precision will not be obligated to perform under such sublicense to the extent such sublicense requires obligations of Precision that exceed the obligations of Precision under this Agreement.
- (c) Within [***] following the expiration of this Agreement or termination of this Agreement in its entirety, each Party shall, at the request of the other Party, (i) deliver to the other Party, or certify the destruction of any and all tangible Confidential Information of the other Party in such Party's possession, (ii) to the extent practicable, remove Confidential Information of the other Party from all databases and systems and in those instances where removal is not practicable, segregate or otherwise indicate that such Confidential Information is restricted, and/or (iii) treat all Confidential Information of the other Party contained in lab notebooks in accordance with such Party's then current procedure for the status of the project and properly note that such Confidential Information contained in such lab notebooks is restricted. Notwithstanding the foregoing, Precision may retain such Confidential Information of Baxalta as is necessary or useful for the practice of the rights granted to it under Section 14.3.1.
- (d) With respect to termination of this Agreement in part, this Agreement shall continue in full force and effect with respect to the Development, manufacture and Commercialization of any remaining Licensed Product Candidates and Licensed Products until the effective date of expiration or termination with respect thereto, without any modification to this Agreement unless otherwise mutually agreed between the Parties.
- 14.3.4 <u>Survival</u>. The following Articles and Sections of this Agreement, as well as remedies for breach of this Agreement, shall survive expiration or termination of this Agreement for any reason: <u>ARTICLE I</u> (solely to the extent required to enforce any other surviving rights or obligations of the Parties), <u>Section 2.6.1(c)</u>, <u>Section 4.5</u>, <u>ARTICLE VIII</u> (solely to the extent any right to payment has accrued prior to the effective date of termination), <u>Section 9.1</u>, <u>Section 12.1</u>, <u>Section 12.2</u>, <u>Section 12.4</u>, <u>ARTICLE XIII</u>, <u>Section 14.3</u>, <u>ARTICLE XV</u> and <u>ARTICLE XVII</u>.
- 14.3.5 Other Remedies. Termination or expiration of this Agreement for any reason shall not release any Party from any liability or obligation that has accrued prior to such termination or expiration, nor affect the survival of any provision hereof to the extent it is expressly stated to survive termination or expiration. Termination or expiration of this Agreement for any reason shall not constitute a waiver or release of, or otherwise be deemed to prejudice or adversely affect, any rights, remedies, or claims, whether for damages or otherwise, that a Party may have hereunder with respect to the period prior to such termination or expiration or that may arise out of or in connection with such termination or expiration.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE XV. DISPUTE RESOLUTION

- 15.1 Resolution by Executive Officers; Mediation. The Parties shall attempt to resolve any and all disputes, claims or controversies arising out of or relating to this Agreement promptly by negotiation between Executive Officers of the respective Parties, in each case who have authority to settle the dispute, claim or controversy. If a Party provides written notice to the other Party regarding any such dispute, claim or controversy and such dispute, claim or controversy is not resolved through such negotiation procedures within [***] after receipt of such written notice by the other Party (or immediately if the Executive Officers have been unable to resolve such dispute, claim or controversy pursuant to Section 3.1.6(e)), then the Parties shall endeavor in good faith to settle the dispute, claim or controversy by confidential mediation by the International Institute for Conflict Prevention & Resolution ("CPR") pursuant to the CPR Mediation Procedure for a period not to exceed [***]. The Parties will select a mediator from the CPR Panels of Distinguished Neutrals ("CPR Panels"). If such dispute, claim or controversy is not resolved by the Executive Officers or by mediation, then it shall be submitted for final and binding arbitration pursuant to Section 15.2.
- 15.2 <u>Arbitration</u>. To the extent not resolved pursuant to <u>Section 15.1</u>, any dispute, claim or controversy arising out of or relating to this Agreement (for the avoidance of doubt, including the Development Plans) or the alleged breach, termination, enforcement, interpretation or validity hereof, including the determination of the scope or applicability of this agreement to arbitrate, shall be determined solely by arbitration in New York, New York, in the language in which this Agreement is written. Notwithstanding anything to the contrary in this Agreement, (a) any disputes, claims or controversies arising out of, or for which resolution depends in whole or in part on a determination of the interpretation, validity, enforceability or infringement of, Patent Rights or the misappropriation of trade secrets, shall not be subject to arbitration under this Agreement and (b) each Party may apply to any court of law or equity of competent jurisdiction for temporary injunctive or other interim relief, pending completion of arbitration, to enforce or prevent any violation of this Agreement.
- 15.2.1 <u>Arbitration Format</u>. Any arbitration hereunder shall be administered by the CPR pursuant to its Rules for Administered Arbitration. References herein to any arbitration or mediation rules or procedures mean such rules or procedures as amended from time to time, including any successor rules or procedures, and references herein to the CPR include any successor thereto. The arbitration shall be before three (3) arbitrators, selected within [***] after the date such dispute, claim or controversy is referred for arbitration. Each Party shall designate one (1) arbitrator from the CPR Panels in accordance with the "screened" appointment procedure

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

provided in Rule 5.4 of the CPR Rules. The two Party-appointed arbitrators will select the third arbitrator from the CPR Panels and such third arbitrator will serve as the arbitration panel's chair or president. All three (3) arbitrators shall have at least ten (10) years' experience in biotechnology licensing, discovery, development or commercialization matters; provided that, in the event of a dispute, claim or controversy relating to the ownership of Patent Rights and Know-How developed under this Agreement, the arbitrators also shall be patent attorneys with at least ten (10) years' experience in the relevant subject matter area. This Section 15.2, and the arbitration itself, shall be governed by the laws of the State of New York and the Federal Arbitration Act, 9 U.S.C. §§ 1-16.

15.2.2 Arbitration Procedures. Consistent with the expedited nature of arbitration, each Party will, upon the written request of the other Party, promptly provide the other Party with copies of documents on which the producing Party may rely in support of or in opposition to any claim or defense and will identify all individuals the Party may call to testify in support of that Party's claim or defense. Each Party may obtain production of documents from the other Party in response to document requests not to exceed [***] in number, answers to interrogatories by the other Party not to exceed [***] in number, and the examination by deposition of witnesses as set forth in this Section 15.2.2. The arbitrators may permit each Party up to [***] additional document requests and [***] additional interrogatories upon a showing that such additional document requests and interrogatories are relevant and appropriate. Unless the Parties otherwise agree, depositions will occur and documents will be produced in New York, New York. Neither Party shall object to the production of documents on the grounds that documents are located outside the United States or are in the possession, custody or control of an Affiliate of the producing Party. Depositions of fact witnesses shall be limited to a maximum of [***]. Additional depositions may be taken only with the permission of the arbitrators, and for good cause shown. In addition to depositions of fact witnesses as provided above, each Party shall have the right to depose all testifying experts designated by the other Party. Each deposition shall be limited to a maximum of eight (8) hours of testimony conducted during [***] duration, except that any deposition involving an interpreter shall be limited to a maximum of [***] of testimony. If agreed by the Parties, any deposition or testimony may be conducted by videoconference. All objections are reserved for the arbitration hearing except for objections based on privilege, the form of questions, and proprietary or confidential information. Any dispute regarding discovery, or the relevance or scope thereof, shall be determined by the arbitrators. All discovery shall be completed within [***] following the appointment of the arbitrators unless the Party initiating such arbitration elects by written notice to the other Party to expand such period to a period of up to [***]. Each Party may present at the hearing a total of [***] of direct testimony. Testimony by sworn affidavit filed by a Party will be permitted provided the witness is made available by that Party at the hearing for cross-examination by the other Party. In the arbitrators' discretion, costs or fees relating to the retrieval, review and production of electronic discovery may be assessed in whole or in part against the Party requesting such discovery. All briefing, hearings, posthearing briefing, and the arbitral award shall be completed within [***] following the completion of discovery. A record that includes all hearings and all evidence (including exhibits, deposition transcripts, and affidavits admitted into evidence) shall be maintained in any arbitration in which [***].

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

15.2.3 <u>Arbitral Awards</u>. The panel of arbitrators shall have no power to award non-monetary or equitable relief of any sort, other than injunctions and specific performance. The arbitrators will have no authority to award punitive or other damages not measured by the prevailing Party's actual damages, except as may be required by applicable statute. The arbitrators shall have no power or authority, under the CPR Rules or otherwise, to relieve the Parties from their agreement hereunder to arbitrate or otherwise to amend or disregard any provision of this Agreement or to award any damages or other relief that conflicts with the express terms of this Agreement. In any arbitration in which [***], the arbitrators shall render a written decision setting forth the factual and legal bases of the award. The arbitrators also shall render a written decision setting forth the factual and legal bases of any award that [***]. In any case, the award of the arbitrators shall be final and binding on the Parties, except as such award may be modified pursuant to appeal as set forth in Section 15.2.4. Either Party may seek to confirm and enforce any final award entered in arbitration in any court of competent jurisdiction as provided in Section 17.13. The cost of the arbitration, including the fees of the arbitrators (excluding each Party's attorneys' fees), shall be borne by the Party the arbitrators determine has not prevailed in the arbitration; provided, if a Party does not prevail on all issues, the arbitrators shall award costs in reasonable proportion to the issues on which such Party prevails.

15.2.4 <u>Arbitral Appeals</u>. The arbitral award may be appealed to a tribunal of appellate arbitrators pursuant to the CPR Arbitration Appeal Procedure if [***]. Otherwise, the award shall not be appealable and shall be subject only to such challenges as are permitted under the Federal Arbitration Act, 9 U.S.C. §§ 1-16. In any appeal, the arbitration tribunal and the appeal tribunal may take such interim measures as it deems necessary. The Parties agree to use commercially reasonable efforts to conclude the appeal within [***] after its commencement. Any appeal will be conducted in New York, New York in the language in which this Agreement is written. The cost of the appeal, including the fees of the tribunal of appellate arbitrators (excluding each Party's attorneys' fees), shall be borne by the Party the appellate arbitrators determine has not prevailed in the arbitration; provided, if a Party does not prevail on all issues, the appellate arbitrators shall award costs in reasonable proportion to the issues on which such Party prevails.

15.2.5 <u>Confidentiality</u>. Except as may be required to confirm or enforce a final award, or as may be required by applicable Law, neither a Party nor an arbitrator may disclose the existence, content, or results of any arbitration hereunder without the prior written consent of both Parties.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE XVI. CERTAIN ADDITIONAL MATTERS

16.1 Change of Control of Precision.

16.1.1 Notice Regarding Potential or Completed Change of Control of Precision.

- (a) During the Nomination Period, Precision will provide prompt written notice to Baxalta in the event that (i) Precision executes or intends to execute a non-binding term sheet or letter of intent relating to the terms of a potential Change of Control of Precision, (ii) Precision has entered into or intends to enter into a period of exclusive negotiation with respect to a potential Change of Control of Precision, (iii) Precision receives an unsolicited offer proposing a potential Change of Control of Precision that Precision intends to accept or with respect to which Precision intends to initiate discussions with the Third Party making such a proposal, or (iv) Precision initiates an external process seeking offers from Third Parties with respect to a potential Change of Control of Precision, in each case subject to confidentiality obligations to Third Parties, the fiduciary obligations of Precision's executive officers and board of directors and applicable Laws, and it being understood and agreed that Precision shall not be required to disclose any further information (including the identity of any potential Acquirer) other than the fact that one of the foregoing events has occurred. Precision represents and warrants that, as of the Effective Date, none of the events described in subsections (i)—(iv) above has occurred and is currently in effect or ongoing.
- (b) Precision will provide Baxalta with prompt written notice prior to, contemporaneous with or promptly after the closing of any Change of Control of Precision.
- 16.1.2 [***]. Following the closing of a Change of Control of Precision, the following provisions shall apply with respect to each Included Target:

(a) Firewall.

- (i) Promptly following the first to occur of any of the following events in relation to [***]: (A) the closing of a Change of Control of Precision, or [***], in either case that results in [***], or (B) the initiation of [***] (each of (A) and (B), with respect to [***], the "Firewall Event"), Precision shall establish and enforce reasonable processes, policies, procedures and systems (the "Firewall") to segregate [***]. Such measures will include, without limitation, ensuring, during the term the Firewall is required to be in effect, that [***]. Precision will present the details of the Firewall for the JSC's review and approval, not to be unreasonably withheld, conditioned or delayed, and Precision will implement any reasonable Baxalta comments in a timely manner, but no later than [***] after receipt of such comments. If the Parties are unable to agree upon the appropriate processes, policies, procedures and systems for the Firewall, such dispute will be resolved in accordance with ARTICLE XV, however, during the pendency of any such dispute, Precision will implement the Firewall in the manner originally proposed by Precision, with any undisputed modifications proposed by Baxalta. Notwithstanding the foregoing, to the extent the foregoing provisions would otherwise prohibit such disclosure, Precision may disclose [***], to the extent that such disclosure is necessary to enable executive officers and the board of directors to comply with their fiduciary obligations to Precision or its Affiliates or to comply with applicable Laws.
- (ii) During the term the Firewall is required to be in effect, Precision shall conduct internal audits, at least annually, to verify compliance with the Firewall, identify any vulnerabilities or breaches in the Firewall, implement changes to the Firewall necessary to address any vulnerabilities identified and to determine whether [***]. If Precision identifies any vulnerability or breach in the Firewall that has resulted in [***], Precision shall promptly report to Baxalta [***], and will provide Baxalta with reasonable assurance that such action has been taken and has been effective. Otherwise, Precision will report to Baxalta that it has conducted the audit required by this Section 16.1.2(a)(ii), that [***], and describing any vulnerabilities or breaches of the Firewall identified, the action taken to remedy such vulnerabilities or breaches, and any changes to the Firewall that have been made as a result of the audit.
- (iii) Baxalta shall have the right to cause an independent consultant reasonably acceptable to Precision to audit the Firewall and its implementation and enforcement for purposes of confirming compliance with the Firewall and identifying any vulnerabilities or breaches. Precision may require such consultant to execute a reasonable confidentiality agreement with Precision prior to commencing the audit, provided that the results of such audit (excluding all Third Party information and all other information that Baxalta does not otherwise have the right to access under this Agreement) may be shared with Baxalta. Such audits may be

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

conducted during Precision's normal business hours, upon reasonable prior written notice to Precision, no more frequently than [***]. In addition, if the auditor identifies [***], Baxalta and/or the auditor will notify Precision, and Precision shall promptly take all action necessary and within Precision's reasonable control, including working with the Acquirer as necessary, to ensure that [***], and will provide Baxalta with reasonable assurance that such action has been taken and has been effective. Precision will implement any reasonable changes to the Firewall that are suggested by Baxalta and/or the auditor as a result of the audit.

- (b) <u>Additional Limitation for Included Targets [***]</u>. Without limiting the obligations of Precision set forth in <u>Section 16.1.2(a)</u>, from and after the designation by Baxalta of a Target as an Included Target in accordance with this Agreement, and regardless of whether [***], Precision covenants that it will not [***].
- (c) <u>Cessation of Certain Obligations</u>. The notification and update obligations set forth in <u>Section 2.4.1</u> and <u>Section 2.4.3</u> shall continue solely with respect to any Inactive Targets for which Precision has initiated Development prior to the date of closing of the Change of Control of Precision. For clarity, the rights and restrictions in connection with the Inactive Target Non-Disclosure Period shall not apply with respect to any sharing of any Inactive Target Evaluation Data with the Acquirer.
- (d) Effect on CDCP Option and Existing CDCP Agreements. Precision shall not have the right to exercise the CDCP Option pursuant to Section 7.3.2 with respect to any Licensed Product Directed to an Included Target for which [***]. If a CDCP Agreement has previously been executed with respect to any Licensed Product Directed to an Included Target for which [***], Baxalta may (at its option) terminate such CDCP Agreement within [***] and the Parties will agree upon terms for the orderly transition of Precision's activities under such CDCP Agreement to Baxalta (which may include, without limitation, acquisition by Baxalta of any sales force dedicated to the applicable Licensed Product). [***].
- 16.2 Effect of Final Determination of [***]. If a Final Determination is made that Precision has [***] and Precision has failed to cure such breach within [***] after receipt of Baxalta's written notice of such breach (or, in the case of a good faith dispute as to whether a breach occurred, within [***] after a Final Determination that such breach occurred), then upon Baxalta's written request, [***] the following provisions shall apply with respect to [***]
- 16.2.1 General. Except as modified pursuant to this Section 16.2, all provisions of this Agreement applicable to Licensed Product Candidates or Licensed Products, respectively, shall apply equally to [***], as applicable.
- 16.2.2 <u>Delivery of [***]</u>. Precision shall deliver to Baxalta [***]. Precision shall be deemed to no longer be in continuing breach of this Agreement effective as of the completion of the obligations contemplated by this Section 16.2.2.
 - 16.2.3 <u>Transfer of [***]</u>. Upon completion of the obligations set forth in Section 16.2.2, [***].
- 16.2.4 <u>Commercial Option Exercise</u>. Within [***], Baxalta may (but shall not be required to) provide notice to Precision that it intends to exercise the Commercial Option, and, after provision of such notice [***]. In the event such notice is provided, upon the first date on which [***] and any required Antitrust Clearance has been obtained, (a) the Commercial Option for [***] shall be deemed exercised [***], and (b) unless Baxalta has previously paid to Precision the Commercial License Fee with respect to a Licensed Product Candidate or [***] with respect to [***].
 - 16.2.5 Role of JSC. The JSC's role with respect to [***].
 - 16.2.6 Regulatory Matters. Precision's Regulatory Role with respect to [***].
 - 16.2.7 Manufacturing and Supply. Precision's Manufacturing and Supply Role with respect to [***].
 - 16.2.8 Adjustments to [***]. With respect to [***].
 - 16.2.9 [***]
- 16.3 Effect of Final Determination of [***]. If a Final Determination is made that (a) Precision has [***], (b) such breach has resulted in [***], and (c) Precision has failed to take all action necessary and within Precision's reasonable control, [***], or it is otherwise not possible through the use of Commercially Reasonable Efforts to [***] within [***] after receipt of Baxalta's written notice of such breach (or, in the case of a good faith dispute as to whether a breach occurred, within [***] after a Final Determination that such breach occurred), then upon Baxalta's written request, [***], the following provisions shall apply with respect to [***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 16.3.1 General. Except as modified pursuant to this Section 16.3, all provisions of this Agreement applicable to Included Targets shall apply equally to [***], and all provisions of this Agreement applicable to Licensed Product Candidates or Licensed Products, respectively, shall apply equally to [***], as applicable
- 16.3.2 <u>Delivery of [***]</u>. Precision shall deliver to Baxalta [***]. Precision shall be deemed to no longer be in continuing breach of this Agreement effective as of the completion of the obligations contemplated by this Section 16.3.2.
 - 16.3.3 <u>Transfer of [***</u>]. Upon completion of the obligations set forth in Section 16.3.2, [***].
- 16.3.4 <u>Commercial Option Exercise</u>. Within [***], Baxalta may (but shall not be required to) provide notice to Precision that it intends to exercise the Commercial Option, and, after provision of such notice [***]. In the event such notice is provided, upon the first date on which [***] and any required Antitrust Clearance has been obtained, (a) the Commercial Option for [***] shall be deemed exercised [***], and (b) unless Baxalta has previously paid to Precision the Commercial License Fee with respect to [***], Baxalta shall pay to Precision the Commercial License Fee with respect to [***].
 - 16.3.5 Role of JSC. The JSC's role with respect to [***].

 16.3.6 Regulatory Matters. Precision's Regulatory Role with respect to [***].

 16.3.7 Manufacturing and Supply. Precision's Manufacturing and Supply Role with respect to [***].

 16.3.8 Adjustments to [***]. [***].

16.4 Other Remedies. Nothing in this Article XVI shall limit any other rights or remedies that a Party would otherwise have at law or in equity.

ARTICLE XVII. GENERAL PROVISIONS

17.1 Assignment. This Agreement is binding upon and will inure to the benefit of the Parties and their respective permitted assignees or successors in interest, including those that may succeed by assignment, transfer or otherwise to the ownership of the assets necessary to the conduct of the business to which this Agreement relates. This Agreement may not be assigned or otherwise transferred by either Party without the prior written consent of the other Party; provided, however, that either Party may, without such consent, assign or otherwise transfer this Agreement, together with all of its rights and obligations hereunder, to any of its Affiliates, or to a successor in interest in connection with the transfer or sale of all or substantially all of its business to which this Agreement relates (including by transfer or sale of all or any portion of such Party's assets, equity or business), or in the event of its merger or consolidation or similar business combination transaction. Notwithstanding the foregoing, Precision shall not be required to obtain Baxalta's consent to sell, assign, pledge as security, contribute, or otherwise transfer, in whole or in part, to any Third Party its rights to receive any payment under this Agreement, and, as it relates to any such transfer, Precision may provide to such Third Party (i) a copy of reports received from Baxalta pursuant to Section 8.4.6 and (ii) the result of any audit conducted in pursuant to Section 8.4.7, in each case of (i) and (ii), provided that any such Third Party is bound by confidentiality and non-use obligations equivalent in scope to those set forth in ARTICLE XII of this Agreement. Any purported assignment in violation of the preceding sentences in this Section shall be void. Any permitted assignee or successor shall assume and be bound by all obligations of its assignor or predecessor under this Agreement. Notwithstanding anything to the contrary in this Agreement, the Patent Rights and Know-How Controlled by an entity that is a permitted assignee, transferee or successor of a Party under this Agreement, or an entity who becomes an Affiliate of a Party during the Term (or any Affiliate of any such entity), whether such Patent Rights and Know-How were developed prior to or after the transaction that was the basis for such assignment, transfer or succession or resulted in such entity becoming an Affiliate, shall be automatically excluded from

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

the rights licensed to the other Party under this Agreement; provided, however, that following such a transaction involving Precision, if at Precision's sole option (a) Precision uses or commences to use any of such excluded Know-How, or (b) Precision comes to Control and uses or commences to use any such excluded Patent Right, in each case for purposes of performing Precision's Development and manufacturing responsibilities with respect to a Licensed Product Candidate or Licensed Product, such intellectual property will be included within the Patent Rights or Know-How licensed to Baxalta pursuant to this Agreement with respect to such Licensed Product Candidate or Licensed Product, as Precision Patents, Precision Know-How, Precision Platform Patents or Precision Platform Technology, as applicable.

- 17.2 <u>Allocation of Costs</u>. Without limiting Baxalta's payment obligations under <u>ARTICLE VIII</u> of this Agreement, each of Baxalta and Precision shall be solely responsible for all costs and expenses it incurs in connection with their activities under this Agreement except as otherwise expressly set forth in this Agreement.
- 17.3 Headings; Rules of Construction. Headings are inserted for convenience and shall not affect the meaning or interpretation of this Agreement. Each Party agrees that this Agreement shall be interpreted without regard to any presumption or rule requiring construction against the Party causing this Agreement to be drafted. Except as otherwise explicitly specified to the contrary in this Agreement, (a) the words "hereof," "herein," "hereby," "hereunder" and words of similar import shall refer to this Agreement as a whole and not to any particular section or subsection of this Agreement and reference to a particular section of this Agreement shall include all subsections thereof, (b) references to a section, exhibit or schedule means a section of, or exhibit or schedule to, this Agreement, (c) definitions shall be equally applicable to both the singular and plural forms of the terms defined, and references to the masculine, feminine or neuter gender shall include each other gender, (d) the words "include," "includes" and "including" shall be deemed to be followed by the words "without limitation," (e) references to a rule, statute or regulation (including CPR rules and procedures) include all rules and regulations thereunder and any successor statute, rule or regulation, in each case as amended or otherwise modified from time to time, (f) references to a particular Governmental Authority include any successor agency or body to such Governmental Authority and (g) references to "days" means calendar days, unless specified as Business Days.
- 17.4 <u>No Implied Waiver</u>. No waiver of any default hereunder by either Party or any failure to enforce, or delay in enforcing, any rights hereunder shall be deemed to constitute a waiver of any subsequent default with respect to the same or any other provision hereof.
- 17.5 <u>Notices</u>. Any notice or other communication given by one Party to the other Party under this Agreement must be in writing and shall be sufficient if (a) delivered personally or (b) sent by registered or certified mail, return receipt requested, reputable overnight business courier, email or fax, in each case properly addressed to the receiving Party as set forth below. The effective date of any notice or other communication given hereunder shall be the actual date of receipt by the receiving Party.

If to Precision:

Precision BioSciences, Inc. 302 East Pettigrew Street, Suite A-100

Durham, NC 27701 Facsimile: (480) 393-5553

Attention: Michael Dombeck, Vice President, Business Development

with a copy (which copy shall not constitute legal notice to Precision) to:

Smith, Anderson, Blount, Dorsett, Mitchell & Jernigan, LLP 150 Fayetteville Street, Suite 2300

Raleigh, NC 27601 Facsimile: (919) 821-6800 Attention: John R. Therien, Esq.

If to Baxalta:

Baxalta Incorporated 1200 Lakeside Drive Bannockburn, IL 60015 Facsimile: (224) 940-8294

Email: legal.operations@baxalta.com

Attention: Legal Department

and

Baxalta GmbH Thurgauerstrasse 130 8152 Glattpark (Opfikon) Switzerland

Attention: Legal Department

with a copy (which copy shall not constitute legal notice to Baxalta) to counsel:

Baxalta Incorporated 1200 Lakeside Drive Bannockburn, IL 60015

Email: general.counsel@baxalta.com

Attention: General Counsel

Any Party may change its notification address by giving notice to the other Party in the manner herein provided.

17.6 Severability. Whenever possible, each term and provision of this Agreement shall be interpreted in such manner as to be valid and effective under applicable Laws, but, if any term or provision of this Agreement is held to be invalid or unenforceable under applicable Laws, such term or provision shall be invalid and ineffective only to the extent of such invalidity or unenforceability, without invalidating or making unenforceable the remainder of this Agreement. In the event of such invalidity or unenforceability, the Parties shall use reasonable efforts to seek and agree on an alternative valid and enforceable provision that preserves the original purpose and intent of the Agreement.

- 17.7 Entire Agreement. This Agreement constitutes the entire agreement between the Parties and shall cancel and supersede any and all prior and contemporaneous negotiations, correspondence, understandings and agreements, whether oral or written, between the Parties respecting the subject matter hereof, including the Confidentiality Agreement and that certain non-binding term sheet exchanged by the Parties prior to the Effective Date.
- 17.8 <u>Amendment; Waiver</u>. Any amendment or modification to this Agreement shall only be made in writing and shall only be valid when signed by an authorized representative of each Party. No term or provision of this Agreement, including the Parties' respective obligations, may be waived except by a writing signed by the Party against which such waiver is sought to be enforced.
- 17.9 <u>Counterparts</u>. This Agreement may be executed in more than one counterpart (including by electronic transmission), each of which shall be deemed an original, but all of such counterparts taken together shall constitute one and the same agreement.
- 17.10 <u>Agency</u>. Neither Party is, nor shall be deemed to be, an employee, agent, co-venturer, or legal representative of the other Party for any purpose. Neither Party shall be entitled to enter into any contracts in the name of, or on behalf of the other Party, nor shall either Party be entitled to pledge the credit of the other Party in any way or hold itself out as having the authority to do so.
- 17.11 <u>Further Actions</u>. Each Party agrees to execute, acknowledge, and deliver such further instruments, and to do all such other acts, as may be necessary or appropriate in order to carry out the purpose and intent of this Agreement.
- 17.12 <u>Compliance with Laws</u>. Each Party will comply with all applicable Laws in performing its obligations and exercising its rights hereunder, including all applicable Laws relating to the export, re-export or other transfer of any Know-How transferred pursuant to this Agreement.
- 17.13 Governing Law; Jurisdiction. Any dispute, claim or controversy arising under or related to this Agreement, including the construction, validity and performance of this Agreement, shall be governed in all respects by the substantive laws of the state of New York, excluding its provisions regarding conflicts of law; provided, however, that any issue relating to the interpretation, construction, validity, enforceability or infringement of Patent Rights shall be determined according to the patent laws of the country (or countries) in which the relevant Patent (or Patents) issued. Subject to Section 15.2, this Agreement will be subject to the exclusive jurisdiction of the United States District Court for the Southern District of New York or, if such court is not of competent jurisdiction, a court of competent jurisdiction located in New York, New York. Each Party hereby consents to the personal jurisdiction of such courts for the purposes of any action brought consistent with this Agreement.

17.14 Rights in Bankruptcy.

- 17.14.1 Effect on Licenses. All rights and licenses granted under or pursuant to this Agreement by Precision or Baxalta are, and shall otherwise be deemed to be, for purposes of Section 365(n) of the U.S. Bankruptcy Code and of any similar provisions of applicable Laws under any other jurisdiction (collectively, the "Bankruptcy Laws"), licenses of rights to "intellectual property" as defined under the Bankruptcy Laws. Each Party agrees that the other Party, as licensee of such rights under this Agreement shall retain and may fully exercise all of its rights and elections under the Bankruptcy Laws. Each Party agrees during the Term, to create and maintain current copies or, if not amenable to copying, detailed descriptions or other appropriate embodiments, to the extent feasible, of all such intellectual property. If a case is commenced by or against the Debtor Party under the Bankruptcy Laws (excluding a reorganization proceeding under Chapter 11 of the U.S. Bankruptcy Code if the Debtor Party is continuing to perform all of its obligations under this Agreement), the Debtor Party (in any capacity, including debtor-in-possession) and its successors and assigns (including a trustee under the Bankruptcy Laws) shall, as the Non-Debtor Party may elect in a written request, immediately upon such request:
- (a) perform all of the obligations provided in this Agreement to be performed by the Debtor Party including, where applicable, providing to the Non-Debtor Party portions of such intellectual property (including embodiments thereof) held by the Debtor Party and such successors and assigns or otherwise available to them and to which the Non-Debtor Party is entitled to have access under this Agreement; or
- (b) provide to the Non-Debtor Party all such intellectual property (including embodiments thereof) held by the Debtor Party and such successors and assigns or otherwise available to them and to which the Non-Debtor Party is entitled to have access under this Agreement; and
- (c) not interfere with the rights of the Non-Debtor Party under this Agreement, or any agreement supplemental hereto, to such intellectual property (including such embodiments), including any right to obtain such intellectual property (or such embodiments) from another entity, to the extent provided in the Bankruptcy Laws.
- 17.14.2 Rights to Intellectual Property. If (a) a case under the Bankruptcy Laws is commenced against the Debtor Party, (b) this Agreement is rejected by the trustee pursuant to the Bankruptcy Laws, and (c) the Non-Debtor Party elects to retain its rights hereunder as provided in the Bankruptcy Laws, then the Debtor Party (in any capacity, including debtor-in-possession) and its successors and assigns (including a trustee under the Bankruptcy Laws) shall provide to the Non-Debtor Party all such intellectual property (including embodiments thereof) held by the Debtor Party and such successors and assigns, or otherwise available to them, and to which the Non-Debtor Party is entitled to have access under this Agreement, immediately upon the Non-Debtor Party's written request. Whenever the Debtor Party or any of its successors or assigns provides to the Non-Debtor Party any of the intellectual property licensed hereunder (or any embodiment thereof) pursuant to this Section 17.14.2, the Non-Debtor Party shall have the right to perform the obligations of the Debtor Party hereunder with respect to such intellectual property, but neither such provision nor such performance by the Non-Debtor Party shall release the Debtor Party from any such obligation or liability for failing to perform it.

- 17.14.3 Additional Rights. All rights, powers and remedies of the Non-Debtor Party provided herein are in addition to and not in substitution for any and all other rights, powers and remedies now or hereafter existing at law or in equity (including under the Bankruptcy Laws) in the event of the commencement of a case under the Bankruptcy Laws by or against the Debtor Party. The Non-Debtor Party, in addition to the rights, power and remedies expressly provided herein, shall be entitled to exercise all other such rights and powers and resort to all other such remedies as may now or hereafter exist at law or in equity (including under the Bankruptcy Laws) in such event. The Parties agree that they intend the foregoing rights to extend to the maximum extent permitted by applicable Laws, including for purposes of the Bankruptcy Laws:
- (a) The right of access to any intellectual property (including embodiments thereof) of the Debtor Party to which the Non-Debtor Party is entitled to have access under this Agreement, or any Third Party with whom the Debtor Party contracts to perform an obligation of the Debtor Party under this Agreement, and, in the case of the Third Party, which is necessary for the Development, manufacture or Commercialization of Licensed Products; and
 - (b) The right to contract directly with any Third Party described in Section 17.14.3(a) to complete the contracted work.

17.15 Force Majeure.

17.15.1 No failure or delay by either Party in the performance of any obligation hereunder (other than any obligation to make a payment to the other Party) shall be deemed a breach of this Agreement nor create any liability for any damages, increased cost or losses which the other Party may sustain by reason of such failure or delay of performance, if the same shall arise from any cause or causes beyond the control of that Party, such as earthquake, storm, flood, fire, other acts of nature, epidemic, war, riot, hostility, public disturbance, cessation of transport, act of public enemies, prohibition or act by a Governmental Authority or public agency, strike or other labor dispute or work stoppage (collectively "Force Majeure"); provided, however, that the failing or delaying Party shall (a) without undue delay, notify the other Party in writing of the applicable failure or delay and (b) continue to take all commercially reasonable actions within its power to comply with its obligations hereunder as fully as possible and to mitigate possible damages.

17.15.2 Should an event of Force Majeure continue for more than [***], the Parties shall promptly discuss their further performance under this Agreement and whether to modify or terminate this Agreement in view of the effect of the event of Force Majeure. Any such modification or termination of this Agreement shall be effective only upon mutual written agreement of the Parties.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

17.16 Consideration. The Parties acknowledge that each of the licenses and rights granted by Precision in this Agreement and each of the provisions of this Agreement for efforts or assistance by Precision and access to Precision Patents, Precision Know-How and Precision Platform Patents, individually and collectively, constitute good, valuable and sufficient consideration for each and all of the fees and payments called for hereunder and for each and all of the other obligations of Baxalta, its Affiliates and Sublicensees; and the Parties further acknowledge that the individual and collective rights under and access to Precision Patents, Precision Know-How and Precision Platform Patents renders the way in which those fees and payments hereunder are determined, and their duration, appropriate and desirable as a matter of convenience.

17.17 Antitrust.

17.17.1 To the extent required by the HSR Act, each Party will (a) file or cause to be filed, as promptly as practicable after Baxalta's provision of a Commercial Option Exercise Notice (but not later than [***], except in the case of provision of the Commercial Option Exercise Notice in accordance with Section 16.2.4 or Section 16.3.4 which shall be not later than [***]), with the United States Federal Trade Commission ("FTC") and the United States Department of Justice ("DOJ"), all reports and other documents required to be filed by such Party under the HSR Act concerning the exercise of the applicable Commercial Option and (b) promptly comply with or cause to be complied with any requests by the FTC or DOJ for additional information concerning the exercise of the applicable Commercial Option, in each case so that the waiting period applicable to the applicable Commercial Option under the HSR Act will expire as soon as practicable after the date hereof. Each Party agrees to request, and to cooperate with the other Party in requesting, early termination of any applicable waiting period under the HSR Act. The Parties acknowledge that compliance with the timelines set forth in this paragraph will require the Parties to begin to prepare such filings sufficiently in advance of provision of the Commercial Option Exercise Notice, and if it is reasonably expected that the Commercial Option Exercise Notice will be provided then each Party agrees to commence any required actions as soon as possible in advance of provision of the Commercial Option Exercise Notice.

17.17.2 From the date of each Commercial Option Exercise Notice through receipt of Antitrust Clearance (but for no longer than [***] after the date of the Commercial Option Exercise Notice unless otherwise agreed between the Parties), Baxalta and Precision agree (and shall cause each of their respective Affiliates) to use their respective commercially reasonable efforts, (a) to cooperate to obtain any authorizations, clearances, orders or approvals required for the exercise of the Commercial Option under the HSR Act and any other federal, state or foreign Law, regulation or decree designed to prohibit, restrict or regulate actions intended to or having the effect of reducing competition or monopolizing or restraining trade (collectively, "Antitrust Laws"); (b) to promptly respond to any request by any Governmental Authority for information under any Antitrust Law; (c) to contest and resist any action, including any legislative, administrative or judicial action, and to have vacated, lifted, reversed or overturned any decree,

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

FINAL EXECUTION VERSION CONFIDENTIAL

judgment, injunction or other order (whether temporary, preliminary or permanent) that restricts, prevents or prohibits the exercise of any Commercial Option under any Antitrust Law; (d) to promptly inform the other Party upon receipt of any material communication from the FTC, the DOJ or any other Governmental Authority regarding the exercise of any Commercial Option; (e) subject to applicable legal limitations and the instructions of any Governmental Authority, keep the other Party apprised of the status of matters relating to the exercise of any Commercial Option, including promptly furnishing the other Party with copies of material notices or other material communications received by such Party or any of their respective Affiliates, as the case may be, from any Third Party and/or any Governmental Authority with respect to the exercise of any Commercial Option, and (f) refrain from taking any action the purpose or effect of which could reasonably be expected delay, impair or impede the termination of any waiting period or the receipt of any required consents, permits, authorizations or approvals of any Governmental Authority. The Parties shall each permit legal counsel for the other Party a reasonable opportunity to review in advance, and consider in good faith the views of the other Party in connection with, any proposed material written communication to any Governmental Authority. Each of the Parties agrees, subject to applicable legal limitations and the instructions of any Governmental Authority, not to participate in any substantive meeting or discussion, either in person or by telephone, with any Governmental Authority in connection with the exercise of any Commercial Option unless it consults with the other Party in advance and gives the other Party the opportunity to attend and participate. Notwithstanding the foregoing, Baxalta shall, on behalf of the Parties, control and lead all communications and strategy relating to Antitrust Laws and litigation matters relating to the Antitrust Laws (provided that Precision is not prohibited from complying with applicable Law), subject to good faith consultations with Precision and the inclusion of Precision at meetings with Governmental Authorities with respect to any discussion related to the exercise of any Commercial Options under the Antitrust Laws.

17.17.3 Baxalta shall be responsible for all costs, expenses (other than Precision's legal expenses), and filing fees in connection with this Section 17.17.

17.17.4 Notwithstanding the foregoing, nothing in this <u>Section 17.17</u> or otherwise in this Agreement shall require Baxalta to propose, negotiate, effect or agree to, the sale, divestiture, license or other disposition of any assets or businesses or otherwise take any action that limits the freedom of action with respect to, or its ability to retain any of the businesses, product lines or assets.

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

PRECISION BIOSCIENCES, INC.

By: /s/ Matthew Kane

Name: Matthew Kane

Title: CEO

[signatures continue on following page]

Signature Page to Development and Commercial License Agreement

BAXALTA INCORPORATED

By: /s/ Robert J. Hombach

Name: Robert J. Hombach

Title: Executive Vice President, Chief

Financial Officer and Chief

Operations Officer

BAXALTA US INC.

By: /s/ Robert J. Hombach

Name: Robert J. Hombach
Title: Executive Vice President, Chief

Financial Officer and Chief

Operations Officer

[signatures continue on following page]

Signature Page to Development and Commercial License Agreement

BAXALTA GMBH

By: /s/ René Büchel
Name: Dr. René Büchel

Title: Dir. Plasma Procurement

By: /s/ Yvo Aebli

Name: Yvo Aebli

Title: Controller Switzerland & ECG

Signature Page to Development and Commercial License Agreement

[***

EXHIBIT A DUKE IP

[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
	[***]		[***]
	[***]		[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
	[***]		[***]
[***]	[***]	[***]	[***]
[***]	[***]		[***]
	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]		[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]		[***]
	[***]		[***]
	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***]

[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]

[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

EXHIBIT B

SUPPLY AGREEMENT TERM SHEET

OVERVIEW

PARTIES Baxalta and Precision

SUBJECT MATTER This "Supply Agreement Term Sheet" sets forth the key terms that will be included in the supply agreement that the Parties will

negotiate as contemplated by Section 6.2 of the Agreement (the "<u>Supply Agreement</u>"). Pursuant to the Supply Agreement, Precision or its CMO will manufacture and Precision will supply clinical trial materials for use in Phase II Clinical Trials, including Phase II Ready Batches, to Baxalta. Baxalta will retain the right to engage or appoint additional suppliers and/or CMOs from time to time in its sole discretion to produce Product for Phase II Clinical Trials or for any further Development

and Commercialization.

TERRITORY As defined in the Agreement.

DEFINITIONS All capitalized terms used but not otherwise defined in this Supply Agreement Term Sheet shall have the meaning set forth in

the Development and Commercial License Agreement between Precision and Baxalta of which this Exhibit is a part (referred to

as the "Agreement" for purposes of this Supply Agreement Term Sheet).

MANUFACTURING OBLIGATIONS

MANUFACTURING OF PRODUCT

Precision shall supply to Baxalta formulated Licensed Product Candidates or Licensed <u>Product</u> (as applicable, the "Product") for use by Baxalta in Phase II Clinical Trials.

Prior to manufacture by Precision of any Product for use in Phase I or Phase II Clinical Trials, the Parties will establish (and the JSC shall approve) a written specification for each Product, setting forth Product attributes and the corresponding test methods and quality systems adequate to ensure quality and consistency of the Product (the "Specification").

Precision shall manufacture all Product for Clinical Trial use in accordance with cGMP, the Specification and the Quality Agreement. Precision may not make any material changes to the manufacturing process for Product without JSC approval. Precision shall label Product in accordance with the JSC's instructions in a manner appropriate for clinical use. Precision shall ensure that all Product supplied to Baxalta has the minimum shelf life specified in the Specification for the applicable Product.

All manufacturers and suppliers of materials to manufacture Product shall be Baxalta certified and approved prior to the initiation of manufacture.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

FORECASTS AND ORDERS

The quantity of the Phase II Ready Batch shall be established for each Product by the JSC in accordance with the Agreement. Baxalta shall deliver a binding order for the Phase II Ready Batch not less than [***] before its requested delivery date.

If Baxalta determines that it requires additional quantities of Product for use in Phase II Clinical Trials after supply of the Phase II Ready Batch, Baxalta shall provide Precision with a non-binding forecast of its requirements as soon as practicable after making such determination and shall adjust its forecast from time to time as its expected requirements of such additional Product change. Baxalta shall issue a binding order of its requirements of Product not less than [***] before its requested delivery date. The Supply Agreement will acknowledge that the process for manufacture of the Products has variable yields and will provide appropriate accommodation parameters regarding ordered and delivered quantities of Products.

CERTIFICATE OF ANALYSIS

In conjunction with each delivery of Product, Precision shall provide a signed Certificate of Analysis certifying that the Product in such shipment meets all requirements of the final Specification.

DELIVERY

EXW (ICC Incoterms 2010) Precision's manufacturing facility.

FINANCIAL TERMS

SUPPLY PRICE AND PAYMENT

Baxalta shall purchase Product (other than Phase II Ready Batches) from Precision at a supply price equal to [***].

Precision may invoice Baxalta upon delivery, and Baxalta will pay all correctly issued invoices within [***] days after receipt.

Precision will not charge [***] Baxalta for the Phase II Ready Batches.

MISCELLANEOUS

TERM AND TERMINATION

The term of the Supply Agreement will commence upon execution of the Supply Agreement and shall continue until six (6) months after completion of manufacture of the Phase II Ready Batch of the last Licensed Product Candidate Baxalta may pursue under the Agreement. The Parties can mutually agree to extend the Supply Agreement. The Supply Agreement will be in place prior to Precision's manufacture of any Licensed Product Candidate for use in Clinical Trials.

The Supply Agreement will automatically terminate if the Agreement terminates. In addition, either Party may terminate the Supply Agreement in its entirety immediately upon written notice to the other Party if the other Party materially breaches its obligations under the Supply Agreement (which termination shall be substantially similar to Section 14.2.1 of the Agreement)

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***

REPRESENTATIONS, WARRANTIES / INDEMNIFICATION / LIMITATIONS OF LIABILITY The Supply Agreement shall contain customary representations and warranties relating to the manufacture and supply of products in the pharmaceutical industry, including representations and warranties that (i) Products are manufactured in compliance with the Specification, cGMP and the Quality Agreement, (ii) Products are not adulterated or misbranded within the meaning of the Federal Food, Drug and Cosmetic Act (21 USC §321, as amended), (iii) Precision will comply with all applicable Laws in connection with its manufacturing activities, and (iv) that Precision will not use any personnel that to its knowledge have been debarred or disqualified or committed any act subject to debarment or disqualification.

In addition, the Supply Agreement shall include appropriate indemnification provisions (including for breach of warranty) and limitations of liability, in each case as are reasonable and customary for agreements for supply of clinical study material.

CONFIDENTIALITY Article XII of the Agreement shall govern the Parties' obligations with respect to confidentiality.

ASSIGNMENT AND CHANGE OF CONTROL

These matters will be subject to the same terms as agreed by the Parties with respect to such matters in the Agreement.

GOVERNING LAW New York

DISPUTE RESOLUTION Article XV of the Agreement shall govern resolution of disputes arising under the Supply Agreement.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

EXHIBIT C

QUALITY AGREEMENT TERM SHEET

OVERVIEW

PARTIES Baxalta and Precision

SUBJECT MATTER This "Quality Agreement Term Sheet" sets forth the key terms that will be included in the quality agreement that the Parties

will negotiate as contemplated by Section 6.3 of the Agreement (the "Quality Agreement").

TERRITORY As defined in the Agreement.

DEFINITIONSAll capitalized terms used but not otherwise defined in this Quality Agreement Term Sheet shall have the meaning set forth

in the Development and Commercial License Agreement between Precision and Baxalta of which this Exhibit is a part

(referred to as the "Agreement" for purposes of this Quality Agreement Term Sheet).

QUALITY AGREEMENT SCOPE

GENERAL SCOPE AND RESPONSIBILITIES

The Quality Agreement will define the responsibilities and interactions of Baxalta and Precision for the clinical production of Product by Precision pursuant to the Supply Agreement, specifically relating to quality control, quality assurance and validation of clinical supply of Product to ensure that Product complies with the Specification (as defined in the Supply Agreement Term Sheet), licenses, and marketing authorizations.

The Quality Agreement is intended to comply with the guidance and directives set forth in current Good Manufacturing Practice (cGMP) requirements for the United States and the European Union. The Parties will be responsible for ensuring compliance to cGMPs for the roles outlined in the Quality Agreement.

TERM The Quality Agreement will be coterminous with the Supply Agreement, except for those provisions that may survive

termination or expiration, which include, without limitation, the provisions regarding complaints, recalls, regulatory inquires

and inspections, completion of defined testing, product-specific document storage, and reserve samples.

JOINT QUALITY TEAM

Baxalta and Precision will implement a joint quality team (the "JQT") to facilitate communication and consensus on issues related to Product quality. Meetings will be held at a frequency that is adequate to assist the JQT responsible for final functional quality decision-making authority with respect to the clinical trial materials. Minimally, the following metrics will be reviewed each meeting:

- Significant changes (change controls)
- Out of Specifications (OOS)

Additional subjects include but are not limited to review of major nonconformities to the Specification, cGMP or the Quality Agreement, and logistics related to Product Candidate testing and release.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ADDITIONAL TERMS

The Quality Agreement will contain provisions relating to the following, which are reasonable and customary consistent with Baxalta's practices:

- Manufacturing cGMP compliance
- Primary packaging
- · Qualification/validation/calibration
- · Quality control
- Quality assurance
- · Change management
- · Third Party subcontracting
- · Vendor management
- · Regulatory responsibilities
- Traceability
- · Regulatory compliance
- Analytical method transfer
- · Technology transfer

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

EXHIBIT D

FORM CDCP AGREEMENT TERM SHEET

OVERVIEW

PARTIES Baxalta and Precision.

SUBJECT MATTER This "CDCPA Term Sheet" sets forth the key terms that will be included in the form CDCP Agreement that the Parties will

negotiate as contemplated by Section 7.3.3 of the Agreement.

TERRITORY United States, including its territories and possessions

DEFINITIONS All capitalized terms used but not otherwise defined in this CDCPA Term Sheet shall have the meaning set forth in the

Development and Commercial License Agreement between Precision and Baxalta of which this Exhibit is a part (referred to as

the "Agreement" for purposes of this CDCPA Term Sheet).

"Regional Product" means [***].

GOVERNANCE AND ALLOCATION OF RESPONSIBILITY FOR CDCP PRODUCT

GOVERNANCE

The Parties will establish a joint co-development and co-promotion committee (for purposes of this CDCPA Term Sheet, the "<u>Committee</u>") which shall oversee the clinical Development, manufacturing, and Commercialization of each Licensed Product for which Precision has exercised the CDCP Option (each, a "<u>CDCP Product</u>") for the Territory. The Committee shall be put in place immediately upon execution of the CDCP Agreement. The Committee will operate in substantially the same manner as the JSC under Section 3.1 of the Agreement, except that the matters over which the Committee will have oversight shall be limited to issues relating to Development, manufacturing, and Commercialization of the CDCP Product(s).

Any disagreement or inability to reach unanimous consensus on the Committee shall be subject to resolution through procedures substantially the same as those set forth in Section 3.1.6 of the Agreement; provided, however, that if after the process set forth in Section 3.1.6(b) of the Agreement the Executive Officers are unable to resolve a particular issue, Section 3.1.6(c) and Article XV of the Agreement shall not apply and instead Baxalta shall have final decision-making authority on all decisions required to be made by the Committee. Baxalta will covenant not to use bad faith in exercising its final decision making authority.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***

DEVELOPMENT

The Parties will equally share all costs and expenses in connection with all aspects of further Development of the CDCP Product for the Territory after the effective date of the CDCP Agreement.

Promptly after execution of the CDCP Agreement, the Committee will set the required form and contents of a comprehensive Development plan for the CDCP Product for the Territory, which plan will allocate responsibility to the Parties for any further Development of the CDCP Product (the "Joint Development Plan"). The Joint Development Plan shall be appropriately coordinated with Baxalta's global plan for Development, manufacturing and Commercialization for the CDCP Product, with Baxalta being the sponsor of any Clinical Trials, including those that are conducted on a global basis ("Global Studies"). The Joint Development Plan will contain an estimate of the budget for the period covered by the Joint Development Plan. For clarity, the Joint Development Plan shall not require any activity directed to Development of the CDCP Product for Commercialization outside the Territory.

Baxalta will prepare an initial Joint Development Plan in good faith in consultation with Precision and subject to the Committee's review and approval. Baxalta will review and, as necessary, update the Joint Development Plan at least annually, in each case in good faith in consultation with Precision and subject to the Committee's review and approval. Each Party may request at any time that the Committee approve other updates to the Joint Development Plan. All Development activities for the CDCP Product related to the Territory shall be performed in accordance with the Joint Development Plan. To the extent Third Parties are engaged under contract by either Party to perform Development activities consistent with the Joint Development Plan, such contracts shall provide for appropriate indemnification rights that are equally protective of each Party, and to the extent Precision engages under contract any such Third Party, all such contracts will be subject to Baxalta's approval.

Promptly after executing the CDCP Agreement and concurrently with preparing the Joint Development Plan, Baxalta will prepare the initial budget for the first year of the activities set forth in the Joint Development Plan (the "Initial Joint Development Budget") in good faith in consultation with Precision and subject to the Committee's review and approval. Thereafter, Baxalta will prepare a budget for the activities under the Joint Development Plan on an annual basis prior to the beginning of the next year (each, including the Initial Joint Development Budget, an "Annual Joint Development Budget"), in each case in good faith in consultation with Precision and subject to the Committee's review and approval. Together with each Annual Joint Development Budget, for

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

***1

informational purposes only, so as to facilitate Precision's planning, Baxalta will provide a non-binding estimate of the budget for activities remaining under the Joint Development Plan in subsequent years. In the event that Baxalta proposes an update to the Joint Development Plan that would have the effect of deviating from the then-current Annual Joint Development Budget in an amount greater than [***]. For a CDCP Product that is not being Developed as a Regional Product, in the event that any Global Studies are conducted, [***] of the aggregate cost of the Global Studies will be reflected in each Annual Joint Development Budget.

In the event that either Party reasonably expects any deviation from an Annual Joint Development Budget in excess of the percent overage threshold set forth above is necessary to complete its Development activities, the Committee shall meet and consider any necessary adjustments to the Annual Joint Development Budget. Further, the Parties will review and reconcile on a Calendar Quarterly basis costs and expenses of each of the Parties in connection with the Joint Development Plan.

Upon execution of the CDCP Agreement, Precision must either: (a) have cash or cash equivalents on hand, or (b) be able to reasonably demonstrate (i) anticipated cash flows, or (ii) access to cash via credit facility or otherwise, in each case sufficient to cover its share of Development costs (in accordance with the reasonably anticipated Initial Joint Development Budget) for at least [***].

COMMERCIALIZATION

The Parties will equally share all costs and expenses in connection with all aspects of Commercialization of the CDCP Product in the Territory. At such time as the Committee deems appropriate (with sufficient time to complete the Joint Commercialization Plan as set forth below), the Committee will set the required form and contents of a comprehensive Commercialization plan for the CDCP Product in the Territory, which plan will allocate responsibility to the Parties for Commercialization of the CDCP Product (the "Joint Commercialization Plan"). The Joint Commercialization Plan will contain an estimate of the budget for the period covered by the Joint Commercialization Plan. Without limiting the foregoing, Precision's required Commercialization activities will be limited to establishing and maintaining a sales force, but Precision may participate in other Commercialization activities subject to Baxalta approval. The Joint Commercialization Plan will be finalized at least [***] prior to launch of Commercialization of the CDCP Product. The Joint Commercialization Plan shall be appropriately coordinated with Development, manufacturing and Commercialization activities undertaken with respect to the CDCP Product outside the Territory, and shall not require any activity directed to Commercialization of the CDCP Product with respect to countries outside the Territory.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***]

Baxalta will prepare an initial Joint Commercialization Plan in good faith in consultation with Precision and subject to the Committee's review and approval. Baxalta will review and, as necessary, update the Joint Commercialization Plan at least annually, in each case in good faith in consultation with Precision and subject to the Committee's review and approval. Each Party may request at any time that the Committee approve other updates to the Joint Commercialization Plan. All Commercialization activities for the CDCP Product in the Territory shall be performed in accordance with the Joint Commercialization Plan. To the extent Third Parties are engaged under contract by either Party to perform Commercialization activities consistent with the Joint Commercialization Plan, such contracts shall provide for appropriate indemnification rights that are equally protective of each Party, and to the extent Precision engages under contract any such Third Party, all such contracts will be subject to Baxalta's approval.

Concurrently with preparing the Joint Commercialization Plan, Baxalta will prepare the initial budget for the first year of the activities set forth in the Joint Commercialization Plan (the "Initial Joint Commercialization Budget") in good faith in consultation with Precision and subject to the Committee's review and approval. Thereafter, Baxalta will prepare a budget for the activities under the Joint Commercialization Plan on a annual basis prior to the beginning of the next year (each, including the Initial Joint Commercialization Budget, an "Annual Joint Commercialization Budget"), in each case in good faith in consultation with Precision and subject to the Committee's review and approval. Together with each Annual Joint Commercialization Budget, for informational purposes only, so as to facilitate Precision's planning, Baxalta will provide a non-binding estimate of the budget for activities remaining under the Joint Commercialization Plan in subsequent years. In the event that Baxalta proposes an update to the Joint Commercialization Plan that would have the effect of deviating from the then-current Annual Joint Commercialization Budget in an amount greater than [***].

In the event that either Party reasonably expects any deviation from any Annual Joint Commercialization Budget in excess of the percent overage threshold set forth above is necessary to complete its Commercialization activities, the Committee shall meet and consider any necessary adjustments to the Annual Joint Commercialization Budget. Further, the Parties will review and reconcile on a Calendar Quarterly basis costs and expenses of each of the Parties in connection with the Joint Commercialization Plan.

All personnel (including sales and marketing personnel) performing Commercialization activities will conform to Baxalta's qualifications and requirements, as will be set forth in the form CDCP Agreement.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

SALES FORCE AND PROMOTIONAL MATERIALS

Baxalta will direct the training of both Parties' sales representatives contemplated by the Joint Commercialization Plan and will prepare and implement, in consultation with Precision, a training program and provide training materials for such sales representatives. Baxalta will specify the conduct and content of the CDCP Product training and marketing materials.

Unless otherwise set forth in the Joint Commercialization Plan, each Party is solely responsible for recruiting, hiring and maintaining its sales force for promotion of the CDCP Product in accordance with applicable Law and industry standards, its standard procedures, the Joint Commercialization Plan and the CDCP Agreement. Each Party's sales representatives will utilize only promotional material approved by Baxalta and all dealings shall be consistent with the approved promotional materials.

Baxalta will have sole authority (i) to execute medical and scientific affairs and programs based upon protocols approved by the Committee, (ii) over all medical affairs activities related to the CDCP Product, (iii) to respond to questions and requests for information about the CDCP Product that go beyond the understanding of the sales representatives or are beyond the scope of the CDCP Product labels and inserts. Further, Baxalta will have sole control over all other customer-facing aspects of Commercialization of the CDCP Product, including but not limited to all aspect of the development & execution of the commercial and promotional strategy, brand name, packaging, pricing, reimbursement, contracting, promotional materials, commercial booth structure & promotional content for display at scientific forums and at other approved venues; provided, however, for a Regional Product, development & execution of the foregoing activities shall be handled by the Committee.

Notwithstanding the foregoing, Precision may, by written notice to Baxalta received at least [***], elect not to field a sales force and instead to share the costs of the Baxalta sales force for promotion of the CDCP Product in accordance with the Profit & Loss Share below.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

REGULATORY

The Parties will assume through the Committee joint responsibility for all regulatory matters regarding seeking Regulatory Approval in the Territory, which shall be part of the Joint Development Plan. Baxalta will take the lead and have final authority with respect to regulatory activities for seeking Regulatory Approval for the CDCP Product, provided that Precision will have the right (i) to review and provide comments on all Regulatory Materials in the Territory other than administrative communications with Regulatory Authorities, and (ii) have a limited number of representatives participate in all meetings with any Regulatory Authorities in the Territory regarding such activities to the extent permitted by the applicable Regulatory Authorities and provided that, for any meetings with limited number of attendees, Baxalta shall have the first right to appoint designees to attend meetings in accordance with the goal of the meeting, but in all events at least one Precision representative shall be permitted to attend at Precision's request. All Regulatory Materials for the CDCP Product will be owned by Baxalta. Baxalta will maintain the global adverse event database for the CDCP Product in accordance with the Agreement.

REPORTS

Each Party will prepare and provide to the other Party reports regarding its activities under the CDCP Agreement as the Committee may reasonably require or as the other Party may reasonably request.

Each Party will prepare and maintain reasonably complete and accurate records regarding its Development and Commercialization activities under the CDCP Agreement. Each Party will provide to the other Party a reasonably detailed report regarding such efforts at least once every [***] (or more frequently if required by the Committee), with sufficient detail to enable a Party to assess the other Party's compliance with its obligations under the CDCP Agreement.

PRECISION'S RIGHT TO NEGOTIATE LEAD ROLE

With respect to the fourth, fifth and sixth Included Targets designated by Baxalta under the Agreement, if applicable, for up to two (2) CDCP Products that are anticipated to be Developed as Regional Products, Precision may elect by written notice to Baxalta to negotiate in good faith additional terms of the applicable CDCP Agreement relating to a lead or increased role for Precision with respect to the following:

- (a) [***]
- (b) [***]
- (c) [***]
- (d) [***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

***1

Notwithstanding anything to the contrary, Baxalta shall be required only to discuss with Precision in good faith the terms under which it may agree to a lead or increased role by Precision, but shall not be required to agree to Precision taking a lead or increased role.

FINANCIAL TERMS

PROFIT & LOSS SHARE

The Parties will share equally all operating profits and all operating losses arising from the CDCP Product for and in the Territory, which shall be calculated in accordance with an exhibit to be attached to the CDCP Agreement (the "Profit and Loss").

The Profit and Loss calculation will include the following: [***].

Each Party shall prepare quarterly reports of their Profits and Losses for the immediately prior Calendar Quarter and submit it to the representatives from each Party selected by the Committee. The representatives shall prepare a reconciliation statement comparing the Parties' actual Profits and Losses. If the agreed reconciliation statement reflects that a Party has incurred more than its share of Losses (or if the other Party received more than its share of Profits) in the applicable calendar quarter, it may submit an invoice to the other Party for the difference, and the other Party shall pay the amount owed within [***] after receipt of the invoice.

The Parties will agree upon a commercially reasonable FTE rate for Development and Commercialization activities that will be included in the form CDCP Agreement.

MISCELLANEOUS

TERMINATION

The CDCP Agreement will automatically terminate if the Agreement terminates with respect to any CDCP Product in the Territory, and may be terminated in accordance with any provision of the Agreement that provides for termination of the CDCP Agreement. In addition, either Party may terminate the CDCP Agreement in its entirety immediately upon written notice to the other Party if the other Party materially breaches its obligations under the CDCP Agreement (which termination shall be substantially similar to Section 14.2.1 of the Agreement).

Precision may terminate the CDCP Agreement by delivering written notice to Baxalta [***]. Upon receipt of notice of the Opt-Out, the Committee will promptly meet and agree upon revisions to the Joint Development

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***]

Plan and Joint Commercialization Plan as necessary for an orderly wind-down of Precision's activities prior to the effective date of the Opt-Out, with any additional cost directly related to wind-down of such activities allocated to Precision to be borne by Precision. The effective date of the Opt-Out will be [***] of the Calendar Year after Baxalta's receipt of the Opt-Out notice. After the effective date of the Opt-Out, Precision will have no further responsibilities for conducting Development or Commercialization activities for the CDCP Product except as set forth in the Agreement, and will no longer share in the Profit and Loss with respect to such CDCP Product. Upon exercising the Opt-Out, the adjusted financials set forth in Section 7.3.5 of the Agreement will no longer apply, and the CDCP Product will again be considered a Licensed Product subject to milestone payments and royalties under the Agreement.

[***]

At any time during the term of the CDCP Agreement, if Precision believes that it may not have access to cash sufficient to cover its share of Development or Commercialization costs (in accordance with the Annual Joint Development Budget or Annual Joint Commercialization Budget, as applicable) for the then-following [***], Precision shall promptly notify Baxalta, and the Parties shall promptly meet and discuss in good faith to resolve the issue. If the Parties agree that Precision will not have sufficient cash for such [***] period, then the Parties shall promptly meet and agree upon revisions to the Joint Development Plan and Joint Commercialization Plan as necessary for an orderly wind-down of Precision's activities within the [***] following such meeting, with any additional cost directly related to wind-down of such activities allocated to Precision to be borne by Precision. Such cessation of activities shall be deemed an Opt-Out, effective on the date of the last day of the mutually-agreed wind-down activities.

INTELLECTUAL PROPERTY

The intellectual property rights and responsibilities described in the Agreement shall govern the CDCP Agreement.

TRADEMARKS

Baxalta will select trademarks and trade dress for CDCP Products. Precision shall have the right to include its logo on Baxalta-approved marketing materials that are used by Precision's salesforce in a manner approved by the Committee.

PRESS RELEASE

The Parties may, upon mutual agreement, jointly issue a press release to reflect the co-development and co-promotion of the CDCP Product.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***]

INDEMNIFICATION The CDCP Agreement would contain commercially reasonable indemnification provisions customary for an agreement of

this type, consistent with the Agreement.

ASSIGNMENT AND

CHANGE OF CONTROL

These matters will be subject to the same terms as agreed by the Parties with respect to such matters in the Agreement.

GOVERNING LAW New York

DISPUTE RESOLUTION Executive escalation, followed by binding arbitration with a neutral arbitrator pursuant to the arbitration process set forth in

Article XV of the Agreement.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

EXHIBIT E PRECISION PATENTS AND PRECISION PLATFORM PATENTS

[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

г	*	*	*	٦

[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	

[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
	. ,
[***]	[***]
[***] [***]	[***] [***]

[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

CONFIDENTIAL

EXHIBIT F

PRESS RELEASE

[Follows on next page.]

F-1





Baxalta and Precision BioSciences form Global Genome Editing Collaboration in Immuno-Oncology

- Baxalta and Precision BioSciences to utilize proprietary ARCUS genome editing technology to develop an allogeneic CAR T cell therapeutic pipeline
- Precision BioSciences to receive \$105 million upfront, in addition to potential future milestone payments and royalties

BANNOCKBURN, Ill. and DURHAM, N.C., Feb [__], 2016 – Baxalta Incorporated (NYSE: BXLT), a global biopharmaceutical leader dedicated to delivering transformative therapies to patients with orphan diseases and underserved conditions, and Precision BioSciences, the genome editing company, today announced a global collaboration to develop a broad series of allogeneic chimeric antigen receptor (CAR) T cell therapies directed towards areas of major unmet need in multiple cancers.

CAR T is widely recognized as a breakthrough technology with the potential to become a curative option for certain malignancies. Most CAR T cell therapy technologies isolate cells from cancer patients' blood and re-engineer them to specifically target receptors on tumor cells. The reprogrammed cells are multiplied in a laboratory and then returned to the patient to target the tumor. This approach has had initial success in clinical trials for certain tumor types, but persistent scaling challenges remain based on the highly personalized nature of the therapy. Precision BioSciences' proprietary ARCUS genome editing technology enables the production of CAR T cells derived from healthy donors rather than relying on the patient. This approach aims to overcome the manufacturing-related limitations with existing CAR T therapies and enable a broader range of malignancies to be targeted.

"Collaborating with Precision BioSciences enables Baxalta to accelerate innovation in immuno-oncology with a next-generation, donor-derived CAR T strategy using a proprietary combination of genome editing expertise and technology," said David Meek, executive vice president and president, Oncology, Baxalta. "Combining Precision BioSciences' ARCUS technology with Baxalta's global infrastructure, expertise and growing immuno-oncology portfolio is a synergistic approach that we believe has the potential to make disruptive approaches available to people with a range of underserved cancers."

"Baxalta is an ideal partner in CAR T for Precision and our ARCUS genome editing platform because of their global reach, collaborative business model, and long-term commitment to succeeding in immuno-oncology," said Matthew Kane, CEO of Precision BioSciences. "We look forward to working closely with the team at Baxalta to develop novel CAR T therapeutics that could transform the treatment of cancer."

Under the terms of the agreement, Baxalta and Precision BioSciences will develop CAR T therapies for up to six unique targets, with the first program expected to enter clinical studies in late 2017. Precision BioSciences will be responsible for performing early-stage research activities up to Phase 2, following which Baxalta has the exclusive right to opt in for late-stage development and commercialization. Precision BioSciences will receive an upfront payment of \$105 million from Baxalta, with additional option fees, developmental, clinical, regulatory, and sales milestones, potentially totaling up to \$1.6 billion, in addition to royalties on worldwide sales. Precision also has the right to participate in the development and commercialization of any licensed products resulting from the collaboration through a 50/50 co-development and co-promotion option in the United States. Additional terms and initial targets were not disclosed.

The agreement follows another recently established Baxalta collaboration to advance novel therapeutics against checkpoint targets, advancing the company's strategic commitment to investing in immuno-oncology and building an innovative portfolio of cancer immunotherapies.

About Baxalta

Baxalta Incorporated (NYSE: BXLT) is a \$6 billion global biopharmaceutical leader developing, manufacturing and commercializing therapies for orphan diseases and underserved conditions in hematology, oncology and immunology. Driven by passion to make a meaningful impact on patients' lives, Baxalta's broad and diverse pipeline includes biologics with novel mechanisms and advanced technology platforms such as gene therapy. The Baxalta Global Innovation and R&D Center is located in Cambridge, Massachusetts. Launched in 2015 following separation from Baxter International Inc, Baxalta's heritage in biopharmaceuticals spans decades. Baxalta's therapies are available in more than 100 countries and it has advanced biological manufacturing operations across 12 facilities, including state-of-the-art recombinant production and plasma fractionation. Headquartered in Northern Illinois, Baxalta employs 16,000 employees worldwide.

About Precision BioSciences

Precision BioSciences, the genome editing company, is dedicated to improving life. Our team seeks to solve significant problems in oncology, genetic disease, agriculture, and beyond via its ARCUS genome editing platform. ARCUS is Precision's proprietary, nuclease-based genome editing platform which now encompasses an industry leading combination of site specificity, ease of delivery, and breadth of editing capabilities. For additional information, please visit www.precisionbiosciences.com.

Forward-Looking Statements

This release includes forward-looking statements concerning a collaboration between Baxalta Incorporated and Precision BioSciences, including expectations with regard to potential impact of the CAR T technology to patients, future clinical studies and commercial launches, as well as the potential financial impact of the arrangements. Such statements are made of the date that they were first issued and are based on current expectations, beliefs and assumptions of management. Forward-looking statements are subject to a number of risks and uncertainties, many of which involve factors or circumstances that are beyond Baxalta's control and which could cause actual results to differ materially from those in the forward-looking statements, including the following: clinical trial results; satisfaction of regulatory and other requirements; actions of regulatory bodies and other governmental authorities; changes in laws and regulations; product quality, manufacturing or supply issues; patient safety issues; and other risks identified in Baxalta's filings with the Securities and Exchange Commission, all of which are available on Baxalta's website. Baxalta expressly disclaims any intent or obligation to update these forward-looking statements except as required by law.

###

Baxalta Media Relations

Kellie Hotz, +1-224-940-2202, media@baxalta.com

Baxalta Investor Relations

Mary Kay Ladone, +1-224-940-3371, mary.kay.ladone@baxalta.com Lorna Williams, +1-224-940-3511, lorna.williams@baxalta.com

Precision BioSciences Media RelationsChelsea Lynam, +1-919-314-5512, chelsea.lynam@precisionbiosciences.com

Precision BioSciences Inquiries partner@precisionbiosciences.com

CONFIDENTIAL

[***] [***] [***]

[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***]

[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***]		
[***]		
	•	[***]
	•	[***]
[***]		
		[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
[***]		
	•	[***]
	•	[***]
[***]		
		[***]
[***]		
[***]		
	•	[***]
	•	[***]
[***]		
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
[***]		

[***]

[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

	•	[***]
[***]		
	•	[***]
	•	[***]
[***]		
	•	[***]
	•	[***]
[***]		
	•	[***]
[***]		
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
	•	[***]
[***]		
	•	[***]

. .

• [***]

· [...]

• [***]

• [***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

EXHIBIT H PRECISION WIRE INSTRUCTIONS

NAME: Precision BioSciences, Inc.

BANK: Square 1 Bank

406 Blackwell Street, Suite 240

Durham, NC 27701

Account No.: [***]
Routing No.: [***]
Swift-BIC: [***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

AMENDMENT NO. 1 TO DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT

This AMENDMENT NO. 1 TO DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT (this "Amendment"), effective as of February 24, 2017, amends that certain Development and Commercial License Agreement, effective February 24, 2016 (the "DCLA"), by and among BAXALTA INCORPORATED, a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("BI"), BAXALTA US INC., a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("BUSI"), BAXALTA GMBH, a company organized under the laws of Switzerland with its principal place of business at Postfach 8010, Zurich, Switzerland ("BGMBH" and, together jointly and severally with BI and BUSI, collectively, "Baxalta"), and PRECISION BIOSCIENCES, INC., a Delaware corporation with its principal place of business at 302 E Pettigrew St A-100, Durham, NC 27701 ("Precision"). Baxalta and Precision may each be referred to herein individually as a "Party" and collectively as the "Parties."

RECITALS

WHEREAS, the Parties desire to amend the DCLA as set forth below in accordance with Section 17.8 of the DCLA.

AGREEMENT

NOW THEREFORE, in consideration of the mutual promises and covenants set forth below and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

- 1. Amendment of the DCLA. Section 7.3.3 of the DCLA is hereby amended and restated in its entirety as follows:
 - 7.3.3 The key terms of the CDCP Agreement are set forth in Exhibit D. The Parties will negotiate in good faith a form of CDCP Agreement consistent with such terms and other terms and conditions that are typical for such agreements with respect to similarly situated products between similarly situated parties in the pharmaceutical industry promptly after the Effective Date (and in any event, no later than [***] after the Effective Date), and such form of CDCP Agreement shall be attached to this Agreement as Exhibit D-1 upon mutual agreement between the Parties.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 2. No Other Changes. Except as expressly modified by this Amendment, the DCLA shall remain unchanged and in full force and effect. This Amendment shall become a part of the DCLA as if set forth in full therein, and references to the DCLA shall be to the DCLA as amended. To the extent of any conflict or inconsistency between the terms of this Amendment and the DCLA, the terms of this Amendment shall prevail.
- 3. <u>Dispute Resolution; Governing Law; Jurisdiction</u>. The Parties acknowledge and agree that Article XV and Section 17.13 of the DCLA shall apply to this Amendment as if fully set forth herein.
- 4. <u>Counterparts</u>. This Amendment may be executed in more than one counterpart (including by electronic transmission), each of which shall be deemed an original, but all of such counterparts taken together shall constitute one and the same agreement.

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 1 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

PRECISION BIOSCIENCES, INC.

By: /s/ Matt Kane
Name: Matt Kane
Title: CEO

[signatures continue on following pages]

Signature Page to Amendment No. 1 to Development and Commercial License Agreement

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 1 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

BAXALTA INCORPORATED

By: /s/ Jeffrey E Prowda
Name: Jeffrey E Prowda
Title: Assistant Secretary

BAXALTA US INC.

By: /s/ Jeffrey E Prowda
Name: Jeffrey E Prowda
Title: Assistant Secretary

[signatures continue on following page]

Signature Page to Amendment No. 1 to Development and Commercial License Agreement

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 1 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

BAXALTA GMBH

By:			
Name:			
Title:			
By: /s/	Yvo Aebli		

Name: Yvo Aebli

Title: EU & EEMEA Controller

Signature Page to Amendment No. 1 to Development and Commercial License Agreement

AMENDMENT NO. 2 TO DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT

This AMENDMENT NO. 2 TO DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT (this "Amendment"), effective as of August 21, 2017, amends that certain Development and Commercial License Agreement, effective February 24, 2016 (as previously amended by that certain Amendment No. 1 to Development and Commercial License Agreement, effective as of February 24, 2017, the "DCLA"), by and among BAXALTA INCORPORATED, a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("BI"), BAXALTA US INC., a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("BUSI"), BAXALTA GMBH, a company organized under the laws of Switzerland with its principal place of business at Postfach 8010, Zurich, Switzerland ("BGMBH" and, together jointly and severally with BI and BUSI, collectively, "Baxalta"), and PRECISION BIOSCIENCES, INC., a Delaware corporation with its principal place of business at 302 E Pettigrew St A-100, Durham, NC 27701 ("Precision"). Baxalta and Precision may each be referred to herein individually as a "Party" and collectively as the "Parties."

RECITALS

WHEREAS, the Parties desire to amend the DCLA as set forth below in accordance with Section 17.8 of the DCLA.

AGREEMENT

NOW THEREFORE, in consideration of the mutual promises and covenants set forth below and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

- 1. Amendment of the DCLA. Section 7.3.3 of the DCLA is hereby amended and restated in its entirety as follows:
 - 7.3.3 The key terms of the CDCP Agreement are set forth in Exhibit D. The Parties will negotiate in good faith a form of CDCP Agreement consistent with such terms and other terms and conditions that are typical for such agreements with respect to similarly situated products between similarly situated parties in the pharmaceutical industry promptly after the Effective Date (and in any event, no later than [***] after the Effective Date), and such form of CDCP Agreement shall be attached to this Agreement as Exhibit D-1 upon mutual agreement between the Parties.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 2. No Other Changes. Except as expressly modified by this Amendment, the DCLA shall remain unchanged and in full force and effect. This Amendment shall become a part of the DCLA as if set forth in full therein, and references to the DCLA shall be to the DCLA as amended. To the extent of any conflict or inconsistency between the terms of this Amendment and the DCLA, the terms of this Amendment shall prevail.
- 3. <u>Dispute Resolution; Governing Law; Jurisdiction</u>. The Parties acknowledge and agree that Article XV and Section 17.13 of the DCLA shall apply to this Amendment as if fully set forth herein.
- 4. <u>Counterparts</u>. This Amendment may be executed in more than one counterpart (including by electronic transmission), each of which shall be deemed an original, but all of such counterparts taken together shall constitute one and the same agreement.

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 2 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

PRECISION BIOSCIENCES, INC.

By: /s/ Matt Kane
Name: Matt Kane
Title: CEO

[signatures continue on following pages]

Signature Page to Amendment No. 2 to Development and Commercial License Agreement

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 2 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

BAXALTA INCORPORATED

By: /s/ Denise M. Serewicz
Name: Denise M. Serewicz
Title: Assistant Secretary

BAXALTA US INC.

By: /s/ Denise M. Serewicz
Name: Denise M. Serewicz
Title: Assistant Secretary

[signatures continue on following page]

Signature Page to Amendment No. 2 to Development and Commercial License Agreement

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 2 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

BAXALTA GMBH

By: /s/ Remco Lemarcq
Name: Remco Lemarcq

Title: Proxy Holder

Signature Page to Amendment No. 2 to Development and Commercial License Agreement

AMENDMENT NO. 3 TO DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT

This AMENDMENT NO. 3 TO DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT (this "Amendment"), effective as of February 5, 2018, amends that certain Development and Commercial License Agreement, effective February 24, 2016 (as previously amended by that certain Amendment No. 1 to Development and Commercial License Agreement, effective as of February 24, 2017, and that certain Amendment No. 2 to Development and Commercial License Agreement, effective as of August 21, 2017, the "DCLA"), by and among BAXALTA INCORPORATED, a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("BI"), BAXALTA US INC., a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("BUSI"), BAXALTA GMBH, a company organized under the laws of Switzerland with its principal place of business at Postfach 8010, Zurich, Switzerland ("BGMBH" and, together jointly and severally with BI and BUSI, collectively, "Baxalta"), and PRECISION BIOSCIENCES, INC., a Delaware corporation with its principal place of business at 302 E Pettigrew St A-100, Durham, NC 27701 ("Precision"). Baxalta and Precision may each be referred to herein individually as a "Party" and collectively as the "Parties."

RECITALS

WHEREAS, the Parties desire to amend the DCLA as set forth below in accordance with Section 17.8 of the DCLA.

AGREEMENT

NOW THEREFORE, in consideration of the mutual promises and covenants set forth below and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

- 1. Amendment of the DCLA. Section 7.3.3 of the DCLA is hereby amended and restated in its entirety as follows:
 - 7.3.3 The key terms of the CDCP Agreement are set forth in <u>Exhibit D</u>. The Parties will negotiate in good faith a form of CDCP Agreement consistent with such terms and other terms and conditions that are typical for such agreements with respect to similarly situated products between similarly situated parties in the pharmaceutical industry promptly after the Effective Date (and in any event, no later [***]), and such form of CDCP Agreement shall be attached to this Agreement as <u>Exhibit D-1</u> upon mutual agreement between the Parties.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 2. No Other Changes. Except as expressly modified by this Amendment, the DCLA shall remain unchanged and in full force and effect. This Amendment shall become a part of the DCLA as if set forth in full therein, and references to the DCLA shall be to the DCLA as amended. To the extent of any conflict or inconsistency between the terms of this Amendment and the DCLA, the terms of this Amendment shall prevail.
- 3. <u>Dispute Resolution; Governing Law; Jurisdiction</u>. The Parties acknowledge and agree that Article XV and Section 17.13 of the DCLA shall apply to this Amendment as if fully set forth herein.
- 4. <u>Counterparts</u>. This Amendment may be executed in more than one counterpart (including by electronic transmission), each of which shall be deemed an original, but all of such counterparts taken together shall constitute one and the same agreement.

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 3 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

PRECISION BIOSCIENCES, INC.

By: /s/ Matt Kane
Name: Matt Kane
Title: CEO

[signatures continue on following pages]

Signature Page to Amendment No. 3 to Development and Commercial License Agreement

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 3 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

BAXALTA INCORPORATED

By:/s/ David BaileyName:David BaileyTitle:Assistant Treasurer

BAXALTA US INC.

By: /s/ David Bailey
Name: David Bailey
Title: Assistant Treasurer

[signatures continue on following page]

Signature Page to Amendment No. 3 to Development and Commercial License Agreement

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 3 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

BAXALTA GMBH

By: /s/ Barbara Lenzlinger

Name: Barbara Lenzlinger
Title: International Controller

Signature Page to Amendment No. 3 to Development and Commercial License Agreement

AMENDMENT NO. 4 TO DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT

This AMENDMENT NO. 4 TO DEVELOPMENT AND COMMERCIAL LICENSE AGREEMENT (this "Amendment"), effective as of May 23, 2018, amends that certain Development and Commercial License Agreement, effective February 24, 2016 (as previously amended by that certain Amendment No. 1 to Development and Commercial License Agreement, effective as of February 24, 2017, that certain Amendment No. 2 to Development and Commercial License Agreement, effective as of August 21, 2017, and that certain Amendment No. 3 to Development and Commercial License Agreement, effective as of February 5, 2018, the "DCLA"), by and among BAXALTA INCORPORATED, a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("BI"), BAXALTA US INC., a Delaware corporation with its principal place of business at 1200 Lakeside Drive, Bannockburn, IL 60015 ("BUSI"), BAXALTA GMBH, a company organized under the laws of Switzerland with its principal place of business at Zahlerweg 4, 6300Zug, Switzerland ("BGMBH" and, together jointly and severally with BI and BUSI, collectively, "Baxalta"), and PRECISION BIOSCIENCES, INC., a Delaware corporation with its principal place of business at 302 E Pettigrew St A-100, Durham, NC 27701 ("Precision"). Baxalta and Precision may each be referred to herein individually as a "Party" and collectively as the "Parties." All capitalized terms not otherwise defined in this Amendment shall have the meaning set forth in the DCLA.

RECITALS

WHEREAS, the Parties have determined to extend the date for negotiation and execution of the CDCP Agreement;

WHEREAS, the Parties have determined that the Isolex Platform Technology will not be used by the Parties with respect to Licensed Product Candidates or Licensed Products, and therefore the Parties wish to remove from the DCLA all references to rights and obligations in connection with such use (but for clarity, retain references to and protection of Isolex Platform Technology and Isolex Platform Patents as Baxalta's Background IP); and

WHEREAS, the Parties desire to amend the DCLA as set forth below in accordance with Section 17.8 of the DCLA.

AGREEMENT

NOW THEREFORE, in consideration of the mutual promises and covenants set forth below and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

- 1. Amendment of Section 7.3.3 of the DCLA. Section 7.3.3 of the DCLA is hereby amended and restated in its entirety as follows:
 - 7.3.3 The key terms of the CDCP Agreement are set forth in Exhibit D. The Parties will negotiate in good faith a form of CDCP Agreement consistent with such terms and other terms and conditions that are typical for such agreements with respect to similarly situated products between similarly situated parties in the pharmaceutical industry promptly after the Effective Date (and in any event, no later than [***]), and such form of CDCP Agreement shall be attached to this Agreement as Exhibit D-1 upon mutual agreement between the Parties.
- 2. Amendment of Section 2.6, Section 14.3.4, and the Table of Contents of the DCLA.
 - a. Section 2.6 of the DCLA is hereby amended and restated in its entirety as follows:
 - 2.6 [Intentionally Omitted]
 - b. Section 14.3.4 of the DCLA is hereby amended and restated in its entirety as follows:
 - 14.3.4 <u>Survival</u>. The following Articles and Sections of this Agreement, as well as remedies for breach of this Agreement, shall survive expiration or termination of this Agreement for any reason: <u>ARTICLE I</u> (solely to the extent required to enforce any other surviving rights or obligations of the Parties), <u>Section 4.5</u>, <u>ARTICLE VIII</u> (solely to the extent any right to payment has accrued prior to the effective date of termination), <u>Section 9.1</u>, <u>Section 12.1</u>, <u>Section 12.2</u>, <u>Section 12.4</u>, <u>ARTICLE XIII</u>, <u>Section 14.3</u>, <u>ARTICLE XV</u> and <u>ARTICLE XVII</u>.
 - c. The Table of Contents of the DCLA is hereby amended to replace "2.6 Isolex Co-Development" with "2.6 [Intentionally Omitted]".

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 3. No Other Changes. Except as expressly modified by this Amendment, the DCLA shall remain unchanged and in full force and effect. This Amendment shall become a part of the DCLA as if set forth in full therein, and references to the DCLA shall be to the DCLA as amended. To the extent of any conflict or inconsistency between the terms of this Amendment and the DCLA, the terms of this Amendment shall prevail.
- 4. <u>Dispute Resolution; Governing Law; Jurisdiction</u>. The Parties acknowledge and agree that Article XV and Section 17.13 of the DCLA shall apply to this Amendment as if fully set forth herein.
- 5. <u>Counterparts</u>. This Amendment may be executed in more than one counterpart (including by electronic transmission), each of which shall be deemed an original, but all of such counterparts taken together shall constitute one and the same agreement.

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 4 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

PRECISION BIOSCIENCES, INC.

By: /s/ Matt Kane
Name: Matt Kane
Title: CEO

[signatures continue on following pages]

Signature Page to Amendment No. 4 to Development and Commercial License Agreement

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 4 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

BAXALTA INCORPORATED

By: /s/ Jeffrey E Prowda

Name: Jeffrey E Prowda

Title: Vice President and Assistant Secretary

BAXALTA US INC.

By: /s/ Jeffrey E Prowda

Name: Jeffrey E. Prowda

Title: Vice President and Assistant Secretary

[signatures continue on following page]

Signature Page to Amendment No. 4 to Development and Commercial License Agreement

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment No. 4 to Development and Commercial License Agreement to be executed as of the date first set forth above by their respective duly authorized officers or representatives.

BAXALTA GMBH

By: /s/ Elisabeth Leimbacher

Name: Elisabeth Leimbacher

Title: Proxy Holder

Signature Page to Amendment No. 4 to Development and Commercial License Agreement

COLLABORATION AND LICENSE AGREEMENT

between

GILEAD SCIENCES, INC.

and

PRECISION BIOSCIENCES, INC.

Dated as of September 10, 2018

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

TABLE OF CONTENTS

ARTICLE 1	DEFINITIONS	Page 1
ARTICLE 2	COLLABORATION MANAGEMENT	15
2.1	Joint Steering Committee	15
2.2	Joint Research and Development Committee	16
2.3	General Provisions Applicable to Joint Committees	17
ARTICLE 3	DEVELOPMENT AND REGULATORY	19
3.1	Collaboration Program	19
3.2	Collaboration R&D Plan	20
3.3	Performance of the Collaboration R&D Plan	21
3.4	Development Following Expiry of the Collaboration Term	21
3.5	Clinical Supply of the Licensed Products	21
3.6	Technology Transfer	22
3.7	Regulatory Matters	23
3.8	Records and JSC Reporting	24
ARTICLE 4	GRANT OF RIGHTS AND COVENANTS	26
4.1	Grants to Gilead	26
4.2	Grants to Precision	27
4.3	Sublicensing and Subcontracting	27
4.4	Retention of Rights	28
4.5	Exclusivity	28
4.6	Existing In-License Agreements	29
4.7	Preservation of Existing In-License Agreements and [***] In-License Agreements	30
ARTICLE 5	PAYMENTS AND RECORDS	30
5.1	Payments during the Collaboration Program	30
5.2	Development and Regulatory Milestones	31
5.3	Commercial Milestones	31
5.4	Royalties	32
5.5	Mode of Payment	35
5.6	Currency	35
5.7	Taxes	35
5.8	Financial Records	36
5.9	Audit	36
5.10	Confidentiality	37

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE 6	INTELLECTUAL PROPERTY	37
6.1	Ownership of Intellectual Property	37
6.2	Maintenance and Prosecution of Patents	39
6.3	Enforcement of Patents	42
6.4	Infringement Claims by Third Parties	45
6.5	Invalidity or Unenforceability Defenses or Actions	46
6.6	Patent Term Extensions in the Territory	48
ARTICLE 7	CONFIDENTIALITY AND NON-DISCLOSURE	48
7.1	Confidentiality Obligations	48
7.2	Permitted Disclosures	49
7.3	Use of Name	51
7.4	Public Announcements	51
7.5	Publications	51
7.6	Return of Confidential Information	52
ARTICLE 8	REPRESENTATIONS AND WARRANTIES	52
8.1	Mutual Representations and Warranties	52
8.2	Additional Representations and Warranties of Precision	53
8.3	Additional Covenants of Precision	55
8.4	DISCLAIMER OF WARRANTIES	56
ARTICLE 9	INDEMNITY	56
9.1	Indemnification of Precision	56
9.2	Indemnification of Gilead	57
9.3	Notice of Claim	57
9.4	Control of Defense	57
9.5	Special, Indirect, and Other Losses	59
9.6	Insurance	59
ARTICLE 10	TERM AND TERMINATION	61
10.1	Term	61
10.2	Termination by Either Party	61
10.3	Additional Termination Rights by Gilead	62
10.4	Additional Termination Rights by Precision	62
10.5	Rights in Bankruptcy	62
10.6	Effects of Termination	63
10.7	Remedies	64
10.8	Accrued Rights; Surviving Obligations	64

ARTICLE 11	MISCELLANEOUS	65
11.1	Force Majeure	65
11.2	Export Control	65
11.3	Assignment	65
11.4	Severability	66
11.5	Governing Law	66
11.6	Dispute Resolution	67
11.7	Notices	67
11.8	Entire Agreement	69
11.9	English Language	69
11.10	Amendments and Waivers	69
11.11	Cumulative Rights	69
11.12	Benefits of Agreement	69
11.13	Further Assurance	70
11.14	Relationship of the Parties	70
11.15	Counterparts	70
11.16	Schedules	70
11.17	Descriptive Headings; Certain Interpretations	70
SCHEDULES		
Schedule 1.45	Duke IP	

Deficatio 1.15	Duke II
Schedule 1.112	Precision Existing Patents
Schedule 1.113	Precision HBV Patents
Schedule 2.1.2	Initial JSC Members
Schedule 2.2.1	Initial JRDC Members
Schedule 3.8.3	Template Report
Schedule 5.1.2	Invoicing Template
Schedule 6.2.8	ARCUS Patents and Precision Patents (To be added when first delivered to Gilead pursuant to Section 6.2.8)
Schedule 7.4	Press Release(s)
Schedule 8.2.12	Precision IP Subject to U.S. Federal Government Rights

COLLABORATION AND LICENSE AGREEMENT

This Collaboration and License Agreement (the "Agreement") is made and entered into effective as of September 10, 2018 (the "Effective Date") by and between Gilead Sciences, Inc., a corporation organized and existing under the laws of the State of Delaware and having its principal place of business at 333 Lakeside Drive, Foster City, California 94404 ("Gilead"), and Precision Biosciences, Inc., a corporation organized and existing under the laws of the State of Delaware and having its principal place of business at 302 East Pettigrew St., Suite A-100, Durham, North Carolina 27701 ("Precision"). Gilead and Precision are sometimes referred to herein individually as a "Party" and collectively as the "Parties."

RECITALS

WHEREAS, Precision has developed a proprietary genome editing platform, the ARCUS Technology (as defined herein), and controls certain intellectual property rights with respect to using the ARCUS Technology to create fully synthetic nucleases derived from homing endonucleases;

WHEREAS, Gilead and Precision desire to collaborate on a research and pre-clinical development program to construct, optimize and develop one or more gene editing therapy(ies) that incorporates or otherwise uses one or more nucleases made using the ARCUS Technology and that targets the hepatitis B virus DNA, as further described below, in accordance with the terms and conditions set forth below; and

WHEREAS, following the end of the collaboration, Gilead wishes to assume sole responsibility for the development and commercialization of such gene editing therapies and products containing such gene editing therapies, as further described below, in accordance with the terms and conditions set forth below.

NOW, THEREFORE, in consideration of the premises and the mutual promises and conditions hereinafter set forth, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, do hereby agree as follows:

ARTICLE 1 DEFINITIONS

Unless otherwise specifically provided herein, the following terms shall have the following meanings:

- 1.1 "Acquirer" means, with respect to a Party, collectively (a) any Third Party that acquires a Party after the Effective Date (whether by transfer or sale of all or any portion of such Party's assets, equity or business, or by a Change of Control or similar business combination transaction or otherwise) and (b) the Affiliates of such Third Party, but excluding such Party and such Party's Affiliates existing immediately prior to the closing of such acquisition of such Party.
- 1.2 "Act" means the United States Federal Food, Drug, and Cosmetic Act, as amended, and the rules, regulations, guidelines, guidance documents and requirements promulgated thereunder, as may be in effect from time to time.

- 1.3 "Action" means any claim or threatened claim, action, suit, arbitration, inquiry, audit, proceeding or investigation (including any investigation by, before or otherwise involving any governmental authority or Regulatory Authority).
- 1.4 "Active Component" means a component that confers a therapeutic effect on a standalone basis or on an incremental or synergistic basis, excluding, for clarity and without limitation, [***] and compounds that potentiate nucleases but which themselves do not confer a therapeutic effect on such basis.
- 1.5 "Affiliate" means, with respect to any Person, any other Person that directly or indirectly Controls, is directly or indirectly Controlled by, or is under direct or indirect common Control with, such first Person. For purposes of this definition, a Person shall be deemed, in any event, to Control another Person if it (a) owns or Controls, directly or indirectly, or has the ability to direct or cause the direction or Control of, more than fifty percent (50%) of the voting equity of such other Person, or (b) has the ability to direct, cause the direction of Control the management or policies of such other Person, whether through direct or indirect ownership of voting equity, by contract or otherwise.
 - 1.6 "Affordable Basis" means sale or other disposition of the Licensed Product by Gilead or its Affiliate or Sublicensee [***].
 - **1.7** "Agreement" has the meaning set forth in the preamble hereto.
 - **1.8** "Alliance Manager" has the meaning set forth in Section 2.3.5.
- 1.9 "Applicable Law" means any applicable federal, state, local or foreign constitution, treaty, law, statute, ordinance, rule, regulation, interpretation, guidance document, directive, policy, order, writ, award, decree, injunction, judgment, stay or restraining order of any governmental authority or Regulatory Authority, the terms of any permit, and any other ruling or decision of, agreement with or by, or any other requirement of, any governmental authority or Regulatory Authority having proper jurisdiction over the matter.
 - 1.10 "ARC Nuclease" means any fully synthetic nuclease derived from a homing endonuclease and made using the ARCUS Technology.
 - 1.11 "ARCUS Assigned IP" [***].
 - 1.12 "ARCUS IP" means the ARCUS Technology and the ARCUS Patents.
- 1.13 "ARCUS Patents" means any and all Patents Controlled by Precision or any of its Affiliates as of the Effective Date or at any time during the Term that claim or cover ARCUS Technology, excluding Patents that claim or cover ARCUS Assigned IP.
- 1.14 "ARCUS Technology" means the proprietary genome editing platform Controlled by Precision, known as ARCUSTM, relating to the design, creation, selection and optimization of fully synthetic enzymes derived from homing endonucleases, including any modifications or improvements to such platform, excluding ARCUS Assigned IP. For the sake of clarity, ARCUS Technology does not include the sequence(s) (including amino acid sequences and mRNA sequences) of any Gilead ARC Nuclease, the use of any Gilead ARC Nuclease, or the formulation of any Gilead ARC Nuclease or the Licensed Products.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.15 "Bayh-Dole Act" means the Patent and Trademark Law Amendments Act of 1980, as amended, codified at 35 U.S.C. §§ 200-212, as amended, as well as any regulations promulgated pursuant thereto, including in 37 C.F.R. Part 401.
 - **1.16** [***].
- 1.17 "Biosimilar Product" means, with respect to a Licensed Product in a country or jurisdiction specified below, any product sold by a Third Party that (a)(i) in the United States, is subject to a license by the FDA under Section 351(k) of the PHSA as a product that is "biosimilar" (as defined in Section 351(i)(2) of the PHSA) to, or "interchangeable" (as defined in Section 351(i)(3) of the PHSA) with, such Licensed Product, (ii) in the EU, has been licensed as a similar biological medicinal product by the EMA pursuant to Directive 2001/83/EC, as may be amended, or any subsequent or superseding law, statute or regulation, or (iii) in any country outside the United States and the EU, has received Regulatory Approval in an abbreviated licensure procedure as a biogeneric, biosimilar or interchangeable product from the applicable Regulatory Authority in such country or jurisdiction, in reliance upon the prior Regulatory Approval (or data therein) of such Licensed Product; and (b) is not an Authorized Biosimilar Version of such Licensed Product; where "Authorized Biosimilar Version" means any product that (1) is sold under the BLA filed by Gilead or its Affiliate or Sublicensee, as applicable.
 - 1.18 "BLA" means Biologics License Application as described in 21 C.F.R §601.2, or equivalent FDA application.
- **1.19** "Business Day" means any day excluding Saturdays, Sundays, December 26 through December 31, and any day that is a legal holiday under the Applicable Laws of the United States or that is a day on which banking institutions located in Durham, North Carolina or San Francisco, California, are authorized or required by Applicable Law or other governmental action to close.
- 1.20 "Calendar Quarter" means each successive period of three (3) calendar months commencing on January 1, April 1, July 1 and October 1, except that the first Calendar Quarter of the Term shall commence on the Effective Date and end on the day immediately prior to the first to occur of January 1, April 1, July 1 or October 1 after the Effective Date, and the last Calendar Quarter shall end on the last day of the Term.
- 1.21 "Calendar Year" means each successive period of twelve (12) calendar months commencing on January 1 and ending on December 31, except that the first Calendar Year of the Term shall commence on the Effective Date and end on December 31 of the year in which the Effective Date occurs and the last Calendar Year of the Term shall commence on January 1 of the year in which the Term ends and end on the last day of the Term.
 - **1.22** "Cellectis Agreement" has the meaning set forth in Section 4.6.1.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.23 "Cellectis Patents" has the meaning set forth in Section 4.6.1.
- 1.24 "Change of Control" means, with respect to a Party, (a) a merger or consolidation of such Party with a Third Party that results in the voting securities of such Party outstanding immediately prior thereto, or any securities into which such voting securities have been converted or exchanged, ceasing to represent at least fifty percent (50%) of the combined voting power of the surviving entity or the parent of the surviving entity immediately after such merger or consolidation, (b) a transaction or series of related transactions in which a Third Party, together with its Affiliates, becomes the beneficial owner of fifty percent (50%) or more of the combined voting power of the outstanding securities of such Party, or (c) the sale or other transfer to a Third Party of all or substantially all of such Party's business. For clarity, an initial public offering of capital stock of Precision that is effected pursuant to a registration statement or an offering statement filed with, and declared effective or qualified, as the case may be, by the Securities and Exchange Commission under the Securities Act of 1933, as amended, shall not in and of itself constitute a Change of Control.
- 1.25 "Clinical Studies" means a Phase I Clinical Study, a Phase II Clinical Study, a Proof of Concept Clinical Study, a Phase III Clinical Study, a Registrational Clinical Study and such other tests and studies in human subjects that are required by Applicable Law, or otherwise recommended by the Regulatory Authorities, to obtain or maintain Regulatory Approvals for a Licensed Product.
 - **1.26** "COGS" means in respect of each Licensed Product, [***].
 - 1.27 "Collaboration Budget" has the meaning set forth in Section 2.1.3(b).
 - **1.28** "Collaboration Program" has the meaning set forth in Section 3.1.
 - 1.29 "Collaboration R&D Plan" has the meaning set forth in Section 3.1.
 - **1.30** "Collaboration Term" has the meaning set forth in Section 3.1.
- 1.31 "Combination Product" means a Licensed Product that contains one or more Gilead ARC Nuclease(s) as one component as well as one or more other Active Components that do not constitute a Gilead ARC Nuclease, whether co-formulated, co-packaged or otherwise sold together for one price.
 - 1.32 "Commercially Reasonable Efforts" means:
- 1.32.1 with respect to the obligations of a Party under this Agreement relating to Development activities, of a Licensed Product, the level of efforts and expenditure of resources typically devoted in the research-based biopharmaceutical industry by a company to Development, of a product of similar commercial potential at a similar stage in its development or product life, in each case taking into account the Relevant Factors and as measured by the facts and circumstances at the time such efforts are due;
- 1.32.2 with respect to the level of obligations of a Party under this Agreement relating to other Exploitation activities, the level of efforts and expenditure of resources typically devoted in the research-based biopharmaceutical industry by a company to a

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

product of similar market potential at a similar stage in its development or product life, taking into account Relevant Factors and as measured by the facts and circumstances at the time such efforts are due; or

1.32.3 with respect to the obligations of a Party under this Agreement relating to any other objective, reasonable, good-faith efforts typically devoted to similar objectives in the research-based pharmaceutical industry by a company, taking into account industry practices;

provided that, [***].

- 1.33 "Competitive Infringement" means any alleged or threatened infringement of the Precision Patents or Joint Collaboration Program Patents, as applicable, by a Third Party (including alleged or threatened infringement based on the development or commercialization of, or an application to market, a Licensed Product) that is based on the manufacture, use or sale of a Gene Editing Therapy.
 - 1.34 "Competitive Program" means [***].
- 1.35 "Competitor" means any Person, other than the Parties and their Affiliates, that is conducting any Competitive Program, for so long as such conduct continues.
 - **1.36** "Confidential Information" has the meaning set forth in Section 7.1.
- **1.37** "Confidentiality Agreement" means that certain Mutual Confidential Disclosure Agreement by and between the Parties, dated September 3, 2015.
- 1.38 "Control" means (a) when used with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such entity, whether through the ownership of voting securities, by contract or otherwise, (b) when used with respect to any security, the possession, directly or indirectly, of the power to vote, or to direct the voting of, such security or the power to dispose of, or to direct the disposition of, such security and (c) when used with respect to any item of Information, Regulatory Documentation, material, Patent, or other IP Rights, possession of the right, whether directly or indirectly, and whether by ownership, license or otherwise, to assign or grant a license, sublicense or other right to or under such Information, Regulatory Documentation, material, Patent or other IP Rights to the extent that it does not violate the terms of any written agreement with any Third Party existing as of the date of such assignment or such grant, as applicable; *provided that*, any such Third Party agreement entered into after the Effective Date and requiring additional payment will meet this definition of Control only if such agreement is entered into in compliance with the requirements set forth in Section 5.4.3. "Controlled" and "Controlling" have corresponding meanings. For clarity, in the case of clause (c), a Person may Control in-licensed IP Rights from a Third Party even if its license to such IP Rights is non-exclusive or otherwise more limited than licenses granted by the licensor Third Party.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.39 "Development" means all activities related to discovery, identification, research, pre-clinical and other non-clinical testing, test method development and stability testing, toxicology, formulation, process development, manufacturing scale-up, qualification and validation, quality assurance/quality control, clinical studies, including manufacturing in support thereof, statistical analysis and report writing, the preparation and submission of applications for Regulatory Approvals, regulatory affairs with respect to the foregoing and all other activities necessary or reasonably useful or otherwise requested or required by a Regulatory Authority as a condition or in support of obtaining or maintaining a Regulatory Approval. "Developing" have corresponding meanings.
 - **1.40** "**Dispute**" has the meaning set forth in Section 11.6.
- 1.41 "Distributor" means any person appointed by Gilead or any of its Affiliates or its or their Sublicensees, and that is not an Affiliate of any of them, to distribute, market and sell the Licensed Products in one or more countries in the Territory, in circumstances where the Person purchases its requirements of the Licensed Products from Gilead or its Affiliates or its or their Sublicensees but does not make any royalty or other payment to Gilead or its Affiliates or its or their Sublicensees for (sub)license rights under Precision Know-How or Precision Patents with respect to such Licensed Products.
- 1.42 "Divest" means, as it relates to a Competitive Program: (a) the sale of all right, title and interest in such Competitive Program, including all technology, intellectual property and other assets relating solely thereto, to a Third Party, without the retention or reservation of any rights, license or interest (other than an economic interest such as a right to receive payments) by the selling entity or its Affiliates; or (b) the complete termination and/or shut-down of such Competitive Program such that no technology, intellectual property or other asset solely relating thereto is used by the terminating entity or its Affiliates for the conduct of such Competitive Program.
 - 1.43 "Dollars" or "\$" means United States Dollars.
- 1.44 "Duke Agreement" means the License Agreement entered into by Precision and Duke University ("Duke") on April 17, 2006, as amended from time to time.
- **1.45** "Duke IP" means all Patents and Information licensed to Precision under the Duke Agreement that constitute ARCUS Technology. The patent numbers and patent application numbers of the Patents that are included within the Duke IP as of the Effective Date are set forth in <u>Schedule</u> 1.45.
 - **1.46** "Effective Date" has the meaning set forth in the preamble hereto.
 - **1.47** "EMA" means the European Medicines Agency and any successor agency thereto.
 - 1.48 "Europe" means the countries of the European Union as constituted on the Effective Date.
 - 1.49 "Existing In-License Agreements" means the Duke Agreement and the Cellectis Agreement.
- **1.50** "Exploit" means to make, have made, import, use, sell, or offer for sale, research, develop, commercialize, register, manufacture, have manufactured, hold, or keep (whether for disposal or otherwise), have used, export, transport, distribute, promote, market, or have sold or otherwise dispose of. "Exploitation" and "Exploiting" have corresponding meanings.

- 1.51 "FDA" means the United States Food and Drug Administration, or any successor agency thereto.
- **1.52** "Field" means the diagnosis, treatment and prevention of all diseases.
- 1.53 "First Commercial Sale" means, on a Licensed Product-by-Licensed Product and country-by-country basis, the first invoiced commercial sale for monetary value for use or consumption by the general public of a Licensed Product in any country in the Territory after the Marketing Approval for such Licensed Product has been obtained in such country. For the avoidance of doubt, sales prior to receipt of all Marketing Approvals necessary to commence regular commercial sales, such as so-called "named patient sales" and "compassionate use sales", shall not be construed as a First Commercial Sale.
- **1.54** "**Formulation and Delivery Combination Patent**" means a Patent that includes a use in combination claim that covers the use of an ARC Nuclease in combination with a formulation or [***] developed by either Party outside the Collaboration Program.
 - 1.55 "FTE" has the meaning set forth in the definition of "FTE Rate."
- 1.56 "FTE Rate" means a rate of [***] based on a total of [***] of work performed by one or more full time employees ("FTE"), to be pro-rated on a daily basis if necessary [***]; such rate to be restricted to scientific work and managerial activities related directly to the Collaboration R&D Plan and included in the Collaboration Budget or otherwise provided for in this Agreement. For the avoidance of doubt (a) such rate includes all benefits, travel, and overhead; and (b) in no event shall any one (1) individual be counted as more than one (1) FTE.
 - 1.57 "GAAP" means generally accepted accounting principles, as applied in the United States.
- **1.58** "Gene Editing Therapy" means any product that functions through a mechanism of action of targeting, editing, deleting or otherwise modifying an HBV Target.
- **1.59** "Generally Applicable Utility" means, with respect to any Patent or Information, that the utility of such Patent or Information is not limited to (a) the field of HBV, (b) any Gilead ARC Nuclease, or (c) any Active Component contained in a Licensed Product.
- 1.60 "Generic Sublicensee" means a Sublicensee with respect to which Gilead's sublicense is non-exclusive, is granted in accordance with Gilead's "Developing World Access" program as consistently applied by Gilead to its hepatitis B product lines or, if such program is not active, then in accordance with Gilead's successor program with respect to its hepatitis B product lines and so applied, and is limited to the right to manufacture and sell a generic version of the Licensed Product in a country in which Gilead generally provides services through its "Developing World Access" program, or if such program is not active, then in a country then listed by the World Bank's latest rankings in the "low" and "lower middle" income group or equivalent.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- **1.61** "Gilead" has the meaning set forth in the preamble hereto.
- **1.62** "Gilead ARC Nuclease" means any ARC Nuclease designed, created, selected or optimized by Precision for Gilead, as disclosed or presented to Gilead pursuant to the Collaboration R&D Plan, [***].
 - 1.63 "Gilead Dual IP" means the Gilead Dual Know-How and the Gilead Dual Patents.
- 1.64 "Gilead Dual Know-How" means any and all Information to the extent Controlled by Gilead or its Affiliates that (a) is conceived, discovered, developed or otherwise made by or on behalf of Gilead or its Affiliates or Sublicensees [***] or (b)(i) Gilead or its Affiliates elect to provide or disclose to Precision under this Agreement at any time during the Term other than in the case of this clause (b) [***] and (ii) [***]. For clarity, Gilead Dual Know-How does not include any Patents.
- **1.65** "Gilead Dual Patents" means (a) any and all Patents (i) that claim or cover Gilead Dual Know-How and (ii) to the extent such Patents are Controlled by Gilead or its Affiliates at any time during the Term; and (b) any and all Patents [***].
 - **1.66** "Gilead Funding Commitment" has the meaning set forth in Section 5.1.1.
- 1.67 "Gilead Know-How" means any and all Information to the extent Controlled by Gilead or any of its Affiliates at any time during the Term that is necessary for Precision to conduct its Development activities under the Collaboration R&D Plan, excluding Gilead Dual Know-How, ARCUS Assigned IP and Joint Collaboration Program Know-How. For clarity, Gilead Know-How does not include any Patents.
- 1.68 "Gilead Patents" means any and all Patents that claim or cover the Gilead Know-How that are Controlled by Gilead or its Affiliates at any time during the Term.
 - **1.69** "Grant-Back Right" has the meaning set forth in Section 4.3.1.
 - 1.70 "HBV" means the hepatitis B virus.
 - 1.71 "HBV Target" means any HBV DNA, [***].
- 1.72 "IND" means (a) an investigational new drug application filed with the FDA for authorization to commence Clinical Studies and its equivalent in other countries or regulatory jurisdictions, and (b) all supplements and amendments that may be filed with respect to the foregoing.
 - **1.73** "Indemnification Claim Notice" has the meaning set forth in Section 9.3.
 - **1.74** "**Indemnified Party**" has the meaning set forth in Section 9.3.
 - **1.75** "**Indemnifying Party**" has the meaning set forth in Section 9.3.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.76 "Information" means all technical, scientific and other know-how and information, trade secrets, ideas, inventions, discoveries, knowledge, technology, means, methods, processes, practices, formulae, instructions, skills, techniques, procedures, experiences, ideas, designs, drawings, assembly procedures, computer programs, specifications, data, results and other information, including: biological, chemical, pharmacological, toxicological, pharmaceutical, physical and analytical, pre-clinical, clinical, safety, manufacturing and quality control data and information, including study designs and protocols, assays and biological methodology, in each case (whether or not confidential, proprietary, patented or patentable) in written, electronic or any other form now known or hereafter developed.
 - **1.77** "**Initial Term**" has the meaning set forth in Section 3.1.
 - 1.78 "Initiation" or "Initiate" means, with respect to a Clinical Study, the first dosing of the [***] human subject in such Clinical Study.
 - **1.79** "IP Assignee" has the meaning set forth in Section 6.1.5.
 - **1.80** "IP Assignor" has the meaning set forth in Section 6.1.5.
- 1.81 "IP Rights" means any and all legal means of establishing rights in and to ideas, inventions, discoveries, Information, data, databases, documentation, reports, materials, writings, designs, computer software, processes, principles, methods, techniques and other information, including Patents, trade secrets, trademarks, service marks, trade names, registered designs, design rights, copyrights (including rights in computer software and database rights) and any rights or property similar to any of the foregoing in any part of the world, whether registered or not, together with the right to apply for the registration of any such rights.
 - 1.82 "Joint Collaboration Program IP" means the Joint Collaboration Program Know-How and the Joint Collaboration Program Patents.
 - **1.83** "Joint Collaboration Program Know-How" has the meaning set forth in Section 6.1.3(a).
 - **1.84** "Joint Collaboration Program Patents" has the meaning set forth in Section 6.1.3(b).
- **1.85** "Joint Committee" means the JSC, the JRDC, or any subcommittee established to carry out the functions of the JSC or JRDC, including the Joint Tech Transfer Team.
 - 1.86 "Joint Research and Development Committee" or "JRDC" has the meaning set forth in Section 2.2.1.
 - **1.87** "Joint Steering Committee" or "JSC" has the meaning set forth in Section 2.1.1.
- **1.88** "Joint Tech Transfer Team" or "JTTT" means the Subcommittee established by the JSC pursuant to Section 2.1.3(f) to oversee, and provide guidance to the Parties regarding the implementation of the Technology Transfer Plan.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- **1.89** "Knowledge" means the actual knowledge of Precision's Chief Executive Officer, Chief Science Officer, Vice President of Business Development and Director of Intellectual Property, with internal due inquiry.
- **1.90** "Licensed Product" means any Gene Editing Therapy that incorporates one or more Gilead ARC Nuclease(s), in any and all forms, presentations, delivery systems, dosages, and formulations.
 - **1.91** "Licensed Product Family" has the meaning set forth in Section 5.4.2.
- 1.92 "[***] In-License Agreements" means any agreements between Precision or its Affiliates and a Third Party entered into as a Third Party License to [***] subject to Section 5.4.3(b) during the Term, in each case under which Precision has obtained rights to [***] for the Exploitation of the Licensed Products and which are used to Exploit the Licensed Products for Gilead.
 - **1.93** "Losses" has the meaning set forth in Section 9.1.
 - 1.94 "Major Market" means any of [***].
- 1.95 "Manufacture" and "Manufacturing" means all activities related to the manufacture, processing, filling, finishing, packaging, labeling, shipping, and holding of any Licensed Product, including process development, process qualification and validation, scale-up, pre-clinical, clinical and commercial manufacture and analytic development, Licensed Product characterization, stability testing, quality assurance, and quality control.
- 1.96 "Marketing Approval" means, with respect to a Licensed Product for a particular country, the grant of a Regulatory Approval that is required in such country from the competent Regulatory Authority to market and sell such Licensed Product in such country, including a BLA in the United States.
 - **1.97** "Net Sales" means [***].
 - **1.98** "Non-Prosecuting Party" means the Party that is not the Prosecuting Party.
- **1.99** "Other Infringement" means any alleged or threatened infringement of the Precision Patents or Joint Collaboration Program Patents, as applicable, by a Third Party and such alleged or threatened infringement is not a Competitive Infringement.
 - 1.100 "Party" and "Parties" has the meaning set forth in the preamble hereto.
 - 1.101 "Patent Challenge" has the meaning set forth in Section 5.4.8.
 - 1.102 "Patent Challenge Criteria" shall have the meaning set forth in Section 5.4.8.
- 1.103 "Patents" means (a) all national, regional and international patents and patent applications, including provisional patent applications; (b) all patent applications filed from such patents, patent applications or provisional applications or from an application claiming

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

priority from either of these, including divisionals, continuations, continuations-in-part, substitutions, provisionals, converted provisionals, and continued prosecution applications; (c) any and all patents that have issued or in the future issue from the foregoing patent applications described in clauses (a) and (b), including utility models, petty patents and design patents and certificates of invention; (d) any and all extensions or restorations by existing or future extension or restoration mechanisms, including revalidations, reissues, re-examinations and extensions (including any supplementary protection certificates and the like) of the foregoing patents or patent applications described in clauses (a), (b), and (c); and (e) any similar rights, including so-called pipeline protection, or any importation, revalidation, confirmation or introduction patent or registration patent or patent of addition to any such foregoing patent applications and patents.

- **1.104** "**Person**" means an individual, corporation, partnership, limited liability company, joint venture, association, trust, unincorporated organization or other entity or any governmental authority or Regulatory Authority.
- 1.105 "Phase I Clinical Study" means a human clinical trial of any product conducted during Phase 1 of a clinical investigation as defined in 21 C.F.R. 312.21(a) or corresponding foreign regulations.
- **1.106** "Phase Ib Clinical Study" means a Phase I Clinical Study within the HBV patient population that is designed to establish an initial indication of efficacy.
- **1.107** "Phase II Clinical Study" means a human clinical trial of any product conducted during Phase 2 of a clinical investigation as defined in 21 C.F.R. 312.21(b) or corresponding foreign regulations, including any such trial conducted as an open label clinical trial.
- 1.108 "Phase III Clinical Study" means a human clinical trial of any product on sufficient numbers of patients that is designed to demonstrate statistically that such product is safe and efficacious for its intended use and to define warnings, precautions and adverse reactions that are associated with such product in the dosage range to be prescribed, as described in 21 C.F.R. 312.21(c) or corresponding foreign regulations, and that is intended to support Marketing Approval of such product.
- **1.109** "PHSA" means the United States Public Health Service Act, as may be amended, or any subsequent or superseding law, statute or regulation.
 - 1.110 [***].
 - **1.111** "**Precision**" has the meaning set forth in the preamble hereto.
 - 1.112 "Precision Existing Patents" means the Patents listed in Schedule 1.112.
- **1.113** "Precision HBV Patents" means any Precision Patent that, without expanding the definition of Precision Patents, [***], including the Patents listed in <u>Schedule 1.113</u>, as such schedule may be updated by Precision during the Term in accordance with this Agreement.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 1.114 "Precision IP" means the Precision Know-How, the Precision Patents and, to the extent not included in the Precision Know-How and the Precision Patents, the ARCUS Assigned IP.
- 1.115 "Precision Know-How" means any and all Information to the extent Controlled by Precision or any of its Affiliates: (a) as of the Effective Date or at any time during the Collaboration Term and resulting from the conduct of the Collaboration Program, in each case that is necessary or reasonably useful for the Exploitation of any Licensed Product or any Gilead ARC Nuclease, or (b) at any time during the Term that Precision or its Affiliates elect to provide or disclose to Gilead under this Agreement, that is necessary or reasonably useful for the Exploitation of any Licensed Product or any Gilead ARC Nuclease; and in each case ((a) and (b)) excluding Joint Collaboration Program Know-How, ARCUS Technology, and any [***]. For clarity, Precision Know-How does not include any Patents.
- 1.116 "Precision Patents" means any and all Patents to the extent Controlled by Precision or any of its Affiliates (a) as of the Effective Date or at any time during the Collaboration Term and resulting from the conduct of the Collaboration Program, in each case that are necessary or reasonably useful for the Exploitation of any Licensed Product or any Gilead ARC Nuclease; (b) at any time during the Term that cover or claim Precision Know-How; or (c) at any time during the Term that cover or claim any Licensed Product or Gilead ARC Nuclease or are necessary for the Exploitation of Licensed Products or Gilead ARC Nucleases, in the case of Licensed Products under this clause (c) in the form in which such Licensed Products exist as of the end of the Collaboration Term (and including any Regulatory Authority-required modifications made thereto after the Collaboration Term) and in any form supplied under the Supply Agreement but excluding in all cases of clause (c) any Patents that claim or cover any formulation or [***] developed or in-licensed by Precision or its Affiliates after the Effective Date outside the Collaboration Program unless such Patent is a Formulation and Delivery Combination Patent, in which case only claims that cover the use, composition or production of such formulation or [***] apart from the combination shall be excluded; and in each of cases (a) through (c), including the Precision Existing Patents, but excluding any Joint Collaboration Program Patents, ARCUS Patents and any [***].
 - 1.117 "Proof of Concept Clinical Study" [***].
- **1.118** "Prosecuting Party" means the Party preparing, filing, prosecuting, maintaining, enforcing or defending the relevant Patent(s), as applicable, in exercise of its rights under, and in accordance, with ARTICLE 6.
 - **1.119** "Publications" has the meaning set forth in Section 7.5.
 - **1.120** "Quality Agreement" has the meaning set forth in Section 3.5.2.
 - 1.121 "Registrational Clinical Study" [***].
- 1.122 "Regulatory Approval" means, with respect to any jurisdiction, any and all approvals (including pricing and reimbursement approvals), licenses, registrations or

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

authorizations of any Regulatory Authority necessary or useful for the Exploitation of any Licensed Product in such jurisdiction, including, where applicable, (a) IND, Marketing Approval applications and supplements and amendments thereto; (b) Marketing Approvals and pre- and post-Marketing Approval marketing authorizations (including any prerequisite manufacturing approval or authorization related thereto); (c) labeling approval; and (d) technical, medical and scientific licenses.

- **1.123** "Regulatory Authority" means any applicable supra-national, federal, national, regional, state, provincial or local regulatory agencies, departments, bureaus, commissions, councils or other government entities regulating or otherwise having legal authority with respect to the Exploitation of Licensed Products in the Territory.
- 1.124 "Regulatory Documentation" means any and all (a) applications, registrations, licenses, authorizations and approvals (including all Regulatory Approvals), and non-clinical and clinical study authorization applications or notifications (including all supporting files, writings, data, studies and reports) prepared for submission to a Regulatory Authority or research ethics committee with a view to the obtaining or maintaining of any Regulatory Approval, (b) correspondence to or with the FDA or any other Regulatory Authority (including minutes and official contact reports relating to any communications with any Regulatory Authority), (c) pharmacovigilance databases, adverse drug experience reports and associated documents, and investigations of adverse drug experience reports, (d) manufacturing records, and (e) nonclinical, clinical and other data contained or referenced in or supporting any of the foregoing.
- 1.125 "Regulatory Exclusivity Period" means, with respect to each Licensed Product in any country in the Territory, a period of exclusivity (other than Patent exclusivity) granted or afforded by Applicable Law or by a Regulatory Authority in such country that confers exclusive marketing rights with respect to such Licensed Product in such country or prevents another Person from using, referencing or otherwise relying on data supporting the Marketing Approval for such Licensed Product without the prior written consent of the Marketing Approval holder, including regulatory data exclusivity, new chemical entity exclusivity, new use or indication exclusivity, new formulation exclusivity, pediatric exclusivity and orphan drug designations.
- 1.126 "Relevant Factors" means all factors that are relevant to the Development, Manufacture or Exploitation of a product, including its safety and efficacy, product profile, cost to develop, cost and availability of supply, the time required to complete Development, the competitiveness of the marketplace (including the proprietary position and anticipated market share of the product), the patent position with respect to such product (including the ability to obtain or enforce, or have obtained or enforced, such patent rights), the Third Party patent landscape relevant to the product, the regulatory structure involved, the likelihood of regulatory approval, the anticipated or actual profitability of the applicable product and other technical, commercial, legal, scientific, regulatory and medical considerations.
- **1.127** "Representatives" means, with respect to any Person, such Person's directors, officers, managers, employees, counsel, accountants, financial advisors, lenders and other agents and representatives.
- 1.128 "Reversion IP" means any Patents and Know-How that are Controlled by Gilead or any of its Affiliates as of the date of termination of this Agreement (in whole or in part,

and including, for clarity, Patents filed or issued at any later date covering or claiming applicable inventions conceived on or prior to such date) that are necessary for Precision to continue the Development, Manufacture, use or Exploitation of Licensed Products in the form existing as of the date of termination of this Agreement (and including any Regulatory Authority-required modifications made thereto after such date), excluding any Patent or Know-How that covers or claims, or in the case of Know-How, relates specifically to, (i) any Active Component of any Licensed Product that is not a Gilead ARC Nuclease or (ii) any use of a Gilead ARC Nuclease in combination with any such other Active Component of such Licensed Product.

- **1.129** "Reversion Patents" means, with respect to any particular Licensed Product, Patents within Reversion IP that come to be Controlled by Gilead or any of its Affiliates as a result of activities conducted, under the Collaboration Program or in connection with this Agreement, [***].
- 1.130 "Royalty Term" means, with respect to each Licensed Product, on a country-by-country basis, the period commencing on the date of First Commercial Sale of such Licensed Product by Gilead, its Affiliate or Sublicensee in such country until the later of (a) the date on which there is no longer a Valid Claim covering the manufacture, use or sale of such Licensed Product in such country; (b) expiration of the Regulatory Exclusivity Period for such Licensed Product in such country; and (c) ten (10) years following the date of First Commercial Sale of the first Licensed Product in such country.
- 1.131 "Senior Officer" means, with respect to Gilead, its Chief Scientific Officer or his or her designee, and with respect to Precision, its Chief Executive Officer or his or her designee.
 - **1.132** "Subcommittee" means any Joint Committee other than the JSC.
- **1.133** "Sublicensee" means a Third Party to which Gilead has granted a sublicense under the licenses granted to Gilead hereunder to Exploit a Licensed Product under Section 4.3.1, but excluding Distributors.
 - **1.134** "Supplied Product" has the meaning set forth in Section 3.5.1.
 - **1.135** "Supply Agreement" has the meaning set forth in Section 3.5.2.
 - **1.136** "**Technology Transfer**" has the meaning set forth in Section 3.6.2.
 - **1.137** "**Technology Transfer Plan**" has the meaning set forth in Section 3.6.2.
 - **1.138** "Term" has the meaning set forth in Section 10.1.
 - **1.139** "Territory" means all countries and territories of the world.
 - 1.140 "Third Party" means any Person other than Gilead, Precision and their respective Affiliates.
 - **1.141** "Third Party Claims" has the meaning set forth in Section 9.1.
 - **1.142** "[***]" has the meaning set forth in Section 5.4.3(b).
 - **1.143** "Third Party License" has the meaning set forth in Section 5.4.3.
- **1.144** "United States" or "U.S." means the United States of America and its territories and possessions (including the District of Columbia and Puerto Rico).

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

1.145 "Valid Claim" means (a) a claim of an issued and unexpired Patent included within the Precision Patents or Joint Collaboration Program Patents which has not been abandoned, cancelled or held permanently revoked, unenforceable or invalid by a decision of a court or other governmental agency of competent jurisdiction unappealed within the time allowed for appeal, or which has not been admitted to be invalid or unenforceable through reissue or disclaimer or otherwise; [***].

ARTICLE 2 COLLABORATION MANAGEMENT

2.1 Joint Steering Committee.

- 2.1.1 **Formation.** Within [***] after the Effective Date, the Parties shall establish a joint steering committee (the "**Joint Steering Committee**" or "**JSC**").
- 2.1.2 **Membership.** The JSC shall consist of two (2) representatives from each of the Parties, each with the requisite experience and seniority to enable such person to make decisions on behalf of the Parties with respect to the issues falling within the jurisdiction of the JSC. The initial JSC members from each Party are set forth on <u>Schedule 2.1.2</u>. From time to time, each Party may substitute one or more of its representatives to the JSC on written notice to the other Party. Each Party shall select from its representatives a co-chairperson for the JSC. From time to time, each Party may change the representative who will serve as its co-chairperson on written notice to the other Party.
- 2.1.3 **Specific Responsibilities during the Collaboration Term.** During the Collaboration Term, the JSC shall oversee the Collaboration Program, and shall in particular:
- (a) monitor and coordinate the activities of the Parties under the Collaboration Program, including overseeing the JRDC, the JTTT, and any other Subcommittees and facilitating communications between the Parties with respect to the Development of the Licensed Products;
- (b) except for the initial budget included in the Collaboration R&D Plan that is to be executed by the Parties contemporaneously with the execution of this Agreement, discuss and facilitate the Parties' agreement on a reasonable budget (the "Collaboration Budget") for the tasks to be completed in each six-month period of the Collaboration Term and for any adjustments to such tasks, subject to Section 3.2.4;
 - (c) approve any amendments to the Collaboration R&D Plan in accordance with Section 3.2.2;
- (d) review and discuss each Party's written reports, including the results of the Development activities, provided to the JSC pursuant to Section 3.8.3;
- (e) approve subcontractors proposed to be used by Precision for the purposes of performing "material" services (as the term "material" is used in Section 4.3.2) in connection with the Collaboration Program, such approval not to be unreasonably withheld, conditioned or delayed, pursuant to Section 4.3.2;

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (f) establish Subcommittees, including a Joint Tech Transfer Team, as appropriate, to carry out its functions and to establish the rules governing and the responsibilities of the Joint Tech Transfer Team;
 - (g) resolve disputes that may arise in any Subcommittee;
- (h) without limiting clause (g) above, resolve disputes that may arise between the Parties or the JTTT regarding the content of the Technology Transfer Plan;
- (i) discuss and consider optimal technologies or methodologies for delivery, Manufacture and administration of Licensed Products, such as [***] and other potential Third Party Licenses pursuant to Section 5.4.3(a), provided however that [***], and provided further that Precision shall timely (A) provide Gilead through the JSC with such information with respect to any [***] that it intends to in-license from a Third Party in reasonable detail to enable Gilead to understand the reasons for the potential selection of such [***], (B) reasonably respond to Gilead's questions relating thereto and (C) consider in good faith Gilead's feedback with respect to such selection; and
 - (j) perform such other functions as may be assigned to the JSC hereunder.
- 2.1.4 **Specific Responsibilities following the Collaboration Term.** The JSC shall automatically be disbanded immediately after the First Commercial Sale of the first Licensed Product. During the period after the Collaboration Term and prior to such First Commercial Sale, the JSC shall serve only (a) as a forum for sharing and discussing information with respect to the Technology Transfer, Manufacture, Development and other Exploitation of the Licensed Products and (b) if the Technology Transfer has not been completed, to resolve disputes between the Parties or the JTTT.

2.2 Joint Research and Development Committee.

2.2.1 **Formation.** Within [***] after the Effective Date, the Parties shall establish a joint research and development committee (the "**Joint Research and Development Committee**" or "**JRDC**"). The JRDC shall consist of two (2) representatives from each of the Parties, each with the requisite experience and seniority to enable such person to make decisions on behalf of the Parties with respect to the issues falling within the jurisdiction of the JRDC. The initial JRDC members from each Party are set forth on <u>Schedule 2.2.1</u>. From time to time, each Party may substitute one or more of its representatives to the JRDC on written notice to the other Party.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

2.2.2 Specific Responsibilities during the Collaboration Term. The JRDC shall:

- (a) provide guidance to the Parties on the implementation of the Collaboration Program;
- (b) propose amendments to the Collaboration R&D Plan to the JSC for its review and approval in accordance with Section 3.2.2;
- (c) discuss the Regulatory Documentation prepared by Gilead pursuant to Section 3.7.1(a);
- (d) discuss the reports to be provided by Precision pursuant to Section 3.7.1(b), including the chemistry, manufacturing and control (CMC) reports; and
 - (e) perform such other functions as may be assigned to the JRDC hereunder.

For clarity, the JRDC shall not have the authority to modify the Collaboration R&D Plan.

2.2.3 **Specific Responsibilities after the Collaboration Term and Disbandment**. The JRDC shall be disbanded and have no further responsibilities or authority under this Agreement upon the expiry of the Collaboration Term.

2.3 General Provisions Applicable to Committees.

2.3.1 Meetings and Minutes.

- (a) Unless otherwise agreed to by the Parties, the JSC shall meet quarterly during every twelve (12) month period after the Effective Date until expiration of the Collaboration Term, and thereafter the JSC shall meet annually until disbanded pursuant to Section 2.1.4, and the JRDC shall meet monthly during every twelve (12) month period after the Effective Date until the expiration of the Collaboration Term; *provided that*, the JSC and the JRDC shall each meet in person once during each such twelve (12) month period after the Effective Date if mutually agreed. The location of such in person meetings shall alternate between locations designated by Gilead and locations designated by Precision.
- (b) The Alliance Manager for each Party shall be responsible for calling meetings with notice provided a reasonable time in advance of such meeting. Each Party shall make all proposals for agenda items and shall provide all appropriate information with respect to such proposed items a reasonable time in advance of the applicable meeting; *provided that*, under exigent circumstances requiring input by the Joint Committee, a Party may provide its agenda items to the other Party within a shorter period of time in advance of the meeting, or may propose that there not be a specific agenda for a particular meeting, so long as the other Party consents to such later addition of such agenda items or the absence of a specific agenda for such meeting, such consent not to be unreasonably withheld, conditioned or delayed. At the conclusion of each meeting, the Parties will decide which Party shall prepare and circulate for review and approval of the Parties minutes of each meeting within ten (10) Business Days after the meeting. The Parties shall agree on the minutes of each meeting promptly, but in no event later than thirty (30) days after such meeting.
- 2.3.2 **Procedural Rules.** Each Joint Committee shall have the right to adopt such standing rules as shall be necessary for its work, to the extent that such rules are not

inconsistent with this Agreement. A quorum of the Joint Committee shall exist whenever there is present at a meeting at least one (1) representative appointed by each Party. Subject to the proviso in Section 2.3.1(a), representatives of the Parties on a Joint Committee may attend a meeting either in person or by telephone, video conference or similar means in which each participant can hear what is said by, and be heard by, the other participants. Representation by proxy shall be allowed. Each Joint Committee shall take action by consensus of the representatives present at a meeting at which a quorum exists, with each Party having a single vote irrespective of the number of representatives of such Party in attendance, or by a written resolution signed by at least one (1) representative appointed by each Party. Employees or consultants of a Party that are not representatives of such Party on a Joint Committee may attend meetings of such Joint Committee with advance written notice to the other Party; *provided, however*, that such attendees (a) shall not vote or otherwise participate in the decision-making process of the Joint Committee, and (b) are bound by obligations of confidentiality and non-disclosure equivalent to those set forth in ARTICLE 7.

2.3.3 **Decision-making and Dispute Resolution.** If the JRDC or any other Subcommittee cannot, or does not, reach consensus on an issue arising within the scope of its responsibilities within a period of [***], then either Party may refer the matter to the JSC for resolution and a special meeting of the JSC may be called for such purpose. If during the Collaboration Term, and thereafter for so long as the Technology Transfer has not been completed, the JSC cannot, or does not, reach consensus on an issue, including any dispute arising in the JRDC or a Subcommittee, within a period of [***] after referral to the JSC, then the JSC shall refer such dispute to the Senior Officers for resolution. If such Senior Officers cannot resolve such dispute within [***] of it being referred to them, then, subject to the remaining provisions of this Section 2.3.3, and without limiting Gilead's diligence obligations under Sections 3.3 and 3.4 of this Agreement, Gilead shall have final decision-making authority with respect to such matter; provided that, Gilead may not exercise such authority (i) to require Precision to license any particular IP Rights for use in the Collaboration Program beyond those already contemplated herein, (ii) to expand the scope of the Collaboration Program beyond HBV Targets, (iii) to expand the scope of the definition of Precision IP, (iv) to accelerate the timelines for Precision Development activities, (v) to establish or modify the Collaboration Budget (for the avoidance of doubt, this clause (v) shall not be construed to limit Gilead's right to modify the Collaboration R&D Plan in a manner that requires the Parties to agree on a modified Collaboration Budget pursuant to Section 3.2.4), (vi) to modify the Collaboration R&D Plan to add additional activities to the Collaboration R&D Plan that Precision reasonably demonstrates would cause Precision's costs or resources for meeting the work plans and timelines set forth in the Collaboration R&D Plan to exceed the Collaboration Budget or the Gilead Funding Commitment, (vii) to dictate the content of the Technology Transfer Plan, or (viii) to select a [***] other than [***] for any Licensed Product. Notwithstanding the foregoing, subject to the terms of this Agreement, Precision shall have final decision-making authority with respect to: (a) the design, creation, and optimization of ARC Nucleases to be proposed to Gilead as Gilead

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARC Nucleases other than site selection for the HBV Target, which shall be within Gilead's final decision-making authority; (b) the selection of specific [***] for any Licensed Products for [***] or such substitute [***] selected by the JSC; and (c) the selection of subcontractors for conducting activities assigned to Precision under the Collaboration R&D Plan other than those subcontractors performing "material" services in connection with the Collaboration R&D Plan, which subcontractors shall be subject to approval of the JSC pursuant to Section 4.3.2; provided however, that with respect to the foregoing (a), Precision shall timely (A) provide Gilead with such information based on which Precision makes its decisions with respect to such design, creation, and optimization of ARC Nucleases under the Collaboration R&D Plan in reasonable detail to enable Gilead to understand the reasons for such decisions, (B) reasonably respond to Gilead's questions relating thereto and (C) consider in good faith Gilead's feedback with respect to such decisions. Following the Collaboration Term, the JSC shall have no decision-making authority with respect to any matter arising under this Agreement and, for clarity, subject to the terms and conditions of this Agreement (including Section 3.4 and Section 4.5), Gilead shall have sole decision-making authority with respect to any matter relating to the Technology Transfer, Manufacture, Development or Exploitation of the Licensed Products.

- 2.3.4 **Limitations on Authority.** Each Party shall retain the rights, powers, and discretion granted to it under this Agreement and no such rights, powers, or discretion shall be delegated to or vested in a Joint Committee unless such delegation or vesting of rights is expressly provided for in this Agreement or the Parties expressly so agree in writing. No Joint Committee shall have the power to amend, modify, or waive compliance with this Agreement, which may only be amended or modified as provided in Section 11.10 or compliance with which may only be waived as provided in Section 11.10.
- 2.3.5 **Alliance Manager.** Each Party shall appoint a person(s) who shall oversee contact between the Parties for all matters between meetings of the Joint Committees and shall have such other responsibilities as the Parties may agree in writing after the Effective Date (each, an "**Alliance Manager**"). Each Party may replace its Alliance Manager at any time by notice in writing to the other Party.

ARTICLE 3 DEVELOPMENT AND REGULATORY

3.1 Collaboration Program. Subject to the terms and conditions of this Agreement, the Parties shall collaborate on a research and pre-clinical program to construct, optimize and develop one or more Gene Editing Therapy(ies) made using the ARCUS Technology that targets the HBV Target (the "Collaboration Program") and conduct the Development activities set forth in the collaboration research and development plan described in Section 3.2.1 (such plan, as amended from time to time, the "Collaboration R&D Plan"). The Collaboration Program shall commence on the Effective Date and will, unless otherwise mutually agreed by the Parties in writing, continue until the third anniversary of the Effective Date or, if earlier, the later of (a) acceptance by the FDA or other competent foreign Regulatory Authority of [***] of the first IND filing for the first Licensed Product and (b) the satisfactory

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

completion (as mutually agreed upon by the Parties) of all tasks and activities required under the Collaboration R&D Plan ("Initial Term"); provided that, if either (a) or (b) has not occurred by the third anniversary of the Effective Date, Gilead, at its option, may extend the Initial Term (or then-current term, as the case may be) for one or more additional six (6) month periods until such time as the later of (a) and (b) has occurred, by serving a written notice(s) to Precision; provided that, Gilead pays to Precision the funding payment for such additional period(s) in accordance with Section 5.1 (the Initial Term together with any such extensions, the "Collaboration Term").

3.2 Collaboration R&D Plan.

- 3.2.1 **Initial Collaboration R&D Plan.** The initial Collaboration R&D Plan as agreed to by the Parties as of the date hereof, including the applicable Collaboration Budget, has been signed and acknowledged by each Party and copies of such signed document have been exchanged between the Parties concurrently with the execution of this Agreement.
- 3.2.2 **Amendments.** During the Collaboration Term, either Party, directly or through its representatives on the JRDC or the JSC, may propose any amendment to the Collaboration R&D Plan, including in light of changed circumstances, and, if the Collaboration Term is extended pursuant to Section 3.1, the Parties shall update the Collaboration R&D Plan to include such additional period(s). Any and all such amendments or updates shall be subject to approval by the JSC, subject to the dispute resolution procedures set forth in Section 2.3.3.
- 3.2.3 **Contents.** The Collaboration R&D Plan shall include, without limitation: (a) the Development activities to be conducted by each Party pursuant to the Collaboration Program; (b) Development goals; (c) the estimated timelines for such activities; (d) the anticipated costs of and resources for the Development activities, which shall, with respect to Precision's Development activities under the Collaboration R&D Plan, not exceed the Gilead Funding Commitment and shall reflect the FTE Rate and the payment schedule set forth in Section 5.1.1; and (e) the preliminary design parameters for pre-clinical studies to be performed under the Collaboration R&D Plan.
- 3.2.4 Collaboration Budget. With respect to each six-month period of the Collaboration Term, Precision shall propose to the JSC a Collaboration Budget for the performance of the activities in the Collaboration R&D Plan including required FTEs at the FTE Rate, materials, Third Party services and any other cost items. In addition, Precision will propose to the JSC an update to the Collaboration Budget to reflect any adjustments made to the Collaboration R&D Plan by the JSC; provided however, that the first Collaboration Budget shall be included in the initial Collaboration R&D Plan to be executed by the Parties contemporaneously with the execution of this Agreement, as described in Section 3.2.1. The Parties, through the JSC, shall use reasonable efforts to agree promptly on each proposed Collaboration Budget. For the avoidance of doubt, the Collaboration Budget must be mutually agreed by Precision and Gilead prior to conducting the relevant activities in the Collaboration Program and will not be subject to final decision-making authority of either Party under Section 2.3.3.
- 3.2.5 **Inconsistency.** If the terms of the Collaboration R&D Plan contradict, or create inconsistencies or ambiguities with, the terms of this Agreement, then the terms of this Agreement shall govern.

- 3.3 Performance of the Collaboration R&D Plan. Precision and Gilead shall each use Commercially Reasonable Efforts to carry out or to cause to be carried out the activities assigned to it under the Collaboration R&D Plan in good scientific manner, in compliance with all Applicable Law and in accordance with the timelines set forth therein, and shall, in particular, use Commercially Reasonable Efforts to perform the activities within the time periods set forth in the Collaboration R&D Plan. In the event of any delay in any payments required under Section 5.1.1 beyond the due date for such payment, and without limiting any right of Precision under Section 10.2, Precision may suspend performance of its activities assigned to it under the Collaboration R&D Plan for so long as such delay continues. Neither Party makes any representation, warranty or guarantee that the Development activities conducted under the Collaboration R&D Plan will be successful or that any particular result will be achieved.
- 3.4 Development Following Expiry of the Collaboration Term. Following expiry of the Collaboration Term, Gilead, at its sole cost, shall be solely responsible for the Development and other Exploitation of the Licensed Products throughout the Territory. Gilead shall provide to Precision an annual high-level summary (consistent with Section B of Schedule 3.8.3) prepared in good faith of its plans to Develop the Licensed Products. Gilead will use Commercially Reasonable Efforts to Develop and otherwise Exploit a Licensed Product in each of the Major Markets, and to comply with such plans, in accordance with the terms of this Agreement and in compliance with all applicable Laws. If Gilead and its Affiliates and Sublicensees have collectively permanently ceased all Development and other Exploitation of Licensed Products under this Agreement, Gilead shall deliver notice to Precision and such notice will be deemed a termination of this Agreement pursuant to Section 10.3.1.

3.5 Clinical Supply of the Licensed Products.

- 3.5.1 With respect to any Licensed Product developed under the Collaboration Program and specified in the Supply Agreement (and any such additional Licensed Products as the parties may mutually agree), until the later of the completion of a [***] for a Licensed Product and the completion of the Technology Transfer for such Licensed Product, Precision shall be solely responsible for manufacturing and supplying clinical supplies of such Licensed Product(s) ("Supplied Product") for use by or on behalf of Gilead under this Agreement, and thereafter Gilead shall assume all manufacturing and supply activities for such Licensed Product (including, for clarity, commercial manufacture and supply).
- 3.5.2 Within ninety (90) days of identifying the Gilead ARC Nucleases to be used in any Licensed Product, the Parties shall discuss in good faith and enter into (a) a reasonable and customary supply agreement pursuant to which Precision shall manufacture and supply and Gilead shall purchase Supplied Products incorporating such Gilead ARC Nucleases, for use in Clinical Studies and under which Gilead would purchase clinical requirements of such Licensed Products for a price equal to [***] of COGS (the "Supply Agreement") and (b) a reasonable and customary quality agreement that shall set forth the terms and conditions upon which Precision shall conduct its quality activities in connection with such supply (the "Quality Agreement"), in each case (a) and (b), in a form reasonably acceptable to Gilead and Precision.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

In the event that by [***] prior to the expected date of the first IND filing for the first Licensed Product the Parties do not mutually agree to the Supply Agreement and the Quality Agreement, Gilead shall have the right to require Precision to conduct an early Technology Transfer to Gilead or its designee so that Gilead or such designee can commence the Manufacture of the Licensed Products.

3.6 Technology Transfer.

- 3.6.1 Upon Gilead's request, and no later than upon the expiration of the Collaboration Term, Precision shall promptly disclose and transfer to Gilead such Precision Know-How and such Information that Precision Controls under the [***] In-License Agreements or other Third Party Licenses that are licensed to Gilead hereunder as is required pursuant to the Technology Transfer Plan; *provided that*, before the expiration of the Collaboration Term, Gilead shall only receive such Precision Know-How as is necessary or reasonably useful for Gilead to meet its obligations under the Collaboration R&D Plan or as the Parties may otherwise reasonably agree in good faith.
- 3.6.2 The Parties shall, no later than upon the expiration of the Collaboration Term, agree to a technology transfer plan with reasonable limitations on access to Precision personnel (including reasonable caps on hours of access) and facilities, for the full technology transfer of the Manufacture of the Licensed Products to any facility of Gilead or its designee, approved in advance in writing by Precision (such approval not to be unreasonably withheld, conditioned or delayed, and will not be required for facilities located in [***], including the transfer of all Precision Know-How relating to the Manufacture of the Supplied Products and any such Information that Precision Controls under the [***] In-License Agreements or other Third Party Licenses that may be obtained under this Agreement, in a form reasonably acceptable to the Parties ("**Technology Transfer Plan**"). Any further transfer by Gilead or its designee following the initial Technology Transfer shall be subject to the approval process set forth above in this Section 3.6.2. The Parties agree that ARCUS Technology will not be transferred to Gilead or its designee under this Agreement. Any disputes regarding the content of the Technology Transfer Plan shall be resolved in accordance with Section 2.3.3. Following expiration of the Collaboration Term, Precision shall conduct such technology transfer to Gilead or its designee in accordance with the Technology Transfer Plan (the "**Technology Transfer**"), under the oversight and guidance of the JTTT (if any).
- 3.6.3 Without limiting the foregoing, for a period of [***] following completion of the Technology Transfer Plan, upon Gilead's request made reasonably in advance of the commencement of anticipated Manufacture by Gilead, Precision shall provide Gilead or its designee with such Precision Know-How relating to the Manufacture of the Supplied Products supplemental to the Technology Transfer Plan (i.e., items inadvertently omitted from the Technology Transfer Plan) as is reasonably necessary for Gilead or its designee to commence the Manufacture of the Licensed Products as permitted by this Agreement and the Supply Agreement. For the avoidance of doubt, Gilead acknowledges that after the [***] period described in this Section 3.6.3, Precision shall have no obligation to provide Gilead any such additional support.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 3.6.4 In the event the Supply Agreement is entered into by the Parties pursuant to Section 3.5.2, the Supply Agreement shall remain in effect until the completion of the Technology Transfer and shall either cover the Parties' respective rights and obligations with respect to the Technology Transfer or the Parties will enter into a separate written technology transfer agreement.
- 3.6.5 All Technology Transfer responsibilities, costs and expenses of Precision and its Affiliates will be set forth in the Collaboration R&D Plan and Collaboration Budget and considered Development activities funded under Section 5.1.1, except for such activities conducted following filing of an IND for any Licensed Product. For any such post-IND activities, (a) Precision will [***] and (b) [***].

3.7 Regulatory Matters.

3.7.1 Regulatory Activities.

- (a) As between the Parties, subject to Section 3.7.1(b) and Section 3.7.1(d), Gilead shall be responsible for (i) preparing and filing all Regulatory Documentation, including the IND filing for the first Licensed Product, (ii) obtaining and maintaining all Regulatory Approvals for the Licensed Products and (iii) conducting communications with the Regulatory Authorities for the Licensed Products. Gilead shall prepare and file all clinical Regulatory Documentation under the Collaboration Program, including the first IND filing for the first Licensed Product, in consultation with the JRDC.
- (b) As between the Parties, during the Collaboration Term, Precision shall be responsible for preparing all non-clinical and chemistry, manufacturing and control (CMC) reports, in each case, as reasonably required by Gilead, for inclusion in the first IND filing for the first Licensed Products. Precision shall prepare all such reports, and provide Gilead with copies of any such reports, in each case, in a timely manner to permit Gilead to make such IND filing without delay, which shall then be discussed in consultation with the JRDC. Without limiting the foregoing, Precision shall support Gilead as may be reasonably necessary in connection with Gilead's preparation of clinical Regulatory Documentation under the Collaboration Program during the Collaboration Term pursuant to Section 3.7.1(a). The responsibilities, costs and expenses of Precision and its Affiliates under this Section 3.7.1(b) during the Collaboration Term will be set forth in the Collaboration R&D Plan and Collaboration Budget and considered Development activities funded under Section 5.1.1. Thereafter, for any such activities, (i) Precision will [***] and (ii) Gilead shall [***].
- (c) All Regulatory Documentation generated under this Agreement, including in the course of conducting the Collaboration Program, shall be owned by Gilead and held in the name of Gilead (or its designee).
- (d) Except as otherwise provided in the Supply Agreement or the Quality Agreement, with respect to each Licensed Product (i) Gilead shall have sole

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

responsibility for attending all meetings (whether occurring in person, by telephone or other remote means) with applicable Regulatory Authorities; provided that, Gilead, where permitted by Applicable Law, shall permit a reasonable number of Precision employees to attend (A) each pre-IND meeting for the Licensed Product, (B) the end of phase 2 meeting for the Licensed Product with the FDA, and (C) any other meeting with FDA or EMA (or other competent Regulatory Authority in the United Kingdom) if such meeting has one or more items on the agenda directed toward the safety or delivery of ARC Nucleases; and (ii) Gilead shall have the sole right and responsibility to correspond with applicable Regulatory Authorities; provided that, Gilead, where permitted by Applicable Law, will provide draft communications with the FDA and EMA (or other competent Regulatory Authority in the United Kingdom) to Precision for review and comment to the extent it relates to the Gilead ARC Nuclease, the ARCUS Technology, ARCUS Assigned IP or any [***] to the extent licensed or sublicensed by Precision to Gilead, and will consider Precision's comments in good faith before submitting the communications to the FDA or EMA (or other competent Regulatory Authority in the United Kingdom). If either Party or its Affiliates or subcontractors receive any material written correspondence or other communication from the Regulatory Authorities in the Major Markets regarding (x) in the case of Precision as the Party receiving such correspondence from the Regulatory Authorities, the Licensed Products or any components thereof, or (y) in the case of Gilead as the Party receiving such correspondence from the Regulatory Authorities, the Gilead ARC Nuclease, any [***] in-licensed by Precision or the ARCUS Technology or ARCUS Assigned IP, such Party shall provide the other Party with access to or copies of all such material written or electronic correspondence promptly after its receipt.

3.7.2 **Pharmacovigilance.** To the extent safety reporting is required by Applicable Law, upon the request of either Party, the Parties shall enter into an agreement to cover the exchange of adverse event safety data in a mutually agreed format in order to monitor the safety of the Licensed Products and to meet reporting requirements with any applicable Regulatory Authority.

3.8 Records and JSC Reporting.

3.8.1 Precision shall, and shall ensure that its subcontractors shall, maintain records in sufficient detail and in good scientific manner appropriate for patent and regulatory purposes, and in compliance with Applicable Law, which shall be complete and accurate and shall properly reflect all work done and results achieved in the performance of its Development activities under the Collaboration Program which shall record only such activities and shall not include or be commingled with records of activities outside the scope of this Agreement. Such records shall be retained by Precision for at least [***] after the expiration or termination of this Agreement, or for such longer period as may be required by Applicable Law. Upon the request of Gilead, Precision shall provide copies of the records it has maintained pursuant to this Section 3.8.1 to Gilead. Gilead shall maintain such records and the information disclosed therein in confidence in accordance with ARTICLE 7.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 3.8.2 Gilead shall have the right, during normal business hours and upon reasonable notice not more than twice annually during the Collaboration Term without Precision's consent (not to be unreasonably withheld, conditioned or delayed), to inspect and copy all records of Precision maintained pursuant to Section 3.8.1, which may at Gilead's reasonable request, be at Precision's facilities, or as permitted by Precision's agreements with its subcontractors, at the facilities of any such subcontractor. Gilead shall maintain such records and the information disclosed therein in confidence in accordance with ARTICLE 7.
- 3.8.3 At each JSC meeting during the Collaboration Term, each Party's members on the JSC shall provide a written report to the JSC of its activities with respect to the Collaboration Program conducted since the last JSC meeting, including a reasonable summary of the results of such activities and the progress of the Collaboration Program. In addition, Precision will provide Gilead with a semi-annual report summarizing its activities under the Collaboration Program, which report shall be in the form set forth on Schedule 3.8.3 and shall include disclosure of Precision Know-How required by Section 6.1.1 that has not been previously disclosed. The JSC may request of a Party any such additional written reports describing its activities with respect to the Collaboration Program, as it determines necessary or useful in its discretion.
- 3.8.4 Without limiting any other rights of Precision hereunder, in the event a Regulatory Authority requests, in connection with a request to obtain regulatory approval for a product (other than a Licensed Product) containing an ARC Nuclease (other than a Gilead ARC Nuclease) that Precision provide clinical data Controlled by Gilead or any of its Affiliates or Sublicensees relating to (a) the safety of (i) the ARCUS Technology generally, or (ii) ARCUS Assigned IP, or (b) if relevant to such other product, a Gilead ARC Nuclease or its delivery, Precision may request that Gilead provide such information within Gilead's Control (and Gilead will use Commercially Reasonable Efforts to obtain and provide such information from its Sublicensees) to Precision solely for use in response to such Regulatory Authority's request with respect to such other product and Gilead shall promptly provide such information for provision to such Regulatory Authority.
- 3.8.5 Without limiting any other rights of Gilead hereunder, in the event a Regulatory Authority requests, in connection with a request to obtain Regulatory Approval for a Licensed Product, that Gilead provide clinical data Controlled by Precision or any of its Affiliates or sublicensees relating to (a) the safety of the Gilead ARC Nuclease or its delivery, or (b) if relevant to a Licensed Product, a Gilead ARC Nuclease or its delivery, Gilead may request that Precision provide such information to Gilead solely for use in response to such Regulatory Authority's request with respect to the Licensed Product and Precision shall promptly provide such information within Precision's Control (and will use Commercially Reasonable Efforts to obtain and provide such information from its sublicensees) for provision to such Regulatory Authority.

ARTICLE 4 GRANT OF RIGHTS AND COVENANTS

- **4.1 Grants to Gilead.** Subject to the terms and conditions of this Agreement and continuing unless and until terminated pursuant to ARTICLE 10, Precision hereby grants to Gilead:
- 4.1.1 a worldwide, royalty-bearing, non-transferable (except pursuant to Section 11.3), exclusive license (or sublicense, as the case may be), with the right to grant sublicenses in accordance with Section 4.3, under the Precision IP, [***] (subject to Section 4.1.3) and Precision's interest in the Joint Collaboration Program IP, to Exploit the Licensed Products (including the Gilead ARC Nucleases incorporated in or for incorporation into a Licensed Product) in the Field in the Territory, including to perform its Development activities under the Collaboration R&D Plan; and
- 4.1.2 a worldwide, royalty-bearing, non-transferable (except pursuant to Section 11.3), non-exclusive license, with the right to grant sublicenses in accordance with Section 4.3, under the ARCUS IP, to the extent necessary for Gilead to Exploit the Licensed Products (including the Gilead ARC Nucleases incorporated in or for incorporation into a Licensed Product) in the Field in the Territory.
- 4.1.3 The license under Section 4.1.1 with respect to [***] shall be (A) effective only upon execution of the applicable [***] license granted to Precision, and (B) exclusive solely as between the Parties. Any licenses granted by Precision under Section 4.1.1 under any [***] license are subject to and limited by any limitation, restriction or additional terms set forth in the agreement under which Precision or its Affiliates obtained rights under such [***] license from such Third Party. Precision shall (i) use Commercially Reasonable Efforts to ensure the terms of such license (a) confer IP Rights sufficient to grant the license to Gilead under Section 4.1.1, (b) do not result in any material expansion of Gilead's obligations under this Agreement, and (c) do not require Gilead to make any additional representations or warranties or to provide any additional indemnities materially different from those set forth herein, and (ii) disclose to Gilead in writing all terms of such agreement which materially limit or otherwise materially impact Gilead's rights or obligations with respect to such [***]. In connection with such disclosure of terms, and prior to entering into any such license, Precision shall provide Gilead reasonable opportunities to review and comment on the draft during negotiations of the draft with the applicable Third Party (which may be reasonably redacted by Precision, including to redact provisions relating to financials, IP Rights not relevant to this Agreement, and targets not relevant to this Agreement), reasonably consider any comments made by Gilead and provide to Gilead a copy of the unexecuted final version of such redacted agreement no later than [***]. Upon receipt of such unexecuted final version, Gilead may elect by written notice to Precision to (1) accept inclusion of the IP Rights licensed to Precision under such agreement in the licenses granted to Gilead herein or (2) reject such inclusion in good faith, in which case, for clarity, Gilead may enter into a license for such [***] or similar [***] directly with the applicable licensor in accordance with Section 5.4.3(b). In the event Gilead elects to accept such license entered into by Precision, Gilead shall comply with the terms or obligations, as disclosed to Gilead in writing in such unexecuted final version shared with Gilead pursuant to the above prior to Gilead's election pursuant to the immediately preceding sentence, that Precision is required to impose under any such agreement for [***]. Precision shall make Commercially Reasonable Efforts to ensure that the license rights to [***] will include rights for Gilead to modify or improve such technology. Gilead acknowledges that, notwithstanding Precision making such Commercially Reasonable Efforts, the license rights to [***] may not

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

include rights for Gilead to modify or improve such technology, or if such rights are included, such rights may be subject to allocation of IP Rights in such modifications or improvements that differ from those contained in this Agreement. For the sake of clarity, if Gilead has accepted its inclusion as set forth above, upon execution the [***] licensed by Precision under the [***] In-License Agreements will fall within the scope of licenses set forth in Section 4.1.1.

- **4.2 Grants to Precision.** Subject to the terms and conditions of this Agreement, Gilead (on behalf of itself and its Affiliates) hereby grants to Precision:
- 4.2.1 a worldwide, fully paid-up, royalty-free, non-exclusive license, with the right to grant sublicenses to subcontractors (that have been, if required, approved by Gilead) in accordance with Section 4.3, under the Gilead Dual IP, Gilead Patents and Gilead Know-How (but Gilead shall not be required to disclose any Know-How except as otherwise specifically provided herein), to perform, or have performed by such subcontractors, its Development activities under the Collaboration R&D Plan during the Collaboration Term; and

4.2.2 [***].

4.3 Sublicensing and Subcontracting.

- 4.3.1 Gilead shall have the right: (a) to grant sublicenses under Section 4.1 through multiple tiers of sublicenses to any Affiliate or Third Party; and (b) to subcontract to any Affiliate or Third Party the performance of any of its obligations under this Agreement. Notwithstanding the foregoing, during the Collaboration Term Gilead shall not grant any sublicenses to a Third Party with respect to any Development of a Licensed Product in or for a Major Market without Precision's prior written consent. Gilead shall provide Precision with written notice of any sublicense under this Agreement within [***] after its execution. [***].
- 4.3.2 Precision shall have the right: (a) to subcontract any of its Development activities under the Collaboration Program to any Affiliate or Third Party; (b) to grant sublicenses under Section 4.2.1 to its subcontractors; and (c) subject to the last sentence of this Section 4.3.2, to grant sublicenses under Section 4.2.2 through multiple tiers of sublicenses to any Affiliates or Third Parties; *provided however*, Precision shall only subcontract activities or services under the Collaboration Program that are designated as "material" in the Collaboration R&D Plan to a vendor that has not been pre-approved by Gilead in the Collaboration R&D Plan after receiving written consent from Gilead, such consent not to be unreasonably withheld, conditioned or delayed. Precision shall provide Gilead with written notice of any sublicense it grants under any issued or published Patents, or Patents that have otherwise been disclosed to Precision, in each case licensed by Gilead to Precision under Section 4.2.2 within [***] after Precision is aware that it has sublicensed such a Patent of Gilead's. In the event Gilead denies consent for such subcontracting, then if Precision requests a recommendation from Gilead for an alternative subcontractor, Gilead shall provide to Precision the name and contact information of at least one subcontractor approved for the conduct of the applicable activity or service within

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

[***] of Precision's request for consent. Notwithstanding anything to the contrary in this Agreement, Precision's right to sublicense under clause (c) above shall be subject to Gilead's prior written consent if [***].

- 4.3.3 Each sublicense or subcontract agreement entered into by a Party under this Section 4.3 shall be consistent with the applicable terms and conditions of this Agreement, including the confidentiality provisions of ARTICLE 7 and the intellectual property provisions of ARTICLE 6, and the applicable Party shall be fully responsible for any breach of this Agreement by any of its Affiliates, Sublicensees or subcontractors. In addition, to the extent required by the Cellectis Agreement, each sublicense granted by Gilead under any Precision Patent must grant the same scope of rights for all Precision Patents and each sublicense granted by Gilead under any ARCUS Patent must grant the same scope of rights for all ARCUS Patents.
- **4.4** Retention of Rights. Subject to Section 4.5, Precision retains the right to (a) practice the Precision IP and its interest in the Joint Collaboration Program IP to exercise its rights and perform its obligations under this Agreement (in each case in a manner consistent with this Agreement), including the Collaboration Program, and under any Supply Agreement or Quality Agreement, (b) conduct research related to the ARCUS Technology and ARCUS Assigned IP; and (c) practice and license ARCUS Patents, ARCUS Assigned IP, Precision Patents and Precision Know-How outside the scope of the licenses granted to Gilead under Section 4.1.1. Gilead hereby retains the right to practice all intellectual property licensed by Gilead to Precision under this Agreement for any and all purposes. Except as set forth herein, neither Party shall acquire any license or other intellectual property interest, by implication or otherwise, under or to any trademarks, patents or patent applications, Information, or other IP Rights owned or Controlled by the other Party. Neither Party grants to the other Party any rights, licenses or covenants in or to any IP Rights, whether by implication, estoppel, or otherwise, other than the license rights that are expressly granted under this Agreement. Each Party shall not, and shall not permit any of its Affiliates or Sublicensees to, practice any Patents or Information licensed to it by the other Party outside of the scope of the license granted to it under this Agreement (other than such practices as would be otherwise permitted by applicable safe harbors under Applicable Law). Without limiting the foregoing, nothing in this Agreement shall be deemed to grant Gilead any right to access or receive any ARCUS Technology or any right to design, create, select, or optimize any ARC Nucleases using the ARCUS Technology or to otherwise use the ARCUS Technology as a genome engineering tool. Except as expressly set forth in this Article 4, the foregoing licenses from Precision to Gilead do not include any rights under the ARCUS Patents or ARCUS Technology. Neither Party grants hereunder any rights with respect to other products or therapies with which a Licensed Product may be combined.

4.5 Exclusivity.

4.5.1 The Parties acknowledge both their possession of confidential or proprietary information and the highly competitive nature of the industry in which they operate and, accordingly, agree that, in consideration of entering into this Agreement and the promises contained herein, in the Territory, [***] Precision shall not, during the Term [***] (a) conduct, participate in, or enable or directly fund, [***] other than [***] in accordance with this

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Agreement, or (b) license, authorize, or appoint or otherwise enable any Third Party (other than subcontractors to the extent permitted herein) to engage in any of the activities set forth in clause (a) of this Section 4.5.1. Notwithstanding the foregoing, in no event shall [***] be prohibited or otherwise restricted by this Section 4.5.1 from engaging in, directly or indirectly, or funding or otherwise enabling, [***]. Each Party acknowledges and agrees that (A) this Section 4.5 has been negotiated by the Parties, (B) the geographical and time limitations on activities set forth in this Section 4.5 are reasonable, valid and necessary in light of the Parties' circumstances and necessary for the adequate protection of the business of Exploiting the Licensed Products and (C) [***]. If, notwithstanding the foregoing, a court of competent jurisdiction determines that the restrictions set forth in this Section 4.5 are too broad or otherwise unreasonable under Applicable Law, including with respect to duration, geographic scope or space, the court is hereby requested and authorized by the Parties to revise this Section 4.5 to include the maximum restrictions allowable under Applicable Law.

- 4.5.2 Notwithstanding anything to the contrary in this Agreement, the restrictions on Precision set forth in Section 4.5.1 shall not apply to any Acquirer of Precision [***].
- 4.5.3 Notwithstanding Section 4.5.1, [***] acquires a Competitor, continuation of the relevant Competitive Program(s) shall not be a breach of this Agreement; *provided that* (i) [***], and (ii) [***].

4.6 Existing In-License Agreements.

- 4.6.1 Cellectis Patents. Gilead acknowledges and agrees that rights under certain ARCUS Patents and/or Precision Patents are licensed to Precision by Cellectis S.A. (the "Cellectis Patents") under that certain Patent Cross-License Agreement between Cellectis S.A. and Precision dated January 23, 2014 (the "Cellectis Agreement"), and, notwithstanding any exclusive license granted to Gilead under this Agreement, (a) Cellectis S.A. retains rights under the Cellectis Patents and is not restricted from granting rights to Third Parties under the Cellectis Patents, (b) any licenses and rights granted by Precision to Gilead under the Cellectis Patents are granted only within the permissible scope of sublicenses granted under the Cellectis Agreement, and (c) pursuant to the Cellectis Agreement, Cellectis S.A. retains non-exclusive rights under certain ARCUS Patents and/or Precision Patents identified in the Cellectis Agreement, which may be further sublicensed by Cellectis S.A. without Precision control or consent. Gilead acknowledges and agrees that any exercise of any right by Cellectis S.A, or by any Third Party through Cellectis S.A, under the Cellectis Agreement shall not constitute a breach of this Agreement by Precision.
- 4.6.2 **Duke IP.** Gilead acknowledges and agrees that any licenses and rights granted by Precision to Gilead under the Duke IP are granted subject to the terms and conditions of the Duke Agreement, including Duke's right to practice under the Duke IP for its own internal, non-commercial, educational, research and clinical purposes, and subject to the rights of the United States Government and applicable limitations under 37 C.F.R. § 401, Public Law 96-517 and Public Law 98-620 resulting from the United States Government's funding of

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

research leading to creation of the Duke IP. Without limiting the foregoing, Gilead agrees to comply with any obligations resulting from such government rights with respect to its practice of the Duke IP (if any) under this Agreement. Duke shall be a third party beneficiary of the Agreement to the extent its terms and conditions apply or relate to the Duke IP.

4.7 Preservation of Existing In-License Agreements and [***] In-License Agreements. To the extent relating to the Gilead ARC Nucleases or the Licensed Products, Precision shall, and shall procure that its Affiliates shall, maintain all licenses to all [***], including the Existing In-License Agreements and any [***] In-License Agreements, in full force and effect in accordance with their terms and conditions and keep Gilead reasonably informed in this regard. Without limiting the foregoing and Section 5.4.9, Precision shall not (a) commit any acts or permit the occurrence of any omissions that would cause breach or termination of any license to [***], including Existing In-License Agreements or any [***] In-License Agreements or (b) amend or otherwise modify or permit to be amended or modified, any license to [***], including the Existing In-License Agreements or any [***] In-License Agreements, in any way that would prejudice Gilead's rights under this Agreement or its ability to continue to Exploit Licensed Products.

ARTICLE 5 PAYMENTS AND RECORDS

5.1 Payments during the Collaboration Program.

- 5.1.1 **Payments.** In consideration of Precision's agreement to perform its Development activities under the Collaboration R&D Plan, Gilead shall pay Precision (a) [***] for the first six (6) month period of the Collaboration Term, (b) [***] for the second six (6) month period of the Collaboration Term thereafter (in aggregate, the "**Gilead Funding Commitment**"); provided that, (i) if the Collaboration Budget at any time is less than the Gilead Funding Commitment for any six-month period, then Precision shall hold the difference for use later in the Collaboration Term if required based on the Collaboration Budget (i.e. if the Collaboration Budget exceeds the Gilead Funding Commitment for a six month period), and (ii) if the Collaboration Budget at any time exceeds the Gilead Funding Commitment for any six-month period, then after application of any amounts held by Precision under clause (i) Gilead shall pay Precision the difference in the then-current period as an advance from later portions of the Gilead Funding Commitment in the Collaboration Term. For clarity, the aggregate Collaboration Budget for the entire Collaboration Term shall not exceed the aggregate Gilead Funding Commitment without the written consent of both Parties.
- 5.1.2 **Invoicing.** On the Effective Date, Precision shall issue to Gilead an invoice for the first [***] tranche referred to in Section 5.1.1 and Gilead shall pay such tranche to Precision within [***] of receipt of invoice by Gilead. Thereafter Precision shall issue

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

an invoice for each [***] tranche or [***] tranche, as applicable, referred to in Section 5.1.1 at least [***] in advance of the relevant six (6) month period. All invoices described in the immediately preceding sentence or this Section 5.1.2 shall be due within [***] of receipt by Gilead. Invoices shall be in accordance with the template set forth on Schedule 5.1.2.

5.2 Development and Regulatory Milestones.

5.2.1 In partial consideration of the rights granted under this Agreement and subject to Section 5.2.2 below, Gilead shall pay to Precision the following one-time milestone payments after the achievement of the following corresponding milestone events with respect to the first Licensed Product to achieve the applicable milestone in the Territory during the Term. Gilead shall notify Precision in writing no later than [***] following the achievement of a milestone event and shall pay to Precision the milestone payment corresponding to such milestone event [***] of receipt of an invoice from Precision.

Development and Regulatory Milestone Event	Milestone Payment
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]
Maximum total:	

5.2.2 Each milestone payment in Section 5.2.1 shall be payable only upon the first achievement of such milestone and no amounts shall be due for subsequent or repeated achievements of such milestone, whether for the same or a different Licensed Product. The maximum aggregate amount payable by Gilead pursuant to Section 5.2.1 is [***].

5.3 Commercial Milestones.

5.3.1 In partial consideration of the rights granted hereunder and subject to Section 5.3.2 below, Gilead shall pay to Precision the following one-time milestone payments after the achievement of the following milestone events with respect to the first occurrence of the Licensed Products achieving the milestone in the Territory during the Term (including any wind-down period following the end of the Term). Gilead shall notify Precision in writing no later than [***] following the end of the Calendar Quarter in which the commercial milestone event occurs and shall pay to Precision the milestone payment corresponding to such milestone event within [***] of receipt of an applicable invoice from Precision.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Commercial Milestone Event	Milestone Payment
The first time aggregate global Net Sales of Licensed Products in a Calendar Year are equal to or greater	
than [***] and less than [***]	[***]
The first time aggregate global Net Sales of Licensed Products in a Calendar Year are equal to or greater	
than [***] and less than [***]	[***]
The first time aggregate global Net Sales of Licensed Products in a Calendar Year are equal to or greater	
than [***] and less than [***]	[***]
The first time aggregate global Net Sales of Licensed Products in a Calendar Year are equal to or greater	
than [***]	[***]
Maximum total:	[***]

- 5.3.2 Each milestone payment in Section 5.3.1 shall be payable only once upon the first achievement of such milestone and no amounts shall be due for subsequent or repeated achievements of such milestone. The maximum aggregate amount payable by Gilead pursuant to Section 5.3.1 is [***].
- 5.3.3 For the avoidance of doubt, for the purposes of this Section 5.3, all Net Sales of Licensed Products shall be aggregated globally for all sales made by Gilead or any of its Affiliates or Sublicensees in any and all forms, presentations, delivery systems, dosages, and formulations for purposes of determining whether the above Net Sales thresholds have been achieved. For clarity, all Licensed Products shall be aggregated for the purposes of this Section 5.3 even if they do not have the same Active Components as one another.

5.4 Royalties.

5.4.1 In partial consideration of the rights granted hereunder and subject to Sections 5.4.2 through 5.4.8, during the applicable Royalty Term, on a Licensed Product-by-Licensed Product and country-by-country basis, Gilead shall pay to Precision royalties on the Net Sales of such Licensed Product at the rates set forth below, as determined by the aggregate annual global Net Sales of all Licensed Products in the Licensed Product Family for such Licensed Product:

Annual Global Net Sales of the Applicable Licensed Product	Royalty Rate
For the portion of Net Sales of such Licensed Product Family in the Territory in any given Calendar Year that is less	
than [***]	[***]
For the portion of Net Sales of such Licensed Product Family in the Territory in any given Calendar Year that is equal	
to or greater than [***] and less than [***]	[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Annual Global Net Sales of the Applicable Licensed Product	Royalty Rate
For the portion of Net Sales of such Licensed Product Family in the Territory in any given Calendar Year that is equal	
to or greater than [***] and less than or equal to [***]	[***]
For the portion of Net Sales of such Licensed Product Family in the Territory in any given Calendar Year that is	
greater than [***]	[***]

- 5.4.3 **Third Party Licenses.** With respect to the Exploitation of any Licensed Product under this Agreement, and without prejudice to any other right of each Party, Gilead and Precision may obtain licenses from Third Parties for any rights for the Exploitation of any Licensed Product in accordance with the following provisions of this Section 5.4.3 (each, a "**Third Party License**"):
- (a) If during the Collaboration Term, either Party considers that a license(s) to additional technology, including [***], or IP Rights of a Third Party are necessary or reasonably useful to Develop, Manufacture or otherwise Exploit a Licensed Product under this Agreement, such Party may refer the matter to the JSC for discussion. If the JSC concludes that such a Third Party License is required or reasonably useful, the Parties shall negotiate in good faith which Party shall enter into the Third Party License and the allocation between the Parties of responsibility for the costs and expenses of obtaining such Third Party License, including any royalty reductions under the payments otherwise payable to Precision under Section 5.4.1. Notwithstanding the foregoing, after the Collaboration Term, if rights to Patents controlled by any Third Party cover or claim the applicable Licensed Product or its Manufacture, then Gilead may negotiate and obtain a Third Party License from such Third Party for Gilead, its Affiliates or Sublicensees to Exploit such Licensed Product in such country in accordance with this Agreement. If, pursuant to this Section 5.4.3(a), Gilead has obtained a Third Party License, and owes a royalty under such Third Party License for sales of a Licensed Product in a particular country, then Gilead shall have the right to reduce the royalty payments otherwise payable to Precision under Section 5.4 based on such sales by up to [***] of such payments under such Third Party License; provided that, no royalty payment to Precision for a Licensed Product hereunder shall be reduced, pursuant to this Section 5.4.3(a), to less than [***] of the royalty payment that would otherwise be due to Precision in the absence of a reduction pursuant to this Section 5.4.3(a).
- (b) Notwithstanding Section 5.4.3(a) above, Precision shall be responsible for obtaining any licenses from Third Parties to Gilead ARC Nuclease delivery technology, including [***], used in the Collaboration Program in the Licensed Products ("[***]"), [***], and following the execution of any such Third Party License shall notify Gilead promptly of the same. Precision shall not incorporate or use any such [***] in the Manufacture

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

of the Licensed Products unless Precision has the right to sublicense such [***] to Gilead consistent with Section 4.1.3 and transfers such [***] to Gilead as set forth in Section 3.6.2. In the event Precision fails to obtain a Third Party License with respect to the [***] that Precision uses in the Collaboration Program in the Licensed Products, or fails to so transfer such [***], or if the terms of such proposed Third Party License as disclosed under Section 4.1.3 are not acceptable to Gilead, then Gilead may do so independently and may in such case [***].

- (c) For clarity, to the extent an agreement includes rights with respect to any active therapeutic ingredient, having a different mode of action than the Gilead ARC Nuclease or a different active component than the Gilead ARC Nuclease, it is not a Third Party License.
- 5.4.4 **No Valid Claim.** In the event that, at the time a Licensed Product is sold in a country, there is no Valid Claim in such country with respect to such Licensed Product and the Regulatory Exclusivity Period has expired in such country with respect to the Licensed Product but the Royalty Term remains in effect, then for the purposes of calculating the royalties owed based on the sale of such Licensed Product in such country under Section 5.4.1 at that time, in such country the royalties that would otherwise be owed and payable under Section 5.4.1 based on such sale shall be reduced by [***]. The calculation of the royalty reduction under this Section 5.4.4 shall be conducted separately for each Licensed Product.
- 5.4.5 **Biosimilar Products.** On a Licensed Product-by-Licensed Product basis, if in any country in the Territory during the Royalty Term for a Licensed Product a Biosimilar Product launches with respect to such Licensed Product in such country, then the royalties that would otherwise be owed and payable under Section 5.4.1 for the Net Sales of such Licensed Product in such country shall be reduced by [***], from the date of launch of such Biosimilar Product in such country until the end of the Royalty Term for such Licensed Product in such country. In the event that Gilead does not learn of such launch until after royalties are paid, Gilead shall be entitled to such adjustment retroactively to such launch date in the form of a credit against future royalty obligations of Gilead under this Agreement.
- 5.4.6 **Royalty Floor.** Under no circumstances will the application of the reductions in Section 5.4.3, Section 5.4.4 and Section 5.4.5 together ever result in a reduction of the royalties payable by Gilead to Precision to less than [***] of the amounts specified in Section 5.4.1.
- 5.4.7 **Compulsory Licensing; Generic Sublicensees.** If a Regulatory Authority requires Gilead or any of its Affiliates or Sublicensees to grant a compulsory license to a Third Party permitting such Third Party to make and sell a Licensed Product in a country in the Territory, or Gilead sublicenses to a Generic Sublicensee, then all amounts received from the compulsory licensee in consideration for grant of the license and received from the Generic Sublicensee in consideration for the sublicense, including any royalties so received, shall be [***]. For the avoidance of doubt, the reductions in Section 5.4.3, Section 5.4.4 and Section 5.4.5 shall not apply to reduce amounts payable to Precision pursuant to this Section 5.4.7.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

5.4.8 Patent Challenge. [***]

- 5.4.9 Payments under Existing In-License Agreements and [***] In-License Agreements. The Parties acknowledge and agree that, without limiting the right of Gilead to take a license to [***] set forth in Section 5.4.3(b), Precision shall be solely responsible for, and shall promptly discharge, any and all payments payable pursuant to the Existing In-License Agreements and [***] In-License Agreements. Notwithstanding anything in this Agreement to the contrary, in the event Precision terminates any of the Existing In-License Agreements and therefore Gilead enters into a license directly with the licensor or otherwise makes payments pursuant to any of the Existing In-License Agreements in order to maintain its rights as a sublicensee under such Existing In-License Agreement, [***].
- 5.4.10 **Royalty Payments and Reporting.** Gilead shall calculate all amounts payable to Precision pursuant to Section 5.4.1 at the end of each Calendar Quarter. Gilead shall pay to Precision the royalty amounts due with respect to a given Calendar Quarter within [***] after the end of such Calendar Quarter. Each payment of royalties due to Precision shall be accompanied by a statement of the amount of gross sales and Net Sales (including applicable deductions) of each Licensed Product, in each country of the Territory during the applicable Calendar Quarter (including such amounts expressed in local currency and as converted to Dollars), a calculation of the amount of royalty payment due on such Net Sales for such Calendar Quarter, and details on the determination of incremental royalty rates including the aggregation of Licensed Products into Licensed Product Families.
- **5.5 Mode of Payment.** All payments to Precision under this Agreement shall be made by deposit of Dollars in the requisite amount to the following bank account of Precision or such other account as Precision may from time to time designate by notice to Gilead:

[***]

ACCOUNT NAME: [***]

5.6 Currency. For the purpose of calculating any sums due under, or otherwise reimbursable pursuant to, this Agreement (including the calculation of Net Sales expressed in currencies other than Dollars), Gilead shall convert any amount expressed in a foreign currency into Dollars equivalents using its, its Affiliate's or Sublicensee's standard conversion methodology consistent with GAAP.

5.7 Taxes.

5.7.1 A Party making payments to the other Party under this Agreement shall make such payments without deduction or withholding for taxes except to the extent that any such deduction or withholding is required by Applicable Law in effect at the time of payment.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 5.7.2 Any tax required to be withheld on amounts payable under this Agreement shall promptly be paid by the applicable paying Party on behalf of the other Party to the appropriate governmental authority or Regulatory Authority, and such paying Party shall furnish the other Party with proof of payment of such tax within [***]. Any such tax required to be withheld shall be an expense of and borne by such other Party.
- 5.7.3 The Parties shall cooperate with respect to all documentation required by any taxing authority or reasonably requested by either Party to secure a reduction in the rate of applicable withholding taxes.
- 5.7.4 If the applicable paying Party had a duty to withhold taxes in connection with any payment it made to the other Party under this Agreement but such paying Party failed to withhold, and such taxes were assessed against and paid by such paying Party, then the other Party shall indemnify and hold harmless such paying Party from and against such taxes (including interest, but not including any related penalties). If such paying Party makes a claim under this Section 5.7.4, it shall comply with the obligations imposed by Section 5.7.2 as if such paying Party had withheld taxes from a payment to the other Party.
- 5.7.5 Notwithstanding the foregoing, if Gilead assigns its rights and obligations hereunder to an Affiliate or successor pursuant to Section 11.3, and if such Affiliate or successor shall be required by applicable law to withhold any additional nonrecoverable taxes from or in respect of any amount payable under this Agreement as a result of such assignment, then any such amount payable under this Agreement shall be increased to take into account the additional taxes withheld so that, after making all required withholdings, Precision receives an amount equal to the sum it would have received had no such assignment been made.
- **5.8 Financial Records.** Gilead shall, and shall cause its Affiliates and Sublicensees to, keep complete and accurate books and records pertaining to its gross sales and Net Sales of the Licensed Products, in sufficient detail to calculate all amounts payable hereunder and to verify compliance with its obligations under this Agreement. Such books and records shall be retained by Gilead and its Affiliates and Sublicensee until the later of (a) three (3) years after the end of the period to which such books and records pertain, and (b) the expiration of the applicable tax statute of limitations (or any extensions thereof), or for such longer period as may be required by Applicable Law.
- **5.9** Audit. At the request of Precision, Gilead shall, and shall cause its Affiliates and Sublicensees to, permit an independent auditor designated by Precision and reasonably acceptable to the Gilead, at reasonable times and upon reasonable notice, to audit the books and records maintained by Gilead pursuant to Section 5.8 to ensure the accuracy of all reports and payments made hereunder. Such examinations may not (a) be conducted for any Calendar Quarter more [***] after the end of such Calendar Quarter, (b) be conducted more than [***] or (c) be repeated for any Calendar Quarter. Except as provided below, the cost of this audit shall be borne by the Precision, unless the audit reveals a variance of more than [***] from

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

the reported amounts, in which case Gilead shall bear the cost of the audit. If such audit concludes that (x) the amount Gilead paid to the Precision for a given period exceeded the amount that was payable to Precision, then Precision shall reimburse Gilead for such variance within [***] of the date on which such audit was completed by Precision, or (y) the amount Gilead paid to the Precision for a given period was less than the amount that was payable to Precision, Gilead shall reimburse Precision for such variance within [***] after the date on which such audit is completed by Precision.

5.10 Confidentiality. The auditing party shall treat all information subject to review under Sections 5.8 and 5.9 in accordance with the confidentiality provisions of ARTICLE 7.

ARTICLE 6 INTELLECTUAL PROPERTY

6.1 Ownership of Intellectual Property.

6.1.1 **Ownership of Arising IP; Disclosure.** As between the Parties, subject to Sections 6.1.2 and 6.1.3, each Party shall own and retain all right, title and interest in and to any and all Information that is conceived, discovered, developed or otherwise made by or on behalf of such Party or its Affiliates under this Agreement, and any and all Patents and other IP Rights with respect thereto. Gilead shall promptly disclose to Precision in writing during the Collaboration Term and shall cause its Affiliates and make Commercially Reasonable Efforts to cause its Sublicensees to so disclose during the Collaboration Term, the conception, discovery, development or making of any Gilead Dual Know-How described in clause (a) of the definition of Gilead Dual Know-How, and shall make available to Precision, in the form that Gilead has available (including by providing copies thereof), all such Gilead Dual Know-How in connection with each JSC meeting. During the Collaboration Term, Precision shall disclose to Gilead in writing and shall cause its Affiliates and make Commercially Reasonable Efforts to cause its sublicensees to so disclose, the conception, discovery, development or making of any Precision Know-How described in clause (a) of the definition of Precision Know-How and shall make available to Gilead, in the form that Precision has available (including by providing copies thereof), all such Precision Know-How, in connection with each JSC Meeting and semi-annual report described in Section 3.8.3. Within [***] following the end of the Collaboration Term, each Party shall disclose to the other Party in writing any additional Gilead Dual Know-How or Precision Know-How required to be disclosed by this Section 6.1.1 that has not previously been disclosed pursuant to this Section 6.1.1.

6.1.2 Ownership of ARCUS Assigned IP; [***].

- (a) [***]
- (b) [***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 6.1.3 **Ownership of Joint Collaboration Program IP.** As between the Parties, each Party shall own an equal, undivided interest in any and all:
- (a) Information which is conceived, discovered, developed or otherwise made jointly by or on behalf of Gilead or its Affiliates, on the one hand, and Precision or its Affiliates or subcontractors, on the other hand, in the performance of the Collaboration Program [***] ("Joint Collaboration Program Know-How"); and
- (b) Patents that claim or cover the Joint Collaboration Program Know-How ("Joint Collaboration Program Patents") or any other IP Rights with respect to the Joint Collaboration Program Know-How.

Subject to the license granted under Sections 4.1.1, the Parties' obligations under Section 4.5, and the payment obligations in ARTICLE 5, each Party shall have the right to Exploit the Joint Collaboration Program Know-How and the Joint Collaboration Program Patents without a duty of seeking consent or accounting to the other Party; *provided however*, that [***]. Each Party shall promptly disclose to the other Party in writing and shall cause its Affiliates to so disclose, the conception, discovery, development or making of any Joint Collaboration Program Know-How or Joint Collaboration Program Patents by or on behalf of such Party or its Affiliates and shall make available to the other Party, in whatever form such other Party may reasonably request (including by providing copies thereof), all such Joint Collaboration Program Know-How within ten (10) Business Days of such generation.

- 6.1.4 **Assignment Obligation.** Each Party shall cause all Persons who perform Collaboration Program activities for such Party, including subcontractors, to be under an obligation to assign their rights in any IP Rights resulting therefrom to such Party, except where Applicable Law requires otherwise and except in the case of governmental, not-for-profit and public institutions which have standard policies against such an assignment (in which case a suitable license, or right to obtain such a license, shall be obtained).
- 6.1.5 **IP Transfer.** The licenses and other rights granted in this Agreement are intended to be and will be binding on any permitted assignee or other transferee of any right, title, or interest with respect to any IP Rights licensed hereunder. Without limiting the generality of the foregoing, and without limiting anything else in this Agreement, if a Party ("**IP Assignor**") assigns or otherwise transfers any right, title or interest to a Third Party ("**IP Assignee**") with respect to any of the IP Rights licensed by IP Assignor to the other Party hereunder, IP Assignor will cause the IP Assignee to agree in writing that such rights are subject to the licenses and other rights granted under or with respect to such IP Rights pursuant to this Agreement, and such assignment or other transfer shall only be effective if such IP Assignee does so agree.
- 6.1.6 **United States Law.** The determination of whether information and inventions are conceived, discovered, developed, or otherwise made by a Party for the purpose of allocating proprietary rights (including Patent, copyright or other IP Rights) therein, shall, for purposes of this Agreement, be made in accordance with Applicable Law in the United States as such law exists as of the Effective Date irrespective of where such conception, discovery, development or making occurs.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

6.2 Maintenance and Prosecution of Patents.

6.2.1 Precision Patents.

- (a) Subject to Section 6.2.4, Precision shall have the first right, but not the obligation, through the use of internal or outside counsel, to prepare, file, prosecute, and maintain the Precision Patents worldwide, at [***] sole cost and expense [***]. Precision shall keep Gilead consulted in a timely fashion with respect to the strategy of such preparation, filing, prosecution and maintenance, and in the event of a disagreement with respect to such strategy in relation to a Precision HBV Patent, Gilead shall have final decision-making authority in accordance with Section 6.2.4 and subject to Section 6.2.5; provided, however, that Gilead may relinquish such final decision-making authority with respect thereto, in which case Gilead shall no longer be obligated to pay for such activities with respect to the applicable Patent.
- (b) In the event that Precision decides not to prepare, file, prosecute, or maintain a Precision Patent pursuant to Section 6.2.1(a), Precision shall provide reasonable prior written notice to Gilead of such intention (which notice shall, in any event, be given no later than [***] prior to the next deadline for any action that may be taken with respect to such Patent). Gilead shall thereupon have the option, in its sole discretion, to assume the control and direction of the preparation, filing, prosecution, and maintenance of such Patent (unless Precision's decision to allow the lapse of any provisional patent application, or abandonment of any patent application, is intended to be in favor of trade secret protection or in favor of a continuation), at [***] cost and expense. The rights and obligations of the Parties shall not otherwise be affected by such assumption of control by Gilead. For the purpose of this ARTICLE 6, "prosecution" shall include any post-grant proceeding including supplemental examination, post grant review proceeding, inter parties review proceeding, patent interference proceeding, opposition proceeding and re-examination.
- (c) Gilead acknowledges and agrees that Precision has no rights or responsibility for preparing, filing, prosecuting or maintaining the Cellectis Patents.
- (d) In the event that Precision decides to allow the lapse of any patent application in favor of trade secret protection as described in Section 6.2.1(b) or Section 6.2.2(b), Precision shall first discuss such decision with Gilead in good faith (including the potential strategic rationale for not disclosing Information in patent filings and preserving such Information as a trade secret). If, following such discussion, Precision proceeds with allowing any such lapse, Precision shall [***].
 - (e) Prior to Precision filing a patent application that would constitute a Precision HBV Patent, [***].

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

6.2.2 Joint Collaboration Program Patents.

- (a) Precision shall have the first right, but not the obligation, through the use of internal or outside counsel, to prepare, file, prosecute, and maintain the Joint Collaboration Program Patents worldwide, with the cost and expense of such activities to be [***].
- (b) In the event that Precision decides not to prepare, file, prosecute, or maintain a Joint Collaboration Program Patent pursuant to Section 6.2.2(a), Precision shall provide reasonable prior written notice to Gilead of such intention (which notice shall, in any event, be given no later than [***] prior to the next deadline for any action that may be taken with respect to such Patent), and Gilead shall thereupon have the option, in its sole discretion, to assume the control and direction of the preparation, filing, prosecution, and maintenance of such Patent (unless Precision's decision to allow the lapse of any provisional patent application, or abandonment of any patent application, is intended to be in favor of trade secret protection or in favor of a continuation), with the cost and expense of such activities to be [***]. In the event Gilead chooses not to assume such control and direction, [***] for the cost and expense of such activities. [***] the costs and expenses of the filing, prosecution, and maintenance of any Joint Collaboration Program Patents (but not its license rights) and, [***] all such costs and expenses, then such other Party shall be deemed the sole owner thereof and the relinquishing Party shall assign its interest in such Joint Collaboration Program Patents to such other Party.

6.2.3 Gilead Dual Patents.

- (a) Gilead shall have the first right, but not the obligation, through the use of internal or outside counsel, to prepare, file, prosecute, maintain and otherwise control the Gilead Dual Patents worldwide, at [***]. Gilead shall provide Precision with notice of the filing of, and a copy of, any patent applications for a Gilead Dual Patent.
- (b) In the event that Gilead decides not to prepare, file, prosecute, or maintain a Gilead Dual Patent pursuant to Section 6.2.3(a), Gilead shall provide reasonable prior written notice to Precision of such intention (which notice shall, in any event, be given no later than [***] prior to the next deadline for any action that may be taken with respect to such Gilead Dual Patent), and Precision shall thereupon have the option, in its sole discretion, to assume the control and direction of the preparation, filing, prosecution, and maintenance of a Patent that covers the patentable subject matter of the Gilead Dual Know-How to the extent relating to ARCUS Technology, ARCUS Assigned IP or ARC Nucleases (unless Gilead's decision to allow the lapse of any provisional patent application, or abandonment of any patent application, is intended to be in favor of trade secret protection or in favor of a continuation), at [***]. The rights and obligations of the Parties shall not otherwise be affected by such assumption of control by Precision.
- 6.2.4 **Cooperation; Precision HBV Patents.** With respect to a Precision Patent, Joint Collaboration Program Patent, or Gilead Dual Patent, the Prosecuting Party shall keep the Non-Prosecuting Party reasonably informed of all steps with regard to such preparation, filing, prosecution, and maintenance, including by providing the Non-Prosecuting

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Party with a copy of material communications to and from any patent authority regarding such Patent, and by providing the drafts to the Non-Prosecuting Party of any material filings or responses to be made to such patent authorities sufficiently in advance of submitting such filings or responses so as to allow for a reasonable opportunity for the Non-Prosecuting Party to review and comment thereon; provided that, with respect to the Precision HBV Patents, and subject to Section 6.2.1(a) and Section 6.2.5, Precision shall obtain Gilead's prior written approval for such activities, and in the event of a dispute Gilead shall have the final decision-making authority with respect to such matter. If Gilead does not provide a response within [***] after a request from Precision with respect to a filing or response, approval shall be deemed to have been provided by Gilead. The Prosecuting Party shall consider in good faith any comments of the Non-Prosecuting Party with respect to such drafts of the Prosecuting Party and with respect to strategies for filing and prosecuting the applicable Precision Patent, Joint Collaboration Program Patent or Gilead Dual Patent; provided that, with respect to the Precision HBV Patents (subject to Section 6.2.1(a) and Section 6.2.5) and Joint Collaboration Program Patents, Precision shall incorporate any and all good faith comments of Gilead, and with respect to Gilead Dual Patents, Gilead shall consider in good faith any comments provided by Precision. Notwithstanding the foregoing, the Prosecuting Party shall promptly inform the Non-Prosecuting Party of any adversarial patent office proceeding or sua sponte filing, including a request for, or filing or declaration of, any interference, opposition, or reexamination relating to the applicable Precision Patents, Joint Collaboration Program Patents or Gilead Dual Patents. The Parties shall thereafter consult and the Prosecuting Party shall consider in good faith all comments, requests and suggestions provided by the Non-Prosecuting Party. Each Party shall provide the other Party all reasonable assistance and cooperation in the patent prosecution efforts under this Section 6.2, including providing any necessary powers of attorney and executing any other required documents or instruments for such prosecution. When a Party assumes the responsibilities for the prosecution and maintenance of a Patent under Section 6.2.1(b), 6.2.2(b) or 6.2.3(b), the other Party shall promptly transfer to such Party the patent prosecution files for such Patent and provide reasonable assistance in the transfer of the prosecution responsibilities. The Party assuming such prosecution and maintenance responsibilities shall have the right to engage its own counsel to do so. In connection with the activities set forth in Section 6.2.1, 6.2.2, and 6.2.3 the Prosecuting Party shall consult with the Non-Prosecuting Party as to the strategy for the prosecution and maintenance of the Precision Patents, Joint Collaboration Program Patents and Gilead Dual Patents, as applicable.

6.2.5 [***].

6.2.6 **Joint Research Agreement.** Each Party shall have the right to make an election under 35 U.S.C. 102(c) when exercising its rights under this ARTICLE 6 without the prior written consent of the other Party; *provided that*, the electing Party shall notify the other Party in advance of any such election and shall consider comments provided by the other Party with respect to such election in good faith. With respect to any such election, the Parties shall coordinate their activities with respect to any submissions, filings, or other activities in support thereof. The Parties acknowledge and agree that this Agreement is a "joint research agreement" as defined in 35 U.S.C. 100(h).

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

6.2.7 **Other Maintenance and Prosecution.** Except as set forth in this Section 6.2, neither Party shall have any rights to prepare, file, prosecute or maintain Patents owned or in-licensed by the other Party or its Affiliates.

6.2.8 Patent Schedules.

- (a) **Precision HBV Patents.** At least [***], Precision shall deliver to Gilead an updated <u>Schedule 1.113</u> listing the Precision HBV Patents as of the date of such delivery. Such updated Schedule 1.113 shall be incorporated into this Agreement upon receipt by Gilead.
- (b) Schedule of ARCUS Patents and Precision Patents. Without limiting clause (a), at least every [***] commencing as of the first anniversary of this Agreement, Precision shall deliver to Gilead a Schedule 6.2.8 listing all ARCUS Patents and all Precision Patents, including Precision HBV Patents, as of the date of such delivery. In addition, without limiting clause (a) or limiting the foregoing obligation in this clause (b), within [***] of Precision delivering to Gilead each Gilead ARC Nuclease under this Agreement, Precision shall deliver to Gilead an updated Schedule 6.2.8 which schedule shall list all ARCUS Patents and all Precision Patents, including Precision HBV Patents, as of the date of such delivery. In addition, within [***] of the conclusion of the Collaboration Term, Precision shall deliver to Gilead an updated Schedule 6.2.8, which updated schedule shall list all ARCUS Patents and all Precision Patents, including Precision HBV Patents as of the date of such delivery. Such updated Schedule 6.2.8 shall be incorporated into this Agreement upon receipt by Gilead.

6.3 Enforcement of Patents.

6.3.1 **Precision Patents.**

(a) Each Party shall promptly notify the other Party in writing of any alleged or threatened infringement of the Precision Patents by a Third Party of which such Party becomes aware (including alleged or threatened infringement based on the development or commercialization of, or an application to market, a Licensed Product), subject in each case to any confidentiality obligations of the applicable Party owed to the Third Party who made such Party aware of such infringement. Neither Party shall prosecute any Competitive Infringement of the Precision Patents prior to [***]. Thereafter, Gilead shall have the first right, but not the obligation, to prosecute any Competitive Infringement of the Precision Patents, at [***] cost and expense, and Gilead shall retain control of the prosecution of such suit. At all times, Precision shall have the sole right, but not the obligation, to prosecute any Other Infringement of the Precision Patents at [***] cost and expense, and Precision shall retain control of the prosecution of such suit.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (b) In the event that Gilead decides not to prosecute any Competitive Infringement of the Precision Patents following [***], Gilead shall provide reasonable prior written notice to Precision of such intention (which notice shall, where reasonably practical, be given no later than [***] prior to the next deadline for any action that may be taken with respect to such Patent), and Precision shall thereupon have the option to assume the control and direction of the prosecution of the Competitive Infringement, at [***] cost and expense; provided that, in deciding whether to exercise its option, and in prosecuting such infringement, Precision shall take into consideration Gilead's business reasons for deciding not to prosecute the infringement of such Precision Patent.
- (c) In the event a Party prosecutes infringement of a Precision Patent against Competitive Infringement pursuant to this Section 6.3.1, the Non-Prosecuting Party shall have the right to join as a party to such claim, suit, or proceeding and participate with its own counsel at [***] cost and expense; provided that the Prosecuting Party shall retain control of the prosecution of such claim, suit, or proceeding. During any such claim, suit, or proceeding, the Prosecuting Party shall: (i) provide the Non-Prosecuting Party with drafts of all official papers and statements (whether written or oral) prior to their submission in such claim, suit, or proceeding, in sufficient time to allow the Non-Prosecuting Party to review, consider and substantively comment thereon; (ii) reasonably consider taking action to incorporate the Non-Prosecuting Party's comments on all such official papers and statements; (iii) allow the Non-Prosecuting Party the opportunity to participate in the preparation of witnesses and other participants in such claim, suit, or proceeding; and (iv) not settle any such claim, suit, or proceeding except in a manner that it believes in good faith is in the best interests of the Licensed Products.
- (d) Gilead acknowledges and agrees that (i) Precision has no rights or responsibility for enforcing the Cellectis Patents, and therefore all references to Precision Patents in this Section 6.3.1 shall be deemed to exclude the Cellectis Patents for all purposes, (ii) prior to initiating enforcement actions against a Third Party with respect to certain Precision Patents which were subject to the non-exclusive license granted by Precision to Cellectis S.A. pursuant to the Cellectis Agreement, Precision is required by the Cellectis Agreement to confirm that Cellectis S.A. has not granted a license to such Third Party under such Precision Patents, and Gilead will cooperate with Precision in taking such actions as required by the Cellectis Agreement, and (iii) Duke retains discretion as to whether to become a party plaintiff and has certain rights with respect to enforcement of Patents contained within the Duke IP in the event Precision does not enforce such Patents.

6.3.2 Joint Collaboration Program Patents.

(a) Each Party shall promptly notify the other Party in writing of any alleged or threatened infringement of the Joint Collaboration Program Patents by a Third Party of which such Party becomes aware (including alleged or threatened infringement based on the development, commercialization, or an application to market a Licensed Product), subject in

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

each case to any confidentiality obligations of the applicable Party owed to the Third Party who made such Party aware of such infringement. Gilead shall have the first right, but not the obligation, to prosecute any Competitive Infringement of the Joint Collaboration Program Patents, at [***] cost and expense, and Precision shall have the first right, but not the obligation, to prosecute any Other Infringement of the Joint Collaboration Program Patents, at [***] cost and expense.

- (b) In the event that Gilead decides not to prosecute any Competitive Infringement of the Joint Collaboration Program Patents, Gilead shall provide reasonable prior written notice to Precision of such intention (which notice shall, in any event, be given no later than [***] prior to the next deadline for any action that may be taken with respect to such Patent), and Precision shall thereupon have the option to assume the control and direction of the prosecution of the Competitive Infringement, at [***] cost and expense; provided that, in deciding whether to exercise its option, and in prosecuting such Competitive Infringement, Precision shall take into consideration Gilead's business reasons for deciding not to prosecute the infringement of such Joint Collaboration Program Patents.
- (c) In the event that Precision decides not to prosecute any Other Infringement of the Joint Collaboration Program Patents, Precision shall provide reasonable prior written notice to Gilead of such intention (which notice shall, where reasonably practical, be given no later than [***] prior to the next deadline for any action that may be taken with respect to such Patent), and Gilead shall thereupon have the option to assume the control and direction of the prosecution of the Other Infringement, at [***] cost and expense; provided that, in deciding whether to exercise its option, and in prosecuting such infringement, Gilead shall take into consideration Precision's business reasons for deciding not to prosecute the infringement of such Joint Collaboration Program Patents.
- (d) In the event a Party prosecutes infringement of a Joint Collaboration Program Patent pursuant to this Section 6.3.2 and (i) the Prosecuting Party finds it necessary or desirable for the Non-Prosecuting Party to join the Prosecuting Party as a party to any such claim, suit or proceeding, the Non-Prosecuting Party shall, at the Prosecuting Party's request, or (ii) the Non-Prosecuting Party otherwise desires to join such claim, suit or proceeding, the Non-Prosecuting Party shall have the right to, join as a party to such claim, suit or proceeding and participate with its own counsel at [***] cost and expense; provided that the Prosecuting Party shall retain control of the prosecution of such claim, suit or proceeding. During any such claim, suit, or proceeding with respect to the Joint Collaboration Program Patents in which the Non-Prosecuting Party has joined pursuant to this Section 6.3.2, the Prosecuting Party shall: (A) provide the Non-Prosecuting Party with drafts of all official papers and statements (whether written or oral) prior to their submission in such claim, suit, or proceeding, in sufficient time to allow the Non-Prosecuting Party to review, consider and substantively comment thereon; (B) reasonably consider taking action to incorporate the Non-Prosecuting Party's comments on all such official papers and statements; and (C) allow the Non-Prosecuting Party the opportunity to participate in the preparation of witnesses and other participants in such claim, suit, or proceeding.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 6.3.3 **Gilead Dual Patents.** Each Party shall promptly notify the other Party in writing of any alleged or threatened infringement of the Gilead Dual Patents by a Third Party of which such Party becomes aware (including alleged or threatened infringement based on the development or commercialization, or an application to market, a Licensed Product or any Licensed Product), subject in each case to any confidentiality obligations of the applicable Party owed to the Third Party who made such Party aware of such infringement. Gilead shall have the right, but not the obligation, to prosecute any such infringement at [***] cost and expense and Gilead shall retain control of the prosecution of such suit. With respect to any such alleged or threatened infringement of the Gilead Dual Patents, if Gilead finds it necessary or desirable for Precision to join Gilead as a party to any such action, Precision shall, at Gilead's request, join as a party to such suit and participate with its own counsel at [***] cost and expense; provided that, Gilead shall retain control of the prosecution of such suit.
- 6.3.4 **Cooperation.** Each Party agrees to cooperate fully with the other Party in any infringement action pursuant to this Section 6.3. Where the Prosecuting Party brings such an action, the Non-Prosecuting Party shall, where necessary, furnish a power of attorney solely for such purpose or shall join in, or be named as a necessary party to, such action. Unless otherwise set forth herein, the Prosecuting Party shall have the right to settle such claim; *provided that*, the Prosecuting Party shall not have the right to settle any Patent infringement litigation under this Section 6.3 in a manner that diminishes or has a material adverse effect on the rights or interest of the Non-Prosecuting Party, or in a manner that imposes any costs or liability on, or involves any admission by, the Non-Prosecuting Party, without the express written consent of the Non-Prosecuting Party. The Prosecuting Party shall provide the Non-Prosecuting Party with copies of all pleadings and other material documents filed with the court and shall consider reasonable input from the Non-Prosecuting Party during the course of the proceedings. In connection with the activities set forth in Sections 6.3.1 and 6.3.2, the Prosecuting Party shall consult with the Non-Prosecuting Party as to the strategy for the enforcement of the Precision Patents against a Competitive Infringement and Joint Collaboration Program Patents, as applicable.
- 6.3.5 **Recovery.** Except as otherwise agreed by the Parties in connection with a cost sharing arrangement, any recovery realized as a result of such litigation described in Section 6.3.1, 6.3.2 and 6.3.3 (whether by way of settlement or otherwise) shall be [***].
- 6.3.6 **Other Enforcement.** Except as set forth in this Section 6.3, neither Party shall have any rights to enforce Patents owned or in-licensed by the other Party or its Affiliates.

6.4 Infringement Claims by Third Parties.

6.4.1 If the Manufacture or use of a Licensed Product pursuant to this Agreement results in, or may result in, any claim, suit, or proceeding by a Third Party alleging

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

patent infringement by either Party, its Affiliates, Sublicensees or subcontractors, such Party shall promptly notify the other Party thereof in writing. Gilead shall have the first right, but not the obligation, to defend and control the defense of any such claim, suit, or proceeding alleging patent infringement against Gilead, its Affiliate or Sublicensee at [***] cost and expense, using counsel of its own choice. Precision may participate in any such claim, suit, or proceeding with counsel of its choice at [***] cost and expense. Without limitation of the foregoing, if Gilead finds it necessary or desirable for Precision to join Gilead as a party to any such action, the Parties shall cooperate to execute all papers and perform such acts as shall be reasonably required for Precision to join such action. If Gilead elects (in a written communication submitted to Precision within a reasonable amount of time after notice of the alleged patent infringement) not to defend or control the defense of, or otherwise fails to initiate and maintain the defense of, any such claim, suit, or proceeding, within such time periods so that Precision is not prejudiced by any delays, Precision may conduct and control the defense of any such claim, suit, or proceeding at [***] cost and expense. Each Party shall keep the other Party reasonably informed of all material developments in connection with any such claim, suit, or proceeding.

- 6.4.2 Except as otherwise agreed by the Parties in connection with a cost sharing arrangement, any recovery realized as a result of such litigation described in Section 6.4 (whether by way of settlement or otherwise) shall be [***]. For avoidance of doubt, any recovery realized for infringement of any Precision Patent, Joint Collaboration Program Patents or Gilead Dual Patents shall be subject to Section 6.3.5.
 - 6.4.3 [***].
 - 6.4.4 Nothing in this Section 6.4 shall be construed to limit any rights or obligations of the Parties under ARTICLE 9.

6.5 Invalidity or Unenforceability Defenses or Actions.

- 6.5.1 **Notice**. Each Party shall promptly notify the other Party in writing of any alleged or threatened assertion of invalidity or unenforceability of any of the Precision Patents, Joint Collaboration Program Patents or Gilead Dual Patents by a Third Party, in each case, of which such Party becomes aware.
- 6.5.2 **Precision Patents.** Gilead shall have the first right, but not the obligation, at [***] cost and expense, to defend and control the defense of the validity and enforceability of the Precision Patents where such defense is pursuant to Gilead's indemnification obligations under Section 9.1, in response to a claim brought against Gilead or its Affiliates or Sublicensees by a Third Party, or where the other party to the action is engaging in Competitive Infringement. In all other cases of defense of Precision Patents, or if Gilead elects not to defend or control such defense, or otherwise fails to initiate and maintain the defense, Precision shall control such defense. Precision may participate in any such claim, suit, or proceeding defended by Gilead with counsel of its choice at [***] cost and expense; *provided that*, Gilead shall retain control of the defense in such claim, suit, or proceeding. With respect to any such action involving the validity or enforceability of the Precision Patents, if the defending Party finds it necessary or desirable for the other Party to join the defending Party as a party to

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

any such action, the other Party shall, at the defending Party's request, join the defending Party as a party to such suit and participate with its own counsel at [***] cost and expense; *provided that*, the defending Party shall retain control of the defense in such claim, suit, or proceeding. If Precision elects not to defend or control the defense of the Precision Patents in a suit brought, or otherwise fails to initiate and maintain the defense of any such claim, suit, or proceeding, then, subject to Precision's rights under Section 6.3.1, Gilead may conduct and control the defense of any such claim, suit, or proceeding at [***] cost and expense. With respect to any such action involving the validity or enforceability of the Precision Patents, if Gilead finds it necessary or desirable for Precision to join Gilead as a party to any such action, Precision shall, at Gilead's request, join Gilead as a party to such suit and participate with its own counsel at [***] cost and expense.

- 6.5.3 **Joint Collaboration Program Patents.** Gilead shall have the first right, but not the obligation, to defend and control the defense of the validity and enforceability of the Joint Collaboration Program Patents at [***] cost and expense. With respect to any such action involving the validity or enforceability of the Joint Collaboration Program Patents, if Gilead finds it necessary or desirable for Precision to join Gilead as a party to any such action, Precision shall, at Gilead's request, join Gilead as a party to such suit and participate with its own counsel at [***] cost and expense; provided that, Gilead shall retain control of the defense in such claim, suit, or proceeding. If Gilead elects not to defend or control the defense of the Joint Collaboration Program Patents in a suit brought, or otherwise fails to initiate and maintain the defense of any such claim, suit, or proceeding, then, subject to Gilead's rights under Section 6.3.1, Precision may conduct and control the defense of any such claim, suit, or proceeding at [***] cost and expense. With respect to any such action involving the validity or enforceability of the Joint Collaboration Program Patents, if Precision finds it necessary or desirable for Gilead to join Precision as a party to any such action, Gilead shall, at Precision's request, join Precision as a party to such suit and participate with its own counsel at [***] cost and expense; provided that, Precision shall retain control of the defense in such claim, suit, or proceeding.
- 6.5.4 **Gilead Dual Patents.** Gilead shall have the right, but not the obligation, to defend and control the defense of the validity and enforceability of the Gilead Dual Patents at [***] cost and expense. With respect to any such action involving the validity or enforceability of the Gilead Dual Patents, if Gilead finds it necessary or desirable for Precision to join Gilead as a party to any such action, Precision shall, at Gilead's request, join Gilead as a party to such suit and participate with its own counsel at [***] cost and expense; *provided that*, Gilead shall retain control of the defense in such claim, suit, or proceeding.
- 6.5.5 **Cooperation.** Each Party shall assist and cooperate with the other Party as such other Party may reasonably request from time to time in connection with its activities set forth in this Section 6.5, including by being joined as a party plaintiff in such action or proceeding, providing access to relevant documents and other evidence, and making its employees available at reasonable business hours. In connection with any such defense or claim or counterclaim, the controlling Party shall consider in good faith any comments from the other

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Party and shall keep the other Party reasonably informed of any steps taken, and shall provide copies of all documents filed, in connection with such defense, claim, or counterclaim. In connection with the activities set forth in Sections 6.5.2 and 6.5.3, the Prosecuting Party shall consult with the Non-Prosecuting Party as to the strategy for the defense of the Precision Patents and Joint Collaboration Program Patents, as applicable.

6.6 Patent Term Extensions in the Territory. As between the Parties, Gilead shall have the sole right to make decisions regarding, and to apply for, patent term extensions in the Territory, including in the United States with respect to extensions pursuant to 35 U.S.C. §156 et. seq. and in other jurisdictions pursuant to supplementary protection certificates, and in all jurisdictions with respect to any other extensions that are now or become available in the future, wherever applicable, in each case for the Precision HBV Patents and any Joint Collaboration Program Patents, with respect to the Licensed Product(s), including whether or not to do so. Prior to Gilead making any such decisions, the patent counsel of each Party shall discuss and recommend for which, if any, of the Precision HBV Patents and Joint Collaboration Program Patents in the Territory the Parties should seek any term extension, supplementary protection certificates, and equivalents thereof. Precision shall provide prompt and reasonable assistance, as requested by Gilead, including by taking such action as patent holder as is required under any Applicable Law to obtain such extension or supplementary protection certificate. Further, and without limiting the foregoing, Precision shall not apply for any patent term extension based on a Marketing Approval for any Gilead ARC Nuclease or Licensed Product without Gilead's prior written consent and further consultation with Gilead. With respect to Precision Patents that are not Precision HBV Patents, Precision shall discuss with Gilead in good faith options for applying for patent term extensions on such Precision Patents based on a Marketing Approval for any Gilead ARC Nuclease or Licensed Product prior to deciding whether to apply for any such extension.

ARTICLE 7 CONFIDENTIALITY AND NON-DISCLOSURE

7.1 Confidentiality Obligations. At all times during the Term and for a period of [***] following the expiration or termination of this Agreement in its entirety, or, with respect to Confidential Information of either Party comprising trade secrets of such Party that have been labeled by such disclosing Party or identified by such disclosing Party to the other Party as being the disclosing Party's trade secrets, for so long as such Confidential Information is a trade secret of such Party, each Party shall, and shall cause its officers, directors, employees and agents to, keep confidential and not publish or otherwise disclose to a Third Party and not use, directly or indirectly, for any purpose, any Confidential Information furnished or otherwise made known to it, directly or indirectly, by the other Party, except to the extent such disclosure or use is expressly permitted by the terms of this Agreement or is reasonably necessary or useful for the performance of, or the exercise of such Party's rights under, this Agreement. "Confidential Information" of a Party means any technical, business, or other information provided by or on behalf of such Party to the other Party in connection with this Agreement,

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

whether prior to, on, or after the Effective Date, including (a) the terms and conditions of this Agreement, (b) the ARCUS Technology and the ARCUS Assigned IP, (c) any unpublished Patents, (d) information relating to the Licensed Products (including the Regulatory Documentation generated pursuant to the Collaboration Program), (e) any Development of Gilead ARC Nucleases or the Licensed Products, and any Information with respect thereto developed by or on behalf of the disclosing Party or its Affiliates (including Gilead Dual Know-How and Precision Know-How, as applicable), and (f) information regarding the scientific, regulatory or business affairs or other activities of either Party. All Joint Collaboration Program Know-How and the terms and conditions of this Agreement shall be deemed to be the Confidential Information of both Parties, and both Parties shall be deemed to be the receiving Party and the disclosing Party with respect thereto. Except with respect to the Precision Know-How contained in the chemistry, manufacturing and control (CMC) section of any Regulatory Documentation, all Information contained in the Regulatory Documentation and the Gilead Dual Know-How shall be deemed to be the Confidential Information of Gilead. All ARCUS IP, ARCUS Assigned IP and Precision Know-How shall be deemed to be the Confidential Information of Precision. Notwithstanding the foregoing, the confidentiality and non-use obligations under this Section 7.1 with respect to any Confidential Information shall not include any information that:

- 7.1.1 is or hereafter becomes part of the public domain by public use, publication, general knowledge or the like through no wrongful act, fault or negligence on the part of the receiving Party;
- 7.1.2 can be demonstrated by documentation or other competent proof to have been in the receiving Party's possession prior to disclosure by the disclosing Party without any obligation of confidentiality with respect to such information; *provided that*, the foregoing exception shall not apply with respect to the Joint Collaboration Program Know-How;
- 7.1.3 is subsequently received by the receiving Party from a Third Party who is not bound by any obligation of confidentiality with respect to such information; or
- 7.1.4 can be demonstrated by documentation or other competent evidence to have been independently developed by or for the receiving Party without use of or reference to the disclosing Party's Confidential Information; *provided that*, the foregoing exception shall not apply with respect to the Joint Collaboration Program Know-How.

Specific aspects or details of Confidential Information shall not be deemed to be within the public domain or in the possession of the receiving Party merely because the Confidential Information is embraced by more general information in the public domain or in the possession of the receiving Party. Further, any combination of Confidential Information shall not be considered in the public domain or in the possession of the receiving Party merely because individual elements of such Confidential Information are in the public domain or in the possession of the receiving Party unless the combination and its principles are in the public domain or in the possession of the receiving Party.

- 7.2 Permitted Disclosures. Each Party may disclose Confidential Information of the other Party to the extent that such disclosure is:
- 7.2.1 made by or on behalf of the receiving Party to the Regulatory Authorities as required in connection with any filing, application or request for Regulatory Approval; *provided, however*, that reasonable measures shall be taken to assure confidential treatment of such information to the extent practicable and consistent with Applicable Law;

- 7.2.2 made by or on behalf of the receiving Party in response to a valid order of a court of competent jurisdiction or other supra-national, federal, national, regional, state, provincial and local governmental or regulatory body of competent jurisdiction or, if in the opinion of the receiving Party's legal counsel and without limiting Section 7.4, such disclosure is otherwise required by Applicable Law (including, for clarity, any disclosure required by Applicable Law on clinicaltrials.gov or disclosure required by reason of filing with securities regulators); provided, however, that the receiving Party shall first have given notice to the disclosing Party and given the disclosing Party (a) a reasonable opportunity to quash any such order or to obtain a protective order or confidential treatment requiring that the Confidential Information and documents that are the subject of any such order be held in confidence by such court or agency or, if disclosed, be used only for the purposes for which the order was issued and (b) a right to review and comment upon such disclosure, which comments shall be considered in good faith by the receiving Party; and provided further that the Confidential Information disclosed in response to any such court or governmental order shall be limited to that information which is legally required to be disclosed in response to such court or governmental order;
- 7.2.3 made by or on behalf of the receiving Party to a patent authority as may be reasonably necessary or useful for purposes of obtaining or enforcing a Patent pursuant to the terms of this Agreement in a manner not inconsistent with Article 6; *provided, however*, that reasonable measures shall be taken to assure confidential treatment of such information, to the extent such protection is available; or
- 7.2.4 made by the receiving Party or its Affiliates, sublicensees or subcontractors to its or their agents, attorneys, auditors, advisors, consultants, contractors, existing or prospective collaboration partners, licensees, sublicensees, investors, insurers or acquirers in connection with the performance of its obligations or exercise of its rights as contemplated by this Agreement; *provided, however*, that such persons shall be subject to obligations of confidentiality and non-use with respect to such Confidential Information substantially similar to the obligations of confidentiality and non-use of the receiving Party pursuant to this ARTICLE 7 (with a duration of confidentiality and non-use obligations as appropriate that is no less than [***] from the date of disclosure); or
- 7.2.5 made by or on behalf of the receiving Party where such disclosure is required by a Regulatory Authority (including in filings with the Securities and Exchange Commission or other agency) of certain material developments or material information generated under this Agreement; provided that, to the extent permitted, the Party seeking such disclosure first provides the other Party a copy of the proposed disclosure; and provided, further, that the receiving Party shall afford to the other Party an opportunity to review and comment, which period shall be no less than [***], and the receiving Party shall accept any reasonable comments so provided; or

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 7.2.6 made by or on behalf of Precision to Duke solely as and to the extent necessary to fulfill Precision's reporting obligations under the Duke Agreement as of the Effective Date so long as such information is disclosed subject to the confidentiality provisions of the Duke Agreement as of the Effective Date.
- **7.3 Use of Name.** Except as expressly provided herein, neither Party shall mention or otherwise use the name, logo, or trademark of the other Party or any of its Affiliates (or any abbreviation or adaptation thereof) in any publication, press release, marketing and promotional material, or other form of publicity without the prior written approval of such other Party in each instance. The restrictions imposed by this Section 7.3 shall not prohibit either Party from making any disclosure identifying the other Party that is required by Applicable Law.
- 7.4 Public Announcements. The Parties have agreed upon the content of press release(s) which shall be issued substantially in the form attached hereto as Schedule 7.4, the release of which the Parties shall coordinate in order to accomplish such release at a time following execution of the Agreement to be agreed upon by the Parties. Neither Party shall issue any other public announcement, press release, or other public disclosure regarding this Agreement or its subject matter without the other Party's prior written consent, except for any such disclosure that is, in the opinion of the disclosing Party's counsel, required by Applicable Law, any Regulatory Authority (including filings with the Securities and Exchange Commission or other agency) or the rules of a stock exchange on which the securities of the disclosing Party are listed (or to which an application for listing has been submitted). In the event a Party is, in the opinion of its counsel, required by Applicable Law, any Regulatory Authority (including filings with the Securities and Exchange Commission or other agency) or the rules of a stock exchange on which its securities are listed (or to which an application for listing has been submitted) to make such a public disclosure, and notwithstanding anything to the contrary in Section 7.2, such Party shall submit the proposed disclosure in writing to the other Party as far in advance as reasonably practicable (and in no event less than [***] prior to the anticipated date of disclosure) so as to provide a reasonable opportunity to comment thereon; provided, however, if a Party is required by Applicable Law, any Regulatory Authority (including filings with the Securities and Exchange Commission or other agency) or the rules of a stock exchange on which the securities of the disclosing Party are listed (or to which an application for listing has been submitted) to disclose this Agreement, such Party shall prepare a proposed redacted version of this Agreement to request confidential treatment for this Agreement, and the other Party may promptly (and in any event, no less than [***] after receipt of such proposed redactions) provide its comments, which comments shall be considered in good faith by the Party required to make such disclosure.
- **7.5 Publications.** Precision shall not publish any papers or make any oral presentations regarding results of, and other information regarding, activities pursuant to the Collaboration Program (such papers and oral presentations, including abstracts of any of the foregoing, "**Publications**") if such papers or presentations disclose any Information relating specifically to HBV, any Gilead ARC Nuclease or the Licensed Product, except as required by Applicable Law, in which case Section 7.2.2 shall apply *mutatis mutandis*. Notwithstanding the

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

foregoing, subject to the obligations set forth in Sections 7.1, 7.2, and 7.3, Precision may make Publications relating specifically to the ARCUS Technology or ARCUS Assigned IP, *provided that* for any Publication that discloses Information relating to HBV, any Gilead ARC Nuclease or the Licensed Product, Precision shall (a) provide Gilead with a draft of such Publication at least [***] prior to submission to the publisher, (b) remove any confidential or sensitive Information as requested by Gilead that Gilead reasonably deems to be of a confidential or sensitive nature, (c) delay the submission for publication of such Publication for an additional [***] period to permit the applicable Party under Section 6.2 to seek patent protection with respect to the content of such Information, and (d) consider in good faith any comments from Gilead with respect to the information contained therein pertaining to Licensed Products, Gilead ARC Nucleases or HBV. Gilead may, in its discretion, publish any Publication; *provided that*, Gilead shall (i) provide Precision with a draft of such Publication at least [***] prior to submission to the publisher, (ii) remove any confidential or sensitive Information of Precision related to ARCUS Technology, ARCUS Assigned IP or ARC Nucleases generally, as requested by Precision, (iii) delay the submission for publication of such Publication for an additional [***] period to permit the applicable Party under Section 6.2 the opportunity to seek patent protection with respect to the content of such Information, and (iv) give Precision a pre-publication right to review and comment upon such Publication, which comments shall be considered in good faith by Gilead.

7.6 Return of Confidential Information. Upon the effective date of the termination of this Agreement for any reason, either Party may request in writing, and the other Party shall either, with respect to Confidential Information to which such first Party does not retain rights under the surviving provisions of this Agreement: (a) promptly destroy all copies of such Confidential Information in the possession of the other Party and confirm such destruction in writing to the requesting Party; or (b) promptly deliver to the requesting Party, at the [***] cost and expense, all copies of such Confidential Information in the possession of the other Party; provided, however, the other Party shall be permitted to retain one (1) copy of such Confidential Information for the sole purpose of performing any continuing obligations hereunder or for archival purposes. Notwithstanding the foregoing, such other Party also shall be permitted to retain such additional copies of or any computer records or files containing such Confidential Information that have been created solely by such Party's automatic archiving and back-up procedures, to the extent created and retained in a manner consistent with such other Party's standard archiving and back-up procedures, but not for any other use or purpose. All Confidential Information shall continue to be subject to the terms of this Agreement for the period set forth in Section 7.1.

ARTICLE 8 REPRESENTATIONS AND WARRANTIES

- **8.1 Mutual Representations and Warranties.** Each Party represents and warrants to the other Party as of the Effective Date that:
- 8.1.1 such Party is an entity duly organized, validly existing and in good standing under the Applicable Law of the state or country (as applicable) of its organization, is

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

qualified to do business and is in good standing as a foreign entity in each jurisdiction in which the conduct of its business or the ownership of its properties requires such qualification and failure to have such would prevent it from performing its obligations under this Agreement, and has full power and authority to enter into this Agreement and to carry out the provisions hereof;

- 8.1.2 such Party is duly authorized, by all requisite action, to execute and deliver this Agreement and the execution, delivery and performance of this Agreement by such Party does not require any shareholder action or approval, and the person executing this Agreement on behalf of such Party is duly authorized to do so by all requisite action;
- 8.1.3 no consent, approval, order or authorization of, or registration, qualification, designation, declaration or filing with, any governmental authority or Regulatory Authority is required on the part of such Party in connection with the valid execution, delivery and performance of this Agreement by it;
- 8.1.4 such Party has not employed (and, to its knowledge, has not used a contractor or consultant that has employed) and in the future shall not employ (or, to its knowledge, use any contractor or consultant that employs; *provided that*, such Party may reasonably rely on a representation made by such contractor or consultant) any person debarred by the FDA (or subject to a similar sanction of a foreign equivalent), or any person which is the subject of an FDA debarment investigation or proceeding (or similar proceeding of a foreign equivalent), in the conduct of its activities under this Agreement;
- 8.1.5 this Agreement is a legal and valid obligation binding upon such Party and enforceable in accordance with its terms except as enforceability may be limited by (a) bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights; and (b) equitable principles of general applicability; and
- 8.1.6 the execution, delivery and performance by it of this Agreement and its compliance with the terms and provisions of this Agreement does not and shall not conflict with or result in a breach of any of the terms or conditions of (a) any other contractual or other obligations of such Party, (b) the provisions of its operating documents or bylaws, or (c) any order, writ, injunction or decree of any governmental authority or Regulatory Authority entered against it or by which it or any of its property is bound.
 - 8.2 Additional Representations and Warranties of Precision. Precision further represents and warrants to Gilead as of the Effective Date that:
- 8.2.1 The Precision Existing Patents comprise all Precision Patents existing as of the Effective Date, other than Cellectis Patents. Precision is the sole and exclusive owner of the entire right, title and interest in, or otherwise Controls, all Precision Existing Patents and all existing ARCUS IP. All Precision Existing Patents owned by Precision, and to the Knowledge of Precision, all Precision Existing Patents in-licensed by Precision, are subsisting and have not been determined by any competent court or other governmental authority to be invalid or unenforceable, in whole or in part. In respect of the Precision Existing Patents owned by Precision, to Precision's Knowledge, Precision and its Affiliates have presented all relevant references, documents, or information to the relevant patent examiner at the relevant patent office as required by any applicable duty of candor. To Precision's Knowledge, each of the Precision Existing Patents properly identifies each and every inventor of the claims thereof as determined in accordance with the laws of the jurisdiction in which such Precision Existing Patent is issued or such application is pending.

- 8.2.2 There are no pending claims or claims threatened in writing (or to its Knowledge, otherwise threatened), judgments, or settlements against, or amounts with respect thereto, owed by Precision or any of its Affiliates relating to the ARCUS IP or the Precision IP. Except as was resolved in connection with the Cellectis Agreement, no claim or litigation has been brought against Precision or its Affiliates, or, to Precision's Knowledge, threatened in writing by any Person alleging that (a) the ARCUS IP or the Precision IP is invalid or unenforceable or misappropriates any intellectual property or proprietary right of any Person other than the Parties, or (b) the Exploitation of the ARCUS Technology as contemplated herein violates, infringes, or otherwise conflicts or interferes with, or would violate, infringe, or otherwise conflict or interfere with, any intellectual property or proprietary right of any Person other than the Parties.
- 8.2.3 To Precision's Knowledge, (a) Precision has the right to use all ARCUS IP and Precision IP as necessary to conduct the Development activities in the Collaboration R&D Plan, and (b) the use of the ARCUS IP and the Precision IP in the Development of the Licensed Products as contemplated herein (i) is not subject to any other license or agreement to which Precision or any of its Affiliates is a party other than the Existing In-License Agreements, (ii) does not infringe any Patent or other intellectual property or proprietary right of any Person other than the Parties, or (iii) does not constitute or involve the misappropriation of trade secrets or other rights or property of any Person other than the Parties, but excluding, in each case ((i), (ii) and (iii)), any intellectual property or proprietary right of any Person other than the Parties relating specifically to any HBV Target. For purposes of the representation and warranty in this Section 8.2.3, "Precision's Knowledge" shall also require review of the foregoing representation and warranty with Precision's external patent counsel within [***] prior to execution of this Agreement.
- 8.2.4 Neither Precision nor any of its Affiliates has previously entered into any agreement, whether written or oral, with respect to, or otherwise assigned, transferred, licensed, conveyed, or otherwise encumbered its right, title, or interest in or to the ARCUS IP, the Precision IP or the Gilead ARC Nucleases (including by granting any covenant not to sue with respect thereto), or any Patent or other intellectual property or proprietary right that would be ARCUS IP or Precision IP but for such assignment, transfer, license, conveyance, or encumbrance, except in each case where such assignment, transfer, license, conveyance, or encumbrance is (a) terminated and no longer in force or effect or (b) not inconsistent with the rights and licenses granted to Gilead under this Agreement.
- 8.2.5 A true, complete and accurate copy of each of the Existing In-License Agreements has been provided or made available to Gilead in an electronic data room maintained by Precision.
 - 8.2.6 To Precision's Knowledge, each of the Existing In-License Agreements is valid, enforceable and binding on the parties thereto.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 8.2.7 Precision (or its Affiliates) has in all material respects performed its obligations under each of the Existing In-License Agreements, including with respect to any obligations relating to funding received under any of the Existing In-License Agreements, and neither Precision nor any other party thereto has given or received any notice to terminate, or asserted any breach of or default under, any Existing In-License Agreement nor, to Precision's Knowledge, are there any grounds for the termination, avoidance, rescission or repudiation of any Existing In-License Agreement.
- 8.2.8 To Precision's Knowledge, no Person is infringing or threatening to infringe, or misappropriating or threatening to misappropriate, the Precision IP.
- 8.2.9 All material adverse information with respect to the safety of the ARCUS Technology within the Knowledge of Precision has been provided or made available to Gilead in an electronic data room maintained by Precision prior to the Effective Date.
- 8.2.10 All current and former officers, employees, agents and consultants of Precision or any of its Affiliates who are inventors of, or have otherwise contributed in a material manner to the creation or development of, the ARCUS IP or any Precision IP have executed and delivered to Precision or such Affiliate, and to Precision's Knowledge are not in violation of, an assignment or other agreement regarding the protection of proprietary information and the assignment to Precision or such Affiliate of the ARCUS IP or any Precision IP, the current form of which has been made available for review by Gilead.
- 8.2.11 Neither Precision nor any of its Affiliates has been debarred or is subject to debarment pursuant to Section 306 of the Act, or is the subject of a conviction described in such section.
- 8.2.12 Except as set forth on Schedule 8.2.12, the inventions claimed or covered by the Precision IP (a) were not conceived, discovered, developed, or otherwise made in connection with any research activities funded, in whole or in part, by the federal government of the United States or any agency thereof, (b) are not a "subject invention" as that term is described in 35 U.S.C. Section 201(f), and (c) are not otherwise subject to the provisions of the Bayh-Dole Act. With regard to any inventions set forth on Schedule 8.2.12 that are subject to the Bayh-Dole Act, Precision and its Affiliates have complied with the applicable provisions of the Bayh-Dole Act, in a manner that protects and preserves Precision's right, title and interest in such inventions to the maximum extent permitted by law.
- 8.2.13 This Agreement satisfies the requirements to be a "Commercial Product Sublicense" under the Duke Agreement, as amended, as that term is used in such agreement.

8.3 Additional Covenants of Precision.

8.3.1 Neither Precision nor any of its Affiliates shall enter into any agreement, whether written or oral, with respect to, or otherwise assign, transfer, license, convey, or otherwise encumber its right, title, or interest in or to the ARCUS IP, the Precision IP, [***] or the Gilead ARC Nucleases (including by granting any covenant not to sue with respect thereto),

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

or any Patent or other intellectual property or proprietary right that would be ARCUS IP or Precision IP but for such assignment, transfer, license, conveyance, or encumbrance, in each case where such assignment, transfer, license, conveyance, or encumbrance is inconsistent with the rights and licenses granted to Gilead under this Agreement.

- 8.3.2 Neither Precision nor any of its Affiliates shall use in any capacity, in connection with the services to be performed under this Agreement, any Person who has been debarred pursuant to Section 306 of the Act, or who is the subject of a conviction described in such section. Precision agrees to inform Gilead in writing immediately if it or any Person who is performing services hereunder is debarred or is the subject of a conviction described in Section 306 of the Act, or if any action, suit, claim, investigation or legal or administrative proceeding is pending or, to Precision's Knowledge, is threatened, relating to the debarment or conviction of Precision or any Person performing such services.
- 8.3.3 With regard to any inventions set forth on <u>Schedule 8.2.12</u> that are subject to the Bayh-Dole Act or other inventions that are subject to the Bayh-Dole Act in the Precision IP, Precision shall, and shall cause its Affiliates to, to comply with the applicable provisions of the Bayh-Dole Act, in a manner that protects and preserves Precision's right, title and interest in such inventions to the maximum extent permitted by law.
- **8.4 DISCLAIMER OF WARRANTIES.** EXCEPT FOR THE EXPRESS REPRESENTATIONS AND WARRANTIES SET FORTH HEREIN, NEITHER PARTY MAKES ANY REPRESENTATIONS OR GRANTS ANY WARRANTIES, EXPRESS OR IMPLIED, EITHER IN FACT OR BY OPERATION OF LAW, BY STATUTE OR OTHERWISE, AND EACH PARTY SPECIFICALLY DISCLAIMS ANY OTHER WARRANTIES, WHETHER WRITTEN OR ORAL, OR EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY OF QUALITY, MERCHANTABILITY, OR FITNESS FOR A PARTICULAR USE OR PURPOSE OR ANY WARRANTY AS TO THE VALIDITY OF ANY PATENTS OR THE NON-INFRINGEMENT OF ANY IP RIGHTS OF THIRD PARTIES OR THE AVAILABILITY OF ANY LICENSES WITH RESPECT TO IP RIGHTS OF THIRD PARTIES.

ARTICLE 9 INDEMNITY

9.1 Indemnification of Precision. Gilead shall indemnify Precision, its Affiliates, Duke, and its and their respective directors, officers, employees, and agents, and defend and save each of them harmless, from and against any and all losses, damages, liabilities, costs, and expenses (including reasonable attorneys' fees and expenses) (collectively, "Losses") in connection with any and all suits, investigations, claims, or demands of Third Party (collectively, "Third Party Claims") arising from or occurring as a result of: (a) the breach by Gilead of any representation, warranty or covenant in this Agreement, (b) the gross negligence or willful misconduct on the part of Gilead or its Affiliates or its or their respective directors, officers, employees, and agents in performing its or their obligations under this Agreement,

- (c) [***], or (d) [***], except, in each case of clause (a) (d), for those Losses for which Precision has an obligation to indemnify Gilead pursuant to Section 9.2 hereof, as to which Losses each Party shall indemnify the other to the extent of their respective liability.
- **9.2 Indemnification of Gilead.** Precision shall indemnify Gilead, its Affiliates and its and their respective directors, officers, employees, and agents, and defend and save each of them harmless, from and against any and all Losses in connection with any and all Third Party Claims arising from or occurring as a result of: (a) the breach by Precision of any representation, warranty or covenant of this Agreement, (b) the gross negligence or willful misconduct on the part of Precision or its Affiliates or its or their respective directors, officers, employees, and agents in performing its obligations under this Agreement, (c) [***], or (d) [***], except, in each case of clause (a) (d), for those Losses for which Gilead has an obligation to indemnify Precision pursuant to Section 9.1 hereof, as to which Losses each Party shall indemnify the other to the extent of their respective liability.
- 9.3 Notice of Claim. All indemnification claims in respect of a Party, its Affiliates, or their respective directors, officers, employees and agents shall be made solely by such Party to this Agreement (the "Indemnified Party"). The Indemnified Party shall give the indemnifying Party (the "Indemnifying Party") written notice (an "Indemnification Claim Notice") of any Losses or discovery of fact upon which such Indemnified Party intends to base a request for indemnification under this ARTICLE 9 within [***] after receipt by such Indemnified Party of actual notice of the Third Party Claim; provided that, failure to give such notification shall not affect the indemnification provided under Section 9.1 or Section 9.2, as applicable, except to the extent the Indemnifying Party has been actually prejudiced as a result of such failure. Each Indemnification Claim Notice must contain a description of the claim and the nature and amount of such Loss (to the extent that the nature and amount of such Loss is known at such time). The Indemnified Party shall furnish promptly to the Indemnifying Party copies of all papers and official documents received in respect of any Losses and Third Party Claims.

9.4 Control of Defense.

9.4.1 **In General.** At its option, the Indemnifying Party may assume the defense of any Third Party Claim by giving written notice to the Indemnified Party within [***] after the Indemnifying Party's receipt of an Indemnification Claim Notice only if the Indemnifying Party has acknowledged to the Indemnified Party in writing that the Indemnifying Party is liable to indemnify the Indemnified Party in respect of such Third Party Claim. Upon assuming the defense of a Third Party Claim, the Indemnifying Party may appoint as lead counsel in the defense of the Third Party Claim any legal counsel selected by the Indemnifying Party. In the event the Indemnifying Party assumes the defense of a Third Party Claim, the Indemnified Party shall immediately deliver to the Indemnifying Party all original notices and documents (including court papers) received by the Indemnified Party in connection with the Third Party Claim. Should the Indemnifying Party assume the defense of a Third Party Claim, except as provided in Section 9.4.2, the Indemnifying Party shall not be liable to the Indemnified Party for any legal expenses subsequently incurred by such Indemnified Party in connection with the analysis, defense or settlement of the Third Party Claim unless specifically requested in

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

writing by the Indemnifying Party. In the event that it is ultimately determined that the Indemnifying Party is not obligated to indemnify, defend or hold harmless the Indemnified Party from and against the Third Party Claim, the Indemnified Party shall reimburse the Indemnifying Party for any and all costs and expenses (including attorneys' fees and costs of suit) and any Losses incurred by the Indemnifying Party in its defense of the Third Party Claim.

- 9.4.2 **Right to Participate in Defense.** Without limiting Section 9.4.1, any Indemnified Party shall be entitled to participate in, but not control, the defense of such Third Party Claim and to employ counsel of its choice for such purpose; *provided, however*, that such employment shall be at the [***] cost and expense unless (a) the employment thereof has been specifically authorized by the Indemnifying Party in writing, (b) the Indemnifying Party has failed to assume the defense and employ counsel in accordance with Section 9.4.1 (in which case the Indemnified Party shall control the defense), or (c) the interests of the indemnitee and the Indemnifying Party with respect to such Third Party Claim are sufficiently adverse to prohibit the representation by the same counsel of both Parties under Applicable Law, ethical rules or equitable principles.
- 9.4.3 **Settlement.** With respect to any Losses relating to a Third Party Claim, the Indemnifying Party shall have the sole right to consent to the entry of any judgment, enter into any settlement or otherwise dispose of such Loss, at [***] cost and expense, on such terms as the Indemnifying Party, in its sole discretion, shall deem appropriate; *provided that*, such settlement shall not result in the Indemnified Party's becoming subject to injunctive or other relief or otherwise adversely affecting the business of the Indemnified Party in any manner. For any other settlement, the Indemnifying Party shall have the right to consent to the entry of any judgment, enter into any settlement or otherwise dispose of such Loss only if it obtains the prior written consent of the Indemnified Party (which consent shall not be unreasonably withheld, conditioned or delayed). If the Indemnifying Party does not assume and conduct the defense of a Third Party Claim as provided above, the Indemnified Party may defend against such Third Party Claim and enter into any settlements of otherwise dispose of such Loss at the [***] cost and expense.
- 9.4.4 **Cooperation.** Regardless of whether the Indemnifying Party chooses to defend any Third Party Claim, the Indemnified Party shall, and shall cause each indemnitee to, cooperate in the defense thereof and shall furnish such records, information and testimony, provide such witnesses and attend such conferences, discovery proceedings, hearings, trials and appeals as may be reasonably requested in connection therewith. Such cooperation shall include access during normal business hours afforded to the Indemnifying Party to, and reasonable retention by the Indemnified Party of, records and information that are reasonably relevant to such Third Party Claim, and making the Indemnified Party and other employees and agents available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder, and the Indemnifying Party shall reimburse the Indemnified Party for all its reasonable out-of-pocket expenses in connection therewith.
- 9.4.5 **Expenses.** Except as provided above, the costs and expenses, including fees and disbursements of counsel, incurred by the Indemnified Party in connection

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

with any claim shall be reimbursed on a Calendar Quarter basis by the Indemnifying Party, without prejudice to the Indemnifying Party's right to contest the Indemnified Party's right to indemnification and subject to refund in the event the Indemnifying Party is ultimately held not to be obligated to indemnify the Indemnified Party.

9.5 Special, Indirect, and Other Losses. EXCEPT IN THE EVENT OF A PARTY'S BREACH OF ITS OBLIGATIONS UNDER ARTICLE 7, AND EXCEPT TO THE EXTENT ANY SUCH DAMAGES ARE REQUIRED TO BE PAID TO A THIRD PARTY AS PART OF A CLAIM FOR WHICH A PARTY PROVIDES INDEMNIFICATION UNDER THIS ARTICLE 9, NEITHER PARTY NOR ANY OF ITS AFFILIATES SHALL BE LIABLE IN CONTRACT, TORT, NEGLIGENCE, BREACH OF STATUTORY DUTY OR OTHERWISE FOR ANY SPECIAL, INDIRECT, INCIDENTAL, PUNITIVE, OR CONSEQUENTIAL DAMAGES (INCLUDING DAMAGES RESULTING FROM LOSS OF USE, LOSS OF PROFITS, INTERRUPTION OR LOSS OF BUSINESS, DIMINUTION IN VALUE, OR OTHER ECONOMIC LOSSES) ARISING OUT OF ANY ACTIVITIES WITHIN THE SCOPE OF THIS AGREEMENT OR WITH RESPECT TO A PARTY'S PERFORMANCE OR NON PERFORMANCE HEREUNDER.

9.6 Insurance.

- 9.6.1 **Insurance Maintained by Precision.** During the Term, Precision shall have and maintain in full force and effect, at its own expense, insurance coverage to include:
- (a) Commercial general liability insurance, including personal and advertising injury, and Licensed Products and completed operations, with limits of liability not less than [***] per occurrence and [***] in the aggregate. General liability limit requirements may be satisfied by a combination of primary and umbrella or excess liability insurance coverage;
- (b) Workers' compensation insurance in compliance with Applicable Law (including the local law requirements of the state or jurisdiction in which the work is to be performed). Employer's liability insurance in amounts not less than [***] for each of (i) bodily injury by accident (each accident), (ii) bodily injury by disease (policy limit), and (iii) bodily injury by disease (each employee). Where permitted by Applicable Law, such policies shall contain a waiver of the insurer's subrogation rights against Gilead; and
- (c) Automobile liability insurance for bodily injury, property damage and automobile contractual liability covering all owned, hired and non-owned automobiles with a combined single limit of liability for each accident of not less than [***].

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- 9.6.2 **Insurance Maintained by Gilead.** During the Term, Gilead shall have and maintain in full force and effect, at its own expense, insurance coverage to include:
 - (a) Commercially purchased insurance in accordance with the following:
- (i) Commercial general liability insurance, including personal and advertising injury, and Licensed Products and completed operations with limits of liability not less than [***] per occurrence and [***] in the aggregate. General liability limit requirements may be satisfied by a combination of primary and umbrella or excess liability insurance coverage;
- (ii) Workers' compensation insurance in compliance with Applicable Law (including the local law requirements of the state or jurisdiction in which the work is to be performed). Employer's liability insurance in amounts not less than [***] for each of (A) bodily injury by accident (each accident), (B) bodily injury by disease (policy limit), and (C) bodily injury by disease (each employee). Where permitted by Applicable Law, such policies shall contain a waiver of the insurer's subrogation rights against Precision;
- (iii) Automobile Liability insurance for bodily injury, property damage and automobile contractual liability covering all owned, hired and non-owned automobiles with a combined single limit of liability for each accident of not less than [***]; and
 - (iv) Clinical trials liability insurance with limits not less than [***] per occurrence; or
 - (b) Self-insurance substantially equivalent to the coverage described in Section 9.6.2(a) above.

9.6.3 Additional Requirements.

- (a) **Additional Insured.** Each Party shall name the other Party as an additional insured on the insurance policies maintained pursuant to Section 9.6.1(a), Section 9.6.1(c), Section 9.6.2(a)(i), Section 9.6.2(a)(iii) and Section 9.6.2(a)(iv), as applicable, either by endorsement or blanket additional insured endorsement.
- (b) **Evidence of Insurance.** Each Party will provide evidence of insurance maintained pursuant to this Section 9.6 on request of the other Party.
- (c) **Notice of Cancellation.** Each Party will provide the other Party a notice of insurance policy cancellation in accordance with the provisions of the applicable insurance policy maintained pursuant to this Section 9.6.
- (d) **Policy Type.** Insurance policies maintained pursuant to this Section 9.6 should be occurrence type. If policies maintained pursuant to this Section 9.6 are claims made, then insurance shall be maintained for at least [***] following expiration or termination of this Agreement.
- (e) **Insurance Carrier Rating.** All insurance maintained pursuant to this Section 9.6 will be underwritten by companies with an AM best rating of at least A-VII.

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

ARTICLE 10 TERM AND TERMINATION

10.1 Term. This Agreement shall commence on the Effective Date and, unless earlier terminated in accordance herewith, shall continue in force and effect, on a Licensed Product-by-Licensed Product and country-by-country basis until the expiration of the Royalty Term for such Licensed Product in such country (the "Term"). Upon expiration of the Royalty Term (but not on early termination of this Agreement), on a Licensed Product-by-Licensed Product and country-by-country basis the license granted to Gilead pursuant to Section 4.1.1 shall be a fully paid-up, irrevocable, perpetual license that will be exclusive unless and until such time as a Biosimilar Product launches or has launched with respect to the applicable Licensed Product in the applicable country (and from such time will be non-exclusive), and the non-exclusive license granted to Gilead pursuant to Section 4.1.2 shall be fully paid-up, irrevocable and perpetual.

10.2 Termination by Either Party.

- may have with respect to this Agreement, either Party may, upon a material breach of this Agreement by the other Party, terminate this Agreement by providing [***] prior written notice (or [***]' prior written notice in the event such material breach is solely based on the breaching Party's failure to pay any amounts due hereunder) to the breaching Party, specifying in such notice the breaching Party's material breach and demanding its cure, with such termination being effective upon the end of such [***] (or [***], as applicable) cure period or, if applicable, the end of the extended cure period set forth in the immediately following sentence, in each case if the applicable material breach has not then been cured. Notwithstanding the foregoing, with respect to a material breach that is not solely based on the breaching Party's failure to pay any amounts due hereunder, if such material breach is not reasonably curable within the [***] cure period, the non-breaching Party's right to terminate this Agreement pursuant to this Section 10.2.1 shall be suspended only if, and only for so long as, (x) the breaching Party has provided to the non-breaching Party a written plan that is reasonably calculated to effect a cure and that includes a proposed extended cure period (not to exceed [***] after the original [***] cure period), (y) the non-breaching Party confirms in writing that such plan is reasonably acceptable to the non-breaching Party and (z) the breaching Party commits to and does carry out such plan no later than the end of the extended cure period set forth in the written plan described in clause (x) of this sentence.
- 10.2.2 **Termination for Insolvency.** In the event that either Party (a) makes an assignment for the benefit of creditors, (b) appoints or suffers appointment of a receiver or trustee over substantially all of its property that is not discharged within [***] after such filing, (c) proposes a written agreement of composition with its creditors, (d) proposes or is a party to any dissolution or liquidation, (e) files a petition under any bankruptcy or insolvency act or has any such petition filed against that is not discharged within [***] of the filing thereof (excluding a reorganization proceeding under Chapter 11 of the U.S. Bankruptcy Code if the

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

debtor Party is continuing to perform all of its obligations under this Agreement), or (f) admits in writing its inability generally to meet its obligations as they fall due in the general course, then the other Party may terminate this Agreement in its entirety effective immediately upon written notice to such Party.

10.3 Additional Termination Rights by Gilead.

- 10.3.1 **For Convenience.** Gilead may terminate this Agreement, for any reason or no reason, upon:
 - (a) [***] prior written notice to Precision, if such notice is provided during the Collaboration Term; and
 - (b) [***] prior written notice to Precision, if such notice is provided on or after the expiry of the Collaboration Term.
- Precision agrees to notify Gilead promptly in writing in the event Precision has entered into a transaction that effects a Change of Control of Precision or assignment or transfer of this Agreement by Precision, or would effect a Change of Control of Precision or assignment or transfer of this Agreement by Precision upon the closing of the transaction, or an event occurs that triggers a Change of Control of Precision or assignment or transfer of this Agreement by Precision. During the Collaboration Term, Gilead may terminate this Agreement in the event of a Change of Control that results in Precision being controlled by, or assignment or transfer of this Agreement by Precision to, a Third Party that is clinically developing or commercializing products in the field of HBV, upon [***] prior written notice to Precision, so long as such notice is sent no later than [***] after Gilead becomes aware of such Change of Control.

10.4 Additional Termination Rights by Precision [***].

10.5 Rights in Bankruptcy. All rights and licenses granted under or pursuant to this Agreement by Gilead or Precision are, and shall otherwise be deemed to be, for purposes of Section 365(n) of the U.S. Bankruptcy Code or any analogous provisions in any other country or jurisdiction, licenses of rights to "intellectual property" as defined under Section 101 of the U.S. Bankruptcy Code. The Parties agree that the Parties, as licensees of such rights under this Agreement, shall retain and may fully exercise all of their rights and elections under the U.S. Bankruptcy Code or any analogous provisions in any other country or jurisdiction. The Parties further agree that, in the event of the commencement of a bankruptcy proceeding by or against either Party under the U.S. Bankruptcy Code or any analogous provisions in any other country or jurisdiction, the Party that is not a party to such proceeding shall be entitled to a complete duplicate of (or complete access to, as appropriate) any such intellectual property and all embodiments of such intellectual property to which such Party is entitled to have access under this Agreement, which, if not already in the non-debtor Party's possession, shall be promptly delivered to it (a) upon any such commencement of a bankruptcy proceeding upon the non-debtor Party's written request therefor, unless the Party subject to such proceeding elects to

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

continue to perform all of its obligations under this Agreement, or (b) if not delivered under clause (a) above, following the rejection of this Agreement by or on behalf of the Party subject to such proceeding upon written request therefor by the non-debtor Party. The Parties acknowledge and agree that payments made under Section 5.1 shall not (x) constitute royalties within the meaning of Section 365(n) of the U.S. Bankruptcy Code or any analogous provisions in any other country or jurisdiction or (y) relate to licenses of intellectual property hereunder.

10.6 Effects of Termination.

- 10.6.1 **Termination by Gilead for Convenience or by Precision for Material Breach or Insolvency.** In the event of a termination of this Agreement by Gilead under Section 10.3.1 or by Precision under Section 10.2.1, Section 10.2.2 or Section 10.4:
- (a) The rights and licenses granted to Gilead under Section 4.1, including any sublicenses, shall be terminated and all such rights shall revert to Precision, except to the extent and for so long as necessary for Gilead to fulfil its responsibilities under the surviving terms of this Agreement as provided in Section 10.8, it being agreed that all such activities shall be discontinued and cease (unless otherwise agreed or required under Applicable Law, by transitioning such activities and responsibilities to Precision) as promptly as possible, subject to Applicable Law.
- (b) If a payment pursuant to the Gilead Funding Commitment is due after notice of termination is given and before the effective date of termination, the payment due by Gilead shall be pro-rated based on the portion of the applicable six (6) month period of the Collaboration Term to have elapsed upon the effective date of termination.
- (c) If at the effective date of termination, Gilead is Manufacturing Licensed Product(s), then, if Precision requests in writing to Gilead within [***] of the effective date of termination, Gilead agrees to Manufacture and supply such Licensed Product(s) to Precision for a reasonable transitional period (not to exceed [***]) from the effective date of termination pursuant to a reasonable and customary transitional supply and quality agreement to be agreed by the Parties in a form reasonably acceptable to Gilead and Precision. Such Licensed Products shall be supplied at [***] of COGS as further defined in the Supply Agreement, plus any amounts due to a Third Party as a result of the sale of such Licensed Product by Gilead or Precision.
 - (d) [***].
- (e) At Precision's request, Gilead agrees to enter into an agreement with Precision, to be negotiated promptly and in good faith, that includes all terms reasonably necessary to transition the Development, use, Manufacture, promotion, marketing and Exploitation of the Licensed Products to Precision, including the following:
- (i) a worldwide, royalty-bearing, non-transferable (except as permitted by such agreement), exclusive or non-exclusive (as requested by Precision), license (or sublicense, as the case may be), with the right to grant sublicenses in accordance with such agreement, to Precision under the Reversion IP (to the extent not licensed pursuant to Section 10.6.1(d)) to Exploit the Licensed Products in the Field in the Territory;

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

- (ii) transition to Precision of all reasonably necessary materials, licenses, Third Party agreements, preclinical and clinical data, safety data, Clinical Studies, Regulatory Documentation, Regulatory Approvals and applications and product trademarks; and
- (iii) provision of technical assistance (including reasonable caps on hours of access) from Gilead relating to the Manufacture, testing and supply of the Licensed Products.
- (f) The Parties shall negotiate in good faith the commercially reasonable compensation to Gilead for the obligations of Gilead under the agreement entered into Section 10.6.1(e), based on Gilead's contributions to the Licensed Products up to the time of termination. In the event the Parties cannot agree on such compensation terms, or any other terms of such agreement, then either Party may refer the disputed terms to one or more arbitrator(s) under the principles of baseball arbitration (i.e., each Party submits a proposed resolution to the arbitrator, and the arbitrator is required to select one of the proposed resolutions), the procedures of which shall be reasonably acceptable to both Parties and shall include express consideration by the arbitrator of the fair market value of the license provided by Gilead in Section 10.6.1(e)(i), the materials transferred under Section 10.6.1(e) and other Relevant Factors.
- 10.6.2 **Termination by Gilead for Material Breach, Insolvency or Change of Control.** If Gilead is entitled to terminate this Agreement pursuant to Section 10.2.1, Section 10.2.2 or 10.3.2, Gilead may elect to terminate this Agreement subject to the provisions set forth in Section 10.6.1(a) and (b) and, in the case of termination pursuant to Section 10.3.2, Section 10.6.1(d).
- 10.7 Remedies. Except as otherwise expressly provided herein, termination of this Agreement in accordance with the provisions hereof shall not limit remedies that may otherwise be available in law or equity.
- 10.8 Accrued Rights; Surviving Obligations. Expiration or termination of this Agreement shall not affect the rights and obligations of each Party under this Agreement that have accrued prior to such date of expiration or termination, and the following provisions shall survive expiration or termination of this Agreement: ARTICLE 1, Sections 4.1.1 and 4.1.2 (solely to the extent provided in Section 10.1 on expiration of this Agreement), Sections 4.2.2 (to the extent that the applicable Information constituted Gilead Dual Know-How or the applicable Patents constituted Gilead Dual Patents, in each case as of the date of termination of this Agreement, including, for clarity, Patents filed or issued at any later date covering or claiming applicable inventions conceived on or prior to such date), 4.3.2, and 4.3.3 (solely to the extent applicable to the license set forth in Section 4.2.2), Sections 4.4, 5.8, 5.9 and 5.10, 6.1 (excluding the proviso in Section 6.1.3(b)), 6.2.2, 6.2.3, 6.2.4, 6.3.2, 6.5.3, ARTICLE 7, ARTICLE 8, ARTICLE 9, Sections 10.6, 10.7 and 10.8 and ARTICLE 11.

ARTICLE 11 MISCELLANEOUS

- 11.1 Force Majeure. Neither Party shall be held liable or responsible to the other Party or be deemed to have defaulted under or breached this Agreement for failure or delay in fulfilling or performing any term of this Agreement (other than an obligation to make payments) when such failure or delay is caused by or results from events beyond the reasonable control of the non-performing Party, including fires, floods, earthquakes, hurricanes, embargoes, shortages, epidemics, quarantines, war, acts of war (whether war be declared or not), terrorist acts, insurrections, riots, civil commotion, strikes, lockouts, or other labor disturbances (whether involving the workforce of the non-performing Party or of any other Person), acts of God or acts, omissions or delays in acting by any governmental authority or Regulatory Authority (except to the extent such delay results from the breach by the non-performing Party or any of its Affiliates of any term or condition of this Agreement). The non-performing Party shall notify the other Party of such force majeure within [***] after such occurrence by giving written notice to the other Party stating the nature of the event, its anticipated duration, and any action being taken to avoid or minimize its effect. The suspension of performance shall be of no greater scope and no longer duration than is necessary and the non-performing Party shall use Commercially Reasonable Efforts to remedy its inability to perform as promptly as possible.
- 11.2 Export Control. This Agreement is made subject to any restrictions concerning the export of Licensed Products or technical information from the United States or other countries that may be imposed on the Parties from time to time. Each Party agrees that it shall not export, directly or indirectly, any technical information acquired from the other Party under this Agreement or any Licensed Products using such technical information to a location or in a manner that at the time of export requires an export license or other governmental approval, without first obtaining the written consent to do so from the appropriate agency or other governmental authority or Regulatory Authority in accordance with Applicable Law.
- 11.3 Assignment. This Agreement and the rights and obligations hereunder shall not be assignable, transferable or delegable by either Party without the prior written consent of the other Party; *provided that*, either Party may assign or transfer any or all of its rights and assign, transfer or delegate any or all of its obligations under this Agreement to (a) any of its Affiliates or (b) a successor to all or substantially all of its business related to this Agreement (including in connection with a merger, consolidation, or sale of all or substantially all of its assets related to this Agreement), in each case (a) and (b), without the prior written consent of the other Party. Notwithstanding the foregoing, Precision shall not be required to obtain Gilead's consent to sell, assign, pledge as security, contribute, or otherwise transfer, in whole or in part, to any Third Party its rights to receive any payment under this Agreement, and, as it relates to any such transfer, Precision may provide to such Third Party (i) a copy of reports received from Gilead pursuant to Section 5.4.10 and (ii) the result of any audit conducted pursuant to Section 5.9, in each case of (i) and (ii), so long as any such Third Party is bound by confidentiality and non-use obligations equivalent in scope to those set forth in ARTICLE 7 of this Agreement. Any attempted assignment, transfer or delegation in violation of this Section 11.3 shall be null and

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

void, except to the extent otherwise permitted under Section 4.3. Notwithstanding anything to the contrary in this Agreement, the Patents and Information Controlled by an Acquirer (or any Affiliate of any such Acquirer existing prior to the closing of the transaction) of a Party under this Agreement shall be automatically excluded from the rights licensed to the other Party under this Agreement to the extent (A) such Patents and Information were filed or developed, respectively, prior to the transaction that was the basis for such assignment, transfer or succession or resulted in such Person becoming an Affiliate, or (B) such Patents and Information were developed after such transaction under a separate and firewalled program not under this Agreement without use of any Patents, Know-How or Confidential Information of the other Party that is licensed hereunder.

11.4 Severability. If any provision of this Agreement is held to be illegal, invalid, or unenforceable under any present or future law, and if the rights or obligations of either Party under this Agreement shall not be materially and adversely affected thereby, (a) such provision shall be fully severable, (b) this Agreement shall be construed and enforced as if such illegal, invalid, or unenforceable provision had never comprised a part hereof, (c) the remaining provisions of this Agreement shall remain in full force and effect and shall not be affected by the illegal, invalid, or unenforceable provision or by its severance herefrom, and (d) in lieu of such illegal, invalid, or unenforceable provision, there shall be added automatically as a part of this Agreement a legal, valid, and enforceable provision as similar in terms to such illegal, invalid, or unenforceable provision as may be possible and reasonably acceptable to the Parties. To the fullest extent permitted by Applicable Law, each Party hereby waives any provision of law that would render any provision hereof illegal, invalid, or unenforceable in any respect.

11.5 Governing Law.

- 11.5.1 **Governing Law.** This Agreement shall be governed by, and construed in accordance with, the laws of the state of New York, regardless of the laws that might otherwise govern under applicable principles of conflicts of laws thereof.
- 11.5.2 **Jurisdiction; Venue; Service of Process.** Each Party irrevocably submits to the exclusive jurisdiction of the state or federal courts located in the State and County of New York for the purposes of any Action arising out of this Agreement. Each Party agrees to commence any such Action either in the state courts of New York or the United States District Court for the Southern District of New York. Each Party further agrees that service of any process, summons, notice or document by the U.S. registered mail to such Party's respective address set forth in Section 11.7 shall be effective service of process for any Action in New York with respect to any matters to which it has submitted to jurisdiction in this Section 11.5.2. Each Party irrevocably and unconditionally waives any objection to the laying of venue of any Action arising out of this Agreement in the state or federal courts of New York, and hereby and thereby further irrevocably and unconditionally waives and agrees not to plead or claim in any such court that any such Action brought in any such court has been brought in an inconvenient forum.
- 11.5.3 **Waiver of Right to Trial by Jury.** EACH PARTY HERETO WAIVES ITS RIGHT TO TRIAL OF ANY ISSUE BY JURY. Each Party hereto (a) certifies that no Representative or attorney of the other Party has represented, expressly or otherwise, that such Party would not, in the event of any Action, seek to enforce the foregoing waiver and (b) acknowledges that it and the other Party have been induced to enter into this Agreement, by, among other things, the mutual waiver and certifications in this Section 11.5.3.

11.5.4 **Equitable Relief.** Each Party acknowledges and agrees that the restrictions set forth in Section 4.5 and ARTICLE 6 and ARTICLE 7 are reasonable and necessary to protect the legitimate interests of the other Party and that such other Party would not have entered into this Agreement in the absence of such restrictions, and that any breach or threatened breach of any provision of such Section or Articles may result in irreparable injury to such other Party for which there will be no adequate remedy at law. In the event of a breach or threatened breach of any provision of such Section or Articles, the non-breaching Party shall be authorized and entitled to seek injunctive relief, whether preliminary or permanent, specific performance, and an equitable accounting of all earnings, profits, and other benefits arising from such breach, which rights shall be cumulative and in addition to any other rights or remedies to which such non-breaching Party may be entitled in law or equity to prevent such breach or threatened breach of this Agreement and to enforce specifically the terms and provisions of such Section or Articles of this Agreement in the state and federal courts of New York. Both Parties agree to waive any requirement that the other (a) post a bond or other security as a condition for obtaining any such relief, and (b) show irreparable harm, balancing of harms, consideration of the public interest, or inadequacy of monetary damages as a remedy. Nothing in this Section 11.5.4 is intended, or should be construed, to limit either Party's right to equitable relief or any other remedy for a breach of any other provision of this Agreement.

11.6 Dispute Resolution. The Parties recognize that disputes as to certain matters may from time to time arise that relate to either Party's rights or obligations hereunder, including the interpretation, alleged breach, enforcement, termination or validity of this Agreement (a "Dispute"). For clarity, Dispute shall not include matters within the JSC's authority, which are resolved under Section 2.3.3 including through the exercise by Gilead or Precision of its final decision making authority in accordance therewith. Any Dispute shall be referred to the Senior Officers of the Parties, who shall confer in good faith on the resolution of the issue. Any final decision mutually agreed to by the Senior Officers shall be conclusive and binding on the Parties. If the Senior Officers are not able to agree on the resolution of any such issue within [***] after such issue was first referred to them, then, either Party may initiate litigation in accordance with Section 11.5 or with respect to Disputes that involve the infringement or validity of any Precision Patents, Joint Collaboration Program Patents or Gilead Dual Patents outside the United States, such Dispute shall be resolved by a court of competent jurisdiction, notwithstanding Section 11.5, in any country in which such rights apply. Notwithstanding anything herein to the contrary, nothing in this Section 11.6 shall preclude either Party from seeking interim or provisional relief, including a temporary restraining order, preliminary injunction or other interim equitable relief concerning a Dispute, if necessary to protect the interests of such Party.

11.7 Notices.

11.7.1 **Notice Requirements.** Any notice, request, demand, waiver, consent, approval, or other communication permitted or required under this Agreement shall be in writing, shall refer specifically to this Agreement and shall be deemed given only if delivered by hand or sent by facsimile transmission (with transmission confirmed) or by internationally

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

recognized overnight delivery service that maintains records of delivery, addressed to the Parties at their respective addresses specified in Section 11.7.2 or to such other address as the Party to whom notice is to be given may have provided to the other Party in accordance with this Section 11.7.1. Such notice shall be deemed to have been given as of the date delivered by hand or transmitted by facsimile (with transmission confirmed) or on the second Business Day (at the place of delivery) after deposit with an internationally recognized overnight delivery service. Any notice delivered by facsimile shall be confirmed by a hard copy delivered as soon as practicable thereafter. This Section 11.7.1 is not intended to govern the day-to-day business communications necessary between the Parties in performing their obligations under the terms of this Agreement.

11.7.2 Address for Notice.

If to Gilead, to:

Gilead Sciences, Inc. 333 Lakeside Drive Foster City, CA 94404 Attn: Alliance Management Email: [***]

with a copy (which shall not constitute notice) to:

Gilead Sciences, Inc. 333 Lakeside Drive Foster City, CA 94404 Attn: General Counsel Email: [***]

If to Precision, to:

Precision BioSciences, Inc. 302 East Pettigrew Street, Suite A-100 Durham, NC 27701 Attention: Michael Dombeck, Vice President, Business Development

Facsimile: (480) 393-5553

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

with a copy (which shall not constitute notice) to:

Smith, Anderson, Blount, Dorsett, Mitchell & Jernigan, LLP 150 Fayetteville Street, Suite 2300 Raleigh, NC 27601 Attention: John R. Therien, Esq.

Facsimile: (919) 821-6800

- 11.8 Entire Agreement. This Agreement, together with the Schedules expressly contemplated hereby and attached hereto, and together with the Supply Agreement and the Quality Agreement (once executed), contains the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements or understandings between the Parties with respect to the subject matter hereof, including the Confidentiality Agreement and the Material Transfer Agreement between the Parties dated as of February 22, 2016 (as amended). Each Party confirms that it is not relying on any representations or warranties of the other Party except as specifically set forth in this Agreement.
- 11.9 English Language. This Agreement shall be written and executed in, and all other communications under or in connection with this Agreement shall be in, the English language. Any translation into any other language shall not be an official version thereof, and in the event of any conflict in interpretation between the English version and such translation, the English version shall control.

11.10 Amendments and Waivers.

- 11.10.1 No modification, amendment or waiver of any provision of, or consent or approval required by, this Agreement, nor any consent to or approval of any departure herefrom, shall be effective unless it is in writing and signed by the Party against whom enforcement of any such modification, amendment, waiver, consent or approval is sought. Such modification, amendment, waiver, consent or approval shall be effective only in the specific instance and for the purpose for which given.
- 11.10.2 Neither the failure of either Party to enforce, nor the delay of either Party in enforcing, any condition or part of this Agreement at any time shall be construed as a waiver of that condition or part or forfeit any rights to future enforcement thereof. No action taken pursuant to this Agreement, including any investigation by or on behalf of either Party, shall be deemed to constitute a waiver by the Party taking action of compliance by the other Party with any representation, warranty, covenant, agreement or obligation contained herein.
- 11.11 Cumulative Rights. Except as expressly provided herein, the various rights under this Agreement shall be construed as cumulative, and no one of them is exclusive of any other or exclusive of any rights allowed by Applicable Law.
- 11.12 **Benefits of Agreement.** All of the terms and provisions of this Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. Except for the provisions of ARTICLE 9, this Agreement is for the sole benefit of the Parties and not for the benefit of any other Person other than Duke to the extent required by the Duke Agreement.

- 11.13 Further Assurance. Each Party shall duly execute and deliver, or cause to be duly executed and delivered, such further instruments and do and cause to be done such further acts and things, including the filing of such assignments, agreements, documents, and instruments, as may be necessary or as the other Party may reasonably request in connection with this Agreement or to carry out more effectively the provisions and purposes hereof, or to better assure and confirm unto such other Party its rights and remedies under this Agreement.
- 11.14 Relationship of the Parties. It is expressly agreed that Gilead, on the one hand, and Precision, on the other hand, shall be independent contractors and that the relationship between the two Parties shall not constitute a partnership, joint venture, or agency. Neither Gilead, on the one hand, nor Precision, on the other hand, shall have the authority to make any statements, representations, or commitments of any kind, or to take any action, which shall be binding on the other, without the prior written consent of the other Party to do so. All persons employed by a Party shall be employees of such Party and not of the other Party and all costs and obligations incurred by reason of any such employment shall be for the account and expense of such Party.
- 11.15 Counterparts. This Agreement may be executed in two counterparts, and each such counterpart hereof shall be deemed to be an original instrument, but both such counterparts together shall constitute but one agreement. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or other electronic transmission shall be effective as delivery of a manually executed original counterpart of this Agreement.
- 11.16 Schedules. In the event of any inconsistencies between this Agreement and any Schedules or other attachments hereto, the terms of this Agreement shall control.

11.17 Descriptive Headings; Certain Interpretations.

- 11.17.1 Descriptive headings are for convenience only and shall not control or affect the meaning or construction of any provision of this Agreement.
- 11.17.2 Except as otherwise expressly provided in this Agreement or as the context otherwise requires, the following rules of interpretation apply to this Agreement: (a) the singular includes the plural and the plural includes the singular; (b) "or" and "any" are not exclusive and the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation;" (c) a reference to any contract includes permitted supplements and amendments; (d) a reference to Applicable Law includes any amendment or modification to such Applicable Law; (e) a reference to a Person includes its successors, heirs and permitted assigns; (f) a reference to one gender shall include any other gender; (g) a reference in this Agreement to an Article, Section, Exhibit or Schedule is to the referenced Article, Section, Exhibit or Schedule of this Agreement, unless expressly specified otherwise; (h) "hereunder," "hereof," and words of similar import shall be deemed references to this Agreement as a whole and not to any particular Article, Section or other provision; (i) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if" and (j) a reference to the right to "approve" includes the right to reject.
- 11.17.3 The Parties agree that they have been represented by counsel during the negotiation, drafting, preparation and execution of this Agreement and, therefore, waive the application of any Applicable Law or rule of construction providing that ambiguities in an agreement or other document shall be construed against the Party drafting such agreement or document.

THIS A	THIS AGREEMENT IS EXECUTED by the authorized representatives of the Parties as of the date first written above.				
GILE	AD SCIENCES, INC.	PREC	ISION BIOSCIENCES, INC.		
By:	/s/ John F. Milligan	By:	/s/ Matthew Kane		

[SIGNATURE PAGE TO COLLABORATION AND LICENSE AGREEMENT]

Name: Matthew Kane

Title: Chief Executive Officer

Name: John F. Milligan

Title: President & Chief Executive Officer

Schedule 1.45 Duke IP

recision Docket No.	<u>Country</u>	Application No.	Patent No.	Status
***]	Edutaba	Edutation	Edutate	Education 1
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***]	[***]
]	[]	[***]	[***] [***]	[***]
] ***]	[]	[***]		[***]
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
***	[***]	[***]	[***]	[***
] ***]	[]	[***]	[***]	[*** [***
· * * · • • 1	[***]	[***]	[***]	
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***] [***]	[***
] ***]	[]	[***] [***]	L J	[*** [***
`TT]	[***]		[***]	L
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[*** [***
]	[]	[***]	[***]	[***
***	[***]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
]	[]	[***]	[***]	[***
] ***]	[]	[***]	[***]	[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Schedule 1.112 Precision Existing Patents

Precision Docket No.	Country	Application No.	Patent No.	Status
[***]				
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]		[***]
[***]				
[***]				
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]	[***]	[***]
L J	L J	L J	L J	L J

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Precision Docket No.	Country	Application No.	Patent No.	Status
[***]	[***]	[***]	[***]	[***]
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]				
[***]				
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]				
[***]	[***]	[***]		[***]
[***]				
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Schedule 1.113 Precision HBV Patents

Precision Docket No.	Country	Application No.	Patent No.	Status
[***]				
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]
[***]				
[***]	[***]	[***]		[***]
[***]				
[***]	[***]	[***]		[***]
[***]	[***]	[***]		[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Schedule 2.1.2 Initial JSC Members

For Precision:
[***]
For Gilead:

[***]

[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Schedule 2.2.1 Initial JRDC Members

For Precision:	
[***]	
For Gilead:	

[***]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Schedule 3.8.3

Template Report

GILEAD SCIENCES - PRECISION BIOSCIENCES COLLABORATION REPORT

A. Preclinical R&D

Pre-Clinical R&D Progress: Months 1 – 6

Executive Summary:

Include:

- Key strategic goals
- · Milestones achieved
- GO/NO GO decisions
- Plan amendments (scope, timelines, resources, other)

Details:

Gilead Funding: [***]

Activity Goal
1. Activity Details Activity Status Timeline Costs, FTEs (Budgeted/Actual)

2.

• Pre-Clinical R&D Progress: Months 7 – 12

Executive Summary:

Include:

- · Key strategic goals
- Milestones achieved
- GO/NO GO decisions
- Plan amendments (scope, timelines, resources, other)

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Det	ails:				
Gile	ead Funding: [***]				
	Activity Goal 1.	Activity Details	Activity Status	Timeline	Costs, FTEs (Budgeted/Actual)
	2.				
•	Pre-Clinical R&D Progress: Months 13 – 18	3			
Exe	ecutive Summary:				
Incl	ude:				
•	Key strategic goals				
•	Milestones achieved				
•	GO/NO GO decisions				
•	Plan amendments (scope, timelines, resources,	other)			
Det	ails:				
Gile	ead Funding: [***]				
	Activity Goal 1.	Activity Details	Activity Status	Timeline	Costs, FTEs (Budgeted/Actual)
	2.				
•	Pre-Clinical R&D Progress: Months 19 – 24	I			
Exe	ecutive Summary:				
Incl	ude:				
•	Key strategic goals				

- Milestones achieved
- GO/NO GO decisions
- Plan amendments (scope, timelines, resources, other)

Details:

Gilead Funding: [***]

Activity Goal	Activity Details	Activity Status	<u>Timeline</u>	Costs, FTEs (Budgeted/Actual)
1.				
2.				

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

3.	Clinical Development		
	Development and Regulatory Activities Progress		
	Activity Phase 1 Trial	<u>Status</u>	Details
	Phase 2 Trial		
	Phase 3 Trial		
	Development Milestones Progress (as described in Section 5.2.1 of the Agreement)		

Development Milestones [***]	<u>Status</u>	Comments
[***]		
[***]		
[***]		
[***]		

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Commercialization Milestones Progress (as described in Section 5.3.1 of the Agreement)

Status Comments

<u>Commercial Milestones</u>
The first time aggregate global Net Sales of Licensed Products in a Calendar Year are equal to or greater than [***] and less [***]

The first time aggregate global Net Sales of Licensed Products in a Calendar Year are equal to or greater than [***] and less [***]

The first time aggregate global Net Sales of Licensed Products in a Calendar Year are equal to or greater than [***] and [***]

The first time aggregate global Net Sales of Licensed Products in a Calendar Year are equal to or greater than [***]

Functional Group Resource Allocation:

Representatives from the following functional groups worked on the program during this semiannual period:

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Schedule 5.1.2 Invoicing Template

[Company Name]

INVOICE

[Street Address]
[City, ST ZIP]
Phone: (000) 000-0000

INVOICE # DATE [123456] 5/1/201

BILL TO

Accounts Payable
Gilead Sciences, Inc.
333 Lakeside Drive
Foster City, California 94404
APInvoices@gilead.com

<u>DESCRIPTION</u>	AMOUNT
Gilead Funding Commitment Pursuant to Section 5.1.1 of the Collaboration and License Agreement between Gilead	XX
Sciences, Inc. and Precision Biosciences, Inc. dated September 7, 2018, for the six (6) month period to	
TOT	AL \$ XX

Payment Terms: Payment shall be due within [***] of receipt by Gilead.

If you have any questions about this invoice, please contact [Name, Phone, email@address.com]

^[***] Confidential treatment requested pursuant to a request for confidential treatment filed with the Securities and Exchange Commission. Omitted portions have been filed separately with the Commission.

Schedule 6.2.8

ARCUS Patents and Precision Patents

Schedule to be provided upon the timing set forth in Section 6.2.8.

Schedule 7.4 Press Release(s)

Gilead & Precision Biosciences logos

CONTACTS: Sung Lee, Gilead Investors

650-524-7792

Amy Flood, Gilead Media

650-522-5643

Heather King, Precision Media

919-314-5512 x1332

GILEAD SCIENCES AND PRECISION BIOSCIENCES ANNOUNCE COLLABORATION TO DEVELOP THERAPIES AGAINST HEPATITIS B VIRUS USING ARCUS GENOME EDITING

FOSTER CITY, Calif. and DURHAM, NC, Sept XX, 2018 - Gilead Sciences (Nasdaq: GILD) and Precision Biosciences announced today that the companies have entered into a strategic collaboration to develop therapies targeting the *in vivo* elimination of hepatitis B virus (HBV) with Precision's proprietary genome editing platform, ARCUS.

An estimated 257 million people are living with HBV infection around the world. Current HBV treatments suppress HBV viral replication but do not completely clear the virus. The presence of covalently closed circular DNA (cccDNA) enables HBV replication if treatment is stopped. Preliminary studies at Gilead using ARCUS nucleases to target HBV cccDNA *in vitro* have demonstrated significant activity against cccDNA and integrated HBV DNA in human hepatocytes.

"Gilead is committed to developing innovative therapies to achieve functional cure for patients with chronic Hepatitis B virus infection," said John McHutchison, MD, Chief Scientific Officer and Head of Research and Development at Gilead. "We are excited about the potential of genome editing and Precision's ARCUS technology, which has demonstrated promising *in vitro* activity. We look forward to exploring this technology as an important component of our HBV cure research efforts."

Under the terms of the collaboration agreement, Precision will be primarily responsible for the development, formulation, and preclinical evaluation of the investigational nucleases, and Gilead will be responsible for the clinical development and commercialization of potential therapies. Gilead will fully fund the research and development. Precision is eligible to receive milestone payments of up to an aggregate of \$445 million and tiered royalties that go up to the mid-teens for commercial products developed through the collaboration.

Precision Chief Scientific Officer Derek Jantz commented, "Gilead's cure-based approach to Hepatitis B is comprehensive and exciting. Precision is pleased that initial studies with our ARCUS platform have established an important role for genome editing in their HBV program. This is an excellent application for our technology, which has made notable progress toward therapeutic *in vivo* editing in relevant models over the last year."

About Precision BioSciences

Precision BioSciences is dedicated to improving life. Our mission is to cure genetic disease, overcome cancer, and feed the planet. We are striving to achieve this goal with ARCUS, our therapeutic-grade, naturally-derived genome editing system that combines both specificity and efficacy to help overcome life's greatest genetic challenges. For additional information, please visit www.precisionbiosciences.com

About Gilead Sciences

Gilead Sciences, Inc. is a research-based biopharmaceutical company that discovers, develops and commercializes innovative medicines in areas of unmet medical need. The company strives to transform and simplify care for people with life-threatening illnesses around the world. Gilead has operations in more than 35 countries worldwide, with headquarters in Foster City, California.

Gilead Forward-Looking Statement

This press release includes forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995 that are subject to risks, uncertainties and other factors, including the risk that the parties may not realize the potential benefits of this collaboration. All statements other than statements of historical fact are statements that could be deemed forward-looking statements. These risks, uncertainties and other factors could cause actual results to differ materially from those referred to in the forward-looking statements. The reader is cautioned not to rely on these forward-looking statements. These and other risks are described in detail in Gilead's Annual Report on Form 10-K for the year ended December 31, 2017, as filed with the U.S. Securities and Exchange Commission. All forward-looking statements are based on information currently available to Gilead, and Gilead assumes no obligation to update any such forward-looking statements.

###

For more information on Gilead Sciences, please visit the company's website at www.gilead.com, follow Gilead on Twitter (@GileadSciences) or call Gilead Public Affairs at 1-800-GILEAD-5 or 1-650-574-3000.

Schedule 8.2.12 Precision IP Subject to U.S. Federal Government Rights

None.

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the use in the Registration Statement on Form S-1 of Precision Biosciences, Inc. filed on March 1, 2019 of our report dated February 21, 2019 (March 1, 2019 as to Note 15) relating to the financial statements of Precision BioSciences, Inc., appearing in the Prospectus, which is part of that Registration Statement, and to the reference to us under the headings "Experts" in such Prospectus.

/s/ Deloitte & Touche LLP

Raleigh, North Carolina March 13, 2019